CHAPTER ONE:

INTRODUCTION

1.1. Background to the Study

The universal characteristic feature of a registered or an incorporated company as contained in the various legislations on companies is that from the date in the Certificate of Incorporation, the subscribers of the memorandum together with such other persons as may from time to time become members of the company, shall be a body corporate¹ by the name stated in the Certificate of Incorporation and capable of exercising all powers and functions of an incorporated company.

The notion of a company being a body corporate from the date of incorporation in the Certificate of Incorporation implies in simple terms, that a company as from the date of incorporation acquires in law an existence and personality, which is distinct, separate and independent from that of its members² and indeed all other entities in the world. It becomes a legal or juristic person with its own name,³ capable of suing and being sued in that name, entering into contracts in that name with any person, holding or disposing of land or other property and of living in perpetuity as it were,⁴ whether or not the members who formed it are themselves, living or dead.

One of the incidences of incorporation is that every company shall for the furtherance of its authorised business or objects, have all powers of a natural person of full capacity.⁵Beyond a company being vested with all the powers of a natural

¹Companies and Allied Matters Act, Cap C20 Laws of the Federation of Nigeria 2004 (CAMA), s. 37; Companies Act 2006, s. 16 (2) (UK); Companies Act 2013. s. 14 (2) Ghana.

²Salomon v Salomon & Co Ltd [1897] AC 22; Aso Motel Kaduna Ltd v Deyemo[2006]7 N.W.L.R (pt. 978) 87; New Resources Int'l Ltd & Anor v Oranusi[2011] 2 N.W.L.R (pt. 1230) 102; [2011] 3 CLRN 52.

³*ibid*; Companies Act, 2008 (as amended), s. 19 (1) (a) (South Africa).

⁴CAMA, *op cit*, s.37.

 $^{^{5}}$ *Ibid*, s. 38 (1).

person, it has the power to borrow money, charge property and to issue debentures.⁶ These powers are primarily conferred on companies for purposes of raising capital for the furtherance of the business of the company. Although the incorporation of a company inures it with supposedly a life in perpetuity, this incident of incorporation does not pretend to assert that a company cannot 'die'. Therefore, there may be circumstances that the existence of a company can be brought to an end.⁷ This is usually exemplified by the process of winding up or liquidation of a company on the ground that it cannot pay its debts or discharge its obligations.⁸

It has been observed over time that the consequence of winding up or liquidation of an insolvent company do not ultimately engender business efficacy or endow the creditor, whether secured or unsecured, with any business or commercial advantage. Rather the process of winding up or liquidation produces 'victims' whose loss or injury may not be remedied. While an insolvent company fixed with a winding up order may eventually be dissolved,⁹ the creditor may not for certain reasons be able to reap any benefit out of the liquidation of the insolvent company.¹⁰

Arising from the ineffectual effects of winding up of an insolvent company is the clamour for alternative options to winding up. Ajavi¹¹ posited, thus:

> ... the global trend is to develop and allow for the development and survival of companies especially in our country where death of a company means that of

⁶*Ibid*, s. 166.

⁷ P L Davies & S Worthington, *Gower & Davies' Principles of Modern Company Law* (9thedn, London: Sweet & Maxwell, 2012) p. 1271.

⁸ CAMA, op cit, ss. 408 (d) & 409; Companies Act 2006 (UK), op cit, ss. 122 (i) (f) & 123; Bodies Corporate (Official Liquidations) Act 1963 [Act 180] [Ghana] ss. 3 (3) & 4 (2) (d).

C.C.B Plc v Mbakwe(2000) F.W.L.R (pt 14) 2351.

¹⁰ G C Nwakoby, The Law and Practice of Commercial Arbitration in Nigeria (2ndedn, Enugu: Snaap Press Limited, 2014) p. 467. This prolific writer commented on the delays in the Nigerian Court system; O Ajayi, Legal Aspects of Finance in Emerging Markets (London: LexisNexis ButterworthsTolley, 2005) p 603, where he stated that since assets of the company realised during winding up is shared among the creditors, winding up on the ground of the company's inability to pay debt may not be a viable option. ¹¹Ajayi, *ibid*.

many families and those dependent on the employees

of the affected company....

While the above-thought for enthronement of corporate rescue mechanisms remain topical, the recent and current development¹² in the corporate world and the state of the Nigerian economy demands for an urgent reform and strengthening of corporate rescue and insolvency law and practice in Nigeria. Without more, the essence of a sound legislation was captured at a public hearing by the Senate Committee on Banking, Insurance and other Financial Institutions on the Bankruptcy and Insolvency (Repeal and Re-enactment) Bill¹³ by the Senate President thus, 'Insolvency system and practice always plays an important role in attracting both domestic and foreign investments as well as promoting investments and entrepreneurial development ... given these opportunities, there is urgent need for us to repeal and re-enact this Act, which has become obsolete and out-dated'.¹⁴

It is pertinent to state that while the stakeholders at the said Senate Committee public hearing on the "Bankruptcy and Insolvency (Repeal and Re-enactment) Bill" did approve the repeal and re-enactment of the 37 years old Bankruptcy/Insolvency Act, they were vociferous in pointing out the flaws, contradictions and shortcomings of the proposed Act.

Finally, the need for urgent reform of corporate rescue and insolvency procedure in Nigeria cannot be over emphasised. The economy is in recession and companies, whether in manufacturing or non-manufacturing sectors, are facing

¹²Channels Television Headline News of March 17th, 2016 "*AMCON Liabilities Rise to 6.6 Trillion Naira*". Prior to this, AMCON had given notice of its delinquent debtors. The only remedy as it stands today in Nigeria, is for AMCON, to appoint a receiver (AMCON Act 2010 s. 48), obtain Order of Possession of the debtor's property (AMCON Act s. 49(1)), freezing of the debtor company's account (AMCON Act s. 50 (1)), Commencement of debt recovery action (AMCON Act, part V) and Winding up (AMCON Act, s. 52);

¹³2015. The Bill was passed into Law by the Senate on the 25th of May, 2016.

¹⁴O Agbajileke, 'Stakeholders Applaud Repeal of 37 years old Bankruptcy/Insolvency Act', Business Day Newspaper, Tuesday, 15th March, 2016. <u>www.businessdayonline.com</u> accessed 15 March 2016.

business decline and predisposed to imminent collapse, banks are overwhelmed with delinquent debtors;¹⁵ the level of unemployment is on an astronomical scale; crude oil price is on an all-time low; State Governments are facing illiquidity; job losses, redundancy and pay-cuts are the prevailing phenomenon in the labour market, Family disharmony, societal breakdown, incidence of violence and disorientation in values are common place. Arising from this, is the agitation for diversification of the Nigerian economy and call for investments whether foreign or domestic. This can veritably be achieved if there is a sound and robust corporate rescue and insolvency law and procedure in Nigeria.

1.2. Statement of Problem

There are two major modes for a company to raise capital. A company limited by shares may raise capital through equity finance, that is, by obtaining capital through allotment of its shares.¹⁶ This mode of raising capital by allotment of shares attaches certain rights and liabilities on the members or shareholders¹⁷ including the right to dividend, if and when declared out of the distributable profits of the company.¹⁸ Another mode of raising capital is through debt finance. The power of every company registered in Nigeria to borrow money, to charge property and to issue debentures for purposes of its business or objects and to provide security for any debt, liability or obligation of the company or any third party is clearly provided in section 166 of the CAMA. However, there are obvious problems arising from a company raising capital through debt finance. The core problem that typify such form of raising capital is mostly tied to the inadequate provisions and mechanisms for creditors,

 ¹⁵Skye Bank Plc publication of List of Loan Debtors, BusinessDay, February 11, 2016, pp. 33-35;
 Fidelity Bank Plc publication of Names of Delinquent Customers, BusinessDay, February 15, 2016, p.19;
 AMCON publication of List of Delinquent Debtors, BusinessDay, February 15, 2016, p. 13.
 ¹⁶CAMA, *op cit*, s. 117.

 $^{^{17}}Ibid$, s. 114

¹⁸*Ibid*, s. 379 (5).

whether secured or unsecured, to realise the credit provided to the debtor company for the furtherance of its business or to enforce the security so provided by the debtor company.

It is common to insert contractual protections in the debt instrument in the form of covenants that generally requires the borrowing company to provide the creditor with periodic accounting information, perhaps including credit ratings, to conduct its operations so as to maintain predetermined financial ratios between assets and liabilities, insertion of negative pledge clauses and to refrain from certain defined acts or activities without the prior consent of the lender. This list of prohibitions usually extends to the disposal of substantial assets, changing the primary activities of the company, taking on additional loans, granting further security, distributing dividends above a nominated level of returns or changing the management or ownership structure. However, there seems to be a problem of enforcement of these contractual covenants when there is a breach.¹⁹

The oft-relied upon protection for a creditor in debt financing is proprietary security. This is a situation where a borrower is obliged to provide security for the repayment of its debts, mostly by way of a charge. In Nigeria, there are two forms of charge, to wit, fixed and floating charges.²⁰ By the very ambulatory²¹ features of a floating charge, a borrowing company or the floating chargor can deplete, if not, extinguish the assets that are subject to the floating charge. In such a situation, the creditor is eventually left with an empty security. This scenario avowedly throws the

¹⁹Davies & Worthington, *op cit*, p. 1203.

²⁰CAMA, op cit, s 178

²¹CAMA, op cit,s. 178; Siebe Gorman & Co Ltd v Barclays Bank Ltd [1997] 2 Lloyds Rep 142; Re Yorkshire Woolcombers Association [1803] 2 Ch 284.

problem of the enforcement of proprietary security by way of floating charge and raises certain salient questions.²²

While in other jurisdictions such as the United Kingdom, the secured creditors have the "right of pursuit",²³ which means that where a company wrongfully and in violation of the rights of a chargee disposes of the property subject to the charge, the chargee is then entitled to enforce the security against any identifiable proceeds of the disposition, such right is not available to a chargee under relevant Nigerian laws. Hence, in the event that the right of pursuit is not provided for in the debt instrument, the security becomes a mere fluke.

The most common modes of enforcement of propriety security in Nigeria are Receivership²⁴and Winding up. However, there are doubts on the efficacy of these modes. At present, the primary legislation on receivership and winding up, even on the ground of insolvency, in Nigeria, is the Companies and Allied Matters Act. It is obvious that the provisions of the CAMA is out-dated²⁵ and not in conformity with international best practices as it relates to receiverships and insolvency proceedings. There is absence of well-defined and adequate provisions on qualifications for a person to be appointed a receiver and/or manager, provisional liquidator, liquidator, etc in the process of receivership and winding up proceedings under the CAMA. Consequently, receivership and winding up proceedings as mechanisms for the enforcement of proprietary security seem not to have achieved much. Suffice to state that there is the problem of out-dated, inflexible and disparate legislations on

²²What happens when the assets, subject of a floating charge, has been depleted or extinguished? What are the remedies open to a creditor in such circumstances?

²³Davies & Worthington, *op cit*, p. 1213.

 ²⁴A A Aina, 'Legal and Institutional Framework for Debt Finance by Companies in Nigeria', A PhD Pre-Field Seminar presented at the Faculty of Law, University of Ibadan, Ibadan, Nigeria on October, 2014, p. 1.
 ²⁵ A Idigbe, 'Using Existing Insolvency Framework to Drive Business Recovery in Nigeria: The Role of the Judges' A Paper delivered at the Federal High Court Judges Conference at Sankuru Hotel Sokoto, 2011, p. 11.

corporate insolvency in Nigeria. Furthermore, there is a problem arising from the absence of professional standards and regulation of persons who practise as corporate rescue and insolvency practitioners in Nigeria.

Some of the provisions in the CAMA as relating to winding up, particularly the definition of the phrase 'company's inability to pay debts', that is, insolvency, as when a company is indebted in a sum exceeding N2, 000.00 (Two thousand Naira),²⁶ is for short of words, confusing, laughable, out-rightly obsolete and absurd. According to Idigbe, the definition of 'insolvent person' in the CAMA is "not only inadequate but inefficient".²⁷ Thus, the definition of insolvency, which problem has made the law and practice of insolvency in Nigeria to be confusing, complex and uncertain.

There is the problem of inadequate corporate rescue mechanisms in the CAMA and other relevant laws in Nigeria. The global trend is to develop and allow for the development and survival of companies.²⁸ Unfortunately, the available corporate rescue mechanisms in Nigeria under the extant laws are limited to Arrangements and Compromise,²⁹ Mergers³⁰ and Receivership.³¹ Regulatory Rescue and other informal corporate rescue mechanism such as cherry picking, purchase and Assumption and Debt for Equity Swap. These mechanisms are not only defective and deficient because of the absence of moratorium in them, but are also inadequate and limited when juxtaposed with the expansive corporate rescue mechanisms in other

²⁶ CAMA, *op cit*, s. 409 (a).

²⁷Idigbe, *op cit*, p. 5.

²⁸CAMA, *op cit*, Part XVI.; UNCITRAL

²⁹Agbajileke, *op cit*.

³⁰Investment and Securities Act (ISA) 2007, ss. 119-130.

³¹CAMA, op cit, Part XIV, ss. 387-400.

jurisdictions.³² Hence, there is the challenge of the absence of dynamic, functional and effective corporate rescue mechanisms in the extant legislations.

However, these challenges associated with the archaic and out-dated provisions in the CAMA as relating to corporate rescue and insolvency may soon be over with the on-going deliberation and consideration of the "Bankruptcy and Insolvency (Repeal and Re-enactment) Bill by the Senate of the Federal Republic of Nigeria. This Bill on Corporate and Individual insolvency seeks to provide for a harmonised code on the rehabilitation of the insolvent debtor as well as create the office of the Supervisor of Insolvency.

While there is every reason to applaud and express support for the proposed Bill that repeals and re-enacts the 37 years old Bankruptcy Act, the contradictions and inadequacies³³ observed in the proposed Bill cannot be ignored. Foremost, the issues and challenges working against a viable and credible corporate rescue and insolvency law and practice in Nigeria seems not to have been holistically dealt with in the Bill as corporate insolvency matters were not provided for. Thus, the quest to reform the corporate rescue and insolvency law and practice in Nigeria must be holistic, and, consistent.

Flowing from the statement of problem, the following research questions are elicited:

1. What are the extant law and procedure on corporate rescue and insolvency

in Nigeria?

2. What are the available corporate rescue mechanisms in Nigeria?

³²Insolvency Act 1986, Part 1 (Company Voluntary Arrangements (CVAs)), Part 2 (Administration) (United Kingdom); Companies Act 2008, Chapter 6 ss. 128-155 (South Africa); Bankruptcy Code, Chapter 11 (United States of America); The Insolvency and Bankruptcy Code 2016, Chapter 2, ss. 6-32 (Republic of India).

³³Agbajileke, *op cit.* Some of the provisions of the proposed Bill, particularly ss. 4, 5, 12, 13 and 68 purportedly contradicts AMCON Act ss. 51, 48, 35 (4) & 35 (2). Also, the inclusion of Dollars instead of Naira.

- 3. Are the extant corporate rescue and insolvency law and procedure in Nigeria adequate and effective for addressing corporate insolvency matters?
- 4. Are the extant law and procedure on corporate rescue and insolvency in Nigeria in line with international best practices and principles in resolving corporate insolvency?
- 5. What are the standards and qualifications regulating persons who practice as corporate rescue and insolvency practitioners in Nigeria?
- 6. What are the issues and challenges working against the orderly, effective and functional corporate rescue and insolvency in Nigeria?

1.3. Purpose of Study

The purpose of this study is to carry out a comprehensive examination of the extant corporate rescue and insolvency law and practice in Nigeria.

The specific objectives are to:

- Examine and analyse the extant corporate rescue and insolvency law and practice in Nigeria;
- 2. Evaluate the effectiveness or otherwise of the extant legal, regulatory and Institutional frameworks for corporate rescue and insolvency in Nigeria;
- Identify and examine the challenges and constraints to effective and functional corporate rescue and insolvency law and practice in Nigeria;
- Establish professional standards and benchmarks for regulation of corporate rescue and insolvency practitioners in Nigeria;
- Suggest reforms and new perspectives and approaches toward improving corporate rescue and insolvency law and practice in Nigeria.

1.4. Scope of Study

Winding up or liquidation³⁴ of a company as provided in the CAMA³⁵ can be triggered or activated if:

- The company has by a special resolution resolved that the company be wound up by the court;
- 2. Default is made in delivering the statutory report to the Corporate Affairs Commission (CAC) or in holding the statutory meeting;
- 3. The number of members is reduced below two;
- 4. The company is unable to pay its debts;
- 5. The court is of the opinion that it is just and equitable that the company should be wound up;
- The affairs of the company are being conducted in an unfairly, prejudicial, illegal or oppressive manner.³⁶

The scope of this study shall primarily be limited to winding up or liquidation of a company on the ground of inability to pay debts. This is ordinarily known as insolvency.³⁷ Thus, the scope of study is limited to discussion of winding up by the Court on the ground of a company's inability to pay debts or discharge obligations and creditors' voluntary winding up. In other words, members' voluntary winding up will not be examined in details in this study. Furthermore, this study shall be limited to the examination of corporate rescue and insolvency law and practice in Nigeria and other jurisdictions such as the United Kingdom, the Republic of South Africa, the Republic of India and the United States of America, with the aim of drawing from

³⁴N E Mustoe, *Bankruptcy Liquidation and Receivership* (London: Butterworth & Co (Publishers) Ltd, 1939) p. 157 – the word "Liquidation" is frequently used as a synonym for winding up.

³⁵CAMA, *op cit*, s. 408

 $^{^{36}}Ibid$, s. 312 (2) (a).

³⁷ *Ibid*, s. 409.

their peculiar experiences toward the quest for an effective and viable corporate rescue and insolvency law and practice in Nigeria.

1.5. Significance of Study

Every company needs capital to carry out its business or to function effectively, but as is most often the case; companies do not realize much of the needed capital from equity contribution. Thus, it becomes inevitable to resort to borrowed capital and in this instance; borrowers are often obliged to provide security for the repayment of their debts.³⁸

However, the extraction or placement of security does not completely protect the interest of the creditor or lender. This is because the mechanisms for the enforcement of the security so as to realize the borrowed funds seem to be contentious, complex, technical, if not cumbersome.

The benefits of having an effective and functional corporate rescue and insolvency law and practice for the growth of any economy and the attraction of foreign investment cannot be underscored. The Senate President of the National Assembly, Senator Saraki, while declaring open the Public Hearing on the "Bankruptcy and Insolvency (Repeal and Re-enactment) Bill 2015", assured that the reposition of the Nigerian economy to effectively meet the challenges of the 21st century was a major priority of the current Senate and that 'insolvency system and practice would play an important role in attracting both domestic and foreign investments as well as promoting investments and entrepreneurial development, adding that, given these opportunities, there is urgent need to repeal and re-enact the Bankruptcy Act which has become obsolete and out-dated'.³⁹

Therefore, the significance of this study are:

³⁸Davies and Worthington, *op cit*, p. 1207.

³⁹Agbajileke, op cit.

- To provide information and further insight into prevailing theories, models and principles that would engender a friendlier and dynamic corporate rescue and insolvency system in Nigeria;
- 2. To be a handbook or resource material for students, legal practitioners, magistrates, judges, governments, regulatory agencies, policy makers, legislators, companies, investors, creditors and debtors of contemporary information on best practices as relating to corporate rescue and insolvency law and practice; and
- To be an addition or contribution to the body of literature on corporate rescue and insolvency law and practice

1.6. Methodology

This study will apply the doctrinal research methodology.⁴⁰This research will review and analyse both primary and secondary sources of information. The primary sources of information is domestic statutes such as the Companies and Allied Matters Act;⁴¹Investment and Securities Act;⁴²Asset Management Corporation of Nigeria Act.⁴³Banks and other Financial Institutions Act,⁴⁴ Nigeria Deposit Insurance Corporation Act,⁴⁵ Bankruptcy and Insolvency (Repeal and Re-enactment) Bill⁴⁶and subsidiary legislation such as the Companies Winding Up Rules;⁴⁷ Securities and Exchange Commission Rules and Regulations⁴⁸ and Bankruptcy Rules.

 ⁴⁰ C Morris & C Murphy, *Getting a PhD in Law* (Oxford: Hart Publishing Limited, 2011) pp. 30 – 31; M
 O U Gasiokwu, *Legal Research and Methodology* (Enugu: Chenglo Limited, 2006) pp. 37, 81; A Taiwo,
 Basic Concepts in Legal Research Methodology (Ibadan: St. Paul's Publishing House, 2011) pp. 43 – 44.
 ⁴¹ Cap C20 Laws of the Federation of Nigeria 2004.

⁴² 2007.

⁴³ 2010 (as amended).

⁴⁴ Cap B3 Laws of the Federation of Nigeria 2004

⁴⁵ Cap N102 LFN 2004

⁴⁶ 2015

⁴⁷ 2001.

⁴⁸ 2013 (as amended).

Furthermore, this study undertakes an examination, review and analysis of relevant Statutes on corporate rescue and insolvency of other jurisdictions, such as the Companies Act;⁴⁹ the Insolvency Act;⁵⁰ the Enterprise Act,⁵¹ Small Business, Enterprise and Employment Act⁵² and subsidiary legislation such as the Insolvency Rules.⁵³ Also, Chapter 11 of the United States of America Bankruptcy Code, Companies Act and Bodies Corporate (Official Liquidations) Act,⁵⁴ Companies Act and the Insolvency Act⁵⁵ and Companies Act and the Insolvency and Bankruptcy Code⁵⁶ so as to draw from the international best practices on corporate rescue and insolvency embedded in them.

In addition to the primary sources of information, this study will rely on secondary sources of information that entails a review and analysis of judicial precedents, articles, textbooks, newspapers, academic and conference papers, reports and opinion as related to or connected with corporate rescue and insolvency law and practice.

1.7. Literature Review

We will review a plethora of literature relating to corporate rescue and insolvency law and practice from a theoretical and conceptual framework.

Korobkin, a protagonist of the 'Wider Interest Perspective' on corporate rescue and insolvency argued that, the preservation of employment should be an independent goal of corporate rescue. This perspective holds that in corporate rescue and insolvency proceedings, the law should seek to protect employment and general

 ⁴⁹ 2006 (United Kingdom).
 ⁵⁰ 1986 (as amended) (United Kingdom).

⁵¹ 2002 (United Kingdom).

⁵² 2015 (United Kingdom).

⁵³ 1986 (as amended 2010 & 2015) (United Kingdom).

⁵⁴ 2013 (Companies Act) & 1963 (Bodies Corporate (Official Liquidation) Act (Ghana).

⁵⁵ 2008 as amended 2011 (Companies Act) & 1936 as amended 2015 (Insolvency Act) (South Africa).

⁵⁶ 2013 (Companies Act) & 2016 (Insolvency and Bankruptcy Code) (Republic of India).

community interest as well as providing equality among creditors since a "rescued" or "rehabilitated" company brings benefit to a variety of external constituencies including the government in the form of more taxes; existing and future employees in the form of continued or new employment and /or higher salary levels; local or host communities in the form that the wealth in the locality is increased. Thus, the argument of the protagonist of this perspective is that it will not be unreasonable for secured creditors to bear some of the cost of rescue and reorganization of the insolvent company.⁵⁷

McCormack made a detailed comparative critique of the Anglo-American Corporate Bankruptcy Law when he challenged the prevailing view that United States Law in the area of corporate bankruptcy is "Pro-Debtor" and that of the United Kingdom Law is "Pro-Creditor" and concluded that there is a functional convergence in practice between the two jurisdictions instead of divergence. Although, McCormack acknowledged that corporate rescue model is different from business rescue model, he still held the view that the business rescue model still plays a larger role in corporate restructuring in the United States.⁵⁸

Jackson, a proponent of the 'Creditors' Bargain Theory' contended that the cost of reorganisation or corporate rescue and insolvency should not be imposed on secured creditors, who do not benefit from the reorganisation or rescue of an insolvent company, but instead, be imposed on the company itself and on unsecured creditors who may benefit therefrom. The creditors bargain theorist argued that corporate rescue mechanisms and insolvency proceedings exist solely for the protection and benefit of creditors and shareholders and the interest of employees,

⁵⁷ D R Korobkin, 'Contractarianism and the Normative Foundations of Bankruptcy Law' (1993) Tex. L. Rev, 541-554; G McCormack, *Corporate Rescue Law - An Anglo-American Perspective* (London: Edward Elgar Publishing, 2008) p. 18.

⁵⁸ McCormack, *ibid*.

suppliers, customers and communities, should be taken into account only to the extent that particular members of those constituencies are creditors with enforceable legal rights against company assets under general law.⁵⁹

The thrust of the creditors' bargain theory seem to be premised on a 'hypothetical agreement' that creditors would reach if they were to bargain amongst themselves before extending credit to the now insolvent company. Simply, the proponents of the creditors' bargain theory held the view that corporate rescue mechanisms and insolvency law should not create rights but instead, act to ensure that pre-existing rights are observed and enforced to the greatest extent possible.⁶⁰

However, Creditors' Bargain theory has been criticised as being notional, hypothetical and an abstraction. In this respect, Goode took a swipe on the theory in the following words:

> ... the parties presented are not in truth real world parties at all.... But what is the source of identification of these qualities and how do we know that they have any relationship with the qualities of the actual creditors? Since in practice creditors do not act collectively in taking credit decisions, we have no idea how they would proceed, or what range of factors they will take into consideration if they were coming together....⁶¹

The 'Procedural Theory' is significantly same as the 'Creditors' Bargain Theory' to the extent that it posits that the cardinal purpose of insolvency is to

⁵⁹ TH Jackson, 'Bankruptcy Non-Bankruptcy Entitlements, and the Creditors' Bargain' (1982) 91 (5) Yale Law Journal, 857-907; McCormack, *op cit*, p. 22.

⁶⁰ McCormack, *ibid*, p. 24.

⁶¹R Goode, *Principles of Corporate Insolvency Law*(4thedn, London: Sweet & Maxwell, 2011)pp 75-76.

maximise recoveries and benefits for those with rights against company's assets. The procedural theorist contended that the secured creditors' proprietary rights (Property rights) should not be sacrificed for the benefit of the unsecured creditors and that "where secured creditors are prevented from enforcing their collaterals during rescue proceedings without being provided with full compensation, the effect is not merely wrong, it is outrageous".⁶²

However, the postulation of the procedural theorist that the cardinal purpose of insolvency proceedings is to maximise returns to creditors by protecting preinsolvency entitlements has been criticised as there are certain issues that emerge in the course of insolvency proceedings and hence proper and right to be addressed at the stage of insolvency proceedings. For instance, issues of fraudulent and reckless trading,⁶³ acts of delinquent directors,⁶⁴ officers⁶⁵ and members⁶⁶ usually crop up in the course of winding up of a company and logically can only be addressed at that point and not before.

Korobkin, a proponent of the 'More Inclusive Bargain Model', posited a normative framework for insolvency law, hinged on a hypothetical bargain to be devised by the representatives of all interests that might be affected by a debtor's financial distress. The hypothesis is built on the assumption that all those involved in the bargain, including creditors, all know that they may be affected by the insolvency, without being aware, whether as a debtor, an unsecured creditor (whether contractual or involuntary), a secured creditor, an ordinary employee, a member of the

 ⁶² McCormack, *op cit*, p. 26.
 ⁶³ CAMA, *op cit*, s. 506.

⁶⁴ Goode, *op cit*, p. 82.

⁶⁵ CAMA, op cit, s. 507.

⁶⁶ *Ibid.* s. 508.

community that is otherwise unconnected to the debtor company or somebody in a different kind of relationship.⁶⁷

Mokal, the proponent of the 'Authentic Consent Model' (ACM), posits that although participation in the imaginary negotiation process is extended to parties other than creditors as argued in the 'More Inclusive Bargain Model', the focus on creditors is what makes insolvency law, special. Mokal developed the concept of "dramatic ignorance", which depicts the fact that the parties to the creditors' bargain are held to be ignorant not only of insolvency outcomes, but how other creditors would behave. Therefore, He of the assumption that all parties in insolvency proceedings are free and equal and enter into a bargain that is fair and just.⁶⁸

One of the implicit assumptions of the 'Creditors' Bargain Theory' is that there is a common pool of assets to which creditors have a pre-insolvency entitlement rights. This is not completely correct as in most insolvency proceedings, creditors are often settled from the anticipated stream of income produced by the on-going corporate entity. Hence, LoPucki, developed a 'Team Production Theory' of corporate reorganisation law and contended that corporate reorganisation is not a regulation imposed by Government but instead, an implicit agreement, under which creditors and shareholders agree to subsume their legal rights to the preservation of the company as a going concern.⁶⁹ The 'Team Production Theory' has been argued to be the theoretical underpinning of the United States of America Insolvency system.⁷⁰

⁶⁷ McCormack, *op cit*, p. 28.

⁶⁸R J Mokal, *Corporate Insolvency Law: Theory and Application* (Oxford: OUP, 2005) pp. 70-87.

⁶⁹ McCormack, *op cit*, p. 30; L M LoPucki, 'A Team Production Theory of Bankruptcy Reorganization' (2003) UCLAW Law School, Law and Econ Research Paper, pp.1-12; MM Blair & LA Stout, 'A Team Production Theory of Corporate Law' (1999) 85 Virginia Law Review,247.

⁷⁰ The features of Chapter 11 of the United States of America Bankruptcy Code which allows the Board of Directors to continue to manage the business of the company, unlike the United Kingdom, where Administration operates as a management displacement device with the administrator assuming the management task formally entrusted to the Board of Directors.

Warren and Finch, held the view that a single unifying theory of corporate rescue law, while intellectually and theoretically attractive, cannot adequately explain all the issues arising from corporate rescue proceedings. Finch contends that modern corporate insolvency law cannot be explained by a description of existing statutory rules and case law but by a critical appraisal of the concept of modern corporate law from a threefold aim, (i) to outline the law on corporate insolvency... and the procedures and enforcement mechanisms used in giving effect to that law...,(ii) to set out a theoretical framework for corporate insolvency law that will establish benchmarks for evaluating that law and any proposed reforms... and (iii) to move beyond an appraisal of current laws and processes and to consider whether any new approaches to insolvency institutions and rules are called for.⁷¹

Finch therefore propounded a theoretical framework, which she called the "Explicit Values Approach"⁷² or as the "Communitarian Theory", a broad-based, interdisciplinary approach that considers community and public interest in addition to the interest of creditors and set out benchmarks for evaluating the efficacy of corporate and insolvency rescue law, to wit; Efficiency, Expertise, Accountability and Fairness.⁷³ However, the 'Multiple Values or Eclectic Approach' has been criticised for being vague, uncertain and indeterminate.⁷⁴

Goode and Adebola pointed out that in corporate rescue of an insolvent company, it is the company's business that are capable of contributing value to the economy that is the real subject of rescue, not the entire company's shell. Simply put, corporate rescue is all about protecting businesses of an insolvent company, which

⁷¹ V Finch, *Corporate Insolvency Law: Perspective and Principles* (2ndedn, Cambridge: Cambridge University Press, 2009) p. 54.

⁷²*Ibid;* A Keay & P Walton, *Insolvency Law: Corporate and Personal* (Essex: Pearson Education Limited, 2003).

⁷³*Ibid*.

⁷⁴ McCormack, *op cit*, p. 35.

requires a '*hiving-off*' of potentially viable portions of the business for sale as a going concern to another company.⁷⁵ This, Business Rescue Model presupposes that the transfer of the business of a distressed and/or failing company to a financially healthy corporate entity may grant the once distressed or failing business a new opportunity to bounce back to buoyancy, if not, liquidity but may also enable a quick resolution of the failed company's distress while giving the previous owners a second chance to succeed at the business through a new corporate entity.⁷⁶

Mokal, stated that the preservation of the corporate entity, or some part, after distress has been remedied should be an objective of corporate rescue proceedings. Mokal posits that the corporate rescue of an insolvent company should not be restricted to the preservation of only the business of the insolvent company but also the preservation of the company as an entity. An arguable advantage of this Model is that by preserving the relationship of pre-distressed owners and managers, there could be initiation of rescue measures in a distressed company at the early stage.⁷⁷ However, the notion that rescue, even of the business, was more likely to succeed only when initiated early and that such early initiation depended on the actions of those in charge of the company, is without doubt, correct, as it accords with common sense that persons are unlikely to trigger rescue mechanisms if they would lose their control of the distressed entity during the process.⁷⁸

Frisby, drew a distinction between rescuing the company and rescuing the business of the company, thus:

⁷⁵Insolvency Law and Practice: Report of the Review Committee (The 'Cork Report') CMND 8558(London; HMSO, 1982).

⁷⁶ Goode, *op cit*, p. 81; B Adebola, 'A Few Shades of Rescue: Towards an Understanding of the Corporate Rescue Concept in England and Wales', a paper delivered at INSOL Academics' Conference, 9 - 12 October 2014, Istanbul (2014) 6.

⁷⁷Mokal, *op cit*, p. 210.

⁷⁸ Ibid.

The former, which might be described as "pure rescue" would involve the corporate entity emerging from the rehabilitation endeavour intact, so as to continue substantially the same operations, with the same work force and in the ownership of the same people. The latter is perhaps most accurately expressed as corporate recycling. The company's business or a viable part of that business is sold as a going concern to a third party.....⁷⁹

Frisby further elaborated on collectivity, fairness and corporate rescue as the ideals or characteristics of a good modern insolvency law.⁸⁰ Collectivity refers to the components of a regime that inhibits individual debt enforcement amongst creditors by the imposition of a mandatory moratorium on the bringing of legal proceedings or the enforcement of security interest or other property rights against the insolvent company during the currency of the insolvency procedure.⁸¹

Goode discussed the historical development of insolvency law, its meaning and proceeded in an illuminating and articulate manner to espouse the framework and fundamental principles of corporate insolvency law.⁸² Although, the discourse is situated on the corporate rescue and insolvency practice in the United Kingdom and the European Community, the principles of corporate insolvency set out by Goode, not only meets international best practices⁸³ but would undoubtedly be relevant in the

⁷⁹ S Frisby, 'In Search of a Rescue Regime: The Enterprise Act 2002', (2004) 67 (2) MLR, 248-249. ⁸⁰*Ibid*, 247.

⁸¹ *Ibid*, 249-250.

⁸² Goode, op cit, pp. 1-60; Cork Report, op cit.

⁸³ United Nations Commission on International Trade (UNCITRAL) Legislative Guide on Insolvency Law 2005.

quest for a reform of the corporate rescue and insolvency law and practice in Nigeria, which principles and rules are:

1. **Recognition of Pre-Insolvency Rights**

The general rule is that liquidation does not of itself terminate contracts or extinguish rights, though it may inhibit the pursuit of remedies. Hence, corporate rescue and insolvency proceedings respect the rights that accrued prior to the proceedings. In Cambridge Gas Transportation Corporation v Official Committee of Unsecured Creditors of Navigator Holdings Plc.,⁸⁴ the court held that "the important point is that bankruptcy whether personal or corporate, is a collective proceeding to enforce rights and not to establish them".

2. **Attachable Assets**

It is only the property of the company at the time of liquidation or winding up that comes into its possession thereafter that is available for its creditors. In view of this, assets held by the company on trust do not form part of its estate nor the goods supplied to it under reservation of title. Thus, it is only the company's assets that are available for its creditors. The liquidator has no power as a liquidator, to sell assets not beneficially owned by the company⁸⁵ or subject to a security interest, although, if the company is a trustee with management powers under an active trust, the liquidator may, with the consent of the beneficiaries or under an order of the court,⁸⁶ manage the trustee property on behalf of the beneficiaries and realize trustee assets as part of that management.

 ⁸⁴ [2007] 1 AC 508, per Lord Hoffman.
 ⁸⁵ CAMA, *op cit*, s. 423.

⁸⁶*Ibid*, s. 424

3. Subsistence of Security Interest and other Pre- Insolvency Real Rights A creditor holding security interest and other pre-insolvency real rights is unaffected by winding up and may proceed to realize his security or assert other rights to property as if the company were not in liquidation. This principle operates to the extent that were the instrument of charge provides for security over acquired property, the secured creditor can assert rights even over monies or other assets falling in after the commencement of the winding up, provided that the consideration for this was already executed before that time, as opposed to being furnished by the liquidator himself. For example, by performance of a contract entered into by the company.⁸⁷

4. Limitations and Defences to the Powers of the Liquidator

The liquidator in exercising rights in the name of the company stands in no better position than the company itself. He takes them as they stand, warts and all.⁸⁸ Simply, the liquidator takes subject to equities. However, it is important to note that the preliquidation right of set-off, which is also equity, stands as an exception to the general rule in that it is not exercisable against a party that has gone into liquidation.⁸⁹

5. Maintenance of the Anti-Deprivation Rule

In the course of insolvency proceedings, there shall not be, whether directly or impliedly, an activation of a contractual provision divesting the company of an asset it previously held. Such a provision contravenes the anti-deprivation rule in that its

⁸⁷ Re Collins [1925] Ch 556; Goode, op cit, p. 96.

⁸⁸ Perpetual Trustee Co. Ltd v B.N.Y Corporate Trustees Services Ltd [2010] Ch 347.

⁸⁹ Under the United Kingdom Laws, the pre-liquidation right of set-off is displaced by the Insolvency Rules of set-off. r. 2.85 (Administration) and r. 4.90 (Liquidation).

effect is to remove from the reach of the general body of creditors, an asset held by the company at the time of liquidation.⁹⁰

It is important to state that this principle is distinct from that of pari passu distribution, which is concerned in ensuring rateable distribution among creditors and to avoid transactions which, although not reducing the net asset value of the company, unfairly benefit one creditor at the expense of others. The anti-deprivation rule is aimed at transactions, which improperly reduce the value of the company's assets to the detriment of all unsecured creditors.⁹¹

6. The PARI PASSU Distribution Rule

This Rule is that all unsecured creditors are required to share and share alike in a common pool of assets. This principle is designed to ensure that one creditor does not get more than his fair slice of the pie.⁹²

However, the *pari passu* rule has been criticised as being irrelevant in corporate rescue and insolvency process as unsecured creditors usually receive little, if anything, by way of distribution of assets. Hence, unsecured creditors are not material and key players in corporate rescue and insolvency proceedings.⁹³

7. **Conversion of Personal Rights into a Right to Dividend in Insolvency** Proceeding

When a company becomes insolvent, so that there are insufficient assets to go round, the creditor with real rights in an asset (for example, a seller of goods under reservation of title, a mortgagee or a chargee of property) can remove it from the pool of general creditors, but the creditor with a purely personal right, although not deprived of the right itself, loses the ability to engage in individual pursuit of

⁹⁰ Goode, *op cit*, p. 98.

⁹¹Goode, *op cit*, p. 98.

⁹² CAMA, op cit, s. 494 (a); For the United Kingdom, see Insolvency Act (Voluntary Winding up) s. 107, Insolvency Rule (Compulsory Winding up) r. 4.181; Insolvency Rules (Administration) r. 2.69; Orojo, *opcit*, p. 499. ⁹³Mokal, *op cit*, p. 210.

satisfaction and is required to lodge his claim with the liquidator in what is called a 'proof of debt', and participates with other unsecured creditors in such *pari passu* distribution of dividend as the liquidator is able to make from realization of the company's assets after allowing for the expenses of the liquidation and the claims of preferential creditors.⁹⁴

8. Cessation of Beneficial Ownership of its Assets upon Liquidation

Although winding up does not of itself divest the company of legal title to its assets, it ceases to be the beneficial owner and the liquidator holds the assets on trust and to apply them in discharge of the company's liabilities in accordance with the statutory scheme of distribution.⁹⁵

This position is contrary to the position under the extant Nigerian law as winding up order does not *ipso facto* vest the company's property on the liquidator, unless and until, the liquidator applies to the court for a vesting order.⁹⁶

9. Acceleration of Creditors' Right to Payment

When a company goes into liquidation, its liabilities for payment of contingent or future debts or prospective debts, becomes fictionally accelerated as if it has already been earned by performance. This principle is predicated on the fact that a creditor is presumed to have an immediate right of proof of debts not already due to him but for those payable in the future or on a contingency.⁹⁷

⁹⁴ CAMA, op cit, s. 493; Companies Winding up Rule (CWR) 2001, r. 74-101; Orojo, opcit, p. 496.

⁹⁵Ayerst (Inspector of Taxes) v C and K (Construction) Ltd (1976) AC 167.

⁹⁶ CAMA, op cit, ss. 423 & 424. This also finds support in the Australian case of Commissioner of Taxation v Linter Textiles Australia Ltd [2005] 215 ALR 1.

⁹⁷ CAMA, *ibid*, s. 410 (1) (b); CWR, *op cit*, r. 84; Insolvency Rule, r. 4.94 & 12.3 (1).

While debts payable in the future has to be established in full, where the debts are not due by the date of dividends distribution, they are discounted⁹⁸ for the purpose(s) of computing the dividend.⁹⁹

10. Liability for Debts

It is a notorious principle of company law, particularly relevant to insolvency, that where a body corporate, contracts, it is her own contracts and not those of its members and in the same vein, liabilities and debts incurred under such contracts are the liabilities and debts of the body corporate,¹⁰⁰ for which the members do not even bore secondary liability, subject and only to the amount, if any unpaid on the shares respectively held by them.¹⁰¹ However, members of a company may incur liabilities where the company acts as their agent or were they undertake a collateral liability, for example, under a guarantee.¹⁰²

It is a universal axiom that theories, principles and policies do not have a life of their own and remain at the level of hypothesis until there is an application of such theories, principles and policies. Thus, this study will examine these theories, principles and policies reviewed in this work so as to evaluate how the extant laws and procedures on corporate rescue and insolvency in Nigeria have lived up to the ideals and standards propounded and contained in the theories, principles and policies.

 $^{^{98}}$ Insolvency Rule, Rule 11.13; CWR, *op cit*, r. 84. While under the CWR, 2001, the rate of discount (rebate of interest) is stipulated (5% per annum), it is left open under the Insolvency Rule of the United Kingdom (although the formula for calculation of the discount is provided).

⁹⁹ Insolvency Rule, *op cit*, r. 4.86 (1).

¹⁰⁰Union Bank of Nigeria Plc v Orharhuge [2000] 2 N.W.L.R (pt 645) 495 at 517; A.I.B Ltd v Lee & Tee Industries Ltd [2001]7 N.W.L.R (pt 819) 366.

¹⁰¹Salomonv Salmon & Co Ltd, supra; CAMA, op cit, s. 21 (1) (a).

¹⁰²AfriBank Nig. Ltd v Moslad Enterprises Ltd & Anor [2008] All FWLR (pt 421) 877. Instances of exception to this can be found under statutory lifting of the veil of incorporation. See CAMA, *op cit*, s. 506 (1); Failed Banks Recovery of Debts and Financial Malpractices in Banks Act, Cap F2, LFN 2004, s. 1 (3) (b).

Fletcher examined all aspects of insolvency law, individual as well as corporate and with a detailed discussion of the impact of the Insolvency Acts 1986 and 2000 (United Kingdom), the Enterprise Act, 2002 (United Kingdom), Cross-Border Insolvency Regulations, 2006 and the Human Rights Act, 1998 on corporate rescue and insolvency law and practice in the United Kingdom and Europe.¹⁰³

While Fletcher's work is expository on the historical development of insolvency law, the nature and incidence of insolvency, company insolvency procedures, nonliquidation and liquidation procedures and alternatives to winding up in the form of corporate rescue mechanisms in the United Kingdom, this study will attempt to transpose the essence of Fletcher's work into the corporate rescue and insolvency law and practice in Nigeria.

Payne critically examined the use of both member and creditor Schemes of Arrangement ('Schemes') as a corporate rescue mechanism and compared the advantages and disadvantages of Schemes with other corporate rescue mechanisms. Payne further discussed the areas and ways for reform of the law and practice of Schemes.¹⁰⁴

Payne's work is very illuminating on Schemes as a corporate rescue vehicle. However, as there are no "one fixes it all" corporate rescue mechanisms, his work is therefore limited as other corporate rescue mechanisms beside Schemes were not examined. Thus, this study will examine other available corporate rescue mechanisms aside Schemes.

Bachner examined the mechanisms for the protection of creditors of private companies from the risk of losing borrowed capital under English and German laws

¹⁰³ I F Fletcher, *The Law of Insolvency* (4thedn, London: Sweet & Maxwell, 2009) pp. 357-478, 493-670. ¹⁰⁴ J Payne, *Schemes of Arrangement, Theory, Structure and Operation* (Cambridge: Cambridge University Press, 2014) pp. 1-305.

through an illuminating discussion of creditor protection mechanisms such as avoidance of pre-insolvency acts, capital maintenance, unlawful distribution, directors' liability for contraventions of capital maintenance rules.¹⁰⁵ His exposition on credit protection mechanism seems defective as it appears a rehash of the principles of insolvency law and not an espousal of corporate rescue mechanisms. Hence this study will examine the available core corporate rescue mechanisms.

Parry carried out an in-depth analysis and digest of relevant case law and legislation on corporate rescue and insolvency in the United Kingdom by providing an outline of the main procedures for corporate rescue by way of Company Voluntary Arrangements (CVAs), Administration, Administrative Receiverships, Schemes of Arrangement; the structure and procedure of creditors' meetings; the main statutory provisions in the Insolvency Act 1986 (United Kingdom); the Insolvency Rules 1986 and the expected impact of the Companies Act 2006 (United Kingdom) on corporate rescue and insolvency law and practice in the United Kingdom.¹⁰⁶

While the concise and prosy style adopted in this work is quite commendable, the corporate rescue mechanisms and procedures examined in the work have undergone legislative changes, therefore this study will examine the thrust of the work in line with the legislative changes.

Lynch-Fannon and Murphy discussed the corporate rescue and insolvency procedures available in Ireland, including Examination, Receivership and Winding-up; the rights and liabilities of parties involved in winding-up and the concept of fraudulent and reckless trading.¹⁰⁷While Lynch-Fannon and Murphy made out an articulate exposition on the corporate rescue and insolvency system in Ireland, this

¹⁰⁵ T Bachner, Creditors Protection in Private Companies: The Anglo-German Perspective for a European Legal Discourse (Cambridge: Cambridge University Press, 2009) pp. 39-248.

¹⁰⁶ R Parry, *Corporate Rescue* (London: Sweet and Maxwell,2008) pp. 1-294.

¹⁰⁷ I Lynch-Fannon and G Murphy, *Corporate Insolvency and Rescue* (2ndedn, London: Bloomsbury Professional, 2012) Chapters 7-14.

study will examine the relevance of their work to the cause of corporate rescue and insolvency in Nigeria.

Oladele examined the current law on corporate restructuring and rescue in Nigeria with emphasis on mergers and take-overs as well as the rescue option of receivership. However, the non-discussion of or mention of Arrangements and Compromise as a corporate rescue option is a short fall to the obvious remarkable merits of the work.¹⁰⁸ While the scope of this work focused on the examination of mergers and take-overs as well as receivership as corporate restructuring mechanisms, in this study, other corporate rescue mechanisms such as arrangement and compromise, bail outs and regulatory rescue would be examined.

Okoli examined the circumstances and or definitions under section 409 of the CAMA regarding a corporate body being insolvent. Okoli argued that the said statutory grounds for regarding a corporate body as insolvent are by no means exhaustive as parties are entitled by mutual agreement(s) to insert other conditions under which a borrowing company can be regarded as insolvent.¹⁰⁹Curiously though, Okoli, suggested that lifting the veil of incorporation in order to fix personal liability on members and employees of an insolvent company is an unexploited option/strategy in addressing insolvency. Since there are already provisions in the CAMA for lifting the veil of incorporate rescue and insolvency law and practice in Nigeria.

Busari identified the different laws in Nigeria that apply to individuals and corporate bodies in cases of insolvency as the Bankruptcy Act 1979 for individuals,

¹⁰⁸ OO Oladele, 'The Legal Intricacies of Corporate Restructuring and Rescue in Nigeria' (2007) *the Calabar Law Journal*, Xi,136-153.

¹⁰⁹ H S N Okoli, 'Corporate Insolvency – Options and Strategies for Lending Institutions', January 1998, *MPJFIL* Vol. 2 No. 1, 55-64.

¹¹⁰ CAMA, op cit, s 506

and the CAMA for Companies. Busari, adopted the definition of insolvency as provided in Black's Law Dictionary and the CAMA, which simply means "the condition of a person or business that is insolvent, inability or lack of means to pay debts...", and proceeded to argue against the general belief that insolvency practice is only relevant when a company or a person is faced with financial difficulties and no more or when the existence of a company is threatened but is relevant in times of expansion or rationalisation or restructuring of a company or its business resources.¹¹¹

Although, Busariis without doubt genuinely interested in the introduction of a robust insolvency practice in Nigeria, he appeared to have misconstrued the fine boundaries between insolvency practice and corporate restructuring. Thus, this study will therefore introduce conceptual clarity to corporate rescue and insolvency law and practice in Nigeria.

Asomugha examined the functions of the Court in schemes of arrangements, reconstructions and mergers under the Nigerian Law and drew the conclusion that while the Courts will still continue to play a prominent part in matters relating to schemes of arrangement, mergers and take-over, notwithstanding the powers now conferred expressly on the Securities and Exchange Commission, the truth is that court procedure is ill-equipped to properly pass judgment on the economic merits of such schemes of arrangement.¹¹² Although, Asomugha, rightly discussed the short comings of the role of the court in corporate rescue and insolvency proceeding in Nigeria, he failed to proffer solutions or an alternative to the court in corporate rescue

¹¹¹ T Busari, 'A Case for Insolvency Law and Practice', October 2000, *MPJFIL* Vol. 4 No. 4, 266-296. ¹¹² E M Asomugha, 'A Re-Appraisal of the Functions of the Court in Schemes of Arrangements, Mergers and Take-Overs' (1991) JUS Vol. 2 No. 1, 41-53.

and insolvency proceedings in Nigeria. This study will therefore suggest alternatives to the courts in corporate rescue and insolvency in Nigeria.

Mayson *et al*, described the nature of company insolvency and liquidation procedures as obtainable in the United Kingdom by outlining the legislation on company insolvency and liquidation. The author gave a summary description and definition of corporate rescue mechanisms such as an administrative receiver, administration, voluntary arrangements and analysing moratorium clauses and the necessary qualification to act as an insolvency practitioner.¹¹³ While this work is very instructive and elucidative on corporate rescue and insolvency law and practice in the United Kingdom, some of the statements on the position of the law have been subjected to legislative repeals, amendments or modifications. Thus, this study will examine the contemporary and extant corporate rescue and insolvency law and practice in the United Kingdom.

Farrar & Hannigan did not limit their articulate discussions on corporate rescue and insolvency to just a description of its nature and procedure, definition and explanation of some of the corporate rescue and insolvency concepts in the glossary of terms but elegantly outlined and discussed the causes or theories of business failure.¹¹⁴ However, Farrar and Hannigan' work did not incorporate the recent modifications or amendments to the corporate rescue law and practice in the United Kingdom. Hence, this study will examine the corporate rescue and insolvency system in the United Kingdom in line with the extant legislation.

Hicks & Goo described the sources, scope and development of corporate insolvency law in the United Kingdom; the purpose of corporate insolvency law, the

¹¹³ S Mayson, *et al, Mayson, French and Ryan on Company Law* (25thedn, Oxford: OUP, 2008) pp. 627-683.

¹¹⁴ J H Farrar & B M Hannigan, *Farrar's Company Law* (4thedn, London: LexisNexis ButterworthsTolley, 1998) pp. 621-746.

requisite qualification to act as insolvency practitioner, the statutory definitions of insolvency, which definitions, they rightly argued were not comprehensive.¹¹⁵ Although this literature was situated in the 1990's, the contents remain seminal, contemporaneous and illuminating on the subject of corporate rescue and insolvency law and practice. However, the principles, statutory provisions and conclusions should be understood and applied with caution because of the subsequent modifications or abrogation of the law relied upon at the time of publication of their work. Hence this study will examine the corporate rescue and insolvency system in line with contemporary realities and statutory provisions in the United Kingdom.

Sealy & Worthington relied on extant legislations and appropriate case laws to succinctly discuss the nature of corporate rescue and insolvency procedures, analyse the mechanisms and concepts on corporate rescue and insolvency as obtainable in the United Kingdom.¹¹⁶ While the content of their work is robust, remarkable and profound, this study will proceed to transpose the narration in their work to the corporate rescue and insolvency environment in Nigeria.

Bourne discussed the nature and features of corporate rescue mechanisms such as Receiverships, Voluntary Arrangement and Liquidation generally and argued that where the receivership is sought to enforce the terms of a debenture in the case of a default, the appropriate remedy would be generally to secure the appointment of a receiver, who need not be a qualified insolvency practitioner but where the receivership is to appoint a person under the terms of a floating charge, that person so appointed is generally an administrative receiver, who must be a qualified insolvency

¹¹⁵ A Hicks & S H Goo, *Cases & Materials on Company Law* (2nd edn, London: Blackstone Press Limited, 1997) pp. 615-705.

¹¹⁶ L Sealy & S Worthington, *Cases and Materials in Company Law* (10th edn, Oxford: OUP, 2013) pp. 708-825.

practitioner.¹¹⁷ Furthermore, Bourne defined voluntary arrangement as 'a simple procedure whereby a company which is in financial difficulty may enter into a voluntary arrangement with its creditors, which arrangement may involve either a composition in satisfaction of its debts, that is, provision for creditors to receive a percentage of what is due to them, or a scheme of arrangement of its affairs', and defined Administration as 'a procedure to facilitate the rehabilitation or reorganisation of a company by placing the management of an insolvent company in the hands of an administrator and that for as long as the administration is in force, it is not possible to commence winding up proceedings or any other process against the company or to enforce any charge, hire purchase or retention of title provisions against the company without the leave of the court'.¹¹⁸

While Bourne's work remains seminal and elucidating on corporate rescue mechanisms in the United Kingdom, this study will examine and analyse the impact of subsequent legislation on some of the postulation in the work as it relates to corporate rescue and insolvency proceedings in the United Kingdom and the relevance to the development of an effectual corporate rescue and insolvency law and practice in Nigeria.

McLaughlin defined the following concepts: Insolvent winding up to mean when a company is unable to pay its debts; Secured Creditors are those who, in addition to having a contractual right to sue the company for the return of any money owed to them, have taken a property interest in one or more items of the company's property as security for the credit they have made available to the company. Remarkably though, McLaughlin examined loan security in details by putting out

¹¹⁷ N Bourne, *Principles of Company Law* (3rdedn, London: Cavendish Publishing Limited, 1998) pp. 243-286.

¹¹⁸N Bourne, *op cit*, pp. 251, 253.

classification according to a number of criteria, namely, mortgages and charges; legal and equitable property interests and fixed and floating charges and further argued that while a distinction is often made in property law between a mortgage and a charge, this distinction is not usually important in company law as the term 'charge' is regularly used to describe a loan security whether the security established is a charge or a mortgage; analysed and explained concepts or phrases such as 'Avoidance of transactions', 'misfeasance by officers of the company', 'wrongful trading' and 'fraudulent trading'.¹¹⁹

Unfortunately, McLaughlin's discourse did not embrace any of the corporate rescue mechanisms. Hence, our discussion in this study will include corporate rescue mechanisms and procedure.

Ajogwu, examined the role of mergers and acquisitions to the growth and health of an economy by being vital tools used by companies for one purpose of expanding their business operations with objectives ranging from increasing their size, long term profitability and relevance within a particular market. Ajogwu stated that although 'mergers' and 'acquisition' are often uttered in the same breath and used as though they were synonymous, the terms mean slightly different things. Thus, Ajogwu, adopting the provision of the Investment and Securities Act 2007 defined a Merger as 'the amalgamation of the undertakings or any part of the undertakings or interest of two or more companies or the undertakings or any part of the undertakings or interest of one or more companies and one or more bodies corporate'. But, 'acquisition' he describes as 'the act of gaining effective control over the assets or management and ownership (of shares in the capital) of another company without any combination of companies', and further posit that 'acquisition' is sometimes referred to as 'take-

¹¹⁹ S McLaughlin, Unlocking Company Law (London: Hodder Education, 2009) pp. 398-434.

over'. Ajogwu further elaborated on the categories of mergers and the threshold thereof.¹²⁰

While the explanation of the concept of 'mergers' and 'acquisitions' is apt and commendable, the definition and description of the nature and procedure of acquisition, categories of merger and the thresholds seem not in line with the current position of the law. This study will therefore fill in the gap on mergers and acquisitions by reviewing and analysing the provisions of the extant legislation and rules¹²¹ as relating to mergers and acquisition in Nigeria.

Sofowora argued that a company having been created by a legal process, that is, by registration, can be brought to an end by a legal process too and further stated four methods of dissolving a company, namely, Winding up, Mergers and Take-up, Reconstruction or Re-organisation and Striking off a defunct company by the Corporate Affairs Commission.¹²²

Although, Sofowora provided in a nutshell the procedure for winding up of companies in Nigeria, the contention that winding up, mergers and take-overs means ipso facto the dissolution of a company, seems incorrect. This study will therefore attempt to show that it is not in all cases that winding up or mergers and take-overs ultimately lead to the dissolution of a company.

Barber in a simple and erudite manner, defined a scheme of arrangement as 'a flexible and versatile method of resolving problems of variations of rights attached to shares, which cannot be dealt with under variation of class rights',¹²³ Also, Barber defined voluntary arrangement as 'a composition in satisfaction of its debts or scheme

¹²⁰ F Ajogwu, 'Mergers and Acquisition Opportunities and Pitfalls' Unib Law Journal VOL. 1, NO 1, OCTOBER 2011, 1-35.

¹²¹ Securities and Exchange Commission Rules and Regulations 2013, Rule 433 defines 'Acquisition' to mean where a person or group of persons buy most (if not all) of a company's ownership stake in order to assume control of the target company; Rules 434-439 (Procedure for acquisition).

 ¹²² M O Sofowora, *Modern Nigerian Company Law* (Lagos: Soft Associates, 1992) pp. 505-533.
 ¹²³ S Barber, *Company Law* (4thedn, London: Old Bailey Press, 2003) p. 303.

of arrangement of its affairs'.¹²⁴ Beyond the definition of scheme of arrangement, Voluntary arrangement and Administration, Barber, aptly discussed the advantages and disadvantages of the respective corporate rescue mechanism of scheme of arrangement, voluntary arrangement and administration and the qualifications to act as an insolvency practitioner.¹²⁵

Although Barber's espousal on corporate rescue and insolvency law and procedure is very enlightening, it was predicated on the Insolvency Act 1986, Enterprise Act 2002 and Companies Act 2006, all laws of the United Kingdom and would therefore have no relevance to corporate rescue and insolvency law and practice in Nigeria if same is not transposed to the Nigerian environment. Thus, this study will attempt to knit the very defining thoughts of Barber and the legal framework of corporate rescue and insolvency law and practice in Nigeria.

Solomon and Palmiter discussed charter amendments, recapitalisations, corporate combinations and dissolution that are often referred to as "organic changes" in the United States of America, which organic changes can be internal or external. Solomon and Palmiter described recapitalisation as an alteration to the corporation's capital structure which can be accomplished by merger. Significantly, they argued that re-capitalisation supplies a flexible tool for adjusting the corporation's financial direction, sometimes making new financing possible or offering shareholders an improved package of rights.¹²⁶ Solomon and Palmiter defined some dissolution terminologies¹²⁷ such as "dissolution" as the formal extinguishment of the corporation's legal life; "liquidation" as the process of reducing the corporation's assets to cash or to liquid assets after-which the corporation becomes a holding shell,

¹²⁴*Ibid*, p. 314.

¹²⁵S Barber, *op cit*, pp. 315, 318-319, 323-324.

¹²⁶ L D Solomon & A R Palmiter, *Corporations Examples and Explanations* (2ndedn, Toronto: Little, Brown & Company, 1994) pp. 520-542.

¹²⁷ *Ibid*, p. 520.

"winding up" as the whole process of liquidating the assets, paying off creditors, and distributing what remains to shareholders.

Against this background, they argued that "although dissolution results in the death of the corporation, it does not necessarily mean the death of the business. Because of its going-concern value, the business (meaning its assets) usually would be sold intact and dissolution simply serves as a redistribution device that follows the sale of assets to an outside party or to a shareholder faction. In dissolution, shareholders receive a pro rata share from the proceeds of the sale".

Solomon and Palmiter' work is very illuminating on the explanation of corporate rescue and insolvency concepts in America. However, the meanings ascribed to some of the concepts, for example, dissolution differs from the definitions and interpretations under the extant laws in Nigeria. This study will therefore provide conceptual clarity to these concepts.

Vagts' work is quite profound on the question of fairness in the consideration of mergers and other basic changes.¹²⁸ Vagts contended that the question of fairness can be looked at from different aspects. Firstly, the financial issue, that is, are the shareholders in each corporation (or in each class) being given a fair equivalent in value to what they are giving up? This is otherwise known as the appraisal problem. Secondly, is the issue of retroactivity and finally, the aspect of motivation, upon which he posed the question, "if the controlling group in the corporation comes into a change with the sole purpose of eliminating or neutralising an awkward minority, should that motivation initiate the transaction even if the minority receives a payment reasonably equivalent to the value of the investment they are forced to surrender?¹²⁹

¹²⁸ D F Vagts, Basic Corporation law: Materials-Cases-Text (10thedn, New York: The Foundation Press, Inc, 1989) pp. 684-725. ¹²⁹*Ibid*, p. 725.

Vagts' work is seminal. However, the analysis of the question of fairness and appraisal remedy was limited to the issue of merger and not to any other corporate rescue remedy such as arrangement and compromise as is the case in the extant law in Nigeria where the question of fairness comes to bear in the determination of the Court's sanction of the scheme of arrangement and compromise.¹³⁰ Thus, this study will examine the question of fairness as it relates to arrangements and compromise in Nigeria.

Ford and Austin discussed the advantages of debt finance in raising capital for a company.¹³¹ Ford and Austin drew a contrast between a mortgage and security interest in the form of a charge, in that, whereas in a mortgage, the mortgagor transfers, or is treated as transferring the ownership to the mortgagee, a charge does not contemplate such a transfer. A charge is a form of hypothecation rather than assignment. The owner of the property who creates a charge over specific property subjects that property, which he or she continues to own, to a burden in favour of another person. But a 'charge' in company law, is defined as any form of security for discharge of a debt.¹³² In other words, a charge in company law could be a mortgage (either legal or equitable) or lien and pledges or pawns. Ford and Austin further discussed reconstructions and amalgamations in the form of schemes of compromise and arrangement. Reconstruction, they defined as a process whereby a scheme is used to transfer the undertaking, assets and liabilities of a company to a new company on terms that shares in the old company will be cancelled and shares in the new company

¹³⁰ CAMA, *op cit*, ss. 539 – 540.

¹³¹ H A J Ford & R P Austin, Ford's Principles of Corporations Law (6thedn, London: Butterworth & Co (Publishers) Ltd, 1992) pp. 674-960. ¹³² H A J Ford & R P Austin, *op cit*, p. 326.

will be allotted to former holders in proportion to their former holdings. This is akin to Arrangement under Sale.¹³³

While Ford and Austin did examine scheme of arrangement under section 411 of the Corporations Law of the United States of America, outlined the distinct features of moratorium, role of insolvency practitioners, this study will necessarily examine scheme of arrangements as presently provided for under the extant laws in Nigeria and evaluate the adequacy of the provisions and suggest reforms, if any.

Baxt and Fletcher defined the phrase, 'a company is unable to pay its debts' by looking at the provisions of s 460 (2) of the Corporations Law of Australia, 1989, which is in *pari materia* with s 409 of the CAMA. In the work, Baxt and Fletcher, set out the number of matters that needs to be taken into account in establishing that a company is insolvent for inability to pay its debt, that is, the person making the demand must be able to give a valid discharge of the debt, the debt must be due and owing at the time and the company's liability for the debt must not be in dispute.¹³⁴

Their examination of the remedies or preventative actions which may be taken where companies find themselves in difficulties are noteworthy. Amongst the remedies inspections, receiverships are investigations and and official management.¹³⁵ According to Baxt and Fletcher, 'Official management' in Australia, ^c contemplates being given a breathing space or moratorium in which to rehabilitate itself and meet the claims of its creditors.

Baxt and Fletcher's work is a very illuminating, simple and concise statement of the corporate rescue and insolvency law and practice in Australia. However, the authors pointed out the limits of their work when they stated that 'at a time of

¹³³ CAMA, *op cit*, s. 538.

¹³⁴ R Baxt and K Fletcher, Afterman and Baxt's Cases and Materials on Corporations and Associations (6thedn, London: Butterworth & Co Publisher Ltd, 1992) pp. 809-928. ¹³⁵ *Ibid*, p. 860.

financial difficulty. The Australian Law Reform Commission recommended a new form of voluntary administration for insolvent companies (see Report No. 45, para 57) to be accompanied by the removal of the official management'. Hence, this study will not only domesticate but equally update Baxt and Fletcher's literature on corporate rescue and insolvency law and procedure.

Davidson, et al, examined the alternatives to straight bankruptcy in the United States of America and argued that while chapter 7 of the Bankruptcy Reform Act of the United States of America deals with liquidation proceedings, commonly referred to as 'a straight bankruptcy'; Chapter 11 of the same Bankruptcy Reform Act provides for a bankruptcy proceedings known as 'reorganisations' used by debtors to avoid liquidations.¹³⁶ Furthermore, they pointed out that one of the major advantage of a reorganisation is that it allows a business to continue.¹³⁷

Davidson, et al, did not only state that individuals can avail of chapter 11 Bankruptcy proceedings but in a concise, simple and straightforward approach, discussed the chapter 11 proceedings from filing the petition; entering an order for relief; appointment of a trustee or an examiner in the unlikely event that no one asks of the appointment of a trustee; the functions of a trustee or examiner, which usually is to investigate the debtor and the debtor's business activities and business potential; make recommendation to the Court, either in the form of a reorganization plan or a transfer from Chapter 11 to Chapter 7. They further examined the features of Chapter 11 bankruptcy proceedings in the United States of America such as the debtor being in possession where no interested party asks for the appointment of a trustee; the appointment of credit committees; the repayment plan and the role of the court which

¹³⁶ D V Davidson, et al, Business Law: Principles and Cases (4thedn, California: Wadsworth Publishing Company, 1993) pp. 630-643. ¹³⁷ *Ibid* p. 631.

highlight is that once the court approves a plan, it becomes binding on everyone affected by it.

Davidson *et al* work is remarkable. However, this study will attempt to switch the features of the chapter 11 bankruptcy proceedings to corporate rescue and insolvency law and practice in Nigeria.

Jennings defined a 'security' as a contract, transaction, or scheme whereby a person invests his money in a common enterprise and is led to expect profits solely from the efforts of a promoter or a third party and in the discussion of types of business combinations, Jennings, defined a merger as a combination of two or more corporations, after which only one corporation continues to exist and outlined the features of three types of mergers, namely, horizontal, vertical and conglomerate.¹³⁸

Although, Jennings's, effort to examine the concepts of security, merger and types of merger stands as a laudable contribution to the body of literature on corporate rescue and insolvency, the work is flawed for being sketchy and not comprehensive enough on the examination of the concepts set out to be explained. In this study, the gaps observed in Jennings' work would be filled up.

Keenan examined receiverships and administration generally in the United Kingdom; the appointment of an administrative receiver; qualifications and procedure for appointment of administrative receiver; effect of appointment of receiver; duties of a receiver; the effect of a winding up on a receiver; methods of appointment of an administrator; purpose of administration; powers and duties of an administrator.¹³⁹ Although Keenan's contribution to the literature on receiverships and administration cannot be ignored, the legal justifications and conclusions in the work were premised

¹³⁸ M Jennings, *Business, Its Legal, Ethical and Global Environment* (4thedn, Ohio: International Thomson Publishing, 1997) pp. 758-819.

¹³⁹ D Keenan, *Smith & Keenan's Company Law* (7thedn, London: Pitman Publishing, 1987) pp. 425-502.

on repealed and otiose laws.¹⁴⁰ Thus, this study will examine and review the extant legislations that did not form the contents of Keenan's work.

Clark *et al* examined the corporate rescue and insolvency law and procedure in the United States of America through a methodical step by step outline of the procedures for straight Bankruptcy under Chapter 7 of the Bankruptcy Code (that is, winding up in the CAMA) and reorganisation under Chapter 11 of the Bankruptcy code.¹⁴¹ More incisively, this work is a reference guide¹⁴² on chapter 11 Bankruptcy procedure of the United States of America. However, this study will examine the potentials of the Chapter 11 Bankruptcy proceedings being integrated into the corporate rescue system in Nigeria.

Beatty & Samuelson set out an incisive overview of the Bankruptcy Code of the United States of America that is structured into 8 Chapters with odd numbers except one, while Chapters 1, 3 and 5 provide for administrative rules that generally apply to all types of bankruptcy proceedings; Chapters 7, 9, 11, 12 and 13 provide the substantive rules for the different types of Bankruptcies, which objectives can either be for rehabilitation or liquidation. Beatty and Samuelson submitted that the objectives of chapter 11 and 13 are to rehabilitate the debtor and Chapter 7 provides for liquidation, also known as "Straight Bankruptcy".¹⁴³

Beatty & Samuelson, further contend that in Bankruptcy proceedings under the Bankruptcy Code, debtors are eligible to file under more than one chapter and that no choice is irrevocable as both the debtors and creditors have the right to ask the Court to convert a case from one chapter to another at any time during the proceedings.

¹⁴⁰ Enterprise Act 2002 has limited the use of Administrative Receiver and also amended the provisions of the Insolvency Act 1986 as it relates to Administration.

¹⁴¹ L S Clark, *et al, Law and Business: The Regulatory Environment* (4thedn, New York: McGraw-Hill, Inc, 1994) pp. 953-1000.

¹⁴² *Ibid* pp. 994 – 995.

¹⁴³ J F Beatty & S S Samuelson, *Business Law for New Century* (New York: Little, Brown and Company, 1996) pp. 623-645.

Beatty & Samuelson's work is remarkable and commendable for the depth of research and analysis, the coloured highlights and striking style of language. However, this study will examine the implication of Chapter 11 procedure in the United States of America on the quest for reform of corporate rescue law and practice in Nigeria.

Donnell et al examined insolvency and reorganisation measures in the United States of America.¹⁴⁴ Donnell et al posited that 'debtors are considered insolvent if they are unable or fail to pay their debts as they become due'.¹⁴⁵ They discussed on the types of bankruptcy proceedings under the Bankruptcy Code of the United States of America, liquidation proceedings, which are usually begun by the filing of a petition, whether voluntary or involuntary. Liquidation proceedings by voluntary petition they argued is not necessarily filed by a person who is insolvent. However the person must be able to allege that he or she has debts.¹⁴⁶

Furthermore, Donnell et al examined Chapter 11 of the Bankruptcy Act, a procedure whereby the debtor's financial affairs can be reorganized, rather than liquidated under the supervision of the Federal Bankruptcy Court in the United States of America. They posited that Chapter 11 proceedings are available to virtually all enterprises, including individual proprietorships, partnerships and business corporations (except Banks, Savings and Loan Associations, insurance companies, commodities brokers and stock brokers), which chapter 11 proceedings are begun by petition filed either voluntarily by the debtor or involuntarily by its creditors. Donnell et al work suffers from the same deficiency as Beatty and Samuelson' work.

¹⁴⁴ JD Donnell, et al, Law for Business (Rev edn, Illinois: Irwin Inc, 1983) pp. 709-724.

¹⁴⁵*Ibid*, p. 709. ¹⁴⁶*Ibid*, p. 711.

Morse discussed the new frame work for corporate insolvency with aims or themes focused on the tightening up of the controls on the abuses of corporate insolvency system, the need for genuine alternatives to the drastic remedy of liquidation for an insolvent Company, an explanation of the regulation of insolvency practitioners, the concept of wrongful trading, receiverships, administration and voluntary arrangements and winding up by the Court.¹⁴⁷ While Morse's writing is quite expository or illustrative on conceptual terms in corporate insolvency in the United Kingdom, the subsequent amendments, if not repeal of the Laws relied upon in the work, vitiated the relevance of some of the postulations. This study will further contribute to the body of literature on corporate insolvency through an examination of the extant insolvency legislation that did not form part of Morse's writing.

Bhadmus defined corporate restructuring as "the process of rearranging one or more aspects of the company and that Restructuring happens in a company when there is a significant modification made to the debt, operations or structure of a company'.¹⁴⁸ Bhadmus simplified the options for corporate restructuring into two that is, internal and external and did contend that the idea behind internal corporate restructuring is to keep the company as a going concern in any way ... as the focus is on survival.

Bhadmus' writing is elucidative on the concepts of arrangement and compromise, merger and other corporate restructuring options as the expositions were predicated on current and up to date statutory provisions. However, Bhadmus' writing on corporate restructuring options were neither backed with rigorous and in depth legal analysis. This study would therefore subject the law and procedure on

¹⁴⁷ G Morse, *Charlesworth & Morse: Company Law*, (15thedn, London: Sweet & Maxwell Limited, 1996) pp. 657-775.

¹⁴⁸ H Y Bhadmus, *Bhadmus on Corporate Law Practice* (2ndedn, Enugu: Chenglo Limited, 2013) pp. 545-585.

arrangement and compromise, merger and takeover in Nigeria to rigorous academic scrutiny with the intention of formulating a functional corporate rescue law and procedure.

Bhadmus in another work examined in a succinct manner the concept of corporate rescue and the procedure for winding up of insolvent companies. Remarkably, Bhadmus devoted a chapter of the book, albeit one and a half page, for discussion on the need for a new legal framework for corporate insolvency. He opined that the extant corporate insolvency law in Nigeria does not secure the rescue of a company as a going concern. Without gainsaying the laudable contribution of this work to understanding of corporate rescue and practice in Nigeria, his recommendations for a new legal framework for corporate insolvency appears not indepth and embracive. Hence, this study will further make recommendations towards the reform of corporate rescue and insolvency in Nigeria.¹⁴⁹

Ogbuanya in profound discussion on corporate restructuring stated that corporate restructuring options can either be internal or external to the company or a combination of both. Ogbuanya identified the major internal corporate restructuring options as Arrangement and Compromise, Arrangement on Sale and Management Buy-out and some of the main types of external restructuring options being Merger and Acquisition, Take-over, External Restructuring, Purchase and Assumption and Cherry Picking.¹⁵⁰ The work is explicative of corporate restructuring in Nigeria. However, the work seems defective for lack of rigorous analysis.

Chianu critically examined winding up on the ground of a company being unable to pay its debts and explained concepts in Insolvency Law and Practice such

¹⁴⁹ H Y Bhadmus, *Corporate Insolvency Law and Practice in Nigeria* (Enugu: Chenglo Limited, 2009) pp. 1 – 304.

¹⁵⁰ N C S Ogbuanya, *Essentials of Corporate Law Practice in Nigeria* (2ndedn, Lagos: Novena Publishers Ltd, 2014) pp. 579-603.

as the definition of the phrase, 'a Company being unable to pay its debts; "Who may Petition for Winding up"; "Debt"; "Demand Notice"; "Court-Ordered Settlement"; "Judgment Creditor"; "Contingent or Prospective Liability" and "Judge's Discretion".¹⁵¹Remarkably, having adopted the definition of 'winding up for inability to pay debt' as stated in Section 409 of the CAMA, Chianu proceeded to define 'debt' as 'a definite sum of money fixed by the agreement of the parties as payable by one party in return for the performance of a specified event or condition'.

Thus, Chianu rightly argued that a winding up petition cannot be based on a debt that is bona fide disputed. Chianu, further examined the provisions of Section 409 (c) of the CAMA by defining 'contingent or prospective liability' to mean commercial insolvency, that is, where a company may have wealth locked up in investments not presently realisable, but although this be so, yet it has no assets available to meet its current liabilities. Hence it is commercially insolvent and may be wound up. Against the backdrop of the negative repercussions of winding up for inability to pay the debt on the insolvent company, Government and society at large, Chianu argued for a Court ordered settlement that is activated by the Federal High Court exercising its jurisdiction¹⁵² to encourage and facilitate the amicable settlement of disputes by urging petitioners in a winding up petition to withdraw the petition and consider settling its affairs with the company. Unfortunately, this contention seems not to represent the correct position of the law. Alternative dispute resolution cannot under any guise be foisted on parties as the unanimous pivot on which any of the alternative dispute resolution method stands is the voluntary agreement or consensus of the parties.

¹⁵¹ E Chianu, *Company Law* (Abuja: LawLords Publications, 2012) pp. 626-637.

¹⁵² Federal High Court Act, cap F12, Laws of the Federation of Nigeria 2004,s. 17.

While the depth of analysis in Chianu' work is profound and remarkable, the definition of insolvency particularly in section 409 of the CAMA still remained foggy and the argument for a Court ordered settlement as a corporate rescue mechanism is still very much inchoate. Thus, this study will provide a better understanding of the concept of insolvency and comprehensively discuss effectual corporate rescue mechanisms.

Orojo discussed in details the law and procedure on Corporate Insolvency, Receivership, Arrangement and Compromise, and Mergers and Take-overs in Nigeria. While stating the essential ingredients or conditions for a Court to make an order of winding up a company, that is:

- 1. There must be a debt;
- 2. Which is due, and,
- 3. The company is unable to pay the debt. 153

Orojo contended that "once these conditions are satisfied, the company will be wound up". This postulation seems not the correct position under Nigerian law.¹⁵⁴ Chianu had argued to the contrary thus "… in essence, there is no inflexible obligation of a Court to make a winding up order merely because one of the three conditions set out in the section is fulfilled".¹⁵⁵

Although, this work is a valuable compendium on corporate rescue and insolvency law and practice in Nigeria, it appears bereft of rigorous and in-depth legal analysis¹⁵⁶ and most of the legislation relied upon appears out-dated. This study will therefore carry out an in-depth analysis and discussion of the corporate rescue and insolvency concepts in line with the extant laws in Nigeria.

¹⁵³ J O Orojo, *Company Law and Practice in Nigeria* (5thedn, London: LexisNexis ButterworthsTolley, 2008) pp. 435-503, 505-575.

¹⁵⁴ CAMA, *op cit*, s. 411 (1) – powers of the Court.

¹⁵⁵Chianu, *op cit*, p. 636.

¹⁵⁶ Securities and Exchange Commission Rules and Regulations 2013 as amended 2015.

Akume examined the causes of corporate collapse from an economic perspective and a legal perspective.¹⁵⁷Akume posits that economic perspective views corporate collapse as being a result of the unproductivity and unviability of the company and the legal perspective in the case of winding up of a company considers the interest of the investors and creditors and whether the company is buoyant or not.¹⁵⁸ Furthermore, Akume contends that in view of the colossal negative impact of corporate failure on the totality of interest involved in the existence of the company, 'winding up should be a matter of last resort where all the other alternatives such as schemes of arrangements, mergers and take-overs, receiverships, etc are inadequate in the circumstances'.¹⁵⁹

However, Akume did not in the work examine or attempt to examine any of the other alternatives as outlined in his work. Hence, this study will therefore examine in detail all the available alternatives to winding up as mentioned.

Abdulai discussed mergers, acquisition and take-overs as a mechanism by which companies that are under threats of foreclosure could be "bailed out" by a willing purchaser, thus, absolving them of liabilities to their bankers, ensuring labour retention and with the shareholders ending up with some financial consideration for their shares.¹⁶⁰ Furthermore, Abdulai analysed the constraints to merger, acquisition and take-overs in Nigeria such as the absence of "Monopolies or Anti-Trust Laws", technical incompetence of the Securities and Exchange Commission to handle such transactions that should be for an independent body established for that purpose¹⁶¹

¹⁵⁷AA Akume, 'The Phenomenon of Corporate Collapse and its Impact on Developing Economies', ABUJCL, VOL. 1 NO. 1 of 2002, 40-53.

¹⁵⁸*Ibid*, 42.

¹⁵⁹*Ibid*, 48.

 ¹⁶⁰A Abdulai, 'Mergers, Acquisitions and Take-Overs in Nigeria: The Guiding Philosophy, Laws, Regulations and Practice' [1997] 2 MILBQ, 21-34.
 ¹⁶¹*Ibid*, 34.

and insufficient expertise of the Federal High Court to deal with corporate restructuring transactions.

However, Abdulai' work predated the Investment and Securities Act, 2007 and the Securities and Exchange Commission Rules 2013 as amended 2015. Thus, this study will therefore examine merger, acquisition and take-over transactions as potential corporate rescue mechanisms in Nigeria in line with the extant laws.

Ofili attributed the recent wave of mergers witnessed in the Nigerian corporate scene to the mismanagement of many companies, the regulatory fiat of some government agencies and the prevailing harsh economic conditions which demanded the pulling of resources to ensure the survival of companies.¹⁶² Ofili' work was restricted to an analysis of the provisions of the Investment and Securities Act 2007 regulating merger of companies in Nigeria, definition of merger and types of mergers under the Common Law. Instructively, Ofili, suggested the harmonisation of the provisions of the Investment and Securities Act 2007 and the SEC Rules and Regulations made pursuant to the now repealed Investments and Securities Act 1999.

It is against Ofili's suggestion that this study will examine the provisions of the extant Securities and Exchange Commission Rules 2013 as amended 2015 as it relates to the procedure on mergers, acquisition and take-overs.

Tettenborn discussed whether the seller's interest under a "Romalpa Clause"¹⁶³ in sale of goods is a floating charge that transforms sellers of goods from unsecured creditors to secured creditors in cases of insolvency.¹⁶⁴ Tettenborn did not comprehensively discuss the effects of "Romalpa Clauses" in insolvency proceedings.

¹⁶² P N Ofili, 'Mergers of Companies under the Investment and Securities Act, 2007'(2012) 1 JCL, 177-193.

¹⁶³ Aluminium IndustrieVaasen B.V v Romalpa Aluminium Ltd [1976] 1 WLR 676; G Etomi, An Introduction to Commercial Law in Nigeria: Texts, Cases and Materials (Lagos: MIJ Professional Publishers, 2014) pp. 167-168.

¹⁶⁴A M Tettenborn, 'Reservation of Title: Insolvency and Priority Problems', The Journal of Business Law 1981 May, 173-177.

Hence, this study will examine the effects and application of a Romalpa Clause in insolvency proceedings.

Fletcher discussed the problems associated with the maintenance of two distinct legal and procedural codes for personal and corporate insolvency in the United Kingdom, the piecemeal and antiquated nature of the insolvency legislations in the United Kingdom and argued that any law and procedure on insolvency is to be guided by the philosophy of it being simple, effective and inexpensive.¹⁶⁵

Fletcher's work is a very resourceful exposition on the principal recommendation of the Cork Report 1982. However, this study will proceed to examine the extant Insolvency Act of the United Kingdom and other legislations and rules on Insolvency that were enacted after the Cork Report.

Asomugha examined the powers of a company to borrow and charge property in the CAMA, defined fixed and floating charges, their distinctive features and the mode of enforcement of a fixed charge by the appointment of a receiver and in the case of a floating charge, by the appointment of a receiver and manager.¹⁶⁶

Beyond the cursory examination of the effects of receivership, Asomugha did not elaborate much on the law and procedure on receiverships. Against this drawback, this study will comprehensively examine the law and procedure on receivership with the aim of suggesting reforms.

Fletcher¹⁶⁷ discussed the effects of the legislative changes to the corporate insolvency law of the United Kingdom and the impact upon corporate rescue. Fletcher argued that in spite of the legislative changes, the 'UK Insolvency Law has

 ¹⁶⁵ I F Fletcher, 'The Reform of Insolvency Law- 1', The Journal of Business Law 1983 March, 97.
 ¹⁶⁶ E M Asomugha, 'A Review of the Borrowing Powers of the Company under the 1990 Companies and Allied Matters Decree' (1990/1991) 13, 14 & 15 J.P.P.L, 41-52.

¹⁶⁷ I F Fletcher, 'UK Corporate Rescue: Recent Developments- Changes to Administrative Receivership, Administration and Company Voluntary Arrangements- The Insolvency Act 2000, The White Paper 2001 and the Enterprise Act 2002' (2004/5) 5 European Organizational Law Review, 119-151.

been unable to embrace the "American way" of corporate rescue, with debtor-inpossession as its core principle, but has instead opted for a "rescue" model in which creditor interests continue to assert a dominant influence'.¹⁶⁸

Although Fletcher lucidly discussed the impact of legislative changes on corporate rescue procedure in the United Kingdom, this study will proceed further to examine the impact of contemporary legislative changes on corporate rescue and insolvency law and practice in the United Kingdom and the possibility of transposition of the positive impact of the legislative changes to corporate rescue and insolvency system in Nigeria.

Bolodeoku examined of the general nature of receivership, particularly in the context of tax debt enforcement, the distinctive features of receivership under the CAMA and the 'non-CAMA' receivership.¹⁶⁹ Bolodeoku seemed to have treated receivership as an end in itself instead of a process or a vehicle for the enforcement of debt security. Thus, Bolodeoku did not examine the challenges that stymied the use of receivership in the enforcement of debt security, particularly in Nigeria. Hence, this study will robustly examine the law and practice of receivership, particularly in Nigeria.

Candide-Johnson and Alex-Adedipe examined the concept of corporate insolvency, proffered definition of corporate insolvency from the renowned categories of cash flow insolvency and balance sheet insolvency, the effects of corporate insolvency on debtor companies and remedies available to creditors by an exhaustive discussion of the procedures for receivership and winding up in Nigeria.¹⁷⁰

¹⁶⁸ Fletcher, *art cit*, 120.

¹⁶⁹ I O Bolodeoku, 'Receivership and Enforcement of Tax Liability' (2004) Vol. 24 The Journal of Private and Property Law, University of Lagos, 29-43.

¹⁷⁰ C A Candide-Johnson & A Alex-Adedipe, 'Debt Recovery: Corporate Insolvency – Receivership, Winding Up and other Arrangements' in O Olanipekun (ed), *Banking: Theory, Regulation, Law and Practice* (Lagos: Au Courant, 2016) pp 430-455.

They were of the view that the focus of corporate insolvency in modern day law deviates from the previous trend of dissolution through winding up of insolvent companies and currently focuses on restructuring of insolvent companies to allow for recovery and continuity in place of dissolution.

While the contemporary perspective of this work is remarkable and profound, it did not examine the flaws in the corporate rescue and insolvency procedures in Nigeria. Thus this study will examine the flaws in the corporate rescue and insolvency procedures in Nigeria.

Oditah examined the limits of the *pari passu* rule in winding up, particularly under English law. Oditah discussed the right of an unsecured creditor in the distribution of assets upon winding up of a company and contended that an unsecured creditor is merely owed an obligation and that at no time both before and after winding up, is the unsecured creditor entitled to a claim in any particular asset of the debtor.¹⁷¹ Thus, Oditah posited that the distinction between proprietary and personal claims is at the heart of insolvency law and if insolvency law respects this distinction, then, it constitutes a significant qualification to the principle of equal treatment of all creditors. Furthermore, since insolvency law do not accord significant respect to pre-insolvency rights, Oditah argued that the determination of who are equals is not a concern of insolvency law but general law.

In this study, Oditah' seminal analysis of priority claims would be comprehensively examined.

51

¹⁷¹ F Oditah, 'Assets and the Treatment of Claims in Winding up', (1992) 108 LQR 459-500.

1.8. Definition of Terms

"Administration"¹⁷² is a procedure introduced by the Insolvency Act 1985 to provide an alternative to receivership and winding up in the case of a company experiencing financial problems. The aim of the procedure is the rehabilitation of the company or its business....

"Administration Order"¹⁷³ is an order directing that, during the period for which the order is in force, the affairs, business and property of the company shall be managed by a person ("the administrator") appointed for the purpose by the court.

"Administrative Receiver"¹⁷⁴ means:

i.A receiver or manager of the whole (or substantially the whole) of a company's property appointed by or on behalf of the holders of any debentures of the company secured by a charge which, as created, was a floating charge, or by such a charge or one or more other securities; or

ii.A person who would be such a receiver or manager but for the appointment of some other person as the receiver of part of the company's property.

"Business Rescue"¹⁷⁵ means proceedings to facilitate the rehabilitation of a company that is financially distressed by providing for –

i.the temporary supervision of the company, and of the management of its affairs, business and property;

ii.a temporary moratorium on the right of claimant against the company or in respect of property in its possession; and

¹⁷² Farrar & Hanningan, *op cit*, Glossary of Terms.

¹⁷³ Insolvency Act, *op cit*, s.8 (2) (UK). There is no equivalent provision in any of the extant laws in Nigeria.

¹⁷⁴*Ibid*, s. 29 (2): CAMA, s. 393 (3) & Eleventh Schedule. It is important to note that in the CAMA it is not described as administrative receiver but receiver and manager.

¹⁷⁵ Companies Act 2008, s. 128 (1) (b) (South Africa). In Nigeria, there is no equivalent provision for business rescue proceeding as it is an indigenous procedure provided in the South African Statute.

iii.the development and implementation, if approved, of a plan to rescue the company by restructuring its affairs, business, property, debt and other liabilities, and equity in a manner that maximises the likelihood of the company continuing in existence on a solvent basis or, if it is not possible for the company to so continue in existence, results in a better return for the company's creditors or shareholders than would result from the immediate liquidation of the company.

"Capital Market Operator"¹⁷⁶ means any person, whether Individual or corporate, duly registered by the Securities and Exchange Commission (SEC) to perform specific function in the capital market.

"Charge" includes mortgage,¹⁷⁷ an encumbrance on the property which is either fixed or floating. A fixed legal charge is a statutory creation; a fixed equitable charge is a creation of equity. A floating charge is a species of equitable charge which does not attach to specific assets until crystallisation.¹⁷⁸

"Commencement of winding up"¹⁷⁹ means the date when a winding up is deemed to have started.

"Committee of inspection"¹⁸⁰ is a committee consisting of creditors and members appointed to assist a liquidator in winding up.

"Company Rescue"¹⁸¹ refers to the preservation of the corporate entity or some part of it, after it has been remedied.

"Compromise"¹⁸² is an agreement which terminates a dispute between parties as to the rights of one or more of them or which modifies the undoubted rights of one or

¹⁷⁶Investment and Securities Act 2007 (ISA), s. 315.

¹⁷⁷ CAMA, *op cit*, s. 197 (11).

¹⁷⁸ Farrar & Hanningan, *op cit*, xviii.

¹⁷⁹ *Ibid*.

¹⁸⁰*Ibid*, xix.

¹⁸¹Adebola, *op cit*, p. 8.

¹⁸²Sneath v Valley Gold Ltd [1891] 1 Ch 447 at 494.

more of them or which modifies the undoubted rights of a party which it has difficulty in enforcing.

"Corporate Rescue"¹⁸³ is a collective process whereby the insolvent company is given breathing space.

"Debt"¹⁸⁴ means the liability or obligation in respect of a claim which is due from any person and includes a financial debt and operational debt.

"Dissolution of a Company"¹⁸⁵ is the corporate equivalent of death. It usually follows winding up.

"Eligible Bank Assets"¹⁸⁶ means assets of eligible financial institutions specified by the Governor of Central Bank of Nigeria as being eligible for acquisition by the AMCON pursuant to section 24 of the AMCON Act.

"Eligible Financial Institution"¹⁸⁷ means a bank duly licensed by the Central Bank of Nigeria to carry on the business of banking in Nigeria under the Banks and other Financial Institutions Act; and shall include a bank or other financial institution, whose banking license has been revoked by the Central Bank of Nigeria pursuant to the Banks and other Financial Institutions Act.

"Hiving-Off"¹⁸⁸ is a method whereby the receiver and manager or the administrator extracts potentially valuable portions of the distressed company and sells them in units to another company either already in existence or incorporated for that purpose.

¹⁸³ O Modupe, 'The Need for Corporate Rescue in Nigeria'<<u>http://modupeoluwalegalconsultant.com></u> accessed on 17 July 2015.

¹⁸⁴ Insolvency and Bankruptcy Code 2016, s. 3 (11) (India); Chianu, *op cit*, p. 628, defined a debt as a definite sum of money fixed by the agreement of the parties as payable by one party in return for the performance of a specified obligation by the other party or upon the occurrence of some specified event or condition.

¹⁸⁵ Farrar & Hanningan, *op cit*, xix.

¹⁸⁶Asset Management Corporation of Nigeria Act 2010 (as amended) (AMCON Act), s.61

¹⁸⁷AMCON Act, *op cit*, s. 61.

¹⁸⁸ B Adebola, 'The Nigerian Business Rescue Model: An Introduction' (2014) NIALS Journal of Legal Studies, 20; also available at <<u>http://papers.ssrn.com/sol3/papers.cfm?abstract_id=2297824></u> accessed on 10 May 2015.

"Insolvency"¹⁸⁹ in relation to a company, includes the approval of a voluntary arrangement under Part 1, the making of an administration order or the appointment of an administrative receiver. A company is also deemed unable to pay its debt if it is proved to the satisfaction of the court that the value of the company's assets is less than the amount of its liabilities, taking into account its contingent and prospective liabilities.¹⁹⁰

"Insolvent Person"¹⁹¹ means any person in Nigeria who, in respect of any judgment, decree or court order against him, is unable to satisfy execution or other process issued thereon in favour of a creditor, and the execution or process remains unsatisfied for not less than six weeks.

"Liquidation"¹⁹² is the process whereby the affairs of the company are wound up and its business and assets are realized for value. A company may be wound up voluntarily by its members if solvent or, alternatively, if it is insolvent, by its creditors or compulsorily by order of the court.

"Moratorium"¹⁹³ is a means by which a company under business rescue is given some breathing space during the subsistence of the business rescue process, and is mainly aimed at preventing a rash of creditors from claiming against what little is left in the company.

¹⁸⁹ Insolvency Act, 1986 (UK), s. 247 (1).

¹⁹⁰ CAMA, *op cit*, s. 409 (c); C S Ola, *Company Law in Nigeria* (Ibadan: Heinemann Educational Books (Nigeria) Plc. 2002)p 237.

¹⁹¹ CAMA, *op cit*, s. 567 (1).

¹⁹² C Mallon (ed), *The Restructuring Review* (7thedn, United Kingdom: Law Business Research Limited, 2014) p. 8.

¹⁹³ M B Halita, 'A Comparative Study of some Issues Relating to Corporate Insolvency Law in Nigeria and South Africa'. Unpublished LL.M Dissertation, University of Pretoria, May 2013, p. 9; Mayson, *et al, op cit*, p. 654.

"Official Receiver"¹⁹⁴ means the officer by whatever name called or known charged with control of affairs in bankruptcy and if the appointment is vacant for any reason whatsoever, means the sheriff.

"Other Financial Institution"¹⁹⁵ means any individual, body, association or group of persons; whether corporate or unincorporated, other than the banks licensed under this Act which carries on the business of a discount house finance company and money brokerage and whose principal object include factoring, project financing, equipment leasing, debt administration, fund management, private ledger services, investment management, local purchase order financing, expert finance, project consultancy, financial consultancy, pension fund management and such other business as the Central Bank of Nigeria may from time to time designate.

"Receiver"¹⁹⁶ is an impartial person appointed by the court to manage, collect, and receive pending the proceedings, rents, issues and profits of land or personal estate which it does not seem reasonable to the court that either party should collect or receive or for the same to be distributed among the persons entitled.

"Scheme of Arrangement"¹⁹⁷ is a restructuring tool that sits outside formal insolvency; that is, the company may become subject to a scheme of arrangement whether it is solvent or insolvent. It is a compromise or arrangement between the company and its creditors, or between the company and its members.

"Secured Creditor"¹⁹⁸ is a person who holds a mortgage, charge or lien on the company's property or any part of it as a security for a debt due him from the company.

¹⁹⁴ CAMA, *op cit*, s. 567 (1).

¹⁹⁵ Banks and other Financial Institutions Act (BOFIA), Revised Edition, Laws of the Federation of Nigeria 2007 (as amended), s.66.

¹⁹⁶*Uwakwe v Odogwe* [1989] 5 NWLR (pt 123) 562 at 579.

¹⁹⁷ Farrar & Hanningan, op cit, xxiii; C Mallon (ed), The Restructuring Review, op cit, p. 8.

¹⁹⁸Omojasola v Plisson Fisko (Nigeria) Ltd [1990] 5 N.W.L.R (pt 151) 434; Orojo, op cit, p. 497.

"Winding up"¹⁹⁹ involves the liquidation of the company/corporation so that its assets are distributed to those entitled to receive them. The process of liquidation of a company is also known as the winding up of the company.²⁰⁰ This can either be compulsory, that is, winding up by the court, or voluntary. Voluntary winding up can either be a members' voluntary winding up in case of solvency, or a creditors' winding up in the case of insolvency.

1.9. Organisational Layout

This Study is structured into Six (6) Chapters.

1.9.1. Chapter One contains the Background to the Study; Statement of Problem; Purpose of Study; Scope of Study; Significance of Study; Research Methodology; Literature Review and Definition of Terms.

1.9.2. Chapter Two discusses the law and procedure relating to corporate insolvency in Nigeria and examines the types of corporate insolvency; modes of and grounds for winding up of companies in Nigeria; legal appraisal of the CAMA, section 408(d); the procedure for winding up of a company by the Court; commencement of winding up proceedings in Nigeria; nature of proceedings before hearing of winding up petition; the effects of winding up order, nature of proceedings after winding up order and the meaning of dissolution of a company and the consequences thereto.

1.9.3. Chapter Three focuses on the law and practice of receiverships with an examination of the appointment of receiver and manager, consequences of appointment of receiver and manager, the powers, rights and obligations of receiver and manager, remuneration of receiver and manager, liabilities of receiver and manager, termination of appointment of receiver and manager, the administrative

¹⁹⁹Amolegbe, Re [2015] 25 JMLR 213 at 229; [2014] 8 N.W.L.R. (pt.1480) 76.

²⁰⁰ Farrar, *op cit*, xxiv.

receiver in the United Kingdom and the regulation of persons to practice as administrative receiver in the United Kingdom.

1.9.4. Chapter Four examines corporate rescue mechanisms and procedure in Nigeria and other selected jurisdictions such as the United Kingdom, the Republic of South Africa, the Republic of India and the United States of America.

1.9.5. Chapter Five deals with the issues and challenges in corporate rescue and insolvency law and practice in Nigeria such as the multifarious statutory frameworks, lack of efficient and pragmatic regulatory institutions, absence of moratorium clauses on corporate rescue provisions in the various statute, inadequate regulation of corporate rescue and insolvency practitioners, negative impact of corruption and other sharp practices, and the absence of uniform guidelines for informal corporate rescue mechanisms.

1.9.6. Chapter Six embodies summary of findings in this Study, conclusion and recommendations.

CHAPTER TWO

THE LAW AND PROCEDURE RELATING TOCORPORATE INSOLVENCY IN NIGERIA

Corporate insolvency is not a term that is subject to a concise or precise definition. However, the concept of insolvency is commonly used when a debtor is unable to meet its financial commitments. Where it is an individual, it is known as personal or individual insolvency and where it involves a corporate entity, it is known as corporate insolvency.²⁰¹ Although personal or individual insolvency and corporate insolvency are used interchangeably, personal or individual insolvency is governed by the law of bankruptcy. Unlike other common law jurisdictions, such as the United Kingdom, there is no one holistic legislation enacted and dedicated to corporate insolvency in Nigeria. Rather, matters of corporate insolvency in Nigeria are principally regulated by the Companies and Allied Matters Act.²⁰² However, insolvency provisions are found in some specific statutes that will be examined in this chapter.

Winding up is a term commonly associated with the ending of the life of a company. It is the process by which the assets of the company are collected in and realized, its liabilities discharged and the net surplus, if there is one, distributed to the persons entitled to it.²⁰³ It is pertinent to state that a company being wound up may be solvent or insolvent. Insolvent winding up occurs essentially when companies are unable to pay their debt.²⁰⁴

²⁰¹ Ian F Fletcher, *The law of Insolvency* (2ndedn, London; Sweet & Maxwell, 1996) pp. 10-11.

²⁰² Cap C20 Laws of the Federation on Nigeria 2004 (CAMA).

²⁰³ J H Farrar & B Hannigan, *Farrar's Company Law* (4thedn, London; LexisNexis Butterworths Tolley, 1998) p. 704.

²⁰⁴ CAMA, *op cit*, ss. 408 (d), 409 and 507 (1); Insolvency Act 1986, ss. 122 (1) (f), 123 (UK); Insolvency and Bankruptcy Code 2016, ss. 6 and 7 (India).

In this chapter, this study will examine the law and procedure relating to winding up in Nigeria.

2.1 Overview of Legal Frameworks and Regulatory Institutions on Insolvency

1. **Companies and Allied Matters Act**

The Companies and Allied Matters Act²⁰⁵ is the principal statute on insolvency generally in Nigeria. The CAMA established a body corporate, the Corporate Affairs Commission²⁰⁶ and vested it with the functions to administer the provisions of the CAMA, including amongst others the winding up of companies.²⁰⁷

There were extensive provisions in the CAMA, on insolvency as contained in Part XV, chapters 1, 2 & 5. The procedure for winding up proceedings is provided in the subsidiary legislation.²⁰⁸ The provisions of the CAMA and the Companies Winding up Rules would be comprehensively examined in later part of this chapter.

2. Asset Management Corporation of Nigeria Act

The Asset Management Corporation of Nigeria Act²⁰⁹ established a body corporate, the Asset Management Corporation of Nigeria²¹⁰ for the purpose of efficiently resolving the non-performing loan assets of banks in Nigeria. The functions and powers of the AMCON included the acquisition of eligible bank assets from eligible financial institutions in accordance with the provision of the AMCON Act,²¹¹ initiation or participation in any enforcement, restructuring, reorganisation,

²⁰⁵CAMA, *op cit*.

 $^{^{206}}$ *Ibid*, s. 1(1) (2) (a). Corporate Affairs Commission is abbreviated in this study as "CAC". ²⁰⁷*Ibid*, s. 7(1) (a).

²⁰⁸Companies Winding up Rules 2001 ("CWR").

²⁰⁹2010 as amended ("AMCON ACT").

²¹⁰*Ibid*, s. 1(1) ("AMCON").

²¹¹ *Ibid*, s. 5(a).

programme of arrangement or other compromise²¹² and to compromise any claim or forgive or forbear any debts or other obligation owed to the AMCON in respect of a specified class of eligible bank debts.²¹³

It is important to state that where compromise, forgiveness, or forbearance will result to recovery of lower price than that paid by the AMCON for the acquisition of the eligible bank assets, such compromise, forgiveness or forbearance is only exercisable with the approval of the Minister of Finance upon the recommendation of the Central Bank of Nigeria.²¹⁴

In order to exercise its functions and powers under the AMCON Act, the AMCON is further conferred with special power to act as or appoint a receiver for a debtor company, whose assets have been charged, mortgaged or pledged as security for an eligible bank asset acquired by the AMCON.²¹⁵ A receiver appointed by the AMCON shall have power to realise asset of the debtor company, enforce the individual liability of the shareholders and directors of the debtor company, and manage the affairs of the debtor company.²¹⁶

Furthermore, the AMCON is vested with special powers to commence winding up proceedings²¹⁷ under the provisions of the CAMA against a body corporate where the Federal High Court have given a decision for the debtor company to pay any sum to the AMCON and such sum is not liquidated or paid over to the AMCON within 90 days of the date of judgment.²¹⁸

3. Banks and Other Financial Institutions Act

²¹²*Ibid*, s. 6(1)(e).

²¹³ *Ibid*, s.6 (1) (1)
²¹⁴AMCON Act, *op cit*, s. 6(5)
²¹⁵*Ibid*, s. 48(1)
²¹⁶*Ibid*, s. 48(2)
²¹⁷*Ibid*, s. 52(1)
²¹⁸*Ibid*

The Banks and other Financial Institutions Act²¹⁹ is an Act to regulate banking and other financial institutions and for matters connected therewith. The BOFIA conferred certain functions on the Central Bank of Nigeria in respect of matters in the BOFIA.²²⁰ Instructively, corporate insolvency provisions are contained in the BOFIA, to wit, the Governor of Central Bank of Nigeria (CBN) may with approval of the Board of Directors and notice published in the Gazette revoke the licence granted to a Bank under BOFIA if such bank amongst other grounds go into liquidation or is wound up or otherwise dissolved²²¹ or has insufficient assets to meet its liabilities;²²² the definition of a failing bank as where it is insolvent²²³ and the making of winding up application to the Federal High Court by Nigeria Deposit Insurance Corporation upon the revocation of licence of the bank.²²⁴

It should be noted that the provisions of the BOFIA apply without prejudice to the provisions of the CAMA insofar as they relate to banks and winding up by the Federal High Court.²²⁵ Where any of the provisions of the CAMA are inconsistent with the provisions of the BOFIA as is it relates to winding up of a bank, the latter shall prevail.²²⁶ In the case of insolvency of other financial institutions, it is also provided in the BOFIA.²²⁷

4. Investment and Securities Act

The Investment and Securities Act²²⁸ is an Act that repealed the Investment and Securities Act 1999 and established a body corporate, the Securities and Exchange

²²⁷*Ibid*, s. 66

²¹⁹Revised Edition, Laws of the Federation of Nigeria 2007, as amended (BOFIA).

²²⁰ BOFIA, op cit, s. 1

²²¹*Ibid*, s. 12(1) (b)

²²²*Ibid*, s. 12(1) (d)

²²³*Ibid*, s. 35 (1) (c)

²²⁴*Ibid*, s. 40 ²²⁵*Ibid*, s. 55 (1)

 $^{^{226}}Ibid$, s. 55 (1)

²²⁸2007 (ISA)

Commission,²²⁹ as the apex regulatory institution for the Nigerian capital market.²³⁰ The functions and powers of the SEC include to regulate investment and securities business in Nigeria as defined in the ISA,²³¹ register and regulate corporate and individual capital market operators as defined in the ISA and to perform such other functions and exercise such other powers not inconsistent with the ISA as are necessary or expedient for giving full effect to the provisions of the ISA.²³²

The ISA provided for insolvency matters.²³³ Furthermore, the ISA established an Investors Protection Fund²³⁴ with the objective to compensate investors who suffer pecuniary loss arising amongst others from the insolvency, bankruptcy and negligence of a dealing member firm of a securities exchange or capital trade point.²³⁵

5. Nigeria Deposit Insurance Corporation Act

The Nigeria Deposit Insurance Corporation Act²³⁶ is an Act that established a body corporate known as the Nigeria Deposit Insurance Corporation²³⁷ and vested on the NDIC the responsibility for insuring all deposit liabilities of licensed banks and such other deposit taking financial institutions, otherwise known as "insured institutions", operating with the meaning of sections 16 and 20 of the NDIC Act so as to engender confidence in the Nigerian banking system²³⁸ and giving financial assistance to

²²⁹*Ibid*, s. 1 (1) (2) (SEC)

²³⁰*Ibid*, s. 13

²³¹ISA, *op cit*, s. 13 (a)

²³²*Ibid*,s. 13 (dd)

²³³ BOFIA, *op cit*, ss.47(1) (c) & 48(1) (c) (Insolvency of a capital market operator); s.53(1) (application to the Federal High Court for winding up by a capital market operator); s.53(2) (Application to the Federal High Court for winding up by the SEC); s.53(3) (powers of the SEC to appoint a provisional liquidator); and s.53(5) (a liquidator to forward to the SEC the copies of any returns which he is required to make under the CAMA)

²³⁴ ISA, *op cit*, s.197(1)

²³⁵*Ibid*, ss. 198(a) 204(a) & 212 (1) (a) & (b); k Udofia, 'Contributions to Investors Protection Fund and the Fraudulent Preference Risk', LAWYER, a This Day Weekly Pull-out, 18.04.2017, p.11.

²³⁶2006 (as amended) ("NDIC ACT")

²³⁷*Ibid*, s.1(1)(2)(a) ("NDIC").

²³⁸*Ibid*, s. 2 (1) (a).

insured institutions in the interest of depositors, in case of imminent or actual financial difficulties.²³⁹

In this Study, the provision of the NDIC Act relating to or connected with insolvency,²⁴⁰ will be relevant.

6. **The Insurance Act**

The Insurance Act²⁴¹ made provisions for presentation of winding up petition against an insurer. Such petition for the winding up of an insurer may be made to the Federal High Court, subject to the approval of the National Insurance Commission by not less than 50 policy-holders, each of whom holds a policy that has been in force for not less than 3 years.²⁴² Furthermore, the provisions of the CAMA on winding up were made subject to the provision of the Insurance Act.²⁴³ In same vein, the Insurance Act provided its own priority list in the settlement of debts.²⁴⁴ In other words, the distribution of assets of an insurer upon winding up is not in accordance with the provisions of section 494 of the CAMA and rule 167 of the CWR.

7. **Bankruptcy Act**

²³⁹NDIC Act, *op cit*, s. 2 (1) (b).

²⁴⁰NDIC Act, op cit, s. 40 (1) (Appointment of the NDIC as deemed provisional liquidator or liquidator of a failed insured institution); s.40(2) (the powers of NDIC to apply to the Federal High Court for winding up of a failed insured institution); s.40(4) (the provisions of sections 427(1) & (2) and 428(1) & (2) of the CAMA shall not apply to the NDIC when acting as liquidator of a failed insured institution): s.40(5) (the provision of the companies winding up Rules shall not apply to the winding up of a failed insured institution by the NDIC as liquidator); s.40(6) (the provisions of the NDIC Act shall apply without prejudice to the provisions of the CAMA in relation to winding up by the Federal High Court and where any provisions of the CAMA are inconsistent with that of NDIC Act, the provision of NDIC Act shall prevail); s.40(7) (the referral to the Court of Appeal of any action challenging the revocation of the licence of an insured institution or a petition for winding up of an insured institution or the appointment of the NDIC as liquidator and there is an application for interim or interlocutory injunction, which application shall be determined by the Court of Appeal within 60 days of such referral) and s.41(2) (powers of the NDIC as a liquidator). ²⁴¹Cap 118 Laws of the Federation of Nigeria 2004 (as amended)

²⁴²*Ibid*, s. 32 (1) (a)

²⁴³*Ibid*, s. 32 (2)

²⁴⁴ *Ibid*, s. 32 (4)

The Bankruptcy Act²⁴⁵ is an Act that made provisions for declaring as bankrupt any person who cannot pay his debts of a specified amount.²⁴⁶ The Bankruptcy Act provide for what constitute acts of bankruptcy in section 1 and on how to commence bankruptcy proceedings.²⁴⁷

While it is unarguable that bankruptcy is the precursor of insolvency, in this study, the focus will be on insolvency and not bankruptcy as provided in the Bankruptcy Act. Although, the notions of corporate insolvency and bankruptcy in the Nigerian context are seemingly synonymous, they differ to the extent that the former relates to corporate entities while the latter relates to individuals.²⁴⁸

However, it is important to state that an understanding of the law of bankruptcy and the application of the bankruptcy rules is relevant in discussion on proof and ranking of claims in winding up of insolvent companies.²⁴⁹

8. The Federal High Court

The Federal High Court is the court vested with jurisdiction²⁵⁰ on insolvency matters.

2.2. Types of Corporate Insolvency

There are many ways of defining insolvency. According to Candide-Johnson and Alex-Adedipe, the definition of corporate insolvency is classified into two renowned categories of Cash flow insolvency and Balance sheet insolvency.²⁵¹ In this study, other type of insolvency, that is, ultimate insolvency and regulatory insolvency will also be discussed. Hence, it is better to define insolvency by way of classification.

1. Cash Flow or Commercial Insolvency

²⁴⁶*Ibid*, preamble

²⁴⁵Cap B2 Laws of the Federation of Nigeria 2004

²⁴⁷ Bankruptcy Act, *op cit*, ss.3, 4, 5, 6, 7, 8 and 9

²⁴⁸C A Candide-Johnson & A Alex-Adedipe, 'Debt Recovery: Corporate Insolvency-Receivership, Winding up and other Arrangements' in O Olanipekun. *Banking: Theory; Regulation, Law and Practice* (Lagos: Au Courant, 2016) p.430

²⁴⁹CAMA, *op cit*, ss. 492 and 493

²⁵⁰ CFRN, s. 251 (1)(e)(j); CAMA, *op cit*, ss.407(1) & 567(1).

²⁵¹S.494

This type of insolvency describes insolvency in a situation where a company is unable to pay its debt as they fall due. In other words, even where its overall assets position may not be in deficit, it has cash flow problems that prevent it from paying its debts when they fall due or upon demand.²⁵² Thus, where a company is indebted to another and such debt is due for payment, the inability of the company to pay on the due date would suggest that the company is insolvent.²⁵³ Cash flow insolvency therefore arises from the extension of credit by the creditor to the borrower.²⁵⁴This is the most often resorted ground for winding up petition in Nigeria.²⁵⁵

2. Balance Sheet Insolvency

This refers to a type of insolvency where the value of the assets of the company is less than the amount of its liabilities. Simply put, this occurs where the liabilities of the company exceed its assets, taking into account, not only current liabilities but also, contingent and prospective liabilities.²⁵⁶

It is pertinent to emphasise that a company that is insolvent in balance sheet terms will not necessarily be commercially insolvent. In other words, a company, for instance, that has a heavy potential liability in tort may still for the time being have a perfectly satisfactory cash flow.²⁵⁷

3. Ultimate Insolvency

This is a type of insolvency where on the liquidation of the assets of the company there is no sufficient assets to satisfy the claim of the creditors.²⁵⁸

²⁵² Ian F Fletcher, op cit, p. 485; Candide-Johnson & Alex-Adedipe, op cit, p. 432.

²⁵³ Candide-Johnson & Alex-Adedipe, *ibid*.

 ²⁵⁴ R Goode, *Principles of Corporate Insolvency Law* (4thedn, London: Sweet & Maxwell, 2011) p. 2.
 ²⁵⁵ CAMA, *op cit*, s. 408 (d); Insolvency Act s. 122(1) (f) (UK); The Insolvency and Bankruptcy Code 2015, ss. 6 and 7 (India).

²⁵⁶ CAMA, *op cit*, s. 409 (c); Insolvency Act, s. 123 (2) (UK); Candide-Johnson & Alex-Adedipe, *op cit*, p. 433.
²⁵⁷ L Sealy & S Worthington, *Cases and Materials in Company Law* (10thedn, Oxford: Oxford)

²⁵⁷ L Sealy & S Worthington, *Cases and Materials in Company Law* (10thedn, Oxford: Oxford University

Press, 2013) p. 768. ²⁵⁸*Ibid*.

4. Regulatory Insolvency

This type of insolvency refers to where though the assets of a corporate entity could exceed its liabilities such corporate entity is still regarded as insolvent because of failure to meet the regulatory thresholds fixed by the applicable law.²⁵⁹

2.3. Modes of Winding up of Companies in Nigeria

The winding up of a company may be commenced by any of the following ways:²⁶⁰

i.by the Court; or

ii.voluntarily; or

iii. subject to the supervision of the Court.

It is pertinent to state that in the Republic of India, the initiation of liquidation of a company can only commence after the initiation of a corporate insolvency resolution process to revive and rehabilitate the distressed or sick company. In other words, the liquidation of a company in India is a fall back process.²⁶¹

1. By the Court (Compulsory Winding up).

It is liquidation or winding up of a company by virtue of an order of a court with competent jurisdiction that the company is wound up. Winding up by the court is also called compulsory winding up.

²⁵⁹K Ekwueme, 'Failure Resolution' in O Olanipekun (ed), op cit, p.493.

²⁶⁰ CAMA, op cit, s. 401 (1); Corporate Affairs Commission v Davis [2008] 1 N.W.L.R (pt 1067) 60 at 64.

²⁶¹ The Insolvency and Bankruptcy Code 2016, s. 33 (India). The liquidation process of a company is determined by an Adjudicating Authority, that is, the National Company Law Tribunal constituted under s. 408 of the Companies Act, 2013.

Compulsory winding up may be applied to a solvent or an insolvent company, but, in practice, its most often invoked in relation to insolvent companies.²⁶² The court vested with jurisdiction to wind up a company in Nigeria is the Federal High Court within whose area of jurisdiction the registered office or head office of the company is situate.²⁶³ It is important to state that the "registered office" or the "head office" mentioned in Section 407 (1) of the CAMA means the place which has been the longest registered office or head office of the company during the six months immediately preceding the presentation of the petition for winding up.²⁶⁴

Winding up by the court is usually invoked upon the circumstances set out in Section 408 and other relevant provisions of the CAMA. However, the circumstance that a company is unable to pay its debt is the most common ground upon which a petition for compulsory winding up is usually presented.²⁶⁵

In the later part of this chapter, the grounds for winding up of an insolvent company will be discussed.

2. Voluntarily

A company may be wound up voluntarily if:

i. the period, if any, fixed for the duration of the company by the articles expires; or the event, if any, occurs, on occurrence of which the articles provided that the company is to be dissolved and the company in general meeting has passed a resolution requiring the company to be wound up voluntarily.²⁶⁶

²⁶² S W Mayson, *et al, Mayson, French & Ryan on Company Law* (25thedn, Oxford: Oxford University Press, 2008 – 2009) p. 662.

²⁶³ CAMA, op cit, s. 407 (1).

²⁶⁴ *Ibid*, s. 407 (2).

 ²⁶⁵ Gbedu v Itie [2007] 10 N.W.L.R (pt. 1202) 227; C S Ola, Company Law in Nigeria (Ibadan: Heinemann Educational Books (Nigeria) Plc. 2002) p. 467.

²⁶⁶ CAMA, op cit, s. 457 (a); Insolvency Act, op cit, s. 84 (1) (UK).

ii. the company resolves by special resolution that the company be wound up voluntarily.²⁶⁷

It is pertinent to state that the type of resolution to adopt in a case where the company is to be wound up voluntarily under section 457 (a) of the CAMA was not expressly stated. Hence, there is the argument as to what type of resolution,²⁶⁸ that is, whether a special resolution or an ordinary resolution that is to be passed at the general meeting.

Usually an Act specifies which type is required and that provision governs. But where the Act is silent about the type of resolution required, then an ordinary resolution is required unless the articles specify a higher level of approval, up to and including unanimity.²⁶⁹ Thus in this Study, it is contended that since it was not stipulated in the CAMA that it has to be by special resolution, voluntary winding up under section 457 (a) of the CAMA can be initiated by an ordinary resolution of the company at a general meeting unless where the article of the company provides otherwise.²⁷⁰

Voluntary winding up can be either a members' voluntary winding up or a creditors' voluntary winding up.²⁷¹ A members' voluntary winding up is initiated when the members of a company adopt a resolution for voluntary winding up following a statutory declaration of solvency²⁷² by the directors of the company in

²⁶⁷CAMA, op cit, s. 457 (b); Insolvency Act, op cit, s. 84 (1) (b).

²⁶⁸ Ordinary resolution is a resolution passed by a simple majority of votes cast by such members of the company as, being entitled to do so, vote in person or by proxy at a general meeting as per CAMA, s. 233 (1); Special resolution is a resolution passed by not less than three-fourth of the votes cast by such members of the company as, being entitled to do so, vote in person or by proxy at a general meeting of which 21 days' notice, specifying the intention to propose the resolution as a special resolution, has been duly given, subject to the proviso as per CAMA, s.233 (2).

²⁶⁹ P L Davies & S Worthington, *Gower and Davies Principles of Modern Company Law* (9thedn, London: Sweet and Maxwell, 2012) p. 460.

²⁷⁰ CAMA, op cit, s. 233 (6); Farrar, op cit, p. 708; Mayson et al, op cit. p. 660.

²⁷¹ *Ibid*, s. 462 (4); Insolvency Act, *op cit*, s. 90 (UK).

²⁷² CAMA, *op cit*, s. 462 (4); S McLaughlin, *Unlocking Company Law* (London: Hodder Education, 2009) p. 399.

accordance with section 462 (1) of the CAMA. Thus, a members' voluntary winding up is generally not regarded as an insolvency procedure as it is mostly invoked in relation to solvent companies. In other respect, a creditors' voluntary winding up is initiated when the members of a company adopt a resolution for winding up without a statutory declaration of solvency by the directors of the company.²⁷³ Creditors' voluntary winding up is an insolvency procedure and is usually invoked in relation to insolvent companies.

The statutory declaration of solvency required under section 462 (1) of the CAMA must state that the declaring directors of the company have made a full enquiry into the affairs of the company and that, having so done, they have formed the opinion that the company will be able to pay its debts in full within such period, not exceeding 12months from the commencement of the winding up as specified in the declaration. A statutory declaration of solvency is of no effect unless it is made within 5weeks immediately preceding the date of the resolution for winding up the company and is delivered to the Corporate Affairs Commission for registration before that date and further embodies a statement of the company's assets and liabilities as at the latest practicable date before the making of the declaration.²⁷⁴ It is not required that the statement be correct in every detail, provided it can be fairly described as being a statement of assets and liabilities.²⁷⁵

In the United Kingdom, the provision for a statutory declaration of solvency to be delivered to the registrar of companies before the expiration of 15days immediately following the date on which the resolution for winding up is passed²⁷⁶ seems reasonable and commercially expedient than the provision under section 462

²⁷³ CAMA, op cit, s. 462 (4).

 $^{^{274}}$ *Ibid*, s. 462 (2).

 ²⁷⁵ De Courcy v Clement [1971] Ch 693.
 ²⁷⁶ Insolvency Act, op cit, s. 89 (3) (UK).

(2) (a) of the CAMA that requires the delivery of the statutory declaration of solvency to the Corporate Affairs Commission for registration within 5weeks before the date of the passing of the resolution for winding up of the company.

It constitutes a criminal offence for any director of a company to make a declaration without having reasonable grounds for the opinion that the company would be able to pay its debts in full within the period specified in the declaration.²⁷⁷ If the company's debts are not paid or provided for in full within the period stated in the declaration, then the declaring directors of the company are presumed not to have had reasonable grounds for making it.²⁷⁸

While the provisions applicable to a members' voluntary winding up is as contained in sections 464 to 470, 480 to 485, 491 to 531 of the CAMA, the provisions applicable to a creditors' voluntary winding up is contained in sections 472 to 485, 491 to 531 of the CAMA.

In the later part of this chapter, the procedure for creditors' voluntary winding up will be discussed.

3. Subject to the Supervision of the Court

Where a company passes a resolution for voluntary winding up, the Court may on petition order that the voluntary winding up shall continue but subject to such supervision of the Court and with such liberty for creditors, contributories, or others to apply to the court and generally on such terms and conditions as the court thinks iust.²⁷⁹

 $^{^{277}}$ CAMA, op cit, s. 462 (3). 278 Ibid

²⁷⁹ *Ibid*, s. 4-86.

The nature of winding up subject to the supervision of the court was rightly pointed out by the court in *Corporate Affairs Commission v Davis*²⁸⁰ thus:

winding up subject to supervision of the Court occupies one unique position in between winding up by the Court and Voluntary winding up as shown in Sections 490 (1) and (2) of the CAMA. Where an order is made for a winding up subject to supervision, the liquidator may, subject to any restrictions imposed by the court, exercise all his powers, without the sanction or intervention of the court, in the same manner as if the company were being wound up voluntarily.... that the powers specified in paragraphs D, E and F of Section 425 (1) shall not be exercised by the liquidator except with the sanction of the court or, in a case where before the order, the winding up was a creditors voluntary winding up, with the sanction of the court or the committee of inspection, or (if there is no committee) a meeting of the creditors....

The provisions of the CAMA that are not applicable to a winding up subject to supervision of the court are provided in the Twelfth Schedule to the CAMA.²⁸¹ In the

²⁸⁰ Supra, at 78 -79, paras. H–F, per Owoade JCA.

²⁸¹ CAMA, *op cit*, s. 490 (2); The 12th Schedule to the CAMA provides for the provisions inapplicable to winding up under the supervision of the court, that is, s. 388 (power to appoint official receiver for debenture holders and others), s. 420 (statement to company's affairs to be submitted to official receiver) s. 422 (appointment, remuneration and title of liquidators except subsection 8 - the acts of a liquidator shall be valid notwithstanding any defect that may afterwards be discovered in his appointment or qualification), s. 427 (exercise and control of liquidator's powers) s. 428 (payments by liquidator into company's liquidation account) s. 429 (audit, etc of liquidators' account), s. 430 (books

United Kingdom, there is no provision for winding up subject to supervision of the court as winding up can be either voluntary or by the court.²⁸²

In the Republic of Ghana, the winding up of a company is either by official liquidation in accordance with the Bodies Corporate (Official Liquidations) Act or by private liquidation in accordance with the Companies Act.²⁸³ In this study, the procedure for winding up of a company by official liquidation will be examined in appropriate circumstance. The modes²⁸⁴ of official winding up of a company are by:

iv. a special resolution of the company;

v.a petition addressed to the Registrar of Companies;

vi.a petition to the court, or

vii.a conversion from a private liquidation.

Against the forgoing, it is important to state that in the Republic of Ghana, while private liquidation, which is akin to members' voluntary winding up in Nigeria, is governed by the provisions of the Companies Act, official liquidation, which is essentially similar to winding up by the Court (Compulsory Winding up) in Nigeria, is governed by the provisions of the Bodies Corporate (Official Liquidations) Act.

2.4 Grounds for Winding up of Companies in Nigeria

The circumstances upon which a company may be wound up by the court in Nigeria are provided in section 408(a) to (e) of the CAMA, to wit:

i.if the company has by special resolution resolved that the company be wound up by the court;

to be kept by liquidators) s. 431 (release of liquidator), s. 432 (control over liquidators) s. 433 (powers to appoint committee of inspection, etc), s. 434 (powers, etc of committee of inspection), s. 435 (powers were no committee of inspection), s. 436 (powers to appoint special manager), s. 450 (power to order public examination of promoters, etc) and s. 453 (delegation to liquidator or of certain powers of the court).

²⁸² Insolvency Act, *op cit*, s. 73.

²⁸³ Companies Act 2013, s. 257 (1) (Ghana).

²⁸⁴ Bodies Corporate (Official Liquidations) Act 1963 (Act 180), s. 1 (1) (Ghana).

ii.if default is made in delivering the statutory report to the Corporate Affairs

Commission or in holding the statutory meeting;

iii.if the number of members is reduced to below two;

iv.if the company is unable to pay its debts;

v.if the court is of the opinion that it is just and equitable that the company should be wound up.

Notwithstanding the provisions of section 408 of the CAMA, a company may be wound up in Nigeria by virtue of section 312 (2) (a) of the CAMA if the court is satisfied that a petition under sections 310 and 311 of the CAMA is well founded. Furthermore, a company may be wound up in Nigeria by reason of the provisions in some extant legislation such as the AMCON Act,²⁸⁵ the ISA on the ground of cancellation of registration of a capital market operator²⁸⁶ and the NDIC Act on the ground of the revocation of the licence of the failed insured institution.²⁸⁷

In the Republic of India, unlike the case in Nigeria and the United Kingdom, the liquidation of a company is not invoked as an originating process. The liquidation of a company can only be triggered after a corporate insolvency resolution process.²⁸⁸ In this vein, liquidation of a company in the Republic of India can only be invoked upon the following circumstance, to wit:

1. Where the Adjudicating Authority, that is National Company Law Tribunal, before the expiration of the insolvency resolution process period or the

²⁸⁵ 2010 (as amended), s. 52

 $^{^{286}}$ ISA, *op cit* s. 53(4) and NDIC Act on the ground of the revocation of the licence of the failed insured institution

 $^{^{287}}_{200}$ NDIC Act, s.40(3)

²⁸⁸ Insolvency and Bankruptcy Code, *op cit*, chapter III.

maximum period permitted for completion of the corporate insolvency resolution process under section 12^{289} or the fast track corporate insolvency resolution process under section 56,²⁹⁰ does not receive a resolution plan under (6) of section 30.²⁹¹

Where the Adjudicating Authority rejects the resolution plan under section
 31 for the non-compliance of the requirements specified therein.²⁹²

3. Where the resolution professional²⁹³ during the corporate insolvency resolution process but before confirmation of the resolution plan, intimate the Adjudicating Authority of the decision of the committee of creditors to liquidate the corporate debtor.²⁹⁴

4. Where any person other than the concerned debtor whose interests are prejudicially affected makes an application to the Adjudicating Authority over the contravention of an approved resolution plan.²⁹⁵

In the Republic of Ghana, the grounds for official liquidation of a company are disparate and depend on the mode of winding up of the company. For instance, where the official winding up of a company is by a petition addressed to the Registrar of Companies, it can be on the ground that the company is unable to pay its debts.²⁹⁶ The meaning of a company being unable to pay its debt is essentially similar to the

 $^{^{289}}$ *Ibid*, s. 12 – the time limit for completion of insolvency resolution process is a period of 180 days from the date of admission of the application to initiate the process. However, the insolvency resolution process period of 180 days can be extended by the Adjudicating Authority upon an application by the resolution professional for such further period as it thinks fit but not exceeding 90 days and any such extension shall not be granted more than once.

²⁹⁰ Insolvency and Bankruptcy Code, *op cit*,, s. 56 – the time limit for completion of fast track corporate insolvency resolution process is a

period of 90 days from the insolvency commencement date. However, the fast track corporate insolvency resolution process period of 90 days can be extended upon an application by the Adjudicating Authority upon an application by the resolution professional for such further period but not exceeding 45 days and any such extension shall not be granted more than once.

²⁹¹ *Ibid*, s. 33 (1) (a).

²⁹² *Ibid*, s. 33 (1) (b).

²⁹³ *Ibid*, ss. 22, 23, 25 & 34 (1) – the resolution professional has a meaning and duties that is akin to that of a Liquidator the in liquidation process in Nigeria and the United Kingdom.

²⁹⁴ *Ibid*, s.33 (2).

 $^{^{295}}$ *Ibid*, s. 33 (3) (4).

²⁹⁶ Bodies Corporate (Official Liquidations) Act, *op cit*, s. 3 (2).

definition of inability to pay debt under section 409 of the CAMA except the amount of indebtedness and the period of statutory demand.²⁹⁷ In the case of official winding up of a company by a petition to the court, it can be on the ground that:²⁹⁸

- The company does not within a year from its incorporation commence to carry on all the businesses which it is authorised by its regulations to carry on or suspends any of such businesses for a whole year;
- ii. The company has no members;
- The business or objects of the company are unlawful or the company is operated for an illegal purpose or the business being carried on by the company is not authorised by its Regulations;
- iv. The company is unable to pay its debts;
- v. The court is of the opinion that it is just and equitable that the company should be wound up.

While the grounds for winding up of a company seems to be more expansive in the Republic of Ghana and the United Kingdom, in this chapter, the discussion on the grounds of winding up of a company will be restricted to the provisions of section 408 (d) of the CAMA, which relates to the inability of a company to pay its debt,²⁹⁹ that is, corporate insolvency.

2.5 Determining Insolvency

The often resorted ground for a petition for winding up of a company is that the company is unable to pay its debts. However, the circumstances when a company can

 $^{^{297}}$ Bodies Corporate (Official Liquidations) Act, *op cit*, s. 3 (3) – the amount of indebtedness is put in a sum exceeding fifty pounds; the period of statutory demand is twenty-one days as against three weeks stipulated in the CAMA.

²⁹⁸ *Ibid*, s. 4 (2) (3) – it is instructive that the Attorney-General of the Republic of Ghana can petition for winding up of a company but only where the ground is that the business or objects of the company are unlawful or the company is operated for an illegal purpose or the business being carried on by the company is not authorised by its Regulations.

²⁹⁹ C. B. D. I. v COBEC (Nig.) Ltd [2004] 13 N.W.L.R (pt 890) 376; Unifam Ind. Ltd v Oceanic Bank Int'l Nigeria Ltd [2005] 3 N.W.L.R (pt 911) 83.

be held unable to pay its debts are not susceptible to a concise meaning. When a company does not have cash to pay its debts as they fall due, it is easily understandable that such a company is insolvent. There are other scenarios that make it difficult to conceptualise when a company is unable to pay its debts. For instance, where a company refuses to pay a single debt that it could perfectly afford to discharge, can it be held to be unable to pay its debts? It is likewise possible for a company to have sufficient assets but not sufficient liquidity to provide an adequate amount of cash to meet present or even contingent liabilities as they fall due. Can such a company be held to be unable to pay its debts?

2.5.1 Essential Elements of the phrase "Inability to Pay Debts"

There is no express provision in the CAMA or the Companies Winding up Rules on the definition of a debt or what constitutes a *bona fide* dispute of a debt or dispute of a debt on substantial grounds. However, a debt has been defined as a definite sum of money fixed by agreement of the parties as payable by one party in return for the performance of a specified obligation by the other party or upon the occurrence of some specified event or condition.³⁰⁰

In the relevant legislation of the Republic of India, a debt is defined as "a liability or obligation in respect of a claim which is due from any person and includes a financial debt and operational debt".³⁰¹ In a definitive and unambiguous manner, the words "financial debt",³⁰² "operational debt"³⁰³ and "dispute",³⁰⁴ were respectively

³⁰⁰ Chianu, *op cit* p. 628; *Hansa International Construction Ltd v Mobil Producing Nigeria* [1994] 9 N.W.L.R (pt 366) 76 at 86, per Uwaifo JCA (as he then was). 'A debt is a sum of money acknowledged as owing by the company or at any rate seen objectively from the facts as owing'; *International Merchant Bank Nigeria Ltd v Speegaffs Company Nigeria Ltd* [1997] 3 N.W.L.R (pt 494) 423. ³⁰¹ Insolvency and Bankruptcy Code, *op cit.* s. 3 (11) (India).

³⁰²Insolvency and Bankruptcy Code, *op cit*, s. 5 (8) (India).

³⁰³ *Ibid*, s. 5 (21) – means a claim in respect of the provision of goods or services including employment or a debt in respect of the repayment of dues arising under any law for the time being in force and payable to the Central Government, any State Government or any local authority. ³⁰⁴ *Ibid*, s. 5 (6) – includes a suit or arbitration proceedings relating to – (a) the existence of the amount

 $^{^{304}}$ *Ibid*, s. 5 (6) – includes a suit or arbitration proceedings relating to – (a) the existence of the amount of debt; (b) the quality of goods or service; or (c) the breach of a representation or warranty.

defined in the Indian legislation. Financial debt means a debt along with interest, if any, which is disbursed against the consideration for the time value of money and includes: -

- a. money borrowed against the payment of interest;
- any amount raised by acceptance under any acceptance credit facility or its dematerialised equivalent;
- any amount raised pursuant to any note purchase facility or the issue of bonds, notes, debentures, loan stock or any similar instrument;
- d. the amount of any liability in respect of any lease or hirer purchase contract which is deemed as a finance or capital lease under the Indian Accounting Standards or such other Accounting Standards as may be prescribed;
- e. receivables sold or discounted other than receivables sold on non-recourse basis;
- f. any amount raised under any other transaction, including any forward sale or purchase agreement, having the commercial effect of borrowing;
- g. any derivative transaction entered into in connection with protection against or benefit from fluctuation in any rate or price and for calculating the value of any derivative transaction, only the market value of such transaction shall be taken into account;
- h. any counter-indemnity obligation in respect of a guarantee, indemnity,
 bond, documentary letter of credit or any other instrument issued by a bank
 or financial institution;
- i. the amount of any liability in respect of any of the guarantee of indemnity for any of the items referred to in sub-clauses (a) to (h) of this clause.

The phrase "inability to pay debts" is defined in section 567 (1) of the CAMA as having the meaning assigned by section 409 of the CAMA. By virtue of Section 409 of the CAMA, a company is deemed to be unable to pay its debts in three circumstances, namely:

1. CAMA, section 409 (a) - the Statutory Demand

A creditor, by assignment or otherwise, to whom the company is indebted in a sum exceeding N2000 then due, has served on the company by leaving it at its registered office or head office, a demand under his hand requiring the company to pay the sum due, and the company has for three weeks thereafter neglected to pay the sum or to secure or compound for it to the reasonable satisfaction of the creditor.³⁰⁵

With regards to section 409 (a) of the CAMA, it has been held in a similar provision in the United Kingdom legislation,³⁰⁶ that the use of the words "is indebted" ... "then due" ... and "so due" have the consequence that a debt can only furnish the basis for a statutory demand served under the United Kingdom legislation if the debt is presently due at the date when the creditor seeks to utilise it for this purpose.³⁰⁷

A company is deemed to be unable to pay its debt if the court is satisfied that the company is unable to pay its debt, taking into account its contingent and prospective liabilities as well as its debts which are immediately payable. It seems that there are two tests of insolvency here. In other words, a company would be unable to pay its debts if it cannot pay them as they fall due, out of cash or readily realisable assets in its hands, and it will be immaterial that it could pay them over a lengthy period by a steady realization of all its assets. The company will also be unable to pay its debts if it has no reasonable prospects of paying all of them, both

³⁰⁵ CAMA, op cit, s.409 (a); Gateway Holdings Ltd v Sterling Asset Management & Trustees Limited [2016] 9 N.W.L.R (pt 1518) 490 at 514 – 515, paras E – C.

³⁰⁶ Insolvency Act, *op cit*, s. 123 (1) (a) (UK). ³⁰⁷ *Re Bryant Investment Co. Ltd* [1974] 1 W.L.R 826.

accrued and prospective, by a steady realization of all its assets, and in this case, it will be immaterial that it can pay its accrued debt out of its liquid resources.

In order to determine whether the company is insolvent in this second sense, it must be supposed that the company will cease carrying on its business immediately, except for fulfilling contracts which it has already entered into and it must then be asked whether the company would be able to satisfy its accrued liabilities and its liabilities under those contracts by steadily realizing all its assets. No regard must be paid to the profits which the company might earn or to any further liabilities which it might incur under fresh contracts if it continues to carry on its business.³⁰⁸

It is submitted that where a debt is *bona fide* and disputed by a company upon substantial grounds, the company's failure to pay will not afford a creditor the basis to petition for a winding up order, since it has been held that the words "neglected to pay" are to be understood to mean "omitted to pay without reasonable excuse".³⁰⁹ Hence, a petitioner in a winding up proceeding must establish not only that there is a debt owed but that the company is not disputing the debt in good faith. In a situation where the company disputes a debt, it is important to ascertain whether the dispute is one that is *bona fide* or substantial.³¹⁰

The elaborate interpretation of the word "debt" and "dispute" in the extant legislation of the Republic of India would afford a better understanding of the phrase "inability to pay debts". In view of the obvious lacuna in respect of the definition of these fundamental terms in the CAMA, reliance is ordinarily placed on the pronouncements from the courts in Nigeria as relating to section 409 of the CAMA.

 ³⁰⁸ In *Re: Medipharm Publications (Nig.) Ltd* (1970) (2) African LR (Comm.) 287 at 293-294; E
 Chianu, *Company Law* (Abuja: Law Lords Publication, 2012) pp. 635-636.
 ³⁰⁹ *Re Lympne Investments Ltd* [1972] 2 All ER 385.

³¹⁰ Fletcher, *op cit*, p. 524; *Pharma-Deko Plc v Financial Derivates Co Ltd* [2015] 10 N.W.L.R (pt 1467) 225 at 252, paras C – E.

However, there is a consensus that where a debt is disputed in good faith or on substantial grounds, winding up of a company based on its inability to pay debt is not the appropriate remedy. In *Oriental Airlines Limited v Air Via Ltd*,³¹¹ the Court of Appeal held thus:

But where there is excuse for non-payment within the prescribed time limit has been held not to amount to neglect. So, where the debt is disputed bona fide, an order for winding up will not be made. Where the debt is *bona fide* disputed by the company, the creditor must seek his remedy in action to establish the debt. The procedure should not be taken in order to extort the payment of a disputed debt. The machinery of a winding up petition should not be converted to an engine for debt collection in circumvention of the established legal procedure for instituting action in appropriate court for the collection of debts.... Where there is a genuine dispute as to the indebtedness of the company, that issue must first be resolved before the petition proceedings are continued. A winding up order of a company would be refused where the insolvency or the inability to pay the debt is the ground for the petition if the debt alleged is in dispute....

In determining the substantiality of the debt in a winding up petition, it does not matter whether they were raised early enough to deny the debt or they were put

³¹¹[1998] 12 N.W.L.R (pt 577) 271 at 280-281, per Musdapher JCA (as he then was); *Air Via Ltd v Oriental Airlines Ltd* (2004) 4 SC (pt. 2) 37.

forward after a misguided petition has been instituted. The ground of dispute can be raised at any time.³¹² Furthermore, in *Hansa International Construction Ltd v Mobil Producing Nigeria*, the Court of Appeal also held that where a petition is based on a debt which is disputed on substantial grounds, the petition will fail. The Supreme Court of Nigeria in *Ado Ibrahim & Co. Ltd v BCC Ltd*³¹³ held thus:

.... However, where the debt in issue is colossal such that the assets of the company, if left untouched, will not in the nearest future, having regards to galloping inflation, be sufficient to pay off the debt when they are realised, justice demands that the court exercises restraint in acceding to a winding up prayer and wait for the determination of the suit relating to the disputed debt.³¹⁴

It is unfortunate that the Supreme Court of Nigeria in the determination of *Ado Ibrahim & Co. Ltd v BCC Ltd* failed to take cognisance of the provisions of section 409 (a) of the CAMA to the effect that the quantum of debt required to establish a company's inability to pay debt is when the debt is in a sum exceeding N 2, 000. $00.^{315}$

³¹² Onochie v Alan Dick & Co Ltd [2003] 11 N.W.L.R (pt 832) 451 at 461, paras A – D.

³¹³ [2007] 15 N.W.L.R (pt 1058) 538.

³¹⁴*Ibid*, at 546.

³¹⁵CAMA, *op cit*, s. 409 (a); Insolvency Act, *op cit*, s. 123 (1) (a), the sum of £750 is stipulated in the United Kingdom for England; the Companies Act 2013, s. 271 (2) (a) (India), the sum of one Lakh Rupees is stipulated. However, this provision has been repealed by the Insolvency and Bankruptcy Code, *op cit*, s. 252. Thus, in the Republic of India, the liquidation of a company is invoked not because of the company's inability to pay its debts but by virtue of the circumstances stated in the Insolvency and Bankruptcy Code, s. 33, primarily that the Corporate Insolvency Resolution process with respect to the company failed.

While it seems settled that where a debt is disputed *bona fide* or on substantial grounds, a petition for winding up of a debtor company may not succeed, it still remains arguable if where the dispute is as to the exact amount owed by the debtor company, whether the petition for winding up on the ground of the company's inability to pay its debt will succeed. In Weide & Co. (Nig) Ltd v Weide & Co. Hamburg,³¹⁶ the Court of Appeal held that "... where a debt is in dispute, the burden is on the creditor to prove not only that a debt is owed but the particular amount owed".³¹⁷

However, the same Court of Appeal in Unifam Industries Ltd v Oceanic Bank International (Nig) Ltd³¹⁸ in answer to the question: should a court refuse to proceed with a winding up proceedings because there is a dispute as to the amount owed by respondent (company) to the petitioner, stated that:

> ... moreover, it seems to me that it would be in many cases, be quite unjust to refuse a winding up order to a petitioner who is admittedly owed monies which have not been paid merely because there is a dispute as to the precise amount owing.

It is pertinent to state that the Supreme Court of Nigeria in Ado Ibrahim's case held that "... a court will not decline to entertain a petition from a creditor from winding up a company that is unable to pay its debt because there is a mere dispute as to how much the company is owing the creditor".³¹⁹

In this study, we align with the contention that it would be wrong to refuse a winding up order simply because there is a dispute as to the precise amount owing as

 ³¹⁶ [1992] 6 N.W.L.R (pt 249) 627.
 ³¹⁷ *Ibid*, 640.
 ³¹⁸ [2005] 3 N.W.L.R (pt 911) 83.

³¹⁹ Ado Ibrahim & Co. Ltd v BCC Ltd, supra, at 573.

long as the debt admitted is in excess of N 2, 000. 00 as provided by section 409 (a) of the CAMA and the company is also commercially insolvent.³²⁰

Suffice to say that by virtue of sections 408 (d) and 409 (a) of the CAMA, if a company is unable to pay its debt in excess of the sum of N 2, 000. 00, three weeks after demand has been made, then, such company is deemed unable to pay its debt and may be wound up by the court in Nigeria. It is therefore appropriate to state that a petitioner relying on paragraph (a) of section 409 of the CAMA must establish that:

1. There is an existing debt in the excess of the sum of N 2, 000. 00 and due.

In this study, we have established the presumption of insolvency law in Nigeria that the courts are astute to prevent creditors from relying on the provisions of section 409 (a) of the CAMA if the debt is disputed *bona fide* on substantial grounds by the company.³²¹

2. A demand has been made for its repayment

It is trite that a statutory demand³²² to the debtor company must be invoked in a creditor's claim for a liquidated sum based upon contract or tort. This is so regardless of the magnitude of the claim. Additionally, the statutory demand must be in the hands of the creditor or its assignee.³²³ The issue that was not provided in the CAMA or the Companies Winding up Rules is the prescribed form of the statutory demand.

³²¹ Hansa International Construction Ltd v Mobil Producing Nigeria, supra; Air Via Limited v Oriental Airlines Limited, supra; Unifam Ind. Ltd v Oceanic Bank Int'l Nigeria Ltd, supra, 99; Home Development Ltd v Scancila Constructing Co. Ltd [1994] 8 N.W.L.R (pt 362) 250; C C Wigwe, Introduction to Company Law and Practice (Accra: Mountcrest University Press, 2016) pp. 346 – 347. ³²²CAMA, op cit, s. 409 (a); International Merchant Bank Nigeria Ltd v Speegaffs Company Nigeria Ltd, supra, at 433.

³²⁰ J O Orojo, *Company Law and Practice in Nigeria* (5thedn. London: LexisNexis Butterworths Tolley, 2008) p. 454.

³²³ CAMA, *op cit*, s. 409 (a); *Tate Industries Plc v Devcom Merchant Bank Ltd* [2004] 17 N.W.L.R (pt 901) 182- where the court held that a demand under the hand of an external solicitor to the creditor does not constitute a demand under the hand of the creditor; in Chianu, *op cit*, p. 631 – it was stated that to meet the requirement of 'demand under his hand', in the case of a body corporate, the demand should be signed by a director, company secretary or other authorised officer by reason of the provisions of the CAMA, s. 77.

Hence, it has been submitted that the demand need not be in any form.³²⁴ Unlike the case in Nigeria, in the United Kingdom, the demand has to be in prescribed form.³²⁵ The significance of the statutory demand being in a prescribed form in the United Kingdom is that it makes provision or offers an option to the company to make an application to the court for an injunction restraining the creditor from presenting or advertising a petition for the winding up of the company.

The statutory demand is served by leaving it at the registered office or head office of the company. But where no such registered office exists, it has been contended that the demand may be validly served by leaving it at the company's actual place of business for the time being.³²⁶ Furthermore, it has been argued that a demand sent by post will be discountenanced as the extant provision states unequivocally that a demand is served by leaving it at the company's registered office, which provision is in tandem with the provisions of section 78 of the CAMA. On the other hand, there is the viewpoint that since the provision did not specify who is to leave the demand at the registered office or head office, it can be left there by the agent of the person making the demand including the post office.³²⁷

3. The company has neglected to pay after a period of three weeks from the day of demand and failed to secure or compound the debt to the reasonable satisfaction of the creditor.

It has been submitted that the period of three weeks is calculated in accordance with the rule that, for the purpose of establishing the running of a prescribed legal time limit, fractions of a day are ignored. Therefore, the day on which the demand is

³²⁴ Orojo, *op cit*, p. 452.

³²⁵ Insolvency Rules 1986, op cit, r. 4.6(1) (d) (as amended) by Insolvency (Amendment) Rules 2010, *op cit*, para 140 (2) (b) (UK). ³²⁶ Fletcher, *op cit*, p. 526.

³²⁷ Chianu, *op cit*, p. 633, relying on the provisions of the CAMA, s. 78 that provides thus: ... 'any other document may be served on a company by leaving it at or sending it by post to, the registered office or head office of the company; Mayson et al, op cit, p. 664.

served on the company is ignored and the period of 21 days starts to run from the beginning of the following day.³²⁸ In this study, we respectfully disagree with the contention that the prescribed time limit of three weeks is same as 21 days as the canon of construction of a period of time is as specifically provided in an enactment³²⁹ and not subject to whims.

It is pertinent to state that "secure" in the context of section 409 (a) of the CAMA does not mean the security given at the time of giving the credit. It refers to the security after the debt has become due and the creditor had made a demand to the company to pay the sum due, which the debtor company is expected to pay the sum or secure or compound for it to the reasonable satisfaction of the creditor within the prescribed time limit of three weeks of the demand.³³⁰

It should be noted that it is not necessary that the creditor who utilises the statutory demand procedure of section 409 (a) of the CAMA to first obtain judgment in respect of the debt upon which the demand is based. On the other hand, if the creditor has already obtained judgment against the company, it is not normally a valid objection to a petition for winding up that the debtor company disputes the claim, or is making a claim for even larger amount against the petitioner, or even that an appeal against the judgment is pending.³³¹

However, it is instructive that having grounds for the presentation of a petition does not necessarily entitle the creditor to the making of a winding up order. Winding

³²⁸ Chianu, *op cit*, p. 633.

³²⁹ Interpretation Act, Cap 123, LFN, 2004, s. 15 (day), s. 18 (month, year); contrast with Bodies Corporate (Official Liquidations) Act, op cit, ss. 3 (3), 4 (3) (Ghana) - twenty-one days is the prescribed limit for statutory demand. It is instructive that in the Republic of India and under the Companies Act, 2013, s. 271 (2) (a), now omitted in the Insolvency and Bankruptcy Code, op cit, s. 252, the prescribed time limit for a demand was 21 days.

³³⁰ Nationwide Development Company Ltd v U.B.A Plc [1996] 3 N.W.L.R (pt 437) 435 at 444; J E O Abugu, Principles of Corporate Law in Nigeria (2ndedn, Lagos: MIJ Professional Publishers Limited, 2014) p. 755. It is worthy of note that under Companies Act, op cit, s. 271 (2) (a) (India) (now omitted), instead of the word 'secure', the words 'provide adequate security' or 'restructure' was used. This provision seems a better phraseology than that in the CAMA. ³³¹ *Re Douglas Griggs Engineering Ltd* [1963] 1 Ch 13.

up is a collective or class remedy and an order may be refused if the petitioner is merely seeking to obtain some private advantage.³³² If the purpose of the petition is legitimate and not unreasonable, it does not matter that the motive of the petitioner is malicious. In such circumstance, a petitioner can still be entitled to a winding up order.³³³

2. CAMA, section 409 (b) – Unsatisfied Judgment Execution

A company shall be deemed to be unable to pay its debts if the execution or other processes issued on a judgment, Act or Order of any court in favour of a creditor of the company is returned unsatisfied in whole or in part. This provision presupposes that a creditor of the company has previously obtained judgment against it and has proceeded to seek enforcement of that judgment by levying execution on the company's property. If such a process fails to realise sufficient value of assets to satisfy the judgment debt, this event in itself is to be treated as signifying the company's insolvent state.334

It is pertinent to state that where a creditor relies on section 409 (b) of the CAMA for the winding up petition, the debtor company cannot normally dispute the debt in the winding up proceeding. To raise a valid plea of bona fide dispute in relation to a judgment debt, the debtor company may advance the defence that the judgment was obtained by fraud or entered in absence of jurisdiction or the judgment itself is a complete nullity.³³⁵

However, in Tony-Anthony Holdings Ltd v Commercial Bank for Africa, ³³⁶ the Court of Appeal curiously held that the plea of res judicata applies to winding up

³³² Farrah & Hannigan, op cit, p. 712.

³³³ International Merchant Bank Nigeria Ltd v Speegaffs Company Nigeria Ltd, supra, at 437.

³³⁴ CAMA, op cit, s. 567 (1). The definition of an "Insolvent person".

³³⁵ Nwankwo v Yar'Adua [2010] 12 N.W.L.R (pt 1209) 518; Babatunde v Olatunji [2000] 2 N.W.L.R (pt 646) 557; *Alawiye v Ogunsanya* [2013] 5 N.W.L.R (pt 1348) 570. ³³⁶ [2013] All FWLR (pt 698) 944.

proceedings. In this case, the first appellant was a customer of the respondent bank. It was granted various credit facilities in the course of their transaction. When it failed to pay back the loans, the respondent bank commenced an action in the High Court of Lagos State. It also obtained a *mareva* injunction against the appellant but the High Court granted judgment in the respondent bank's favour. When the appellant failed to pay the judgment debt, the Nigeria Deposit Insurance Corporation (NDIC) appointed Chief Ikoku to recover the debt at Failed Banks Tribunal (Tribunal). The said Chief Ikoku filed an application for the Tribunal to enter final judgment as per his claims. The Tribunal granted the application and also issued a bench warrant against the 2^{nd} appellant. The Federal High Court subsequently became seised of its jurisdiction over the matter upon the return to democracy in 1999. The 2nd appellant thereafter was arrested and remanded in prison. When the appellants still failed to pay the debt, the respondent bank filed an application seeking orders directing confiscation of 1st appellant's properties, sale of same, directing the winding up and committing the 2nd appellant to imprisonment. The appellants in response filed an application seeking order dismissing the action on grounds of *locus standi* of the respondent bank to commence the action and lack of jurisdiction to entertain the suit. The Federal High Court dismissed the application.

The appellants being dissatisfied with the ruling of the Federal High Court filed an appeal to the Court of Appeal challenging the decision of the Federal High Court. They contended that the lower court had no jurisdiction to entertain the suit on grounds that the suit is *res judicata*, the prosecution constituted an abuse of court process and lack of *locus standi* of the respondent bank. The Court of Appeal while allowing the appeal held that: ... where a court of competent jurisdiction has settled the matters in dispute between the parties, none of them are allowed to re-litigate same by bringing a fresh action. The matter is said to be *res judicata*. Thus, a successful plea of *res judicata* ousts the jurisdiction of the court in which it is raised. In the instant case, where there was a subsisting judgment on the recovery of debt between the parties, the Federal High Court erred by assuming jurisdiction on the matter subsequently commenced.³³⁷

In this study, it is submitted that the judgment of the Court of Appeal seems not to represent the correct position of the law. Since the respondent bank's claim before the Federal High Court included winding up of the 1st appellant for the inability to pay its debt even after the respective judgment of the High Court of Lagos State and the Failed Banks Tribunal, is one that rightly comes under section 409 (b) of the CAMA. In other words, the provision of section 409 (b) of the CAMA is an exception to the plea of *res judicata*.

While it has been propounded that prescription of time is inapplicable to a winding up of a company for its inability to pay its debts under section 409 (b) of the CAMA,³³⁸ in this study, it is submitted that by reason of the prescription of time as provided in the definition of an insolvent person in section 567 (1) of the CAMA, the prescribed time limit within which a petition can be brought under section 409 (b) of the CAMA is that the judgment execution or other process has remained unsatisfied for a period of not less than six weeks. In the case of winding up upon the application

³³⁷ Tony-Anthony Holdings Ltd v Commercial Bank for Africa, supra, at 948.

³³⁸ Chianu, op cit, p.634.

of the Asset Management Corporation of Nigeria (AMCON), for non-satisfaction of Judgment debt, the petition can be commenced if the debtor company failed to pay the judgment sum within 90 days from the date of judgment.³³⁹

3. CAMA, section 409 (c) – Commercial Insolvency

A company is deemed to be unable to pay its debt if the court after taking into account any contingent or prospective liability of the company is satisfied that the company is unable to pay its debts.³⁴⁰ A company's inability to pay its debt in this context is ordinarily known as 'commercial insolvency'.

Although this provision has been declared to be somewhat tautologous,³⁴¹ it is important to state that this procedure is rarely resorted to as a ground of winding up of a company for its inability to pay its debt because it is more difficult to apply as the assets of the debtor company will have to be valued, which valuation is often a matter of judgment and speculation, with the liabilities of the company, contingent and prospective, being taken into consideration.³⁴²

However, where a contingent or prospective creditor brings a petition, the court will not hear it until security for costs has been given and a *prima facie* case made for winding up.³⁴³

2.6 Procedure for Winding up by the Court

The power to make rules of court for carrying into effect the objects of the CAMA so far as they relate to the winding up of companies or generally in respect of other applications to a court under the CAMA is vested in the Chief Judge of the Federal

³³⁹AMCON Act, s.52 (5)

³⁴⁰ Insolvency Act, *op cit*, s. 123 (2).

³⁴¹ Fletcher, *op cit*, p. 527. It is said to be tantologous because s 123 (1) (e) of the Insolvency Act 1986 of United Kingdom provides that <u>a company is deemed unable to pay its debts</u> if its proved to the satisfaction of the court that the <u>company is unable to pay its debts</u> as they fall due.

³⁴² Candide-Johnson & Alex-Adedipe, *op cit*, p. 435.

³⁴³ CAMA, *op cit*, s. 410 (2) (c).

High Court³⁴⁴. Thus, the Companies Winding up Rules (CWR)³⁴⁵ is the applicable rule to winding up of companies in Nigeria.³⁴⁶

By virtue of the provisions of the Companies Winding up Rules,³⁴⁷ all proceedings in every winding up under the CAMA shall be governed by the provisions of the Rules. Furthermore, the forms in the appendix, subject to the power of the Chief Judge of the Federal High Court to either alter any forms specified in the appendix or substitute new forms *in lieu* thereof, shall be used where applicable in all winding up proceedings in Nigeria. But where there is no provision in the Rules in respect of any proceedings in or before the court, the Federal High Court (Civil Procedure) Rules shall apply.³⁴⁸

The winding up of a company in Nigeria is commenced by a Petition.³⁴⁹ Every petition shall be in form 2, 3, or 4 in the Appendix with such variations as the circumstances may require.³⁵⁰ The application to the court for a winding up of a company by Petition can be presented³⁵¹ either by:-

- i. The Company;
- ii. A Creditor, including a contingent or prospective creditor of the company;
- iii. The official receiver;
- iv. A Contributory;

³⁴⁴CAMA, *op cit*, s. 552 (1)

³⁴⁵ 2001. It is important to note that prior to the Companies Winding up Rules (CWR) 2001, Companies Winding up Rules 1983 made pursuant to section 375 of the Companies Act 1968 (now repealed) were applicable in all winding up proceedings in Nigeria; In the United Kingdom, the applicable rule is the Insolvency Rules 1986 (as amended).

³⁴⁶Honeywell Flour Mills Plc v Ecobank (Nig) Ltd [2016] 16 N.W.L.R (pt 1539) 387 at 435, paras C – D. ³⁴⁷CWR, op cit, r. 2; Honeywell Flour Mills Plc v Ecobank (Nig) Ltd, ibid.

³⁴⁸CWR, *op cit*, r. 183; *Honeywell Flour Mills Plc v Ecobank (Nig) Ltd*, *ibid* at 427, para E. ³⁴⁹CAMA, *op cit*, s.410 (1); Insolvency Act, *op cit*, s. 124 (1) (UK); Bodies Corporate (Official

Liquidations) Act, op cit, s. 1 (1) (b) and (c) (Ghana); Gbedu v Itie & ors, supra.

³⁵⁰ CWR, *op cit*, r. 15. Form 2 is the general form of a Petition; Form 3 is for a Petition by unpaid creditor on simple contract; Form 4 is for a Petition by minority shareholders; Insolvency Rules, *op cit*, r 4.7 (UK) – there is one form for winding up petition unlike the case in Nigeria.

³⁵¹ CAMA, *op cit*, s. 410 (a) to (h); Insolvency Act, *op cit*, s. 124 (1) (4) (UK). In the Republic of India, the liquidation of a company is not by a petition an originating process but is only invoked after a corporate insolvency resolution process under chapter II of the Insolvency and Bankruptcy Code 2016.

- v. A Trustee in bankruptcy to, or a contributory;
- vi. The Corporate Affairs Commission;
- vii. A Receiver if authorised by the instrument under which he was appointed; or;
- viii. By all or any of those parties, together or separately.

Notwithstanding the above provision of the CAMA, it is important to state that the Asset Management Corporation of Nigeria.³⁵² Nigeria Deposit Insurance Corporation (NDIC), a capital market operator³⁵³ in practice are statutorily empowered to bring an application by petition to the Federal High Court for winding up of a company. However, it is of common knowledge that a winding up petition pursuant to the provisions of section 410 (1) of the CAMA is usually presented by creditors or contributories.³⁵⁴ A contributory has been defined as "every person liable to contribute to the assets of a company in the event of its being wound up..."³⁵⁵ It also includes a 'fully paid up shareholder, provided he or she has a tangible interest in the winding up of the company, which is usually demonstrated by showing that the company has a surplus of assets over liabilities".³⁵⁶

In *First Equity Securities Ltd v Anozie*³⁵⁷ the Court of Appeal while adopting the definition of a contributory as stated in *Ado Ibrahim & Co. Ltd v B.C.C. Ltd*, departed from the pronouncement of the Supreme Court of Nigeria that a contributory shall be entitled to present a winding up petition only when there are assets for distribution to the benefit or interest of the petitioner if the order of winding up was made. Rather, the Court of Appeal held that by virtue of s.410 (4) of the CAMA, a contributory shall

³⁵² Asset Management Corporation of Nigeria (AMCON) Act 2010 (as amended 2015), s. 52. ³⁵³ISA, *op cit*, s.53(1) and the SEC s. *Ibid* s.53(2)

³⁵⁴ Banks and other Financial Institutions Act (BOFIA) Cap B3 LFN 2004 (as amended), s. 38 (1). ³⁵⁵ CAMA, *op cit*, s. 567 (1); Insolvency Act, *op cit*, s. 79 (1) (UK).

³⁵⁶ Ado Ibrahim & Co v B.C.C Ltd, supra, at 574, paras E-F.

³⁵⁷ [2015] 12 N.W.L.R (pt 1473) 337.

be entitled to present a winding up petition not withstanding that there may not be available assets on the winding up for distribution to contributories.³⁵⁸

In this study, the focus will be mainly on Petition presented by a creditor for winding up of a company for the inability to pay its debt. Hence, it is important to underscore the fact that winding up petition proceeding can have far reaching consequences on the company, its shareholder, employees, creditors and the society.³⁵⁹ The courts strictly follow the provisions and procedures specified in the CAMA and the Companies Winding up Rules for the winding up of a company. In this respect, the court in *Pharma-Deko Plc v Financial Derivates Co Ltd*, held that "Companies Winding up proceedings are specialised and unique.... This specialised guidelines and procedures have their own implications for the court and the parties. For example, once a law has prescribed a particular method of exercising a statutory power, any other method of exercise of it is excluded".³⁶⁰

Thus, where an application for winding up of a company is not presented by petition and in the prescribed form but through other modes of commencement of action such as writ of summons, originating summons or originating motion, the court would be robbed of jurisdiction and the application and entire proceedings void *abinitio*.³⁶¹

1. Presentation of Petition:

A Petition is presented by being filed at the Federal High Court's Registry.³⁶² Unlike the procedure in Nigeria, in the United Kingdom and in the case of England and

³⁵⁸ First Equity Securities Ltd v Anozie, supra, at 365 – 366, paras G – A.

³⁵⁹Air Via Limited v Oriental Airlines Limited supra, at 56; Chianu, op cit, p. 632.

 ³⁶⁰248 – 249, paras H – A; Honeywell Flour Mills Plc v Ecobank (Nig) Ltd, supra, at 435, paras C – D.
 ³⁶¹ Obiuweubi v C.B.N [2011] 7 N.W.L.R (pt 1247) 465; Madukolu v Nkemdilim (1962) 2 SCNLR 341; (1962) All NLR 581; Sokoto State Government v Kamdax (Nig) Ltd [2007] 7 N.W.L.R (pt 1034) 466 SC; Bank of Industry Limited v Awojugbagbe Light Industries Limited [2016] 1 N.W.L.R (pt 1493) 280.
 ³⁶² CAMA, op cit, s. 407; CWR, op cit, r. 16; Insolvency Act, op cit, s. 117 (1); Mercantile Bank of Nigeria Plc v Nwobodo [2000] N.W.L.R (pt 648) 297.

Wales, it is the High Court³⁶³ that exercises jurisdiction over winding up petitions. This jurisdiction is exercisable by every judge of the High Court. Although it is the practice for winding up petitions to be assigned to the Chancery Division of the High Court and when the Chancery Division of the High Court exercises jurisdiction under the Companies Act and the Insolvency Act, it is often referred to as the 'Companies Court' as a term of convenience. But where the amount of the company's share capital paid up or credited as paid up, does not exceed £120, then the County Court of the District where the registered office or head office of the company is situate can hear the winding up petition.³⁶⁴

In the Republic of India, liquidation of corporate person is presented and determined by the Adjudicating Authority, that is, the National Company Law Tribunal constituted under section 408 of the Companies Act³⁶⁵ having territorial jurisdiction over the place where the registered office of a company is located.³⁶⁶ Suffice it to say that in the Republic of India, the jurisdiction of a civil court or authority is ousted in respect of any matter on which the National Company Law Tribunal or the National Company Law Appellate Tribunal has jurisdiction.³⁶⁷ Instructively, there are time limits for determination of liquidation issues by the National Company Law Tribunal, National Company Law Appellate Tribunal and the Supreme Court.³⁶⁸

In the Republic of South Africa, the jurisdiction to determine corporate insolvency issues is vested on the Companies Tribunal.³⁶⁹ In the case of the Republic

³⁶³ Insolvency Act, *op cit*, s. 117 (1).
³⁶⁴ Supreme Court Act, 1981 (United Kingdom), s. 61 (1) and Schedule 1, para 1.

³⁶⁵ 2013 (Republic of India).

³⁶⁶ Insolvency and Bankruptcy Code (Republic of India), *op cit*, s. 60 (1).

³⁶⁷ *Ibid*, s. 63.

³⁶⁸ *Ibid*, chapter VI.

³⁶⁹ Companies Act 2008 (Republic of South Africa), ss. 193 – 195, as amended by Companies Act 2011,

ss. 112 and 113.

of Ghana, it is the High Court that has jurisdiction in respect of official winding up of a company.³⁷⁰

It is pertinent to state that where a petition is presented at the Federal High Court Registry in the wrong Judicial Division, it may be heard in the Judicial Division where it was presented, unless the court otherwise directs or the Respondent pleads specially in objection to the jurisdiction and in such a case, the presiding Judge shall transfer the matter to the appropriate Division.³⁷¹

However, it is worthy of note that the transfer of a matter to the appropriate Division of the court upon successful objection by the respondent will not invalidate the prior proceedings conducted before the transfer order.³⁷²

2. Verification of Petition:

It is mandatory that a petition be verified by an affidavit, which affidavit shall be in form 7 or 8 in the appendix with such variations as circumstances may require.³⁷³ Unlike the case in Nigeria, in the United Kingdom, the petition is not verified by an affidavit but a statement of truth.³⁷⁴ It is provided that an affidavit verifying petition has to be deposed by the petitioner, or by one of the petitioners. If the petitioner is a limited liability company, the affidavit verifying the petition has to be made by some director, secretary or other principal officer of the limited liability company.³⁷⁵ An

³⁷⁰ Bodies Corporate (Official Liquidations) Act, *op cit*, s. 66 – interpretation of the court as used in the Act to mean the High Court.

³⁷¹ Federal High Court (Civil Procedure) Rules 2009, Order 2, rr 3 & 4.

³⁷² *Ibid;* J Amadi, *Modern Civil Procedure law and Practice in Nigeria* (Port Harcourt: Pearl Publishers, 2014) p. 168.

 $^{^{373}}$ CWR, r. 18 (1) & (2). While form 7 is used as affidavit verifying petition of an individual, form 8 is used as affidavit verifying petition of a limited liability company; Insolvency Rules, rr. 4.7 (1), 4.12 (1) & (2) – (7); Insolvency (Amendment) Rules, 2010, rr. 142 (2) & 146 (2) & (3) – Affidavit wherever it appeared was substituted with statement of truth. 374 Local Lange Park (2) = (1) + (1)

³⁷⁴ Insolvency Rules, *op cit*, r. 4.12 (1).

³⁷⁵ CWR, *op cit*, r. 18 (1); Insolvency Rules, *op cit*, r. 4.12 (4) (5) (7) (8) – in the United Kingdom, the statement of truth is authenticated and not verified as is the case in Nigeria.

affidavit verifying petition is not to be filed along with the Petition but within four days after the petition is presented.³⁷⁶

It is trite that an affidavit verifying a petition is a sufficient *prima facie* evidence of the statements in the petition.³⁷⁷ Hence, where there is no affidavit verifying a petition, the petition would be struck out³⁷⁸ as it touches on the jurisdiction of the court.

However, there is a different position of the law that where a petition is not verified by an affidavit, it is against the requirements of the Companies Winding up Rules but such non-compliance does not render the petition void but only voidable.³⁷⁹ It is important to note that the Court of Appeal in this case interpreted the provision of the repealed Companies (Winding up) Rules, 1949, rule 30, precursor of the extant Companies Winding up Rules³⁸⁰ which requires any petition to be verified by an affidavit.

In this study, we submit that where a petition is not verified by an affidavit, although it is not in compliance with the Companies winding up Rules, the petition is not void but a defect or an irregularity that can be amended or remedied by an order of the court if such will not occasion injustice to the other party as there are enabling provisions in the Companies Winding up Rules for an amendment of petition and other processes and even extension of time.³⁸¹

³⁷⁶ CWR, *op cit*, r 18; *Gateway Holdings Ltd v Sterling Asset Management & Trust Limited, supra*, at 517, paras E –G; 518, paras D – E.

³⁷⁷ *Ibid*.

³⁷⁸Spectra Limited v Stabilini Vision Limited [1999] 6 N.W.L.R (pt 608) 631 at 638, para C, per Oguntade JCA (as he then was).

³⁷⁹ Melwani v Feed Nation Industry (Nig) Ltd (No.1) [1986] 5 N.W.L.R (pt 143) 587 at 588. ³⁸⁰ CWR, op cit, r. 18.

³⁸¹Ibid, rr 181, 182 (1) and 183; Federal High Court (Civil Procedure) Rules, *op cit*, order 17 (amendment of originating process) *Makinde v Orion Engr. Services (UK) Ltd* [2014] 11 N.W.L.R (pt 1417) 1; *N.D.D.C. v Precision Association Ltd* [2006] 16 N.W.L.R (pt 1006) 527 at 559 – 560, paras E – C; *International Merchant Bank Nigeria Ltd v Speegaffs Company Nigeria Ltd, supra*, at 432, paras G – H; *Gateway Holdings Ltd v Sterling Asset Management & Trust Limited, supra*, at 518 – 519.

3. Service of Petition

Service of Petition is basic or fundamental in the activation of the jurisdiction of the Federal High Court against any respondent. Thus, failure to serve a Petition is fatal to the competence of the Federal High Court to assume jurisdiction to hear and determine the Petition. Service of Petition, therefore, goes to the root of the jurisdiction of the Federal High Court, so that failure to comply with the requirements of service is capable of rendering the proceedings or any step taken in the winding up proceedings to be null and void.³⁸²

Service of Petition in a winding up proceedings, in the case of a company, is generally in the manner provided in the Rules.³⁸³ Service on a company is by leaving the Petition at its registered office, if any, and if there is no registered office, then service is at the principal or last known place of business of the company by leaving a copy of the petition with any member, officer or servant of the company.³⁸⁴ Where no such member, officer or servant can be found at the principal or last known place of business, service of petition is by leaving a copy at such registered office or principal place of business or by serving it on such member, officer or servant of the company as the court may direct and where the company is being wound up voluntarily, by service of the petition on the liquidator, if any, appointed for the purpose of winding up the affairs of the company.³⁸⁵

In the light of the above, there are two major ways by which a petition could be served on a respondent company under the Rules. The first is by service on the

³⁸² Pharma-Deko Plc v Financial Derivates Co Ltd, supra, at 249; Tony-Anthony's case, supra, at 39; Eimskip Ltd v Exquisite Ind (Nig) Ltd [2003] 4 N.W.L.R (pt 809) 88; (2003) FWLR (pt 151) 1842; Sken Consult Ltd v Ukey (1981) 12 NSCC 1; (1981) 1 SC 6.

³⁸³ CAMA, op cit, ss. 78 & 552 (1); CWR, op cit, r. 183.

³⁸⁴ CWR, *op cit*, r. 17 (1); Insolvency Rules, *op cit*, r. 4.8 (1), (2), (3) (UK) – the alternate provision for service on a company is by leaving a copy with a director or other officer or employee of the company instead of member, officer or servant of the company as provided in the Companies Winding up Rules. ³⁸⁵ CWR, *op cit*: Insolvency Rules, *op cit*

respondent company at its registered office or principal or last known place of business. The other way is by leaving a copy of the petition with any member, officer or servant of the company or leaving a copy at the registered office or principal place of business of the company.³⁸⁶ Instructively, there is provision for substituted service of a Petition as there are enabling provisions in the applicable rules that confers discretionary power on the Federal High Court as relating to service of petition.³⁸⁷

Although, the provision of the Rules that provides for service of petition on a company by leaving a copy of the petition with any member, officer or servant seems understandable on the face of it, in this study, it is submitted that the provision for service of petition on a company by leaving a copy of the petition with any "officer", or "servant" appears redundant and obfuscating. While an "officer" is defined in the CAMA in relation to a body corporate to include 'a director, manager or secretary', ³⁸⁸ a servant is not defined in the CAMA or any other statute in force in Nigeria. Simply put, the word 'servant' is a generic term and ordinarily means "a person who works for a company or organisation; a person who works in another person's house and cooks, cleans, etc ...³⁸⁹ was inadvertently included among the persons through which service of a petition on a company can be validly effected. In other words, service of petition on a company can be effected by leaving a copy on junior employees of the company such as, messenger, driver, cleaner or any person found in the company, whether employee or not.

³⁸⁶ F.B.N Plc v Onukwugha [2005] 16 N.W.L.R (pt 950) 120.

³⁸⁷ Federal High Court (Civil Procedure) Rules, op cit, order 5; CWR, op cit, r. 17 (1); Insolvency Rules, *op cit*, r. 4.8 (6) (7) ³⁸⁸ CAMA, *op cit*, s. 567 (1).

³⁸⁹ A S Hornby, Oxford Advanced Learner's Dictionary of Current English (7thedn, Oxford: Oxford University Press, 2005) p. 1334; M Robinson (ed), Chambers 21st Century Dictionary (updated edn, Edinburgh: Chambers Harrap Publishers Ltd, 2004) p. 1281, "Servant" is a person employed by another to do household or menial work for them; a person who acts for the good of others in any capacity; a public servant.

In this study, it is submitted that the provisions of the Rules that validates the service of Petition by service on a servant of the company is unsatisfactory. In a situation where the modern rules of court in civil proceedings have not opened the gate so wide to admit the service of an originating process meant for a company on a junior employee of the company or a "servant",³⁹⁰ it is not readily decipherable the basis for the provision for service on a "servant" to be valid in companies winding up proceedings.

The proof of service of petition is by an Affidavit of Service³⁹¹ as in form 5 or 6 in the Appendix with such variations as circumstances may require.³⁹² The essence of an "affidavit of service" is to prove that the process emanating from the court has been brought to the notice of the litigant whose presence is required in court.³⁹³ In the United Kingdom, service of petition is proved by Certificate of Service.³⁹⁴

4. Advertisement of Petition

The condition precedent for hearing of a petition is the advertisement of the petition fifteen days before the hearing.³⁹⁵ The procedure is for the petitioner to apply by way of motion on notice for an order of the court to advertise the petition. Where an order for advertisement of the petition is made, the petition shall be advertised once or as many times as the court may direct, in the Federal Gazette and in one national daily newspaper and one other newspaper circulating in the State where the registered

³⁹⁰ *Kisari Investment Ltd v La-Terminal Co Ltd* (2001) FWLR (pt 66) 766; *Miden System Ltd v Effiong* [2011] 2 N.W.L.R (pt 1231) 354; *Mark v Eke* [2004] 5 N.W.L.R (pt 865) 54.

³⁹¹ Anyaoha v Chukwu [2008] 4 N.W.L.R (pt 1076) 31.

 $^{^{392}}$ CWR, *op cit*, r. 17 (2) – while Form 5 is used for Affidavit of service of petition on members, officers or servants, Form 6 is used for Affidavit of service of petition on Liquidator.

³⁹³ Martin Schroder & Co v Major & Co (Nig) Ltd [2002] FWLR (pt 128) 1304.

³⁹⁴ Insolvency Rules, *op cit*, r. 4.9A.

³⁹⁵ CWR, op cit, r. 19 (2) (a); Companies Proceedings Rules 2004, r. 9.

office or principal or last known principal place of business of the company is or was situate or in such other newspaper as directed by the court.³⁹⁶

The advertisement must state the day on which the petition was presented and the names and address of the petitioner and of his solicitor and shall contain a note at the foot thereof stating that any person who intends to appear at the hearing of the petition, either to oppose or support, must send notice of his intention to the petitioner or to his solicitor within the time and manner prescribed by the Rule. Where a petition is not advertised within the time prescribed in the Rules or the advertisement did not contain a note, it shall be deemed irregular and liable to be struck out, unless for sufficient reason given, the Court otherwise orders.³⁹⁷ The advertisement of the petition shall be in Form 9 or 10 in the appendix with such variations as the circumstances may require.³⁹⁸

In addition to the fact that the advertisement of a petition notifies all those interested in assets and continued existence of the company, as the contributories, members, directors and creditors, well in advance of the proposals to wind up the company so that they could file in their claims, if they so wish,³⁹⁹ it is a prerequisite or condition precedent for the appointment of Provisional Liquidator.⁴⁰⁰ However, it is important to note that the Court may refuse an application to advertise a petition if satisfied that the petition is not bona-fide but an abuse of the process of the court and

³⁹⁶ CWR, op cit, r. 19 (2) (b); Pharma-Deko Plc v Financial Derivates Co Ltd, supra, at 249; Air Via's case, supra; Fisher v Waste Management Ltd, unreported: Appeal No. CA/L/35/85 delivered on 14/5/87 in Abugu, op cit, pp. 758-759.

³⁹⁷ CWR, *op cit*, r. 19 (2) (c), (3).

³⁹⁸CWR, op cit, r. 19 (4). While Form 9 is for advertisement of petition, Form 10 is for advertisement of petition by minority shareholders. ³⁹⁹ Santana v Ajede (2000) 5 W.R.N 35 at 43.

⁴⁰⁰CWR, op cit, r. 21 (1); Savannah Bank of Nigeria Plc v Nigeria Deposit Insurance Corporation [2006] 9 N.W.L.R (pt 986) 424 at 441, paras E – H; General and Aviation Services Ltd v Thahal (2004) 4 SC (pt 1) 109 at 127.

may proceed to terminate the petition by striking it out or dismissing it.⁴⁰¹

Amendment of Petition 5.

Amendment of petition involves the alteration of the contents of the petition. The alteration may be for purpose of correcting an error or rectifying a mistake or omission in the petition. However, it is arguable whether a petition can be amended, having regards to the specialized and unique nature of companies winding up proceedings.⁴⁰² In this study, it is the position that a petition can be amended.⁴⁰³ The basis for amendment of petition is lent credence and justified by the pronouncement of the court in *Makinde v Orion Engr. Services (UK) Ltd*, ⁴⁰⁴ that:

> The amendment of the petition to join Orion project Services Nigeria Limited as the 1st petitioner for the purpose of complete composition of the action as to parties was properly made by the court below. Even the Almighty God amended creation to bring into existence woman to complement man and complete the sphere of human creation. The essence of the power of amendment is therefore to make whole what has been left undone either inadvertently or by blunder or oversight, in other to avoid injustice.

It is therefore important to state that an application for an amendment of petition is not granted as a matter of course. Although, the Federal High Court has the discretion to grant an application for amendment, such discretion must be exercised

⁴⁰¹ Unifam Ind. Ltd v Oceanic Bank Int'l Nigeria Ltd, supra, at 102, paras A – B; 103, paras F – G; First Equity Securities Ltd v Anozie supra, at 461, paras A - D.

Pharma-Deko Plc v Financial Derivates Co Ltd, supra, at 248.

⁴⁰³ CAMA, op cit, s. 552 (1); CWR, op cit, r. 183; Federal High Court (Civil Procedure) Rules, op cit, r. 17. ⁴⁰⁴ *Supra* at 30.

judicially and judiciously. Thus, where the purpose for an amendment of petition is to clarify the issues in controversy between the parties or to prevent any possible injustice in the matter and is not to overreach the adverse party, an amendment will be granted.⁴⁰⁵ Where however, the amendment is for petition that is fundamentally defective *ab initio*, it is incompetent and does not exist at all in law. Consequently, such incompetent petition or process cannot be amended.⁴⁰⁶

6. **Opposition to Winding Up Petition**:

The Supreme Court of Nigeria in *Air Via Ltd v Oriental Airlines Ltd*,⁴⁰⁷ set out four probable methods of answering an allegation of indebtedness or insolvency, to wit:

- i. To admit the debt;
- ii. To deny the debt;
- iii. To counter-claim against the debt;
- iv. To set-off against the debt.

It is rare for a debtor company in a winding up petition to admit the debt rather, the usual cause for such debtor company is to deny the debt and raise all manner of preliminary objections to the allegation of indebtedness. The procedure for a respondent in a winding up petition that intends to deny the allegation of indebtedness is to file an affidavit in opposition to the petition within ten days of the service of the petition.⁴⁰⁸ But where it is a person other than the respondent, such a person must file an affidavit in opposition to the petition within fifteen days of the date on which the petition was advertised.⁴⁰⁹ Notice of the filing of every affidavit in opposition to the

⁴⁰⁵*Makinde v Orion Engr. Services (UK) Ltd, supra; Ogidi v Egba* [1999] 10 N.W.L.R (pt 621) 42; *Kode v Yussuf* [2001] 4 N.W.L.R (pt 703) 392 at 412.

⁴⁰⁶ N.N.B Plc v Denclag Ltd [2005] 4 N.W.L.R (pt 916) 549; Akaniwo v Nsirim[2008] 9 N.W.L.R (pt 1093) 439; Salisu v Mobolaji [2014] 4 N.W.L.R (pt 1396) 1.

⁴⁰⁷ *Supra* at 52.

⁴⁰⁸ CWR, *op cit*, r. 25 (1).

⁴⁰⁹ *Ibid*; *Baykam Ventures Ltd v Oceanic Bank Int'l Nig. Ltd* [2005]All FWLR (Pt. 286) 648 at 668, paras. A-E.

petition shall be given to the petitioner or his solicitor. In the United Kingdom, the debtor company that intends to oppose the petition files in court a witness statement in opposition not less than five business days before the date fixed for the hearing.⁴¹⁰

However, where the respondent fails to file an affidavit in opposition to the petition, the petition would be regarded as an unopposed petition. The full effect of this is unclear except that an unopposed petition is disposed of.⁴¹¹ In this case,⁴¹² the court unequivocally stated that:

> When a Petition is considered unopposed by the mere fact that an affidavit in opposition was not filed or did not exist, it does not lead to an automatic winding up order. The petitioner or his solicitor shall satisfy the court, on the next adjourned date that the petition has been duly advertised, that the prescribed affidavit verifying the statements therein and the affidavit of service (if any) have been duly complied with the petitioner. No order than the one already made in respect or advertising the petition shall be made on the petition of the petitioner who has not, prior to the hearing of the petition, satisfied the court in the manner required by Rule 22 of the Companies Winding up Rule, 1983.

Unlike writ of summons where pleadings are filed, in a winding up petition, pleadings are not filed. However, in this study, it is the contention that because of the

 ⁴¹⁰ Insolvency Rules, op cit, r. 4.18 (1).
 ⁴¹¹ International Merchant Bank Nigeria Ltd v Speegaffs Company Nigeria Ltd, supra, at 434.

⁴¹² *Ibid*.

sui generis nature of a winding up petition, the petition, affidavit in opposition and affidavit in reply constitutes pleadings in winding up petition.⁴¹³ It is pertinent to state that there are distinctions between affidavit and pleadings. While pleadings are statement of facts, which must be proved on oath in court by a witness before it could constitute evidence, affidavit, on the other hand, though relating to facts sworn on oath, constitutes evidence without the need of oral testimony, subject to exceptions.⁴¹⁴

Since the primary statute governing the use of affidavit in Nigeria is the Evidence Act,⁴¹⁵ its provisions override that of the Rules of Court or any other Statute.⁴¹⁶ Thus, the parties in a winding up petition must strictly follow the provisions of the Evidence Act in preparing the affidavit in opposition to a petition or affidavit in reply. However, a defective or erroneous affidavit may be amended and re- sworn by leave of the court, on such terms as to time, cost or otherwise as seem reasonable.⁴¹⁷ Furthermore, since the Rules did not provide for the form and contents of either the affidavit in opposition or affidavit in reply, in this study, it is respectfully canvassed that the form and contents of affidavit as generally provided in the Evidence Act⁴¹⁸ would apply.

It is worthy to emphasize that while the contravention of the form of affidavit can almost always be cured,⁴¹⁹ the breach of the requirement as to contents of affidavit may not readily be cured.⁴²⁰ In this vein, it is advisable that parties in a

⁴¹³ *P* & *C.H.S. Ltd v Migfor (Nig) Ltd* [2013] 13 N.W.L.R (pt 1333) 555; *Agbakoba v INEC* [2008] 18 N.W.L.R (pt 1119) 489.

⁴¹⁴ Henry Stephens Engineering Ltd v S A Yakubu (Nig) Ltd [2009] 10 N.W.L.R (pt 1149) 416; Nigerian Navy v Garrick [2006] 4 N.W.L.R (pt 969) 69 at 112, para G.

⁴¹⁵ 2011.

⁴¹⁶ Elabanjo v Dawodu [2006] 15 N.W.L.R (pt 1001) 76.

⁴¹⁷ Evidence Act, *op cit*, s. 114; J Amadi, Contemporary Law of Evidence in Nigeria, Volume 1 (Port Harcourt: Pearl Publishers, 2011) pp. 1177 – 1184.

⁴¹⁸ Evidence Act, *ibid*, s. 115 (1) – (4) (contents of affidavit); S. 117 (form of affidavit); *Thahal's case* supra, 119 - 120.

⁴¹⁹ Evidence Act, *op cit*, s. 113.

⁴²⁰ S T Hon, *S.T. Hon's Law of Evidence in Nigeria*, Vol 2 (2ndedn, Port Harcourt: Pearl Publishers, 2013) pp. 1055 – 1072.

winding up petition must adhere to the requirements of the contents of affidavit as provided in the Evidence Act.

2.7 Commencement of Winding up Proceedings in Nigeria

The moment in time prescribed by law as the commencement of winding up is of considerable importance, since it is in relation to the commencement of winding up that many other matters are determined in a winding up proceedings. Where it is a voluntary winding up, the winding up of the company shall be deemed to have commenced at the time of the passing of the resolution by the company for voluntary winding up.⁴²¹ In any other case, the winding up of a company by the court shall be deemed to commence at the time of the presentation of the petition for the winding up.⁴²² However, in the Republic of Ghana, in other cases other than voluntary winding up, the winding up of a company is deemed to have commenced on the making of a winding up order.⁴²³

2.8 Interim Protection Measures

1. Avoidance of Property dispositions

Upon the commencement of winding up by the court, any disposition of the property of the company, including things in action and any transfer of shares or alteration in the status of the members of the company, shall, unless the court otherwise orders, be void.⁴²⁴ Since a compulsory winding up commences with the presentation of the petition, property which should be available to the creditors may have been disposed of in the period between the time of the presentation of the petition and the making of the winding up order. Hence, to preserve such property for the creditors, it is usually

⁴²¹ CAMA, *op cit*, s. 415 (1); Insolvency Act, *op cit*, s. 129 (1) (UK); Bodies Corporate (Official Liquidations) Act, *op cit*, s. 13 (Ghana).

⁴²²CAMA, *ibid*, s. 414 (2); Insolvency Act, *ibid*, s. 129 (2).

⁴²³ Bodies Corporate (Official Liquidations) Act, op cit, s. 13.

⁴²⁴ CAMA, op cit, s. 413; Insolvency Act, op cit, s.127 (UK); Utuk v The Liquidator Utuks Constructions and Marketing Ltd [2011] All FWLR (pt 554) 144 at 159, paras A–B.

provided that all dispositions of the company's property after the commencement of the winding up are void unless the court orders otherwise.

The purpose of preventing the disposal of property after commencement of winding up of a company and the need for court approval of certain transactions was summarised thus:⁴²⁵

this is a wholesome and necessary provision, to prevent during the period which must elapse before a petition can be heard, the improper alienation and dissipation of the property of a company in extremis. But where a company actually trading, which it is in the interest of everyone to preserve, and ultimately to sell, as a going concern, is made, the object of a winding up petition which may fail or may succeed, if it were to be supposed that transactions in the ordinary cause of its current trade, bona fide entered into and completed, would be avoided, and would not, in the discretion given to the court, be maintained, the result would be that the presentation of a petition, groundless or wellfounded, would, ipso facto, paralyse the trade of the company, and great injury, without any counterbalance of advantage, would be done to those interested in the assets of the company.

Unfortunately, the principles governing the exercise of the courts discretion to validate dispositions of property was not provided for in the CAMA or the Rules.

⁴²⁵ A Hicks & S H Goo, *Cases and Materials on Company Law* (2ndedn, London: Blackstone Press Limited, 1997) p. 629.

Hence, the guiding principles considered and approved by the court in *Re Gray's Inn* Construction Co. Ltd⁴²⁶ in the interpretation of section 127 of the Insolvency Act would apply in the court's exercise of discretion in respect of the provisions of section 413 of the CAMA. These guiding principles postulated by the court 427 are as follows:

- i. The discretion vested in the court is entirely at large, subject to the general principles which apply to any kind of discretion, and also subject to a limitation that the discretion must be exercised in the context of the liquidation provisions of the statute.
- ii. The basic principle of law governing the liquidation of insolvent estates, whether in bankruptcy or under the companies' legislation, is that the assets of the insolvent company at the time of the commencement of the liquidation will be distributed *pari passu* among the insolvent's unsecured creditors as at the date of the bankruptcy...
- iii. There are occasions, however, when it may be beneficial not only for the company but also for the unsecured creditors, that the company should be able to dispose of some of its property during the period after the petition has been presented, but before the winding up order has been made. Thus, it may sometimes be beneficial to the company and its creditors that the company should be able to continue the business in its ordinary course.
- iv.In considering whether to make a validating order, the court must always do its best to ensure that the interests of the unsecured creditors will not be prejudiced. v.The desirability of the company being enabled to carry on its business was often speculative. In each case, the court must carry out a balancing exercise.

⁴²⁶ [1980] 1 WLR 711. ⁴²⁷ *Ibid* at 717 – 719.

- vi. The court should not validate any transaction or series of transactions which might result in one or more pre-liquidation creditors being paid in full at the expense of other creditors, who will only receive a dividend, in the absence of special circumstances making such a course desirable in the interest of the creditors generally.
- vii.A disposition carried out in good faith in the ordinary course of business at a time when the parties were unaware that the petition has been presented would usually be validated by the court unless there is ground for thinking that the transaction may involve an attempt to prefer the disponee, in which case, the transaction would not be validated.
- viii.Despite the strength of the principle of securing *pari passu* distribution, the principle has no application to post-liquidation creditors; for example, the sale of an asset at full market value after the presentation of the petition. That is because such a transaction involves no dissipation of the company's assets for it does not reduce the value of its assets.

2. Avoidance of Attachments

Upon the commencement of the winding up of a company by the court in Nigeria, any attachment, sequestration, distress or execution put in force against the estate or effects of the company shall be void.⁴²⁸ While this provision is explicative enough, the provision does not extend to attachment, sequestration, distress or execution put in force before the commencement of the winding up of the company. In this study, it is the position that a party that intends to avoid such sequestration, attachment, distress or execution has to invoke the power of the court to stay or restrain an action or other proceedings against the company.

⁴²⁸ CAMA, op cit, s. 414; NDIC v Ifediegwu [2003] 1 N.W.L.R (pt 800) 56 at 81, paras. D-E.

3. **Stay of Proceedings**

Another consequence of commencement of winding up of a company is the invocation of the power of court to stay or restrain an action or other proceedings instituted or pending in any court (commonly referred to as "the court's concerned") before the making of the winding up order.⁴²⁹ By the express provision in the CAMA, it would appear that upon an application to the court concerned for an order staying or restraining an action or proceedings, the court concerned is bound to stay or restrain proceedings or refer the case to the court hearing the winding up petition.⁴³⁰

It is the trite position of law that the granting of a stay of proceedings is a matter of discretion depending on the facts and circumstances of each case and where all courts of record, whether trial or appellate, possess inherent powers to stay proceedings pending the determination of an appeal so as to preserve the subject matter of the litigation⁴³¹. In the case of *Klifco Ltd v Phillipp Holzmann A. G*, 432 the court affirmed the principle of law that there has to be a pending appeal before the granting of an application for stay of proceedings in a matter arising from a winding up petition. It would be arguable if this principle of having a pending appeal before the granting of an application of stay of proceedings would apply in the activation of section 412 of the CAMA. It is the position of this study, that since the principles adumbrated by the court in Klifco v Phillipp Holzmann A.G was on the stay of proceedings of a winding up order; it would not materially apply in a case of application for stay of proceedings upon the commencement of a winding up petition.

⁴²⁹ CAMA, op cit, s. 412; Insolvency Act, op cit, s. 126 (1).

⁴³⁰ CAMA, *ibid*.

 ⁴³¹ Nzeribe v Dave Engineering Co Ltd [1994] 8 N.W.L.R (pt 361) 124.
 ⁴³² [1996] 3 N.W.L.R (pt 430) 276.

4. Vulnerable Transactions: Fraudulent Preference

Upon the presentation of a winding up petition, where any conveyance, mortgage, delivery of goods, payment, execution or other acts relating to property of a company in liquidation is tinted with a fraudulent preference of its creditors, such transactions shall be invalid.⁴³³Unfortunately, the meaning of "fraudulent preference" is not provided for in the CAMA. However, it has been submitted that a company gives preference to a person if that person is one of the company's creditors or a surety or guarantee for any of the company's debts or other liabilities and the company does anything or suffers anything to be done, which in either case, has the effect of putting that person into a position which in the event of the company going into insolvent liquidation will be better than the position he would have been in if that thing had not been done.⁴³⁴

In this study, it is respectfully submitted that there seems to be an obvious lacuna in the provision on fraudulent preference under section 495 of the CAMA as same appears vague and ambiguous. For instance, it is provided that certain acts, deed or transactions shall be invalid or void but there are no express procedural steps and on whose jurisdiction is vested the power to declare such acts, deed or transactions invalid or void. This is unlike the position in the United Kingdom and the Republic of India where the court and the Adjudicating Authority respectively, have the powers to declare such acts, deeds or transactions to be invalid or void.⁴³⁵ Furthermore, unlike the position in the United Kingdom⁴³⁶ and the Republic of India,⁴³⁷ there is no prescribed time limit for the invocation of section 495 of the CAMA.

⁴³³ CAMA, *op cit*, s. 495.

⁴³⁴ Farrar & Hannigan, *op cit*, p. 731.

⁴³⁵ Insolvency Act, op cit, s. 239 (UK); Insolvency and Bankruptcy Code, op cit, ss. 43, 44 (India).

⁴³⁶ Insolvency Act, *ibid*, s. 240.

⁴³⁷ Insolvency and Bankruptcy Code, *op cit*, s. 43 (4).

5. Avoidance of Floating Charges

It is the law that a floating charge on the undertaking or property of the company created within 3 months of the commencement of the winding up, shall, unless it is proved that the company immediately after the creation of the charge, was solvent, be invalid. This is subject, however, to the exception that the amount of any cash paid to the company at the time of, and in consideration for, the charge together with interest on that amount at the current bank rate, shall not be rendered invalid.⁴³⁸ Thus, this provision is designed to invalidate floating charges given close to insolvency which simply secure past indebtedness and provide no new benefit, no new money, to the company. In other words, only the charge is rendered void. The underlying debt remains valid and if the debt has been repaid before winding up, the fact that the charge would have been void in the winding up does not affect the repayment. It should be noted that in appropriate circumstances, such repayment may be challenged as a fraudulent preference.⁴³⁹

2.9 Nature of Proceedings before Hearing of Winding up Petition

Upon commencement of a winding up petition and after the advertisement⁴⁴⁰ of the petition, on the application of a creditor, or of a contributory, or of the company, the court may appoint a provisional liquidator for the purpose of conducting the proceedings in the winding up a company and performing such duties as the court may impose from time to time. The application for appointment of a provisional liquidator must be supported by an affidavit⁴⁴¹ disclosing sufficient ground for the

⁴³⁸ CAMA, *op cit*, s. 498.

⁴³⁹ Mace Builders (Glasgow) Ltd v Lunn [1987] Ch 191; Power v Sharp Investments Ltd [1994] 1 B.C.L.C 111.

⁴⁴⁰ CWR, op cit, r. 21 (1); Savannah Bank of Nigeria Plc v NDIC, supra at 441.

⁴⁴¹ CWR, op cit, r. 21 (1).

appointment. In United Kingdom, the application to the court for the appointment of a provisional liquidator is supported by a witness statement.⁴⁴²

1. Appointment of a Provisional Liquidator

A provisional liquidator may be appointed by the court upon the application of a creditor or of a contributory of the company after the presentation of a petition and before the making of a winding up order.⁴⁴³ There are certain statutory provisions that empower some regulatory office holders such as the Governor of the Central Bank of Nigeria to appoint the Nigerian Deposit Insurance Corporation as either the provisional liquidator in the winding up proceedings of failed banks.⁴⁴⁴ Where a provisional liquidator is appointed for a company, he is not the agent of any of the parties but an officer of the court.⁴⁴⁵ The appointment of a provisional liquidator is not automatic⁴⁴⁶ simply because a petition has been filed. A court that is faced with an application for appointment of a provisional liquidator must take into consideration:

i.Proof by affidavit of sufficient grounds for the appointment as a provisional

Liquidator;⁴⁴⁷

ii.The previous experience of the provisional liquidator, the particulars of the nature and description of the property with which he had dealt being stated.⁴⁴⁸

It is important to note that the order appointing the provisional liquidator shall bear the number of the petition and state the nature and a short description of the property of which the provisional liquidator is ordered to take possession of and the

⁴⁴² Insolvency Rules, *op cit*, r. 4.25 (2).

⁴⁴³ CAMA, op cit, s. 422 (2); Insolvency Act, op cit, s. 135 (1).

⁴⁴⁴ BOFIA, *op cit*, s. 38 (2).

⁴⁴⁵ Provisional Liquidator of TAPP Industries Ltd v TAPP Industries Ltd &Ors. [1995] 5 N.W.L.R (pt 393) 9 at 33; D Sasegbon (ed), SASEGBON'S Laws of Nigeria Volume 3 (Lagos: DSC Publications, 2005) p. 828.

⁴⁴⁶ General and Aviation Services Ltd v Thahal, supra at 424.

⁴⁴⁷ CWR, *op cit*, r. 21 (1).

⁴⁴⁸*Ibid*, r. 21 (2).

duties to be performed by the provisional liquidator.⁴⁴⁹ The order appointing the provisional liquidator shall be in Form 11 of the appendix with necessary variations.⁴⁵⁰

2. Purpose, Duties, Powers and Effect of the Appointment of a Provisional Liquidator

The primary purpose for the appointment of a liquidator, whether provisional or otherwise, is to preserve the company's assets and to see that such assets before a winding up order is made by the court, is not dissipated. Where, therefore, the court has appointed a provisional liquidator, he shall be entitled to take into his custody or under his control all the property and choses in action to which the company is or appears to be entitled pending the making of a winding up order by the court.⁴⁵¹

The powers of a provisional liquidator are specified in the order appointing the provisional liquidator.⁴⁵² Where the powers of a provisional liquidator is not restricted in the order of appointment, such a provisional liquidator exercises the general powers of a liquidator; the fact that his appointment is "provisional" qualifies the period of his appointment and not the powers conferred on him. However, the exercise by the provisional liquidator of the powers so conferred is at all times subject to the control of the court.⁴⁵³ Notwithstanding, the Provisional Liquidator seems not to have authority to wind up the company or to distribute its assets.⁴⁵⁴

A provisional liquidator is generally appointed where the assets of the company are in jeopardy and the primary object of his appointment is to prevent the directors of the company from dissipating such assets of the company. It follows therefore that

⁴⁴⁹CWR, op cit, r. 21 (2).

⁴⁵⁰ *Ibid*, r. 22(5).

 ⁴⁵¹ Provisional Liquidator of TAPP Industries Ltd v TAPP Industries Ltd & Ors., supra, at 38 – 39.
 ⁴⁵² CAMA, op cit, s. 422 (2); Insolvency Rules, op cit, r 4.26 (1); Provisional Liquidator of TAPP Industries Ltd v TAPP Industries Ltd &Ors, supra; General and Aviation Services Ltd v Thahal, supra.
 ⁴⁵³ CAMA, ibid, s. 425 (3).

⁴⁵⁴ Fletcher, *op cit*, p. 551.

his duty is to preserve the company's assets before a winding up order is made. Unless expressly restricted therefore, the provisional liquidator has the authority of any liquidator to secure the safety of the properties that would appear to belong to the company being wound up.⁴⁵⁵ Notwithstanding, the position of the law seems to be that the power of the provisional liquidator cannot be limited by the instrument of appointment and where there are conditions that may be attached to the appointment of a provisional liquidator, such conditions only go towards strengthening the main objective of the appointment and not derogate therefrom.⁴⁵⁶ It is worthwhile to state that any creditor or contributory or the company may challenge the exercise of the powers by the provisional liquidator.⁴⁵⁷

It is the primary duty of a provisional liquidator to take into his custody or under his control all the property of the company.⁴⁵⁸ Subject to this, his duties depend on the terms of the order appointing him. Most commonly his powers are restricted to taking possession of and protecting the assets of the company. It is further posited that a provisional liquidator has the duty to see to the insurance of the property of the company and to take special care to secure the books of the company. Where the provisional liquidator finds that his powers as contained in the order of court appointing him are not sufficiently wide, he can apply to the Court for liberty to carry on with consequential directions.⁴⁵⁹

3. Effects of the Appointment of a Provisional Liquidator

It is arguable whether on the appointment of a provisional liquidator, all the powers of directors of a company ceases, except so far as the court may by order, sanction the

⁴⁵⁵ CAMA, *op cit*; *Anakwenze v TAPP Industry Ltd* [1992] 7 N.W.L.R (pt 252) 142.

⁴⁵⁶ Anakwenze v TAPP Industry Ltd, supra.

⁴⁵⁷ Ibid; Savannah of Nigeria Plc v NDIC, supra, at 445; NDIC v FNB Ltd [1997] 4 N.W.L.R (pt 501) 519.

⁴⁵⁸*Provisional Liquidator of TAPP Industries Ltd v TAPP Industries Ltd &Ors, supra,* 40. ⁴⁵⁹*Ibid*, paras F–H.

continuance.⁴⁶⁰ However, there is the divergent view that the provisional liquidator, although under the control of the court, stands generally within its restricted powers, in place of the board of directors of the company, the board may still exercise some residual powers.⁴⁶¹

Furthermore, where a provisional liquidator is appointed for a company, no action or proceedings can be proceeded with or commenced against the company except by leave of the Federal High Court given on such terms as the court may impose.⁴⁶² It is pertinent to state that the "court" whose leave is required before proceeding with or commencing any action against a company in liquidation is the Federal High Court, not any other court.⁴⁶³ However, the Supreme Court of Nigeria in *Onwuchekwa v NDIC*⁴⁶⁴ clarified the provisions of section 417 of the CAMA when it held thus:

The provisions of an enactment should not be read so as to deny access to the court. There is nothing in section 417 which prohibits such company as is described in the section from proceeding with action or proceedings against another person. What that section prohibited subject to the leave of the court is proceeding with an action or proceedings against the company...

⁴⁶⁰ CAMA, *op cit*, s. 422 (9); *FMB Ltd v NDIC* [1995] 6 N.W.L.R (pt 400) 226 at 245 – 246. ⁴⁶¹ Anakwenze v TAPP Industry Ltd, supra, at 159.

⁴⁶²NDIC v Akahall & Sons Ltd (2004) All FWLR (pt 229) 897 at 922 – 924, paras. G–A; Progress Bank (Nig) Plc v O.K. Contact Point Holdings Ltd [2008] 1 N.W.L.R (pt 1069) 514 at 529; Agro Allied Development Ent Ltd v MV Northern Reefer [2009] 12 N.W.L.R. (pt 1155) 255 at 268 – 269; Ezeobi v Daily Times of Nigeria Plc [2013] All FWLR (pt 672) 1778 at 1788 – 1789, paras G –A; Atoju v Triumph Bank Plc [2016] 5 N.W.L.R (pt 1505)252 at 312, paras C – D.
⁴⁶³ MV Northern Reefer's case, ibid, 270, paras E – F.

⁴⁶⁴ [2002] FWLR (pt 101) 1015 at 1631, paras. C–D; [2002] 5 N.W.L.R (pt 766) 371 at 393, para F.

Suffice it to say that while section 417 of the CAMA does not prohibit the continuation of proceedings or commencement of an action against a company in liquidation, such proceedings or action can only be continued or commenced only with the leave of the Federal High Court. Where such leave is not obtained, an action commenced or proceedings continued would be incompetent.⁴⁶⁵

Termination of Appointment of Provisional Liquidator 4.

While there is no specific provision on the termination of the appointment of a provisional liquidator in the extant Statute or the Rules in Nigeria, in this study, it is respectfully submitted that the appointment of a provisional liquidator terminates upon the happening of the following events:

The making of a winding up order i.

Upon the making of a winding up order, the life of a provisional liquidator automatically ceases. Since upon the making of a winding up order, the next procedural step is the appointment of a liquidator and if no liquidator is appointed, the official receiver shall by virtue of his office become the liquidator.⁴⁶⁶

ii. Vacation of office

The appointment of a provisional liquidator shall be deemed terminated if the provisional liquidator resigns or is removed by the court upon the application of a creditor or a contributory of the company or for failure to notify his appointment to the Corporate Affairs Commission and to give security.⁴⁶⁷

⁴⁶⁵ Atolagbe v Awoni [1997] 9 N.W.L.R (pt 522) 536; Ezeobi v Daily Times of Nigeria Plc, supra; Omaghoni v Nigeria Airways Ltd [2006] 18 N.W.L.R (pt 1011) 310 at 328, paras E - H; 339, paras D -F. 466 CAMA, *op cit*, s. 422 (3) (b); Insolvency Rules, *op cit*, r. 4.31 (4) (UK).

⁴⁶⁷ CAMA, *ibid*, s. 422 (3) (d), (5); CWR, *op cit*, r. 43 (1); Insolvency Rules, *ibid*, r. 4.31 (1).

iii. Dismissal of the winding up petition

The life of a provisional liquidator terminates upon the dismissal of the winding up petition, either on the merits or otherwise.⁴⁶⁸

It is pertinent to state that there are obvious *lacunae* as it relates to the law and procedure relating to the appointment of a provisional liquidator in winding up proceedings in Nigeria. The most prominent drawback in the appointment of provisional liquidator is that there are no concise qualification requirements for the appointment of provisional liquidators. In other words, the office of a provisional liquidator is made to be an 'all comers' position. This is quite unsatisfactory. Furthermore, the office of the provisional liquidator in winding up proceedings is not satisfactorily regulated.

In other countries such as the United Kingdom, the Republic of India and the Republic of South Africa, to practice or function as an insolvency practitioner, whether as provisional liquidator or liquidator or business rescue practitioner or insolvency professional, the person must be qualified as a member of a recognised professional body, subject to regulation of a regulatory body.⁴⁶⁹ The essence of providing for qualification and regulation of persons that function as insolvency practitioner cannot be ignored. The powers and duties of a provisional liquidator are quite far-reaching and expansive so that if not conferred on fit and proper persons and subjected to regulatory supervision, may be abused.

2.10 Hearing of Winding up Petition

After the advertisement of petition, the petitioner on the next adjourned date has to satisfy the court that the petition has been duly advertised, the prescribed affidavit

⁴⁶⁸ CAMA, *op cit*, s. 411 (1); Insolvency Rules, *op cit*, r. 4.31 (2) (UK).

⁴⁶⁹ Insolvency Act, *op cit*, chapter 13 as amended by Small Business, Enterprise and Employment Act 2015, Part 10, ss. 137 – 146(UK); Insolvency and Bankruptcy Code, *op cit*, chapter 4, ss. 207,

²⁰⁸⁽India); Companies Act, *op cit*, ss. 138 – 140 as amended Companies Amendment Act 2011, ss. 88 – 90 (South Africa).

verifying the statements therein and the affidavit of service have been duly complied with.⁴⁷⁰ Subsequently, at the hearing of a petition, the court may dismiss it or adjourn the hearing conditionally or unconditionally, or make any interim other, or any other order that it thinks fit.⁴⁷¹ Where the court is satisfied that the preconditions to a successful winding up petition for inability to pay debt has been established, the court shall exercise its expansive powers under section 411 (1) of the CAMA to make an order winding up the company.⁴⁷² In other words, the making of a winding up order is not automatic but subject to the petitioner satisfying the court in the manner required by the Rules.⁴⁷³

It is important to state that at the hearing of a petition, upon application, the court can make an order dismissing or striking out the petition or restraining the petitioner from continuing with the winding up petition on the ground that it constitutes an abuse of process or it was not brought *bona fide* particularly in cases where the debt is being disputed.⁴⁷⁴ Unfortunately, in Nigeria, unlike in the United Kingdom, the Republic of South Africa and the Republic of India, there are no expansive and streamlined rescue remedies to winding up in the extant statute that can be activated when a company is insolvent.

2.11 Winding up Order

This is an order of the Federal High Court that the affairs of a company be wound up. Where a winding up order is made, copies of the order shall be forwarded by the

⁴⁷⁴Air Via Limited v Oriental Airlines Limited, supra; Ado Ibrahim & Co. Ltd v B.C.C. Ltd, supra, UBN Ltd v Tropic Foods Ltd [1992] 3 N.W.L.R. (pt 228) 231 at 250; Tony-Anthony's case, supra; F

Uzuegbunam, 'Ecobank Loses in Court against Honeywell', Business Day Newspaper, June 29, 2016, p. 6.

⁴⁷⁰ CWR, *op cit*, r. 22 (1).

⁴⁷¹ CAMA, *op cit*, s. 411 (1).

⁴⁷² Okafor v Igwilo [1997] 11 N.W.L.R. (pt 527) 36 at 51 – 52.

⁴⁷³ International Merchant Bank Nigeria Ltd v Speegaffs Company Nigeria Ltd, supra, at 435.

Registrar to the Official Receiver,⁴⁷⁵ who will then transmit a copy of the order to the Corporate Affairs Commission which shall make a minute thereof in its book relating to the company.⁴⁷⁶ The order for the winding up of a company shall be in Form 16 in the appendix with such variations as circumstances may require.⁴⁷⁷ The making of a winding up order by the court attracts some far-reaching consequences in the life of the company wound up.

2.11.1 Consequences of a Winding up Order

Where a winding up order is made, it elicits certain consequences, to wit;

i.Cessation of the Powers of Board of Directors of the Company

Where a winding up order is made, the powers of the board of directors of the company ceases and such powers is then invested on the liquidator, except the court by order⁴⁷⁸ or the liquidator⁴⁷⁹ sanctions the continuance of the power of the board of directors. It is doubtful if the board of directors of the company can exercise even residual powers without the authorisation of the court.

ii.Commencement, Transfer and Continuation of Actions and Proceedings

Where a winding up order is made, no action or proceeding shall be proceeded with or commenced against the company except by leave of court given on such terms as the court may impose.⁴⁸⁰

The consequence of winding up order as it relates to commencement of action or continuation of proceedings have been discussed in this chapter.

⁴⁷⁵ CWR, *op cit*, rr. 28, 29, 30 (1).

⁴⁷⁶ CAMA, *op cit*, s. 416.

⁴⁷⁷ CWR, *op cit*, r 29 (2).

⁴⁷⁸ CAMA, op cit, s. 422 (9); FMB Ltd v NDIC, supra.

⁴⁷⁹ Bodies Corporate (Official Liquidation) Act, *op cit*, s. 14.

⁴⁸⁰ CAMA, *op cit*, s. 417; Insolvency Act, *op cit*, s. 130 (2); Bodies Corporate (Official Liquidation) Act, *op cit*, s. 17.

iii.Stay of Winding up Proceedings

Upon the making of a winding up order, the Federal High Court on the application of a Liquidator or the Official Receiver, or any Creditor or Contributory, may make an order staying the proceedings either all together or for a limited time, on such terms and conditions as the court thinks fit.⁴⁸¹ Instructively, it should be noted that a copy of every order of stay made after the winding up order shall be forwarded to the Corporate Affairs Commission which shall enter it in its records relating to the company.⁴⁸² As earlier discussed in this chapter, an order for stay of proceedings is not granted as a matter of course.

iv.Submission of Statement of Company's Affairs to Official Receiver

Where the court has made a winding up order and unless the court thinks fit to order otherwise, a statement as to the affairs of the company in the prescribed form, verified by affidavit by one or more of the persons who are the directors of the company, the secretary, or such other persons as the official receiver may direct, is to be submitted to the official receiver within 14 days of the winding up order.⁴⁸³ The requirement to submit statement of affairs may be dispensed with under certain special circumstances.⁴⁸⁴

v.Settlement of List of Contributories and Application of Assets; Making of

Calls:

It is after the making of a winding up order that the court shall settle a list of contributories and may rectify the register of members in all cases where rectification

⁴⁸¹ CAMA, op cit, s. 438 (1) (2); Insolvency Act, op cit, s. 147 (1); CWR, op cit, r. 32 (1).

⁴⁸² CAMA, *ibid*, s. 438 (4); Insolvency Act, *ibid*, s. 147 (3).

⁴⁸³ CAMA, *ibid*, s. 420 (1) (2) (3) (8); Insolvency Act, *ibid*, s. 131(1) (2) (3) (4) (6) (UK) – in this case, the period within which the statement of affairs is to be submitted to the official Receiver is 21 days; Bodies Corporate (Official Liquidation) Act, *op cit*, s. 19 – in the Republic of Ghana, the statement of affairs is submitted to the liquidator, who is the Registrar of Companies; CWR, *op cit*, rr. 35 – 40; Insolvency Rules, *op cit*, chapter 6.

⁴⁸⁴ CWR, *ibid*, r. 40; Insolvency Rules, *ibid*, r. 4.46.

is required and cause the assets of the company to be collected and applied in discharge of its liabilities.⁴⁸⁵ The court may dispense with the settlement of a list of contributories where it appears not necessary to make calls on or adjust the right of contributories.⁴⁸⁶ This power vested in the court to settle a list of contributories, in practice, is exercised by the liquidator as an officer of the court.⁴⁸⁷

Furthermore, the making of a winding up order empowers the court to make calls on all or any of the contributories for the time being settled on the list of contributories to the extent of their liability, for payment of any money which the court considers necessary to satisfy the debts and liabilities of the company and the cost, charges and expenses of the winding up proceedings.⁴⁸⁸ This power vested in the court to make calls on contributories, in practice, is exercised by the liquidator as an officer of the court.⁴⁸⁹

vi.Collective Regime

The making of a winding up order activates the notion of collectivity in insolvency proceedings. In other words, the effects of a winding up order does not inure on the petitioner alone but operates in favour of all the creditors and of all the contributories of the company as if made on the joint petition of a creditor and of a contributory.⁴⁹⁰ Hence, the making of a winding up order imposes a collective regime involving the abatement of all claims of creditors, who are ranked equally under the *pari passu* rule who now share in the distribution under the winding up proceedings.

⁴⁸⁵ CAMA, *op cit*, s. 439 (1); Insolvency Act, *op cit*, s. 148 (1); Bodies Corporate (Official Liquidation) Act, *op cit*, s. 20 (1) – in the Republic of Ghana, it is the liquidator that settles the list of contributories.
⁴⁸⁶ CAMA, *ibid*, 439 (2); Insolvency Act, *ibid*, s. 148 (2); Bodies Corporate (Official Liquidation) Act, *ibid*, s. 20 (2).

⁴⁸⁷ CWR, *op cit*, rr. 63 – 66.

⁴⁸⁸ CAMA, op cit, s. 442 (1); Insolvency Act, op cit, s. 150 (1).

⁴⁸⁹ CWR, *op cit*, rr. 69 – 73.

⁴⁹⁰ CAMA, op cit, s. 418; Insolvency Act, op cit, s.130 (4).

vii.Appointment of Liquidator

It is upon the making of a winding up order by the court that a liquidator would be appointed for the purpose of conducting the proceedings in winding up a company and performing such duties as the court may impose.⁴⁹¹ Where no liquidator is appointed or there is a vacancy, the official receiver shall by virtue of his office become the liquidator or act as liquidator until such time as the vacancy is filled.⁴⁹²

viii.Delivery of Property to Liquidator

On the making of a winding up order, the court may require any contributory for the time being on the list of contributories and any trustee, receiver, banker, agent or officer of the company, to pay, deliver, convey, surrender or transfer immediately or within such time as given by the court, to the liquidator, any money, property or books in his hands to which the company is *prima facie* entitled.⁴⁹³

ix.Contract of Employment

The making of a winding up order constitutes a ground for the termination by operation of law of a contract of employment.⁴⁹⁴ However, in the case of voluntary winding up, the employee may still be entitled to damages.⁴⁹⁵ In this study, it is respectfully submitted that winding up order made by a court does not automatically terminate existing contracts of employment but rather constitute an implied notice to the employees of the close determination of their employment.⁴⁹⁶

2.12 The Official Receiver

The Official Receiver plays a significant role in winding up proceedings. Official Receiver means the Deputy Chief Registrar of the Federal High Court or an officer

⁴⁹¹ CAMA, op cit, s. 422 (1); CWR, op cit, r. 41.

⁴⁹² CAMA, *ibid*, s. 422 (1) (3) (b).

⁴⁹³*Ibid*, s. 440; CWR, *op cit*, r. 62.

⁴⁹⁴ E E Ama Oji & O D Amucheazi, *Employment & Labour Law in Nigeria* (Lagos: Mbeyi & Associates (Nig) Ltd, 2015) p. 341.

⁴⁹⁵ *Ibid*.

⁴⁹⁶ *Gbedu v Itie & ors, supra,* at 261 – 266.

designated for the purpose by the Chief Judge of the Court.⁴⁹⁷ In the United Kingdom, official receivers are officers of the insolvency service, an Executive Agency of the Department for Business Innovation and Skills, attached to courts having bankruptcy jurisdiction. Official receivers in the United Kingdom have the unique distinction of being entitled to act as liquidators notwithstanding that they are not licensed insolvency practitioners under the Insolvency Act.⁴⁹⁸

1. Appointment of the Official Receiver

Where upon the commencement of petition and before the making of a winding up order, no provisional liquidator is appointed, the official receiver is appointed as the provisional liquidator.⁴⁹⁹ Furthermore, where upon the making of a winding up order, no liquidator(s) is appointed, the official receiver shall by virtue of his office, act as liquidator until such time as a liquidator is appointed.⁵⁰⁰ It is important to state that where the official receiver is appointed liquidator, he is not required before acting in that capacity to give notice of his appointment to the Corporate Affairs Commission and to give security in the prescribed manner to the satisfaction of the court.⁵⁰¹ Where the official receiver is a liquidator of a company, he is described after his individual name as "Official Receiver and Liquidator of (name of the company)".⁵⁰² It is worthy to note that though the official receiver as either a provisional liquidator or liquidator, is an officer of the court.⁵⁰³ The official receiver can be appointed a receiver on behalf of the debenture holders or other creditors of a company being wound up by the

⁴⁹⁷ CAMA, *op cit*, s. 419.

⁴⁹⁸ Insolvency Act, op cit, ss. 388 (5), 389 (2); Davies & Worthington, op cit, p. 1276.

⁴⁹⁹ CAMA, *op cit*, s. 422 (3) (a).

⁵⁰⁰ *Ibid*, s. 422 (1).

⁵⁰¹*Ibid*, s. 422 (3) (d).

⁵⁰² *Ibid*, s. 422 (7) (a).

⁵⁰³ *Ibid*, ss. 419, 567 (1).

court.⁵⁰⁴ In this chapter, the discussion will be limited to the official receiver as a liquidator, whether provisionally or otherwise.

2. **Powers and Duties of the Official Receiver**

The official receiver is conferred with expansive powers and duties in winding up proceedings. Official receiver is vested with all the powers of a liquidator while acting as a liquidator in winding up proceedings. Some of the powers and duties of the official receiver include:

i. Summon of meetings of creditors and contributories

Since the appointment of official receiver, whether as provisional liquidator or liquidator, is at best temporary, the official receiver is vested with the power to summon meetings of creditors and contributories of the company for the purpose of appointing another liquidator.⁵⁰⁵

ii. Statement of Affairs

Official receiver is empowered to request or direct any person as mentioned under section 420 of the CAMA to submit and verify a statement of affairs of a company,⁵⁰⁶ which shall be in Form 19 in the appendix.⁵⁰⁷ The statement of affairs is to be submitted within 14 days from the relevant date or within such extended time as the official receiver or the court may, for special reasons grant.⁵⁰⁸ Where any person requires an extension of time for submitting the statement of affairs, such person shall apply to the official receiver, who may grant the application as he thinks fit.⁵⁰⁹

⁵⁰⁴CAMA, *op cit*, s. 437.

⁵⁰⁵*Ibid*, s. 422 (3) (c); CWR, *op cit*, r. 95; Davies & Worthington, *op cit*, p. 1277.

⁵⁰⁶ CAMA, *ibid*, s. 420 (1) (2).

⁵⁰⁷ CWR, *op cit*, r. 35 (2).

⁵⁰⁸ CAMA, op cit, s. 420 (3) (8); the "relevant date" means, in a case where a provisional liquidator is appointed, the date of his appointment and in a case where no such appointment is made, the date of the winding up order. ⁵⁰⁹ CWR, *op cit*, r. 36.

iii. Investigatory Powers

The official receiver is vested with the powers to investigate the causes of the failure of the company and make a report to the court as he or she thinks fit.⁵¹⁰ Furthermore, the official receiver where in his further report under section 421 of the CAMA, is of the opinion that fraud has been committed by the promoters or by any director or other officer of the company, may apply to the court for public examination of such persons.⁵¹¹

 Application for Declaration of Liability for Fraudulent Trading and of any Misfeasance

The official receiver is among the persons empowered to bring an application to the court to declare that persons who were knowingly parties to the carrying on of the business of the company in a reckless manner or with intent to defraud creditors of the company or creditors of any other person for any fraudulent purpose, shall be personally responsible, without any limitation of liability for all or any of the debts or other liabilities of the company.⁵¹²

v. Application for Appointment of Special Manager

The official receiver is vested with the power, where satisfied that the nature of the estate or business of the company or the interest of the creditors or contributories generally, require the appointment of a special manager of the estate or business of the company other than himself, apply to the court for an order appointing a special manager.⁵¹³

The powers of a special manager are usually contained in the order of appointment. The remuneration of the special manager is usually stated in the order

⁵¹⁰ CAMA, *op cit*, s.421 (1).

⁵¹¹*Ibid*, ss. 421 (2) (3), 450; CWR, *op cit*, rr. 44 – 50.

⁵¹² CAMA, *ibid*, ss. 506, 507.

⁵¹³CAMA, *op cit*, s. 436 (1); CWR, *op cit*, r. 33 (1).

appointing him, but the court may, at any subsequent time, for good cause shown, make an order for payment of the special manager for further remuneration.⁵¹⁴ In addition to the powers of the special manager to manage the estate or business of the company, the special manager is required to give security⁵¹⁵ and submit accounts to the official receiver, which accounts shall be verified by affidavit as in Form 18 in the Appendix to the Rules.⁵¹⁶

vi. Reports

The official receiver, upon receipt of the statement of affairs of a company, owes the duty to make a report and a further report, if any, to the court on the affairs of the company concerning the amount of capital issued, subscribed and paid-up, the estimated amount of assets and liabilities; and if the company has failed, the causes of the failure and his general opinion on the desirability of conducting any further enquiry into any matter relating to the promotion, formation or failure of the company.⁵¹⁷

vii. Book Keeping

The official receiver owes the duty to keep a book to be called the "Record Book" in which he shall record all minutes, all proceedings had and resolutions passed at any meeting of creditors or contributories or of the committee of inspection. The official receiver shall further keep a book to be called the "Cash Book" in which he shall enter the day to day receipts and payments made by him.⁵¹⁸

⁵¹⁴ CAMA, *op cit*, s. 436 (2); CWR, *op cit*, r. 33 (2).

⁵¹⁵ CAMA, *ibid*; CWR, *ibid*, rr. 42, 43 (consequences of failure to give required security).

⁵¹⁶ CAMA, *ibid*; CWR, *ibid*, r. 34.

⁵¹⁷ CAMA, *ibid*, s.421.

⁵¹⁸ CWR, *op cit*, rr. 154, 155.

3. Termination of Appointment of the Official Receiver

Appointment of the official receiver as liquidator of a company ceases immediately upon the appointment of a liquidator.⁵¹⁹ However, the official receiver stands to act as liquidator of a company if there is a re-occurrence of a vacancy in the office of a liquidator.

While official receiver is eligible to act as a liquidator of a company, whether provisionally or otherwise, in practice, an official receiver is rarely appointed to act as a liquidator in compulsory winding up proceedings in Nigeria, which may not be unconnected to the incapacitation⁵²⁰ of the office of the official receiver.

2.13 Committee of Inspection

The Committee of Inspection plays a key role in winding up proceedings. It consists of creditors and contributories of the company or persons holding general powers of attorney from creditors and contributories in such proportion as may be agreed on by meeting of creditors and contributories or as, in case of difference may be determined by the court.⁵²¹ The committee of inspection meets at the time(s) appointed in so far as at least a meeting is held once in every month during its existence but the liquidator or any member of the committee may convene a meeting as and when necessary.⁵²² The quorum of a Committee of Inspection is a majority of members and decisions are taken by majority of members present.⁵²³

⁵¹⁹ CAMA, *op cit*, ss. 419, 422 (3) (b).

⁵²⁰ Official receiver being a Deputy Chief Registrar of the Federal High Court or an officer of the court designated as such by the Chief Judge of the Court, has other duties that makes it difficult for him to function as official receiver. It is doubtful if the official receiver will possess commensurate expertise, skills and experience required of a liquidator.

⁵²¹ CAMA, *op cit*, s. 434 (1).

⁵²²*Ibid*, s. 434 (2).

⁵²³ *Ibid*, s. 434 (3).

A member of the committee of inspection loses his membership if he resigns by notice in writing, signed by him and delivered to the liquidator,⁵²⁴ becomes bankrupt or compounds or arranges with his creditors or is absent from five consecutive meetings of the committee without leave of other members,⁵²⁵ or is removed by an ordinary resolution of creditors or contributories (depending on the one he represents). It is important to note that where no committee has been appointed, the Corporate Affairs Commission may, on the application of the liquidator, do any of those things or give any directions that are required to be done by the Committee of Inspection.⁵²⁶

The members of the Committee of Inspection are not entitled to remuneration except by the express sanction of the court.⁵²⁷ Furthermore, a member of the Committee of Inspection, except by leave of court, either directly or indirectly, by himself or any employee, clerk, agent or servant, purchase the company's assets or any part thereof⁵²⁸ or make any profit or derive any profit from any transaction arising out of the winding up.⁵²⁹

1. Appointment of the Committee of Inspection

The Committee of Inspection is appointed by the court upon the application of the separate meetings of creditors and contributories (usually the first meeting).⁵³⁰In the United Kingdom, the Committee of Inspection is known as the "Liquidation Committee".⁵³¹ In the unlikely event that there is any difference in the resolution of the separate meetings in respect of the appointment of the committee of inspection

⁵²⁴CAMA, *op cit*, s. 434 (4).

⁵²⁵ *Ibid*, s. 434 (5).

⁵²⁶ *Ibid*, s. 435.

⁵²⁷ CWR, *op cit*, r. 148.

⁵²⁸ *Ibid*, r. 144.

⁵²⁹ *Ibid*, r. 146.

⁵³⁰ CAMA, *op cit*, s. 433 (1).

⁵³¹ Insolvency Act, *op cit*, s. 141 (1) (UK); Insolvency Rules, *op cit*, r 4.52 (1) (UK).

and those that are to be members of the committee, it is the court that shall decide and make any order it thinks necessary.⁵³²

2. Functions of the Committee of Inspection

The Committee of Inspection generally functions in a winding up proceedings in Nigeria to sanction the exercise of some of the powers vested on the liquidator;⁵³³ control and give directions to the liquidator in the exercise of his powers in the course of liquidation;⁵³⁴ fixing the remuneration of a liquidator, unless the court shall order otherwise⁵³⁵ and generally to act with the liquidator in the administration and distribution of the assets of the company in liquidation.⁵³⁶

2.14 The Liquidator of a Company

The liquidator plays a prominent role in the winding up of a company. The duties and powers of a liquidator in the winding up of a company are enormous and challenging. Upon the making of a winding up order, the court appoints a liquidator for the purpose of conducting the proceeding in winding up a company and performing such duties in reference thereto.

1. Appointment of a Liquidator

On the making of a winding up order, the court may appoint a liquidator(s) for the purpose of conducting the proceedings in winding up a company and to perform such duties connected with the winding up proceedings.⁵³⁷ However, if the court fails or neglects to appoint a liquidator(s), the official receiver shall by virtue of his office become the liquidator.⁵³⁸ Where the official receiver acts as the liquidator, he shall summon separate meetings of creditors and contributories for the purpose of

⁵³² CAMA, *op cit*, s. 433 (2).

⁵³³ *Ibid*, s. 425 (1).

⁵³⁴*Ibid*, ss. 425 (1), 427 (1).

⁵³⁵₅₃₆ CWR, *op cit*, r. 142 (1).

⁵³⁶ CAMA, *op cit*, s. 433 (1).

 $^{^{537}}_{538}$ *Ibid*, s. 422 (1).

⁵³⁸CAMA, *op cit*, s. 422 (3) (b).

determining whether or not to apply to the court for an order appointing a liquidator in his place.⁵³⁹ If at the separate meetings of creditors and contributories, it was resolved that a liquidator be appointed in place of the official receiver, such official receiver or the chairman of the meeting shall make a report of the result of the separate meetings to the court in Form 20 in the Appendix to the Rules.⁵⁴⁰ Further to the report, the court will appoint a liquidator and make necessary orders to give effect to the resolution of the separate meetings.⁵⁴¹ Upon the appointment of a liquidator, he shall not act in that capacity until he has notified his appointment to the Corporate Affairs Commission⁵⁴² and given security in the prescribed manner to the satisfaction of the court.⁵⁴³

However, if there is a difference in the resolutions of the meetings, the court on application of the official receiver will fix a date for hearing of the resolutions and determinations, which hearing shall be advertised by the official receiver in such manner as the court shall direct in so far as the advertisement shall be published not less than seven days before the date of hearing.⁵⁴⁴ Upon the consideration of the resolutions and determinations of the meetings, the court may, by order in Form 21 in the Appendix to the Rules, appoint a liquidator. Subsequently, the order of the appointment of the liquidator shall be given to the official receiver, who shall then transmit it to the Registrar General of the Corporate Affairs Commission. Upon receipt of the order of the appointment of the liquidator, the Registrar General of the

⁵³⁹CAMA, *op cit*, s. 433 (1).

⁵⁴⁰ CWR, *op cit*, r. 41 (1).

⁵⁴¹ CAMA, op cit, s. 433 (2); CWR, ibid, r. 41.

⁵⁴² Companies Regulation 2012, reg 43 (1) (3).

⁵⁴³ CAMA, *op cit*, s. 422 (3) (d); CWR, *op cit*, r. 42.

⁵⁴⁴ CWR, *ibid*, r. 41 (2) (3).

Corporate Affairs Commission, subject to the liquidator giving security, shall cause notice of the appointment to be gazetted.⁵⁴⁵

Furthermore, the notice of appointment of a liquidator has to be advertised by the liquidator as directed by the court immediately after his appointment and the provision of required security.⁵⁴⁶ Where a liquidator dies, resigns or is removed, another liquidator may be appointed in his place in the same manner as in the case of a first appointment.⁵⁴⁷

While there is no specific provision on the qualifications for a person to be appointed liquidator, certain persons are disqualified from being appointed as liquidator or to act as liquidator of a company.⁵⁴⁸ It appears from the wordings of section 509 (1) of the CAMA that the disqualification provision as provided thereunder is only applicable to persons to be appointed or to act as liquidator of a company being wound up by or under the supervision of the court or in a voluntary winding up. In *Federal Mortgage Bank of Nigeria v Nigerian Deposit Insurance Corporation*,⁵⁴⁹ the court held that the appointment of the N.D.I.C as a liquidator does not conflict with the provision of section 509 (1) (c) of the CAMA in that a body corporate shall not be appointed a liquidator. Relying on section 38 (3) of BOFID, the court further held that N.D.I.C can be appointed, in the public interest as a provisional liquidator and the provision of section 509 (1) (c) of the CAMA is therefore deemed to have been amended to that extent.⁵⁵⁰ This position of the Supreme Court of Nigeria

⁵⁴⁵ Companies Regulation, *op cit*, reg 43 (1) (c); CWR, *op cit*, r 41 (5). However, in practice, it is the responsibility of the liquidator to notify his appointment to the Corporate Affairs Commission. ⁵⁴⁶ CWR, *ibid*, r. 41 (7).

⁵⁴⁷ CAMA, op cit, s. 422 (1) (5); CWR, ibid, rr. 41 (8), 163.

⁵⁴⁸ CAMA, *ibid*, s.509 (1); Orojo, *op cit*, p. 477; A Aliyu & M D Saulawa, 'The Role of a Liquidator in Winding up Proceedings in Nigeria: Examining the Inadequacies under Companies and Allied Matters Act (CAMA)', PHJBL 2.1 (2016).227.

⁵⁴⁹ [1999] 2 N.W.L.R (pt 591) 333.

⁵⁵⁰ H Y Bhadmus, *Corporate Insolvency Law and Practice in Nigeria* (Enugu: Chenglo Limited, 2009) p. 137.

is to say the least, unsatisfactory. The appointment of a provisional liquidator is different from that of a liquidator. Section 509 of the CAMA expressly provides for disqualification of persons for appointment as liquidator. Further, the amendment of the provision of a statute cannot be deemed or implied but must be an express Act of parliament. Hence, in this study, it is submitted that it must be an inadvertent omission on the part of the draftsmen to have excluded or omitted the application of the disqualification provision to persons to be appointed liquidator or to act as a liquidator of a company in a winding up by the court. This position is re-enforced by the fact that the disqualification provision under section 509 (1) of the CAMA is *in tandem* with international best practices that disqualifies bodies corporate from being appointed or to act as liquidators.

2. Effects of the Appointment of a Liquidator

The appointment of a liquidator attracts the following effects on:

i. Power of Directors

The appointment of a liquidator puts an end to the powers of directors⁵⁵¹. However, the directors may continue to exercise powers if the Court by order sanctions the continuance thereof.⁵⁵² Put in another way, the appointment of liquidator of a company signifies the divesture of the powers of its Board of Directors and the investiture of such powers on the liquidator. When this happens, the Board is said to be *funtus officio*.⁵⁵³

ii. Contractual Capacity of the Company

The powers of the company to deal with its assets and of the continued performance of its obligations, contractual or not, comes to an end and by reason of

⁵⁵¹ CAMA, *op cit*, s. 422 (9); *Federal Mortgage Bank of Nigeria v Nigerian Deposit Insurance Corporation, supra*, at 529; M O Sofowora, *Modern Nigerian Company Law* (2ndedn, Lagos: Soft Associates, 2002) p. 522.

⁵⁵² Anakwenze v TAPP Industry Ltd, supra.

⁵⁵³ Gbedu v Itie & ors, supra, 260.

the appointment of a liquidator and the powers conferred on him, the capacity of the company to sustain its contractual obligations is negated.⁵⁵⁴

Furthermore, the company ceases to carry on its business as same can only be done by the liquidator with the sanction of the court or of the committee of inspection, in so far as it is necessary for the beneficial winding up of the company.⁵⁵⁵

iii. Commencement and continuation of action against the company

Upon the appointment of a liquidator, no action or proceedings shall be proceeded with or commenced against the company without the leave of the court.⁵⁵⁶ In this study, we reiterate the arguments canvassed under the effects of appointment of a provisional liquidator as it relates to commencement and continuation of action against the company. Suffice it to say that the provision of section 417 of the CAMA does not debar the company in liquidation from commencing or continuing with actions or proceedings against third parties.⁵⁵⁷

iv. Contract of employment

Contrary to the postulations that the appointment of a liquidator automatically terminates every contract of employment,⁵⁵⁸ the correct position of the law is that the effect of the appointment of liquidator on a contract of employment depends on whether the winding up is compulsory or voluntary.⁵⁵⁹ Where it is a compulsory winding up, the prevailing position of the law was succinctly stated in *Gbedu v Itie* & *ors*,⁵⁶⁰ that:

⁵⁵⁴ CAMA, op cit, s. 425 (1) (e) (f); Gbedu v Itie & ors, supra, at261, paras C-D.

⁵⁵⁵ CAMA, *ibid*, ss. 425 (1) (b), 494 (6) (a); *Gbedu v Itie & ors*, *ibid*, 260, paras F-H; *Palmer's Company Law*, Volume 3 (London: Sweet & Maxwell) p 15134/1.

⁵⁵⁶ CAMA, ibid, s. 417; Onwuchekwa v NDIC, supra; Atojuv Triumph Bank Plc, supra. ⁵⁵⁷ Onwuchekwa v NDC, supra 1631.

⁵⁵⁸ Orojo, *op cit*, pp. 451, 474; Ola, *op cit*, p. 467; Sofowora, *op cit*, p. 232.

⁵⁵⁹ Gbedu v Itie & ors, supra, at 264, paras A-B.

⁵⁶⁰ *Ibid*, at 261 – 262, paras A-B.

A contract of service is a contract in which both parties have continuing obligations: whilst the employee has an obligation to work, the company has a duty to remunerate him for such exertions by way of the payment of his wages. However, a winding up order hampers a company's capacity to sustain its aforesaid obligations. It, simply, comes to this. A winding up order drains the company of its powers: powers of dealing with its assets and of the continued performance of its obligations! By that winding up order, a new officer: an officer of the court [the liquidator] inherits these powers. The servant must, then, be deemed to know the legal effect of a winding up order. In consequence, such an order operates as a notice to him that the company cannot continue its obligations to him under his contract of service and thus constitutes a breach going to the root of their contract.

Simply put, while the making of a winding up order serves as notice to employees of the wound up company that their contract of employment will not be continued, on the appointment of liquidator, the employees are discharged from the date of the winding up order. However, the employees of the wound up company are entitled to make a claim under the winding up proceedings.⁵⁶¹

⁵⁶¹ CAMA, *op cit*, ss. 494, 518.

3. Duties and Powers of Liquidator

The liquidator is appointed for the primary purpose to secure that the assets of the company are got in, realised and distributed to the company's creditors and if there is a surplus to be distributed among the members in accordance to their rights.⁵⁶² In order to fulfil the primary purpose of their appointment, liquidators are usually given a wide range of duties and powers, some of which can be exercised of their own volition entirely, others of which is exercised with the sanction of either the Court or the Committee of Inspection. In the case of a body corporate such as the AMCON, it is important to state that any act, thing, directive or permission that required the sanction of the Committee inspection may be exercised by the Court.⁵⁶³

1. Duties of a Liquidator

The general duties of a liquidator are that as a fiduciary, he must act in good faith and for a proper purpose, he must not fetter his discretion, he must not allow conflict of interest and duty, he must be impartial, he must exercise a degree of care and skill appropriate to the circumstances and he must exercise his discretion personally or where appointed jointly, he must act jointly.⁵⁶⁴ Furthermore, a liquidator owes a general duty to collect in and realise the company's assets.⁵⁶⁵

 ⁵⁶²CAMA, op cit, ss. 422 (1), 439 (1); Insolvency Act, op cit, s. 143 (1) (UK); CWR, op cit, r. 61 (1);
 Abekhe v NDIC [1995] 7 N.W.L.R (pt 406); Davies & Worthington, op cit, p. 1277; J A Dada,
 Principles of Nigerian Company Law (Rev edn, Calabar: Wusen Publishers, 2001) p. 318.
 ⁵⁶³AMCON Act, op cit, s.52(4).

⁵⁶⁴ Palmer's Company Law, op cit, p. 15130.

⁵⁶⁵ Provisional Liquidator of TAPP Industries Ltd v TAPP Industries Ltd & Ors, supra, at 9; Savannah Bank of Nigeria Plc v NDIC, supra, at 440, paras C – E.

The liquidator has specific functions or duties, namely:

i. Collection of Debts

The liquidator, being an officer of the court, has the duty to secure the payment to him or other discharge of all debts of the company and apply them to the discharge of its liabilities.⁵⁶⁶

ii. Custody of Company's Property

The liquidator owes the duty to take into custody or under his control all the property and things or *choses in action* to which the company is, or appears to be entitled.⁵⁶⁷ In the course of carrying out this duty, the liquidator is given the power to disclaim onerous property⁵⁶⁸ and to have the property of the company vested in him.⁵⁶⁹

iii. Settlement of List of Contributories

The liquidator exercises the duty to settle a list of contributories as in Form 31 in the Appendix to the Rules.⁵⁷⁰ In the course of the settlement of list of contributories, the liquidator may rectify the register of members in all cases where rectification is required. It should be noted that it is only after the settlement of contributories, unless where dispensed by the court, that the collected assets of the company can be applied in discharge of its liabilities.⁵⁷¹ Worthy of note is the fact that the duty for settlement of list of contributories is a delegated duty.⁵⁷² However,

⁵⁶⁶ Insolvency Act, *op cit*, s. 143 (1) (UK); Bodies Corporate (Official Liquidations), *op cit*, s. 36 (Ghana); CWR, *op cit*, r. 61 (1).

⁵⁶⁷ CAMA, *op cit*, s. 423; Insolvency Act, *ibid*, s. 144 (1).

⁵⁶⁸ CAMA, *ibid*, s. 499 (1).

⁵⁶⁹ *Ibid*, s. 423.

⁵⁷⁰ CWR, *op cit*, r. 63 (2).

⁵⁷¹ CAMA, *op cit*, s. 439; CWR, *ibid*,, rr. 63 – 68.

⁵⁷² CAMA, *ibid*, s. 453 (1) (b); CWR, *ibid*, r. 63 (1).

the court can dispense with the settlement of the list of contributories where the circumstance does not warrant the making of a call.⁵⁷³

The procedure for the settlement of list of contributories require the liquidator to give notice in writing as in Form 32 in the Appendix to the Rules, of the time and place appointed for the settlement of the list of contributories, to every person whom he proposes to include in the list.⁵⁷⁴ On the day appointed for the settlement of the list of contributories, the liquidator shall hear any person who objects to being settled as a contributory and after such hearing, shall finally settle the list and issue a certificate as in Form 33 in the Appendix to the Rules.⁵⁷⁵ It is the list so settled after hearing that constitutes the list of contributories of the company. Subsequently, the liquidator shall give notice as in Form 34 in the Appendix to the Rules to every person whom he has finally placed on the list of contributories.⁵⁷⁶ It should be noted that the liquidator or the court may vary or set aside the list.⁵⁷⁷

iv. Making of Calls

The liquidator has the duty to make calls on all or any of the contributories in the list of contributories to the extent of their liability; and for payment of any money which the court considers necessary to satisfy the debts and liabilities of the company; and the cost, charges, and expenses of winding up.⁵⁷⁸ Instructively, this duty is exercised by the liquidator as a delegated duty and subject to either the special leave of the court or the sanction of the Committee of Inspection.⁵⁷⁹

It is important to state that in other to obtain the sanction of the committee of inspection to make calls on all or any of the contributories, the liquidator has to

⁵⁷³ CAMA, op cit, s. 439 (1) proviso; CWR, op cit, r. 63 (1).

⁵⁷⁴ CWR, *ibid*, r. 64.

⁵⁷⁵ Ibid, r. 65.

⁵⁷⁶ *Ibid*, r. 66.

⁵⁷⁷ *Ibid*, rr. 67 and 68.

⁵⁷⁸ CAMA, *op cit*, s. 442 (1).

⁵⁷⁹ *Ibid*, s. 453 (1) (d) (2); CWR, *op cit*, r. 69 (a).

summon a meeting of the committee by giving notice of the meeting as in Form 36 in the Appendix to the Rules. Such notice shall be sent to each member of the committee in sufficient time, not less than fourteen days to the day appointed for the meeting, which notice shall contain a statement of the proposed amount of the call and the purpose for the call.⁵⁸⁰ It is further required that the notice of the intended call and the meeting of the Committee of Inspection shall be advertised as in Form 37 in the Appendix to the Rules, in a national newspaper where the proceedings is in the head office of the court⁵⁸¹ and in any other case, in a newspaper circulating in the division of the court in which the proceeding is pending.⁵⁸²

Where there is no Committee of Inspection, the liquidator can only make a call on the contributories with the leave of court,⁵⁸³ which leave is by summons as in Forms 39 and 40 in the Appendix to the Rules. The summons shall state the proposed amount of the call and be served seven clear days before the day appointed for making the call on every contributory. However, the court may direct that notice of such proposed call be given by advertisement as in Form 41 in the Appendix to the Rules, without a separate notice to each contributory.⁵⁸⁴

v. Keeping of Proper Books

It is a duty of the liquidator to keep proper books in which he shall cause to be made entries or minutes of proceedings at meetings.⁵⁸⁵ The book to be kept by the liquidator is the book called the "Record Book" in which he shall record all minutes of proceedings and resolutions passed at any meeting of the creditors or contributories or of the committee of inspection and all such matters that may be necessary to give a

⁵⁸⁰ CWR, *op cit*, r. 69 (b).

 $^{^{581}}$ *Ibid*, r. 184 (2) – it means where the Chief Judge is stationed.

⁵⁸²*Ibid*, r. 69 (c).

⁵⁸³ Ibid, r 69 (g).

⁵⁸⁴ *Ibid*, r. 70.

⁵⁸⁵ CAMA, *op cit*, s. 430.

correct view of his administration of the company's affairs. However, the liquidator is not bound to enter in the Record Book, any document of a confidential nature.⁵⁸⁶ Another book to be kept by the liquidator is that known as the "Cash Book" in which he shall enter from day to day the receipts and payments made by him and in trading account where carrying on the business of the company.⁵⁸⁷

It is pertinent to state that the liquidator owes consequential duty to submit the books to the Committee of Inspection when required which is not less than once every three months.⁵⁸⁸ Furthermore, the Committee of Inspection shall not, less than once every three months, audit the liquidator's Cash Book and certify the audit as in Form 70 in the Appendix to the Rules.⁵⁸⁹ In addition, the liquidator is mandated to transmit to the Registrar-General of Corporate Affairs Commission (CAC) for audit, a copy of the Cash Book in duplicate; vouchers and copies of the certificates or audit by the Committee of Inspection, at the expiration of six months from the date of the winding up order and at the expiration of every succeeding six months thereafter until his release.⁵⁹⁰ It is instructive that the Attorney-General of the Federation may order that the books and papers of a company wound up should not be destroyed for such period not exceeding five years from dissolution.⁵⁹¹

vi. Payments into Company's Liquidation Account

It is a duty of every liquidator of a company to pay money received by him, including unclaimed dividends, into the public fund of the Federation kept by the Corporate Affairs Commission and known as "the Companies Liquidation

⁵⁸⁶ CWR, *op cit*, r. 154.

⁵⁸⁷*Ibid*, rr. 155 (1), 159 (1).

⁵⁸⁸₅₈₀ *Ibid*, r 155 (2).

⁵⁸⁹ *Ibid*, r 157.

⁵⁹⁰ CAMA, *op cit*, s. 429; CWR, *ibid*, r. 158 (1).

⁵⁹¹ CAMA, *ibid*, s. 515 (2) (3) – powers of the Corporate Affairs Commission to prevent disposal of books, etc of the company; CWR, *op cit*, r. 178.

Account".⁵⁹² However, the liquidator can pay moneys received by him into another bank selected by the Committee of Inspection and an account other than the Companies Liquidation Account, where the Committee of Inspection were able to obtain exemption from the Corporate Affairs Commission on the ground of business expediency and beneficial advantage to the creditors and contributories, which account is known as the "Special Bank Account".⁵⁹³ It still remains the position of the law that a liquidator of a company cannot pay any sum received by him into his private bank account.⁵⁹⁴

vii. Statement and Information

The liquidator of a company is under a duty to give to the official receiver such information and access to and facilities for inspecting the books and documents of the company.⁵⁹⁵ Furthermore, the liquidator owes the duty to send a statement in the prescribed form concerning the liquidation, to the Corporate Affairs Commission⁵⁹⁶ and to make available the statement for inspection by a creditor or contributory of the company, on the payment of the prescribed fees.⁵⁹⁷

Distribution of the Company's Assets viii.

Once the liquidator has gathered the company's assets, he is under a duty to distribute surplus among the members according to their rights.⁵⁹⁸

ix. Liquidator's Meetings

The liquidator, subject to the provisions of the CAMA and the control of the court, has the duty to summon, hold and conduct meetings of the creditors or

⁵⁹² CAMA, op cit, ss. 428 (1), 527 – definition of "Companies Liquidation Account"; CWR, op cit, rr 152, 171.

⁵⁹³ CAMA, *ibid*, s. 428 (1) proviso.

⁵⁹⁴ *Ibid*, s. 428 (3).

⁵⁹⁵ Ibid, s. 426.

⁵⁹⁶ Ibid, s. 516 (1). ⁵⁹⁷*Ibid*, s. 516 (2).

⁵⁹⁸ Orojo, *op cit*, p. 484.

contributories from time to time as may be directed or requested⁵⁹⁹ for the purpose of ascertaining their wishes in all matters relating to the winding up. The procedure for meetings including the Liquidator's meetings of creditors and contributories are set out in the Rules,⁶⁰⁰ including the right to attend by proxies.⁶⁰¹ Suffice to state that a minor cannot be appointed a general or special proxy.⁶⁰² While a general proxy is as in Form 64 in the Appendix to the Rules,⁶⁰³ a special proxy is as in Form 65 in the Appendix to the Rules.⁶⁰⁴

Proof of Debt and Priority of Debt x.

One of the significant duties of a liquidator is that of proof of debt. It is entrusted on the liquidator the duty to ascertain, examine, admit or reject the proof of debt lodged with him and to ascertain priority of debt.⁶⁰⁵ As a corollary, the liquidator is to verify debts ranking for dividends in a winding up by the court.⁶⁰⁶ In order to carry out this duty, a liquidator is guided by certain rules such as the "rule against double proof", which seeks to prevent more than one creditor suing the insolvent company for the same debt⁶⁰⁷ and the "mandatory set-off rules" which requires the sum due from the creditor to be set off against the sum due from the company where those sums arise from mutual credits, mutual debts or other mutual dealings between the company and the creditor arising in the liquidation.⁶⁰⁸

xi. Returns

The liquidator is under a duty to make various returns, including:

⁵⁹⁹ CAMA, op cit, ss. 427 (2), 453 (1) (a); CWR, op cit, rr. 110 (1), 112.

⁶⁰⁰ CWR, *ibid*, rr. 111- 124.

⁶⁰¹*Ibid*, rr. 129 to 139.

⁶⁰² *Ibid*, r. 137 (3).

⁶⁰³ *Ibid*, rr. 130, 132 and 137. ⁶⁰⁴ *Ibid*, rr. 130, 133 and 137.

⁶⁰⁵ *Ibid*, rr. 74 to 101.

⁶⁰⁶ *Ibid*, r. 102.

⁶⁰⁷ Davies & Worthington, op cit, p.1290.

⁶⁰⁸ *Ibid*, p. 1291.

- a) Return of capital to contributories.⁶⁰⁹
- b) Notice of his appointment.⁶¹⁰
- c) Account of receipts and payments as liquidator.
- d) Copy of the order of dissolution of the company.⁶¹¹
- e) Statement in the prescribed forms at such intervals as may be prescribed, if the winding up is not concluded within one year after its commencement.⁶¹²

It is pertinent to state that the liquidator's duty to make returns is enforceable by an order of the court.⁶¹³

2. Powers of the Liquidator

In order to carry out the expansive duties, the liquidator is conferred with a wide range of powers, some of which can be exercised on his own volition, others need the leave of court or sanction of the committee of inspection⁶¹⁴ and some exercised with the special leave of court.⁶¹⁵ The liquidator has the power with either the leave of court or sanction of the Committee of Inspection⁶¹⁶ to:

- a) Bring or defend any action or other legal proceedings in the name and on behalf of the company.⁶¹⁷ In other words, where no sanction of either the court or the Committee of Inspection is sought and obtained by the liquidator, no legal action or proceedings can be brought or defended by the liquidator,⁶¹⁸
- b) Carry on the business of the company so far as may be necessary for its

⁶⁰⁹ CWR, *op cit*, r. 103.

⁶¹⁰ CAMA, *op cit*, ss. 422 (3) (d), 491; CWR, *ibid*, r. 41 (7).

⁶¹¹₆₁₂ CAMA, *ibid*, s. 454.

 $^{^{612}}$ *Ibid*, s. 516 (1).

⁶¹³ *Ibid*, s. 511 (1). ⁶¹⁴ *Ibid*, s. 425.

⁶¹⁵*Ibid*, s. 453 (2).

⁶¹⁶ *Ibid*, s. 425 (1).

⁶¹⁷*Ibid*, s. 425 (1) (a); *Onwuchekwa v NDIC*, *supra*.

⁶¹⁸ In Re: Amolegbe (2015) 25 JMLR, 213 at 232, para. E; [2014] N.W.L.R (pt 1408) 76 at 96 – 97, paras G – A.

beneficial winding up,⁶¹⁹

- c) To appoint a Legal Practitioner or any other relevant professionals to assist him in the performance of his duties,⁶²⁰
- d) Pay any classes of creditors in full,⁶²¹
- e) Make any compromise or arrangement with creditors or persons claiming to be creditors or having or alleging themselves to have any claim against the company,⁶²²
- f) Compromise all calls and liabilities to calls, debt and liabilities and all claims between the company and a contributory and all questions relating to the assets of the company,⁶²³
- g) Making of calls,⁶²⁴
- h) Rectifying the register of members where required.⁶²⁵

The liquidator shall have the power without either the leave of court or sanction

of the Committee of Inspection to:

- a) Sell the property of the company of whatever nature by public auction or private contract, with power to transfer the whole thereof to any person or to sell the same in parcels,⁶²⁶
- b) Do and execute all acts, in the name and on behalf of the company, all deeds, receipts and other documents, and for that purpose, when necessary to use the company's seal,⁶²⁷
- c) Prove, rank and claim in the bankruptcy insolvency or sequestration of any

⁶²⁰ CAMA, *ibid*, s. 425 (1) (c).

⁶¹⁹ CAMA, op cit, s. 425 (1) (b); Gbedu v Itie & ors, supra, at 260, paras. F–H.

⁶²¹ Ibid, s. 425 (1) (d); CAC v Davis, supra at 78-79, paras H–F.

⁶²² CAMA, *ibid*, s.425 (1) (e); *CAC v Davis, supra* at 78–79.

⁶²³ CAMA, *ibid*, s.425 (1) (f); *CAC v Davis, ibid*.

⁶²⁴ CAMA, *ibid*, s. 453 (1) (d) and (2).

⁶²⁵*Ibid*, s. 453 (1) (b) and (2).

⁶²⁶*Ibid*, s. 425 (2) (a).

⁶²⁷ *Ibid*, s. 425 (2) (b).

contributory for any balance against his estate,⁶²⁸

- Draw, accept, make and endorse any bill of exchange or promissory note in the name and on behalf of the company,⁶²⁹
- e) Raise on the security of the assets of the company any money requisite;⁶³⁰
- f) Take out in his official name, letters of administration to the estate of any deceased contributory and to do in his official name any other act necessary for obtaining payment of any money due from a contributory or his estate which cannot be conveniently done in the name of the company,⁶³¹
- g) Appoint an agent to do any business which the liquidator is unable to do himself,⁶³² and,
- h) Do all such other things as may be necessary for winding up the company and distributing its assets.⁶³³

4. Limits and Control of the Power of a Liquidator

While the liquidator in a winding up by the court is unquestionably endowed or conferred enormous powers to carry out his duties, the liquidator, however, acts in accordance with the directions, limits and control of the court⁶³⁴ or Committee of Inspection or the creditors and contributories or the Corporate Affairs Commission.

The limits and control of the power⁶³⁵ of a liquidator includes:

a) The power vested on the court to control and to give directions from time to time to the liquidator in the exercise of his powers as a liquidator, being an officer of the court.⁶³⁶

⁶³¹ *Ibid*, s. 425 (2) (f).

 634 *Ibid*, s. 425 (2).

⁶²⁸CAMA, *op cit*, s. 425 (2) (c).

 $^{^{629}}$ *Ibid*, s. 425 (2) (d).

⁶³⁰*Ibid*, s. 425 (2) (e).

⁶³² *Ibid*, s. 425 (2) (g). ⁶³³*Ibid*, s. 425 (2).

 $^{^{635}}$ CAC v Davis, supra, at 78.

- b) The liquidator exercises his powers in the administration and distribution of the assets of the company, subject to directions and approvals of the creditors or contributories at any general meeting or by the committee of inspection.⁶³⁷ However, in the case of conflict between directions given by the creditors or contributories at any general meeting and that of the Committee of Inspection, the directions of the creditors or contributories shall override that of the Committee of Inspection.⁶³⁸
- c) The requirements for payments by liquidator of money received by him into the Company's Liquidation Account,⁶³⁹ audit and investigation of the books of account kept by the liquidator.⁶⁴⁰
- d) The requirements for the release of a liquidator.⁶⁴¹
- e) The right of any creditor or contributory to apply to the court with respect to any exercise or proposed exercise of any of the liquidator's powers.⁶⁴²

5. Cessation of office of the Liquidator

The liquidator ceases to act in a winding up proceedings or vacates the office, if:

- a) He resigns his appointment 643
- b) He is removed by the court or Committee of Inspection 644
- c) He is released⁶⁴⁵ subject to compliance with the requirements of the Rules,⁶⁴⁶ and,

⁶³⁶ CAMA, op cit, ss. 425 (1) (3), 427 (3), 453 (1).

 $^{^{637}}$ *Ibid*, s. 427 (1).

⁶³⁸₆₃₉ *Ibid*, s. 427 (1).

⁶³⁹ *Ibid*, s.428 (1).

⁶⁴⁰ *Ibid*, ss. 429, 432; CWR, *op cit*, rr. 157, 158.

⁶⁴¹ CAMA, *ibid*, s. 431; *CAC v Davis*, *supra*.

⁶⁴² CAMA, *ibid*, s. 425 (3).

⁶⁴³ *Ibid*, s. 422 (5); CWR, *op cit*, r. 150.

⁶⁴⁴ CAMA, *ibid*, s.422(5).

⁶⁴⁵ *Ibid*, s. 431; CWR, *op cit*, r. 177.

⁶⁴⁶ CWR, *ibid*, r. 163.

d) He is fixed with a Receiver's Order in Bankruptcy.⁶⁴⁷

2.15 Distribution of Assets of the Company

Once the liquidator has collected in the company's assets, they must be distributed in an orderly manner to those entitled to them.⁶⁴⁸ There are certain basic rules guiding the distribution of assets. Foremost is that the assets are devoted to first, paying the expenses of the liquidator,⁶⁴⁹ then to the preferred creditors,⁶⁵⁰ the general unsecured creditors, the deferred creditors and if there is anything left, to the members, according to their entitlements associated with their class rights.⁶⁵¹ If the assets are not sufficient to pay all of a particular class in this hierarchy in full, then they share *pro rata*.⁶⁵²

2.16 Completion of Winding up

Winding up is completed by the distribution of assets, filing of the final accounts as in Form 72 in the Appendix to the Rules and the order dissolving the company has been reported by the liquidator to the Corporate Affairs Commission.⁶⁵³

It is important to state that if a winding up is not completed within a year; the liquidator must send to the Corporate Affairs Commission, at specific intervals, a prescribed statement giving particulars as to the position of the liquidation⁶⁵⁴ and a creditor or contributory may examine the statement and take copies or extracts on payment of a fee.⁶⁵⁵

⁶⁴⁷CWR, *op cit*, r. 151.

 $^{^{648}}_{640}$ Okafor v Igwilo, supra at 54.

⁶⁴⁹ CAMA, *op cit*, s. 494 (5); CWR, *op cit*, r. 167.

⁶⁵⁰ CAMA, *ibid*, s. 494 (1) - (4).

⁶⁵¹*Ibid*, s. 446.

⁶⁵² Orojo, *op cit*, p. 499.

⁶⁵³ CWR, *op cit*, r. 168 (a).

⁶⁵⁴ CAMA, op cit, s. 516 (1).

⁶⁵⁵ Ibid, s. 516 (2).

2.17 Dissolution of a Company: Meaning and Consequences.

While winding up of a company involves a liquidation of the company so that its assets are distributed to those entitled to receive them, liquidation is quite distinguishable from dissolution, which is the end of the legal existence of a company.⁶⁵⁶ When a company is under a winding up proceedings, it is not dead but still alive though sick.⁶⁵⁷ Thus, the life of a company comes to an end when the court orders dissolution of the company upon an application by the liquidator.⁶⁵⁸ However, the court may at any time within two years after dissolution of a company, on the application of the liquidator or any interested person, make an order declaring the dissolution void and the winding up proceedings to be of no effect.⁶⁵⁹

It is important to state that dissolution of a company may occur without liquidation. This usually occurs in circumstances of a smooth transition to the company' legal successor through transformation of legal entities, that is, by a merger, division, change of the legal form of the business corporation, a transfer of assets to its shareholders or a cross-border relocation of the registered office.⁶⁶⁰

 ⁶⁵⁶ Amolegbe Re, supra, at 94; Progress Bank (Nig) Plc v O.K. Contact Point Holdings Ltd, supra at 531 – 532; Spring Bank Plc v ACB International Plc [2016] 18 N.W.L.R (pt 1544) 245 at 255, paras C – F.
 ⁶⁵⁷ Onwuchekwa v NDIC, supra; C.C.B (Nig) Plc v Mbakwe [2000] F.W.L R (pt 14) 2351.
 ⁶⁵⁸ CAMA, op cit. s. 454.

⁶⁵⁹ Ibid, s. 524 (1); Abekhe v NDIC, supra; Musa v Elaidiamkhem [1994] 3 N.W.L.R (pt 334) 544. ⁶⁶⁰Spring Bank Plc v ACB International Bank Plc, supra, at 255 – 260.

CHAPTER THREE

RECEIVERSHIP AS A MODE OF ENFORCEMENT OF SECURITY

In order to carry out its business, a company ordinarily requires capital. The needed capital can be equity capital or debt capital. In most cases, equity capital is never sufficient for the flotation of a company; hence, there is understandably a resort to debt capital. The powers of a company to carry out its business through debt capital are provided in the CAMA.⁶⁶¹ Debt capital is in most cases, secured, as unsecured credits often turn out to be a hoax, disappointment and loss to the creditor.⁶⁶² Thus, the essence of security in debt capital transactions is priceless. In the case of credits advanced to a company, security can be in the form of mortgage or charge of its undertaking, property and uncalled capital or any part thereof, creation and issue of debentures, debenture stock and other securities.⁶⁶³

It is usual for companies in Nigeria to obtain debt capital by issuing debentures, which debentures are secured either by a fixed charge mostly on the specific or ascertained company properties or by a floating charge over the whole or a specific part of the company's undertakings and assets or by both.⁶⁶⁴ Notwithstanding the comfort afforded creditors by the provision of security in debt capital transactions, the enforcement of such security are not usually seamless. Worse still, as examined in chapter two of this study, resort to winding up of a company may not ultimately be a beneficial mode of resolving insolvency issues.⁶⁶⁵ Hence, there is need to utilise other

⁶⁶¹ Companies and Allied Matters Act (CAMA), Cap C20, Laws of the Federation of Nigeria, 2004, s. 166.

⁶⁶² I O Smith, Nigerian Law of Secured Credits (Lagos: Ecowatch Publications (Nigeria) Limited, 2001) p. 1. ⁶⁶³ CAMA, *op cit*, s. 166.

⁶⁶⁴ Ibid, s. 173 (2); Smith, op cit, p. 297.

⁶⁶⁵ O Ajayi, Legal Aspects of Finance in Emerging Markets (London: London: LexisNexis Butterworths Tolley, 2005) p. 603.

alternatives to winding up. One of the prominent alternatives to winding up of a company in Nigeria is Receivership.

The institution of receivership is one created by the agreement of parties to secured credit arrangements and also by virtue of the provisions of statute.⁶⁶⁶ Thus, in the realm of corporate law, the appointment of a receiver and/or manager is provided as one of the potent remedies available to a debenture holder in the enforcement or realisation of securities.⁶⁶⁷ Although a receiver and a manager is usually given similar meaning,⁶⁶⁸ there are some fine distinctions between a receiver and realise debts, he is only a receiver but where the person appointed is reserved with powers to manage and continue the business of the company as a going concern, he is not only a receiver but also a manager.⁶⁶⁹

In this chapter of the study, the focus would be on a thorough examination of the law and practice relating to receivership in Nigeria.

3.1 Jurisdiction over Receivership Proceedings

The court with exclusive jurisdiction to appoint a receiver and/or manager for companies in Nigeria is the Federal High Court.⁶⁷⁰ However, where the subject matter of receivership would not entail recourse to either the CAMA or any enactment

⁶⁶⁶ Conveyancing Act 1881, s. 27; Property and Conveyancing Law 1959, s. 131 (Western Region of Nigeria); CAMA, *op cit*, Part XIV, ss. 387 to 400.

⁶⁶⁷ CAMA, *op cit*, s. 209 (1).

⁶⁶⁸ *Ibid*, s. 567 (1).

⁶⁶⁹ Uwakwe v Odogwe [1989] 5 N.W.L.R (pt 123) 562 at 589; PIP Ltd v Trade Bank (Nig) Plc [2009] 13 N.W.L.R (pt 1159) 577 at 637 – 8; Ponson Enterprises Nig Ltd v Njigha (2001) FWLR (pt 16) 1685; Palmer's Company Law, Volume 3 (London: Sweet & Maxwell) Pt 14, p. 14068; J E O Abugu, Company Securities: Law and Practice (2nd edn, Lagos: MIJ Professional Publishers Limited, 2014) p. 227.

 $^{^{670}}$ Constitution of the Federal Republic of Nigeria (CFRN) 1999, s. 251 (1) (e); *Bank of Industry Limited* v Awojubagbe Light Ind. Ltd [2016] 1 N.W.L.R (pt 1493) 280 at 327 – 8.

regulating the operations of companies, a State High Court and not the Federal High Court has jurisdiction in such instance of receivership.⁶⁷¹

3.2 Appointment of Receiver and/or Manager

A receiver and/or manager may be appointed either by the court or out of court, that is, under the express powers of the debentures or trust deed.⁶⁷² In this study, the discussion would be on each mode of appointment.

1 Appointment by the Court

The court is conferred with the powers to appoint a receiver and manager.⁶⁷³ However, the power of the court to appoint a receiver and manager is not exercised as a matter of course. The appointment of a receiver and manager is usually predicated on sound principles of law. The court may appoint a receiver and manager in the following circumstances:⁶⁷⁴

- 1. Where the principal money borrowed by the company or the interest is in arrears.⁶⁷⁵
- 2. Where the security or property of the company is in jeopardy.⁶⁷⁶

In Ceramic Manufacturers Nigeria Plc v Nigerian Industrial Bank,⁶⁷⁷ the

court listed three events that must be established before the grant of an order of appointment of a receiver, thus:-

- 1. That the principal money or the interest thereon is in arrears.
- 2. That the security or the property is in jeopardy.
- 3. That the appointment of the receiver was made under a power contained in

⁶⁷¹ Tanarewa (Nig) Ltd v Plastifarm Ltd [2003] 14 N.W.L.R (pt 840) 355 at 370, paras B –C.

⁶⁷² CAMA, *op cit*, ss. 389, 390.

⁶⁷³ Ibid, s. 389 (1); Okoya & Ors v Santili & Ors [1990] 2 N.W.L.R (pt 131) 172.

⁶⁷⁴ Nashtex Int'l Ltd v Habib Bank (Nig) Ltd [2007] 17 N.W.L.R (pt 1063) 328.

⁶⁷⁵ CAMA, *op cit*, s. 389 (1) (a).

⁶⁷⁶ *Ibid*, s. 389 (1) (b).

⁶⁷⁷ [1999] 11 N.W.L.R (pt 627) 383.

the mortgage deed between the parties.

It seems that the position of the court in *Ceramic's case*, do not reflect the correct position of the law. The pronouncement of the court that among the grounds for appointment by the court of a receiver is that the appointment of the receiver was made under a power contained in the mortgage deed between the parties is not one of the circumstances for appointment by the court of a receiver as expressly provided in the CAMA. Thus, the court seems to have confused the power of the court to appoint a receiver with that of the debenture holder to appoint out of court.⁶⁷⁸ Rather, the third condition upon which the court can appoint a receiver or manager is that the charge to be enforced is secured.⁶⁷⁹

While, it is less difficult to establish the circumstances for appointment by the court of a receiver where the principal money borrowed by the company or the interest is in arrears, the proof of the ground where the security or property of the company is in jeopardy can be daunting and challenging. This is not unconnected to the imprecision in the definition of the word "jeopardy". Worst still, the constituent features of the word 'jeopardy' were not provided in the CAMA. However, some events or happenings that imply that a company or its assets is in jeopardy were set out in *Fasakin*,⁶⁸⁰ thus:-

- Where a company about to be wound up is wholly insolvent and other creditors are threatening action against the company for recovery of their debt, or,⁶⁸¹
- 2. Where a company was insolvent and its books closed,⁶⁸² or,

⁶⁷⁸ K Aina, 'Rethinking the Duties of a Receiver and Powers of Directors of Companies in Receivership under the Nigerian Law' (2015) 6 GRBPL No1, 62.

⁶⁷⁹ CAMA, *op cit*, s.180 (3).

⁶⁸⁰ [1994] 4 N.W.L.R (pt. 340) 597 at 601.

⁶⁸¹ Palmer's, *op cit*, p.14069.

⁶⁸² McMahon v North Kent Iron Works Co. [1891] 2 Ch. 148.

- Where judgment had been entered against a company and execution was likely to issue,⁶⁸³ or,
- 4. Where a company is proposing to distribute among its shareholders a reserved fund which constituted practically its only assets thereby putting the debenture holders interest at risk,⁶⁸⁴ and,
- 5. Where the company's auditors declared in a general meeting and without being challenged by the directors that after providing for liabilities, the company's assets would only cover principal loans secured and that the company's credit and funds were exhausted.⁶⁸⁵

It is important to state that the events that can imply jeopardy to the assets and property of a company and thereby trigger the appointment of a receiver and/or manager by the court are not circumscribed but depends on the fact of each case.⁶⁸⁶ Notwithstanding, it is apt to state that the appointment of a receiver and /or manager by the court is within the province of the equitable jurisdiction of the court. Hence, the granting of same is usually predicated on the fact that it is just or convenient to do so.⁶⁸⁷ In other words, the general ground on which the court will appoint a receiver is the preservation or the protection of the property for the benefit of persons who have interest in it. Furthermore, it should be noted that the words 'just and convenient' have been interpreted as meaning "where is practicable and the interest of justice requires it".⁶⁸⁸

2 Appointment out of Court

⁶⁸³ Edwards v Standard Rolling Stock [1893] 1 Ch. 574; Palmer's, op cit, p.14069.

⁶⁸⁴ Re: Tilt Cove Copper Co. [1913] 2Ch. 588; Palmer's, op cit, p.14069.

⁶⁸⁵ Re: Branstein and Majorline Ltd [1914] 112 L.T. 25.

⁶⁸⁶ H Picarda, *The Law Relating to Receivers and Managers* (London: Butterworths & Co (Publishers) Ltd, 1984) p. 27.

⁶⁸⁷ Atuanya v Atuanya [1994] 1 N.W.L.R (pt 322) 572 at 580, para P; Uwakwe v Odogwe, supra, 579; Fasakin v Fasakin, supra; Okoya & Ors v Santili & Ors, supra; Emodi v Emodi [2007]4 N.W.L.R (pt 1024)412.

⁶⁸⁸ Edwards & Co v Picard [1909] 2 KB 903 at 907; Picarda, op cit, p. 207.

It is common to have the appointment of a receiver and/or manager under the express powers of the debentures or trust deed. The power of a debenture holder or a class of debenture holders entitled to realise his or their security to appoint a receiver of any assets, subject to a mortgage, charge or security is provided in the CAMA.⁶⁸⁹ Just as in the case of appointment by the court of a receiver and/or manager, the appointment of a receiver and/or manager out of court also appears as an option that is not embarked upon as a matter of course. It is mostly triggered on the happening of certain events.

The circumstances upon which the option of appointment of a receiver and/or manager out of court can be invoked include:

- Where the company failed to pay the principal money borrowed of any premium or any instalment of interest, whether in whole or part, within one month after it becomes due.⁶⁹⁰
- 2. Where the company fails to fulfil any of the obligations imposed on it by the debentures or the debentures trust deed.⁶⁹¹
- Where any circumstances provided in the terms of the debentures or debentures trust deed occurs,⁶⁹² or
- 4. Where the company is wound up^{693}
- 5. Where any creditor of the company issues a process of execution against any of its assets or commences proceedings for winding up of the company by order of any court of competent jurisdiction,⁶⁹⁴ or
- 6. Where the company ceases to pay its debts as they fall due;⁶⁹⁵ and

⁶⁸⁹ CAMA, op cit, s. 209(1); Nashtex Int'l Ltd v Habib Bank (Nig) Ltd, supra 328, paras A – F.

⁶⁹⁰ CAMA, *ibid*, s. 208 (1) (a).

⁶⁹¹ *Ibid*, s. 208 (1) (b).

⁶⁹² *Ibid*, s. 208 (1) (c).

⁶⁹³ *Ibid*, s. 208 (1) (d).

⁶⁹⁴ *Ibid*, s. 208 (2) (a).

7. Where any company ceases to carry on business.⁶⁹⁶

It is instructive to note that other additional events not in the CAMA may be specified in the debentures or the debentures trust deed that could entitle the debenture holder or class of debenture holders to appoint a receiver and/or manager. Hence, it is imperative that a would-be receiver should take certain precautions before acceptance of his appointment.⁶⁹⁷

3.3 Status of Receiver and/or Manager

It is pertinent to state that a receiver and/or manager appointed by the court are deemed to be an officer of the court.⁶⁹⁸ In view of this, such a receiver and/or manager exercise his duties and powers in accordance with the directions and instructions of the court.⁶⁹⁹ Consequently, where such a receiver and/or manager is unreasonably hindered from exercising his duties and powers, it shall amount to contempt of court.⁷⁰⁰

In the case of a receiver and/or manager appointed out of court, he is deemed to be an agent of the person or persons on whose behalf he is appointed.⁷⁰¹However, the Court of Appeal in *Fadeyibi v I. H. (Beverages) Ltd* appears to have misconstrued the provisions of the law when it pronounced that a receiver appointed pursuant to a court order is also deemed to be an agent of the person or persons on whose behalf he is appointed.⁷⁰² It is explicit that the receiver appointed by the court is deemed an officer of the court.⁷⁰³ The provisions of section 390 (1) of the CAMA that the court

⁷⁰² Fadeyibi v I. H. (Beverages) Ltd, supra, 374.

⁶⁹⁵ *Ibid*, s. 208 (2) (b).

⁶⁹⁶ *Ibid*, s. 208 (2) (c).

⁶⁹⁷ Picarda, *op cit*, p.43

⁶⁹⁸ CAMA, op cit, s. 389(2); Picarda, op cit, p.272.

⁶⁹⁹ CAMA, *ibid*; *Dagazau v Bokir Int'l Co. Ltd* [2011]14 N.W.L.R (pt 1267)261 at 330, paras B – C; Jukok Int'l Ltd v Diamond Bank Plc [2016] 6 N.W.L.R (pt 1507) 55 at 94.

⁷⁰⁰ Picarda, *op cit*, p.64; Abugu, *op cit*, p. 231.

⁷⁰¹ CAMA, op cit, s. 390 (1); Fadeyibi v I. H. (Beverages) Ltd [2013] 4 N.W.L.R (pt 1344) 353.

⁷⁰³ CAMA, op cit, s. 389 (2); Palmers', op cit, 14072; Abugu, op cit; H Y Bhadmus, Corporate Insolvency Law and Practice in Nigeria (Enugu: Chenglo Limited, 2009) p. 70.

relied upon in the Fadevibi v I. H. (Beverages) Ltd is only limited and concerned with appointment of a receiver out of court and not by court.

Although it should be noted that the deemed status of a receiver appointed out of court as provided under sections 389 and 390 of the CAMA, are rebuttable. Since the aforesaid sections of the CAMA are default provisions, where the status of the receiver and/or manager is expressly declared in the debentures or any instrument or court order, the status of the receiver and/or manager as expressly declared in the debentures or instrument or court order shall prevail.⁷⁰⁴

It is pertinent to state that at Common Law, a receiver and/or manager appointed by the mortgagee is deemed to be the agent for the mortgagor so that the mortgagee may sue the mortgagor for any loss resulting from the negligent act or omission of the receiver.⁷⁰⁵ Put in another way, while the receiver and/or manager is entitled to take possession of the company's asset and operate the business on behalf of the debenture holder, the debenture holder assumes the position of being virtually a mortgagee in possession through the receiver and/or manager and is thereby liable or answerable for the commitment entered into by his receiver and/or manager in the execution of the business.

Notwithstanding that the receiver is deemed to be an agent for the person or persons on whose behalf he is appointed, where such a receiver is also appointed a manager of the whole or any part of the undertakings of a company, he stands in a fiduciary relationship to the company and is duty bound to observe utmost good faith towards it in any transaction with it or on its behalf.⁷⁰⁶

 ⁷⁰⁴ Picarda, *op cit*, pp. 59 – 60; Palmer's, *op cit*, p. 14080/1.
 ⁷⁰⁵ Halsbury's Laws of England (4th edn, reissue, vol 7(2), 1988) p.913, para 1157.

⁷⁰⁶ CAMA, *op cit*, s.390 (1).

In the later part of this chapter, the receiver and/or manager fiduciary relationship with the company will be comprehensively discussed.

3.4 Qualifications for Appointment of Receiver and/or Manager

There are no mandatory qualifications imposed or required for a person to be appointed as a receiver and/or manager in Nigeria. Rather, there is disqualification provision in relation to the appointment of receiver and/or manager in the CAMA.⁷⁰⁷ Thus, the following persons are disqualified from being appointed or to act as a receiver and/or manager of any property or undertaking of any company, namely: -

- 1. an infant;
- 2. any person found by a competent court to be of unsound mind;
- 3. a body corporate;
- 4. an undischarged bankrupt, unless such person has been given leave to act as a receiver or manager by the court that adjudged him bankrupt.
- 5. a director or auditor of the company;
- any person convicted of any offence involving fraud, dishonesty, official corruption or moral turpitude and who is disqualified under section 254 of the CAMA.

Hence, it is important to state that a body corporate may in certain circumstance be qualified to act as a receiver.⁷⁰⁸ This where the statute in force so permits. In this study, the conferment of powers to act as a receiver given to a body corporate such as AMCON is deprecated as it is against international best practices and a receiver and/or manager is a disinterested and impartial person.

It is unarguable that the absence of statutory provision on qualification for appointment of a receiver and/or manager in Nigeria is to say the least, unsatisfactory.

⁷⁰⁷ CAMA, *op cit*, s. 387 (1).

⁷⁰⁸ AMCON Act, s.48(1)

In the face of the wide ranging duties and powers exercisable by a receiver and/or manager in the course of receivership, it is not appropriate for appointment of a receiver and/or manager to be for "all and every person" insofar as the person is not statutorily disqualified. In this study, it is respectfully submitted that there should be a minimum qualification criterion for persons to be appointed or to act as receivers and/or managers.

3.5 Validity of Appointment of Receivers and/or Managers

The ascertainment of the validity of appointment of a receiver and/or manager is immeasurable and not to be ignored. Where an appointment of a receiver and/or manager is in contravention of the requirements of the law or the terms of the debentures, such appointment would be void and subject to challenge by the company.

The validity of the appointment of receivers and/or managers is mostly ascertained on the following grounds:-

1. Formalities of Appointment

There is no set statutory form for the appointment of a receiver and/or manager out of court. However, any formalities stipulated in the relevant debentures or trust deed must be followed. Where the debenture requires the appointment to be made in writing or by deed, that form must be observed. It is important to state that a deed is required not only where the appointment provision expressly states that the appointment shall be by deed, but also, where the receiver and/or manager is given the power to execute deeds in the name of the debenture holders.⁷⁰⁹ Also, where the debenture requires the consent of other debenture holders for an appointment to be made, such consent must be first obtained.

⁷⁰⁹ Picarda, *op cit*, p. 44.

2. Conditions Precedent

Where there are express condition precedent to the appointment of a receiver, such conditions must be faithfully satisfied if the appointment is to be valid. Some of the conditions precedent to the appointment of a receiver as provided in the CAMA or the debenture include:-

(i) Necessity for Formal Demand

It is a moot point whether a formal demand for payment of money due is a prerequisite to the appointment of a receiver. It is submitted that the making of a demand is not a prerequisite to the validity of an appointment of a receiver under the extant law in Nigeria. However, it is worthy to note that express words in a debenture or trust deed may make a demand a crucial element in the appointment of a receiver. But in the absence of express words requiring a demand to be made, the necessity of demand cannot be implied.⁷¹⁰

(ii) Requirement of Time

As a corollary to (i) above, is whether there is a prescribed time limit to be observed prior to the appointment of a receiver. This issue borders more on the time that the realisation of security by the appointment of a receiver and/or manager can be invoked when a company fails to pay the principal sum borrowed or any instalment of interest or any premium.⁷¹¹ The prescribed time limit provided in the CAMA is within one month after it becomes due.⁷¹²

Arising from the above, is the contention as to whether a receiver and/or manager can be appointed to realise the security for the debenture holder without exhaustion of the prescribed time limit of one month after the payment becomes due

⁷¹⁰ H Picarda, 'Receiver and on Demand Debentures' (1984) 5 Bus L Rev, pp105-107.

⁷¹¹ CAMA, *op cit*, s. 208 (1) (a).

⁷¹² *Ibid*.

or the happening of the event specified in the debentures. It is submitted that where there is prescription of time in a statute or the debenture as a condition precedent for the appointment of a receiver and/or manager, it must strictly be complied with. Thus, where appointment of a receiver and/or manager did not comply with the condition precedent as to the time or happening of a specified event as provided in the statute or debenture, such appointment may be challenged on the ground of non-exhaustion of prescribed time limit, hence the appointment is premature and incompetent.

(iii) Authority to Appoint.

The validity of an appointment of a receiver and/or manager can be challenged on the ground that the person making the appointment lacked the necessary authority to make the appointment at the time it was so made. In view of this, it is important that the appointment of a receiver and/or manager be made by persons with the appropriate standing to so do as provided in the statute or debenture.⁷¹³

The persons with the requisite authority to appoint a receiver and/or manager under the extant laws in Nigeria are:-

- a) The trustee of the covering debenture trust deed, 7^{14}
- b) Debenture holder(s),⁷¹⁵
- c) Asset Management Corporation of Nigeria,⁷¹⁶
- d) The court on the application of the trustee 717 or any interested person.⁷¹⁸
- 3. Absence of Registration of the Charge

Another issue that can impugn the validity of appointment of a receiver and/or manager is where the charge under which he was appointed was not registered where

⁷¹³ CAMA, *op cit*, s. 209(1).

⁷¹⁴ *Ibid*, s. 209(1) (a).

⁷¹⁵ *Ibid*, s. 209(1) (b) (c). *op cit*

⁷¹⁶ AMCON Act, s. 48(1).

⁷¹⁷ CAMA, op cit, s. 209(1) (d).

⁷¹⁸ *Ibid*, s. 389 (1).

registration is mandatory. In this respect, the validity of an appointed receiver can be attacked by a liquidator or creditor on the ground that the charge itself is void for want of registration.⁷¹⁹ It should be noted that while the charge remains valid and effective as against the company, in the case of a liquidator and creditors, nonregistration of the charge on which the appointment of a receiver and/or manager is derived from is fatal and may subject the appointment to a challenge or attack.

3.6 Procedure after Appointment of Receiver/Manager.

Upon the appointment of a receiver and/or manager, there are certain procedural steps that must be taken. These are:-

1. Notice of Appointment and Registration

Where the receiver and/or manager is appointed by the court, he shall advertise the appointment in the gazette and two daily newspapers.⁷²⁰ Where the appointment is made out of court, the receiver or manager shall:

- Immediately send notice to the company of his appointment and the a) terms.⁷²¹
- Within 14 days of the appointment, send notice to the Corporate Affairs b) Commission (CAC) indicating the terms of and remuneration for the appointment.⁷²²
- Where a person obtained an order for the appointment of a receiver and/or c) manager of the property of a company or appointed such a receiver and/or manager under any power contained in any instrument to within seven days from the date of the order or the appointment under the said powers to give notice of the fact to the CAC, which notice of order or appointment

⁷¹⁹CAMA, *op cit*, s. 197(1).

⁷²⁰*Ibid*, s. 181.

⁷²¹ *Ibid*, s. 396(1) (a). ⁷²² *Ibid*, s. 392 (1).

on payment of prescribed fee shall be entered in the register of charges.⁷²³

2. **Statement of Affairs**

Upon receipt of the notices as listed above, the following step takes place:

- The company shall within 14 days or such longer period as may be a) allowed by the court or by the receiver, submit to the receiver, statement of the affairs of the company in the prescribed form.⁷²⁴
- The receiver shall within 2 months after the receipt of the above-stated b) statement of affairs, send to:⁷²⁵
- The court, a copy of the statement and his comments on it, if any,⁷²⁶ or, i.
- ii. The CAC, a copy of the statement and a summary of the statement together with his comments thereon, if any,⁷²⁷ or,
- The company, a copy of his comments and where no comments, a note to iii. that effect,⁷²⁸or,
- Any trustees of the debenture and every debenture holder, whose address iv. is known, a copy of the summary of the statement.⁷²⁹

It should be noted that the statement of affairs must include the matters stated in section 397 (1) of the CAMA. Such statement of affairs is to be verified by affidavit of one or more of the directors of the company and by the secretary of the company.⁷³⁰ However, the statement of affairs may be verified by the affidavit, subject to the direction of the court, by other persons than directors and the secretary

⁷²³ *Ibid*, s. 206 (1).

⁷²⁴CAMA, *op cit*, s. 397 (1) (b).

⁷²⁵*Ibid*, s. 396 (1) (c).

⁷²⁶*Ibid*, s. 396 (1) (c) (i).

⁷²⁷ *Ibid*, s. 396 (1) (c) (i).

⁷²⁸ *Ibid*, s. 396 (1) (c) (ii). ⁷²⁹ *Ibid*, s. 396 (1) (c) (iii).

⁷³⁰ *Ibid*, s. 397 (2).

of the company.⁷³¹ It is important to state that any person making the statement of affairs, verified by affidavit, is entitled to recoup the costs and expenses incurred from the receiver or his successor as may be considered reasonable.⁷³²

3. **Publication of Appointment**

There shall be a statement in every invoice, order for goods or business letter issued by or on behalf of the company or the receiver and/or manager or the liquidator of the company, being a document on or in which the company's name appears, the fact of appointment of a receiver and/or manager.⁷³³

4. Delivery and Showing of Accounts, Receipts and Payments

Where a receiver and/or manager is appointed under the powers contained in any instrument, such receiver and/or manager shall within one month or such longer periods as the CAC may allow, after the expiration of 6 months of his appointment and of every subsequent period of 6 months, and within 1 month after he ceases to act as receiver and/or manager, deliver to the CAC for registration an abstract in the prescribed form showing his receipts and his payment during that period of 6 months, or where he ceases to act, between the date of the last abstract and the date of his so ceasing to act, the aggregate amount of his receipts and payments during all preceding periods since his appointment.⁷³⁴

In the case of a receiver and/or manager appointed by court, such receiver and/or manager shall within 2 months or such longer periods as the CAC may allow, after the expiration of 12 months of his appointment and of every subsequent period of 12 months, and within 2 months after he ceases to act as receiver and/or manager, deliver to the CAC for registration an abstract in the prescribed form showing his

⁷³¹ *Ibid*, s. 397 (2) (a) (b) (c) (d).

⁷³²CAMA, *op cit*, s. 397 (3).

⁷³³*Ibid*, s. 392(1).

⁷³⁴*Ibid*, s. 398 (1); Companies Regulations 2012, reg 42.

receipts and his payment during that period of 12 months, or where he ceases to act, between the date of the last abstract and the date of his so ceasing to act, the aggregate amount of his receipts and payments during all preceding periods since his appointment.⁷³⁵

It is imperative to state that failure to observe the specified procedure after appointment of a receiver and/or manager constitutes an offence⁷³⁶ punishable to a fine not exceeding N25 or N50 for every day during which the default continues. However, the non-compliance with the post-appointment procedure does not vitiate or invalidate the appointment.

3.7 Legal Consequences of Appointment of Receiver and/or Manager

There are legal consequences that flow from the appointment of receiver and/or manager, to wit:

1. Legal Personality of a Company

The appointment of a receiver and/or manager does not extinguish the legal personality of the company under receivership. Where a receiver and/or manager are appointed for a company, the company neither loses its legal personality nor title to the goods in receivership.⁷³⁷ The company in receivership still remains in existence.⁷³⁸ The life and the legal personality of a company is extinguished only upon its winding up and dissolution thereafter.⁷³⁹

⁷³⁵ Companies Regulation, *op cit*, reg 41.

⁷³⁶ CAMA, *op cit*, ss. 206 (3), 392 (2), 396 (7), 397(5), 398 (2).

⁷³⁷ Solar Energy Advanced Power Systems Ltd (S.P.E.A.S) v Ogunnaike [2008] 14 N.W.L.R (pt 1106) 1 at 12, para G; Christlieb Plc v Majekodunmi [2008] 16 N.W.L.R (pt 1113) 324 at 345, paras C – E; Union Bank of Nigeria Ltd v Tropic Foods Ltd [1992] 3 N.W.L.R (pt 228) 231; Pharmatek Industries Ltd v Trade Bank (Nig) Ltd [1997]7 N.W.L.R (pt 514) 639.

⁷³⁸ Remalo Ltd v National Bank of Nigeria Ltd [2003] 16 N.W.L.R (pt 846) 235 at 243, para F.

⁷³⁹ Commercial Bank (Credit Lyonnais) Nig Ltd v Okoli [2009] 5 N.W.L.R (pt 1135) 446 at 461, paras F-G; N.D.I.C v Okem Enterprises Ltd [2004] 10 N.W.L.R (pt 880)107; Ehiadimhen v Musa [2000] 8 N.W.L.R (pt 669)540; (2000) 4 SC (pt 11)166 at 184.

Consequently, since the company retains its title to the goods in receivership and its legal personality remains intact, the company has a right to ensure that its property is not dissipated and therefore, has a legal right to seek injunctive reliefs so as to protect its assets.⁷⁴⁰

2. Powers of Directors of the Company

By virtue of the provisions of section 63 (1) and (3) of the CAMA, the board of directors of a company is vested with the powers to manage the business of the company. Hence, it has been argued as to whether the directors of a company may continue to exercise powers over the business, property or undertakings of the company under receivership. It is posited that the effect of the appointment of a receiver and/or manager by the court is that it practically removes the conduct and guidance of the affairs of the company from the directors and places it in the hands of the receiver and/or manager so that the directors become incapable of making any contract on behalf of the company or exercising any control over any part of the property or assets of the company in relation to the particular assets to which the security attaches.

The appointment of the receiver and/or manager supersedes the powers of the company and the authority of its directors in the conduct of its business which remains in abeyance during the appointment.⁷⁴¹ Suffice to state that with the appointment of a receiver and/or manager, the powers of the company and the authority of the directors cease only in respect of those assets within the scope of the charge or where

⁷⁴⁰ Union Bank of Nigeria Ltd v Tropic Foods Ltd, supra, 246, para A.

⁷⁴¹ Orojo, op cit, p. 439 – 440; Palmer's, op cit, p. 14072; Moss Steamship Co v Whinney [1912] A C 254.

the receiver has refused to act, the company and the directors retain their powers.⁷⁴² Simply put, the appointment of a receiver or manager paralyses the powers of the company and authority of its directors in respect of the company's assets in receivership.⁷⁴³

It is therefore indubitable that the appointment of a receiver and/or manager does not end all the powers of the board. The directors still have continuing powers and duties. Thus such appointment does not prevent the company or its directors from suing or being sued in respect of those matters outside the control of the receiver and/or manager.⁷⁴⁴ Hence, the surmise of the effect of the appointment of a receiver and/or manager is that the directors of the company do not become *functus officio* upon the appointment of a receiver.⁷⁴⁵ The powers of the directors over the company's property and management of the business are only temporarily suspended during the receivership.⁷⁴⁶ The company continues to be a going concern and is not thereby liquidated or wound up, the directors are not also removed as directors, and may continue to function as directors of the company over those assets of the company not covered by the debenture.

However, in respect of the assets constituted and charged under the debenture deed, the powers of the directors to deal with the company's assets cease with the

⁷⁴² CAMA, op cit, s. 393 (3); Omojasola v Plisson Fisko (Nigeria) Ltd, supra; Intercontractors Nig. Ltd v N.P.F.M.B (1983) 1 NSCC 759; Fadeyibi v I. H. (Beverages) Ltd, supra, 373, paras F – H; UBA Trustees Ltd v Nigergrob Ceramic Ltd [1987] 3 N.W.L.R (pt 62) 600; Oluyori Bottling Industry Limited v Union Bank of Nigeria Plc [2009] 3 N.W.L.R (pt 1127) 129; Smith, op cit, p. 337.

⁷⁴³ Standard Printing & Publishing Co Ltd v N. A. B Ltd (2003) FWLR (pt 137) 1097 at 1109, para C;
Intercontractors Nig. Ltd v N. P. F. M. B, supra; U.B.N Plc v David Lab Ltd [2014] All FWLR (pt 714)
157 at 171, paras F – G; Unibiz Nig Limited v C.B.C.I Ltd (2003) FWLR (pt 152) 71; Solar Energy
Advanced Power Systems Ltd (S.P.E.A.S) v Ogunnaike, supra; Dagazau v Bokir Int'l Co. Ltd, supra, 325 – 326.

⁷⁴⁴ UBA Trustees Ltd v Nigergrob Ceramic Ltd, supra, at 612; Orojo, op cit, p. 440.

⁷⁴⁵ CAMA, *op cit*, s 390 (1).

⁷⁴⁶ Central London Electricity Ltd v Berners [1985] 1 KBD 160.

appointment of the receiver.⁷⁴⁷ The receiver in view of powers invested in him under schedule 11 to the CAMA⁷⁴⁸ is empowered to bring an action or defend same in the name of the company and remains the only authority to use the name of the company in actions before the court during the receivership. But the directors may maintain action in the name of the company under certain circumstances.⁷⁴⁹

3. Contracts of Employment

Where the receiver and/or manager is appointed by court, upon his appointment, the company's employees are automatically dismissed and may claim damages for breach of contract even when they are subsequently re-employed by the receiver.⁷⁵⁰ But where the receiver and/or manager is appointed out of court and regarded as agent of the company, the appointment does not of itself automatically terminate contracts of employment previously made and subsisting between the company and its employees.⁷⁵¹ However, this proposition of the law as to the effect of appointment of receiver and/or manager as it relates to contracts of employment in a company yields to exceptions. The first exception is that where the appointment is accompanied by a sale or 'hiving down' of the business of the company.⁷⁵² The second exception occurs where simultaneously with or very soon after his appointment, the receiver and/or manager enters into a new agreement with a particular employee that is inconsistent with the old contracts. In other words, the existing contract of employment is where the exception is where the manager of the manager is appointed by a superseded and *ex hypothesi* terminated.⁷⁵³ The third exception is where the

⁷⁴⁷ Moss Steamship Co. v Whinney, supra, at 260; Newhart Developments v Co-Op Commercial Bank [1978] 2 All ER 901; Jukok Int'l Ltd v Diamond Bank Plc, supra at 96

⁷⁴⁸ CAMA, *op cit*, s. 393 (v).

⁷⁴⁹ Aina, art cit, 68.

⁷⁵⁰ Palmer's, *op cit*, p. 14087; Abugu, *op cit*, p. 239.

⁷⁵¹ Palmer's, *ibid*, p. 14088; Picarda, *op cit*, p. 104; Smith, *op cit*, p. 337; Abugu, *ibid*.

⁷⁵² *Re Foster Clark Ltd* [1966] 1 All ER 43.

⁷⁵³ Griffiths v Secretary of State for Social Services [1974] QB 468.

continuation of the employees' employment is inconsistent with the role and functions of the receiver as receiver and manager.

It is arguable what the effect of the appointment of a receiver will be on the existing contracts of employment if he is acting as an agent for the debenture holders. Although there is no settled precedence, the preponderance of opinion⁷⁵⁴ is that such appointment will terminate the company's existing contracts of employment. In *Dagazau v Bokir Int'l Co. Ltd*, the court pronounced that "... the powers of the board automatically vests in the *defacto* sole administrator who determines the continuity of the secretary and indeed, of all other staff".⁷⁵⁵ It should be noted that where the receiver and/or manager adopts the existing contracts of employment, he is bound by them.⁷⁵⁶

4. Commercial Contracts

Upon the appointment of a receiver and/or manager, existing commercial contracts remain binding upon the company but the receiver is not bound to adopt them. The receiver does not, it seems, owe any duty to the other party to adopt the contract. The position of the law is that where the abandonment of an existing commercial contract would affect the goodwill of the business of a company and the beneficial realisation of its assets,⁷⁵⁷ the receiver and/or manager should not recline from adopting such contracts. The court in *Tanarewa (Nig) Ltd v Plastifarm Ltd* appeared to have misconstrued the law when it held that section 393 of the CAMA provides that a contract entered before the appointment of a receiver and/or manager is binding on

⁷⁵⁴ Dagazau v Bokir Int'l Co. Ltd, supra, at 325 – 326, paras H – A; Picarda, op cit, pp. 105 – 106.

⁷⁵⁵ Dagazau v Bokir Int'l Co. Ltd, supra.

⁷⁵⁶ S Barber, *Company Law* (4th edn, London: Old Bailey Press, 2004) p. 185.

⁷⁵⁷ Tanarewa (Nig) Ltd v Plastifarm Ltd, supra, at 380.

him and cannot be renounced.⁷⁵⁸ This position of the court seems to be *obiter* as section 393 of the CAMA never dealt with contracts.

It should be noted that where a receiver and/or manager of a company entered into a contract, he or she shall be personally liable, in the absence of express negation of liability.⁷⁵⁹ A receiver and/or manager is however, entitled to an indemnity in respect of liability suffered as a result of the contract entered by him in receivership.⁷⁶⁰

5. Crystallisation of Floating Charges

Upon the appointment of a receiver and/or manager, the floating charges crystallise and become fixed charges.⁷⁶¹ Thus, the company loses the benefits that were hitherto derivable from its assets being secured as a floating charge as aptly described in *Siebe Gorman & Co. Ltd v Barclays Bank Ltd*.⁷⁶² Hence, all the assets ceases to be available to the company and comes under the control of the receiver and/or manager as the company and its directors loses the right to deal in or use those assets in the normal course of business of the company without the receiver's consent.⁷⁶³

6. Execution

The appointment of a receiver and manager does not operate as a stay of execution.⁷⁶⁴ However, it is pertinent to state that the attachment and sale will be ineffective against the company's assets that have ceased to be in its custody.⁷⁶⁵ Hence, the title of the receiver or manager prevails over those of unconcluded execution creditors.⁷⁶⁶

⁷⁵⁸ *Ibid*, at 379.

⁷⁵⁹ CAMA, *op cit*, s. 394 (1); Abugu, *op cit*, p. 239.

⁷⁶⁰ CAMA, op cit, s. 394 (2) (3); Abugu, op cit.

⁷⁶¹ CAMA, *ibid*, s. 178 (1) (a) (b), (2); *UBA Trustees' v Nigergrob Ceramic Ltd*, *supra; Re: Griffin Hotel Company Ltd* [1941] Ch 129.

⁷⁶² [1979] 2 Lloyd's Report 142; *Re: Yorkshire Wool Combers Association* [1903] 2 Ch 284.

⁷⁶³ Palmers's, *op cit*, p. 14087.

⁷⁶⁴ Intercontractors Nigeria Ltd v UAC Nigeria Ltd (1988) 1 NSCC 737 at 751.

⁷⁶⁵ Omojasola v Plisson Fisko (Nigeria) Ltd, supra, at 434.

⁷⁶⁶ Intercontractors Nigeria Ltd v UAC Nigeria Ltd, supra.

7. Actions

The position of the law before the enactment of the CAMA was that since the receiver and/or manager does not have any legal estate in the assets as the company retained its title to assets and legal personality, the receiver and/or manager requires the leave of court to enable him institute or defend an action in the name of the company with respect to assets involved in the receivership.⁷⁶⁷ Hence, where the receiver is appointed by the court, no action can be brought against him or in respect of the property in his possession without the leave of court since he is an officer of the court⁷⁶⁸ and any interference with the performance of his duties may amount to contempt of court.⁷⁶⁹ Where the receiver and/or manager is appointed by a debenture holder, such a receiver and/or manager can institute and defend actions in the name of the debenture holder or the company entitled to the goods comprised in the debenture.770

It would appear that the incontrovertible position of the law in Nigeria is that leave of Court⁷⁷¹ is not required by a receiver and/or manager appointed out of court to bring or defend an action in the name of the company. This position of the law found support in *Onofowokan v Wema Bank Plc*⁷⁷² thus:

⁷⁶⁷ Solar Energy Advanced Power Systems Ltd (S.P.E.A.S) v Ogunnaike, supra at 15, para D, 16, para F; Bhadmus, op cit, pp. 78 – 79.

⁷⁶⁸ CAMA, op cit, s. 389 (2); Dagazau v Bokir Int'l Co. Ltd, supra, at 330, paras B – C.

⁷⁶⁹ Intercontractors Nigeria Ltd v UAC Nigeria Ltd, supra at 737; Abugu, op cit, p. 231.

⁷⁷⁰ CAMA, op cit, 11th Schedule, item 5; Intercontractors Nigeria Ltd v UAC Nigeria Ltd, ibid; Dagazau v Bokir Int'l Co. Ltd, supra at 331, paras B – C; Brifina Ltd v Intercontinental Bank Ltd [2003] 5 N.W.L.R (pt 814) 540 at 577, paras F – H.

⁷⁷¹ Unibiz (Nig) Ltd v C.B.C.L Ltd, supra; Inter-contractors Nig. Ltd v N.P.F.M.B [1988] 2 N.W.L.R (pt 76) 280; ⁷⁷² [2011] All FWLR (pt. 585) 201 at 219, paras D–F, per Muhammed JSC.

consequently, as the appointment of the 3rd respondent by 1st respondent as receiver/manager over all the properties of the 2nd respondent company charged under the debenture trust deeds was made in substantial compliance with section 393 (3) of the Companies and Allied Matters Act and Schedule 11 of that Act, the 3rd respondent required no leave of court to institute the action against the appellants to recover the loan granted to the 2nd respondent which is still awaiting liquidation. The cases of Intercontractors (Nig) Ltd v N.P.F.M.B;⁷⁷³ Intercontractors (Nig.) Ltd v U.A.C (Nig.) Ltd;⁷⁷⁴ Adegboyega .v. Dina⁷⁷⁵ and Unibiz (Nig.) Ltd v C.B.C.L Ltd^{776} in which the provisions of section 393 (3) of the Companies and Allied Matters Act and Schedule 11 of that Act did not arise for consideration, are not relevant.

However, it is important to state that the leave of court to institute or defend an action in the name of the company is still required where the receiver and/or manager appointed by the court is not for the whole or substantially the whole of the company's property to be entitled to the powers provided in the Eleventh Schedule of the CAMA. Furthermore, a receiver and/or manager is not entitled to bring an action

⁷⁷³ Supra. ⁷⁷⁴ Supra ⁷⁷⁵ [1992] 7 N.W.L.R (pt. 255) 576.

⁷⁷⁶ Supra

in his own name as title in the goods in receivership is not vested in him but the company.⁷⁷⁷

3.8 Duties and Powers of Receiver and/or Manager

The receiver and/or manager is conferred with far-reaching powers under the extant statute or debenture instrument in order to function as a receiver and/or manager. The purpose of the appointment of a receiver and/or manager is primarily to work towards paying outstanding debts or redeeming security or freeing property from some jeopardy for the benefit of creditors or debenture holders on whose behalf the appointment is made.⁷⁷⁸ In order to carry out the purpose for his appointment, a receiver and/or manager exercises some far-reaching duties and powers. While some of the duties and functions of a receiver and/or manager are usually specified in the court order appointing the receiver and/or manager or the debenture instrument, other expansive general duties and powers of receivers and/or managers are contained in the CAMA. Hence, the dictum of the court that the "the duty and function of a receiver appointed by court is not different from that of a receiver appointed by debenture holders"⁷⁷⁹ seems not a correct representation of the law. Although, the general duties and functions of a receiver and/or manager as provided in the CAMA are common to a receiver and/or manager appointed by court and that appointed out of court, in terms of their specific duties and functions, it is circumscribed to the order of appointment and the provisions of the debentures.

In addition to the duties as may be contained in the debenture instrument or order of court appointing him, some of the threshold duties of a receiver and/or manager under statute or common law are:

⁷⁷⁷ Shinning Star Nigeria Ltd v Ask Steel Nig. Ltd (2011) All FWLR (pt 578) 825 at 861, para B; Atoju v Triumph Bank Plc [2016] 5 N.W.L.R (pt 1505) 252 at 308, paras C – D.

⁷⁷⁸ CAMA, op cit, s. 393 (1) (2); Anatogu v Anatogu [1998] 5 N.W.L.R (pt 548) 42; Ajayi, op cit, p. 569; Smith, op cit, p. 343. ⁷⁷⁹ Solar Energy Advanced Power Systems Ltd (S.P.E.A.S) v Ogunnaike, supra, at 15.

1. Duty to Receive

The foremost duty of a receiver is, subject to the rights of prior encumbrances, to take possession of and protect the property, receive the rents and profits and discharge all out goings in respect thereof and realise the security for the benefit of those on whose behalf he is appointed.⁷⁸⁰ It should be noted that a receiver cannot carry on any business or undertaking of the company unless he is also appointed a manager or receiver of the whole or substantially the whole or any part of the undertaking of the company.⁷⁸¹ The business or undertaking shall be managed by the receiver and/or manager with a view to the beneficial realisation of the security for those on whose behalf he is appointed.⁷⁸²Thus, the court in *Ponson Enterprises Nig Ltd v Njigha*⁷⁸³ drew a fine distinction between the functions of a receiver and a manager, to wit, "a receiver has the duty to stop the business, collect the debts and realise the assets. He has no authority to carry on a going concern. But a manager, on the other hand, has powers to continue a business.⁷⁸⁴

2. Duty to Manage and the carrying on of the Business of the Company

Where a receiver is appointed for the whole or substantially the whole or any part of the undertaking of the company with a view to the beneficial realisation of the

⁷⁸⁰ CAMA, op cit, s. 393 (1) (2); Babington-Ashaye v E.M.A Gen. Ent. Nig. Ltd (2012) All FWLR (pt 645) 256 at 295, Paras. D-E; P.I.P. Ltd v Trade Bank (Nig) Plc, supra at 637, paras F - G; Nashtex Int'l Ltd v Habib Bank (Nig) Ltd, supra at 329, paras B - E.

⁷⁸¹ Nashtex Int'l Ltd v Habib Bank (Nig) Ltd, supra, at 329, per Kekere-Ekun JCA (as she then was);
P.I.P. Ltd v Trade Bank (Nig) Plc, supra, at 637 – 638, paras F – G; Intercontractors Nig. Ltd v N. P. F.
M. B, supra.

 ⁷⁸² CAMA, op cit, s. 393 (1) (2); Nashtex Int'l Ltd v Habib Bank (Nig) Ltd, supra; P.I.P. Ltd v Trade Bank (Nig) Plc, supra, at 637 – 638, paras F – G; Intercontractors Nig. Ltd v N. P. F. M. B, supra.
 ⁷⁸³ Supra

⁷⁸⁴Ponson Enterprises Nig Ltd v Njigha supra, at 1700, paras. B–C; Uwakwe v Odogwe, supra at 589; Nashtex Int'l Ltd v Habib Bank (Nig) Ltd, supra, at 329, paras B – E; NBCI v Alfijir (Mining)Nigeria Ltd [1993] 4 N.W.L.R (pt 287) 346; [1999] 14 N.W.L.R (pt 638) 176; P.I.P. Ltd v Trade Bank (Nig) Plc, supra, at 637 – 638. J O Bolodeoku, 'Receivership and Enforcement of Tax Liability', Journal of Private and Property Law, 2004, 40.

security of those on whose behalf he is appointed,⁷⁸⁵ or a manager, he can carry on the business or undertaking of the company with a view to the beneficial realisation of the security for those on whose behalf he is appointed.⁷⁸⁶

It is pertinent to state that in exercising the duty to manage and the carrying on the business of the company, the receiver and/or manager is not bound to receive directions from the directors but to give directions and must always have regards to the purpose for which he has being appointed. In the exercise of the duty to manage and the carrying on the business of the company, it must be exercised for proper purpose and not for collateral purpose or extraneous reasons.⁷⁸⁷

It is worthwhile to state that the receiver and/or manager's duty is to manage and carry on the business of the company with a view to the realisation of the debenture holder's security may entail the invocation of the power to sell off the assets of the company,⁷⁸⁸ which sale can be by public auction or private contract.⁷⁸⁹ Furthermore, in exercising this duty to manage and the carrying on business of the company, the receiver and/or manager must act in good faith in the best interests of the debenture holders and having regards to the interests of the company, its members, employees, creditors and other stakeholders.⁷⁹⁰

3. Fiduciary Duty: to Act in Utmost Good Faith

A receiver and/or manager where appointed out of court or by the court stands in a fiduciary relationship to the company and shall observe utmost good faith towards the

⁷⁸⁵ Union Bank of Nigeria Ltd v Tropic Foods Ltd, supra; Re: London Iron and Steel Co. Ltd [1990] BCLC372.

 ⁷⁸⁶ CAMA, op cit, s 393 (1) (2); Nashtex Int'l Ltd v Habib Bank (Nig) Ltd, supra; Intercontractors Nig. Ltd v N. P. F. M. B, supra; Ponson Enterprises Nig Ltd v Njigha, supra; Abugu, op cit, p. 236.
 ⁷⁸⁷ Picarda, op cit, p. 791.

⁷⁸⁸Uwakwe v Odogwe, supra.

⁷⁸⁹ CAMA, *op cit*, s. 393 (3); Eleventh schedule, para 2.

⁷⁹⁰*Ibid*, ss. 390 (1) (2), 393 (2).

company in any transaction with it or on its behalf.⁷⁹¹ Such a receiver and/or manager shall in any transaction, act in utmost good faith, that is, in the best interest of the company as a whole so as to preserve its assets; further its business and promote the purposes for which it was formed.⁷⁹²Thus, a receiver appointed by the court has been held as "not only fills a fiduciary position towards all the debenture holders, but his appointment to an office of such responsibility presupposes that he will discharge his duties with punctilious rectitude".⁷⁹³

Although, the words "utmost good faith" as used in section 390 (1) of the CAMA was not subjected to precise definition, the Supreme Court of Nigeria in *West African Breweries Ltd v Savannah Ventures Ltd*,⁷⁹⁴ adopting Lord Herschell's dictum in *Kennedy v De Trafford*⁷⁹⁵ stated thus:

Granted that the duty of the mortgagee, or receiver, to act in good faith is capable of precise statement, what amounts to good faith cannot be stated with equal precision. ... It is very difficult to define exhaustively all that would be included in the words 'good faith' but I think I would be unreasonable to require the mortgagee to do more than exercise his power of sale in that fashion. Of course, if he wilfully deals with the property in such manner that the interests of the mortgagor are sacrificed, he had not been exercising his power of sale in good faith.

⁷⁹¹CAMA, op cit, ss. 390 (1) (2), 393 (2); Fadeyibi v I. H. (Beverages) Ltd, supra, at 374, paras B –D; Unibiz (Nig) Ltd v C.B.C.L Ltd, supra.

⁷⁹² CAMA, *ibid*, ss. 390 (1) (2), 393 (2); *Fadeyibi v I. H. (Beverages) Ltd, supra,* at 374; *Unibiz (Nig) Ltd v C.B.C.L Ltd, supra, at 402; Abugu, op cit, p. 232.*

⁷⁹³ Palmer's *op cit*, p. 14074.

⁷⁹⁴ [2002] FWLR (pt 112) 53 at 77, paras A–C.

⁷⁹⁵ [1897] AC 180.

Notwithstanding the impreciseness of the words "utmost good faith", it has been held that a receiver and/or manager has failed to exercise utmost good faith where there is evidence of gross negligence and recklessness⁷⁹⁶ of a receiver in dealing with the properties of a company, and collusion⁷⁹⁷ with the purchaser of a mortgaged property.

Suffice to state that the fiduciary duties of a receiver and/or manager encompass that:-

- 1. He must act in the best interests of the company.⁷⁹⁸
- He must exercise his powers for proper purpose and not for any collateral purpose.
- 3. He must not have personal interest in the subject matter of the receivership, or in any self-dealing transactions whatever.
- 4. He must not be negligent or reckless while exercising his duty.⁷⁹⁹
- He must not be tainted with acts of corruption or collusion in any manner howsoever.⁸⁰⁰

It is arguable whether this fiduciary duty to act in good faith and in best interests is owed only to the debenture holders and not the company or other stakeholders. The quick response would be that while a receiver and/or manager exercises this fiduciary duty in order to achieve the primary purpose for his appointment, that is, the realisation of the security on behalf of persons upon whom he is appointed, in doing that, he must have regards to the interests of the company, its members, employees and creditors.⁸⁰¹

⁷⁹⁶ West African Breweries Ltd v Savannah Ventures Ltd, supra, at 84, paras. A–B. ⁷⁹⁷ Ibid, at 77, para. D.

⁷⁹⁸ CAMA, *op cit*, s 390 (1).

⁷⁹⁹ Fadeyibi v I. H. (Beverages) Ltd, supra.

⁸⁰⁰ West African Breweries Ltd v Savannah Ventures Ltd, supra; Picarda, op cit, p.81.

⁸⁰¹ CAMA, op cit, s. 390 (1) (2); Smith, op cit, p. 344.

4. Duty of Care

A receiver and/or manager in the conduct of his responsibilities have the duty to exercise due care, skill, judgment, diligence, faithfulness and act in a manner an ordinarily skilful manager would do in the circumstances.⁸⁰² In order to underscore this duty, a receiver as a manager is enjoined when considering whether a particular transaction or course of action is in the best interest of the company to have regards to the interest of the employees as well as the members of the company, and when appointed by, or as a representative of, a special class of members or creditors, may give special, but not exclusive consideration to the interest of that class. In other words, the receiver may give special but not exclusive consideration to the interest of the interest of the secured creditors on whose behalf he is appointed.⁸⁰³

5. Duty to Give Notice of Appointment and Make Returns

Upon the appointment of a person as a receiver and/or manager, he shall give notice of his appointment to the company and the terms,⁸⁰⁴ call for a statement in the prescribed form as to the affairs of the company,⁸⁰⁵ which statement shall be submitted within 14 days after receipt of the notice of appointment to the receiver in accordance with section 397 of the CAMA and also give notice to the CAC within 14 days of the appointment, indicating the terms and remuneration for the appointment.⁸⁰⁶

The receiver shall within 2 months after receipt of the statement, send a copy to the CAC (in the case of receiver appointed out of court) or to the Court (in the case of receiver appointed by court), and of any comment he sees fit to make thereon and in

⁸⁰² CAMA, *op cit*, s. 390 (2) (a); *Cuckmere Brick Co. Ltd v Mutual Finance Ltd* [1971] 1 Ch 949; [1971] 2 All ER 633; Abugu, *op cit*, p. 232.

 ⁸⁰³ B Adebola, 'The Duty of the Nigerian Receiver to "Manage" the Company'
 http://papers.ssrn.com/sol3/papers.cfm?abstract_id=2297805 accessed on 10 May 2015.
 ⁸⁰⁴ CAMA, *op cit*, s. 396 (1) (a).

⁸⁰⁵ *Ibid*, s. 396 (1) (b).

⁸⁰⁶*Ibid*, s 392 (1).

the case of the Corporate Affairs Commission, also, a summary of the statement and of his comments if any thereon,⁸⁰⁷ to the company, a copy of any such comments as aforesaid or if he does not see fit to make any such comment, a notice to that effect⁸⁰⁸ and to any trustees for the debenture holders on whose behalf he has been appointed, a copy of the said summary.⁸⁰⁹

The receiver shall within 2 months or such longer period as the court may allow after the expiration of the period of 12 months from the date of his appointment, and of every subsequent period of 12 months and within 2 months or such longer period as the court may allow after he ceases to act as receiver or manager of the property of the company, owe the duty to send to the CAC, to any trustees for the debenture holders on whose behalf he was appointed, so far as he is aware of their addresses; to the company, an abstract in the prescribed form showing his receipts and payments⁸¹⁰ of the receivership. Notwithstanding this provision, a receiver owes the duty to render proper accounts of his receipts and payments as he may be required from time to time⁸¹¹ and deliver to the CAC for registration⁸¹² an abstract in the prescribed form, showing his receipts and payments during that period of 6 months.

Furthermore, a receiver or a manager owes the duty to ensure that every invoice, order for goods or business letter issued by or on behalf of the company shall contain a statement that a receiver or manager has been appointed.⁸¹³

6. Duty to Distribute Company's Assets

It is the duty of the receiver to distribute all moneys realised from sale of assets, in accordance with his instrument of appointment or the order of court. Where a receiver

⁸⁰⁷CAMA, *op cit*, s 396 (1) (c) (i).

⁸⁰⁸*Ibid*, s 396 (1) (c) (ii).

⁸⁰⁹ *Ibid*, s.396 (1) (c) (iii).

⁸¹⁰ *Ibid*, s. 396 (2); Companies Regulation, *op cit*, reg 41.

⁸¹¹CAMA, *ibid*, s. 396 (6).

⁸¹²*Ibid*, s. 398 (1); Companies Regulation, *op cit*, reg 42.

⁸¹³ CAMA, *ibid*, s. 392 (1); Insolvency Act 1986 (as amended), s. 39 (1) (UK).

and/or manager is appointed by an order of court, an application shall be brought for an order of court setting out the manner of distribution among the various claimants.⁸¹⁴ In the absence of express provision in the debenture or the order of court, the receiver and/or manager distribute the proceeds of the receivership as follows⁸¹⁵ :-

- i. The cost of realising the security.
- ii. Other expenses of the receiver and manager including remuneration and assets;
- iii. Costs, expenses, other outgoings and the expenses of trustees under a debenture, trust deed, if any, including their remuneration
- iv. Cost of debenture holder's action, if any preferential debts out of the property, subject to a floating charge in priority to the claims of the debenture holder.⁸¹⁶
- v. The debenture debt with interest accruing therein up to the date of payment.

2. Powers of Receiver and/or Manager

The powers of the receiver and/or manager may be set out in the debentures or order of court. However, the powers that are exercisable by him are now regulated by the CAMA.⁸¹⁷ It should be noted that the debenture instrument or order of court upon which the appointment is made may exclude the applications of the provisions of the CAMA where necessary.⁸¹⁸ Furthermore, all other implied and incidental powers which prior to the CAMA were exercisable by receivers and/or managers are now

⁸¹⁴ Omojasola v Plisson Fisko (Nigeria) Ltd, supra, at 434.

⁸¹⁵ *Ibid*; Abugu, *op cit*, pp. 241 – 242.

⁸¹⁶ CAMA, op cit, s. 182.

⁸¹⁷*Ibid*, s. 393 (3) & Schedule 11.

⁸¹⁸ Abugu, *op cit*, p. 232.

incorporated in the CAMA. Such implied or incidental powers can only be disregarded if expressly excluded.⁸¹⁹

It is indisputable that a person appointed a receiver and/or manager shall, subject to the rights of prior encumbrances, exercise powers,⁸²⁰ namely: -

- a) To take possession of and protect the property, subject to the mortgage, charge or security,
- b) To receive the rents and profits and discharge all out goings in respect of the charged asset,
- c) To sell those assets,
- d) To collect debts owed the company,
- e) To enforce claims vested in the company,
- f) To compromise, settle and enter into arrangements in respect of claims by or against the company on the company's business with a view to selling it on the most favourable terms,
- g) To grant or accept leases of land and licences in respect of patent, design, copyright or trademarks, and,
- h) To recover any instalment unpaid on the company's issued shares.

Where a receiver or manager is appointed for the whole or substantially the whole of a company's property, in addition to the powers contained in the debenture instrument, such a receiver or manager is empowered to exercise the powers specified in the CAMA to the extent that they are not inconsistent with any of the provisions of the debenture instrument. In the CAMA,⁸²¹ a receiver or manager as aforesaid can exercise the following powers:

⁸¹⁹ Abugu, op cit, p. 233; Onofowokan v Wema Bank Plc, supra, at 218.

⁸²⁰ CAMA, *op cit*, ss. 209 (3), 393 (1) (2); Bhadmus, *op cit*, p. 74.

⁸²¹ CAMA, *ibid*, schedule 11.

- Power to take possession of, collect, and get in the property of the company and, for that purpose, to take such proceedings as may seem to him expedient;
- Power to sell or otherwise dispose of the property of the company by public auction or private contract;
- Power to raise or borrow money and grant security therefore over the property of the company;
- 4. Power to appoint a solicitor or accountant or other professionally qualified person to assist him in the performance of his functions;
- 5. Power to bring or defend any action or other legal proceedings in the name and on behalf of the company;
- 6. Power to refer to arbitration any question affecting the company;
- Power to effect and maintain insurances in respect of the business and property of the company;
- 8. Power to use the company's seal;
- 9. Power to do all acts and to execute in the name and on behalf of the company and deed, receipt or other document;
- Power to draw, accept, make and endorse any bill of exchange or promissory note in the name and on behalf of the company;
- Power to appoint any agent to do any business which he is unable to do himself or which can more conveniently be done by an agent and power to employ and dismiss employees;
- 12. Power to do all such things (including the carrying out works) as may be necessary for the realisation of the property of the company;
- 13. Power to make any payments which is necessary or incidental to the

performance of his functions;

- 14. Power to carry on the business of the company;
- 15. Power to establish subsidiaries of the company;
- Power to transfer to subsidiaries of the company the whole or any part of the business, and property of the company;
- 17. Power to grant or accept a surrender of a lease or tenancy of any of the property of the company, and to take a lease or tenancy of any property required or convenient for the business of the company;
- Power to make any arrangement or compromises on behalf of the company;
- 19. Power to call up any uncalled capital of the company;
- 20. Power to rank and claim in the bankruptcy, insolvency, sequestration or liquidation of any person indebted to the company and to receive dividend, and to accede to trust deeds for the creditors of any such person;
- 21. Power to present or defend a petition for the winding up of the company;
- 22. Power to change the situation of the company's registered office;
- Power to do all other things incidental to the exercise of the foregoing powers.

Conversely, where the receiver or manager is not appointed for the whole or substantially the whole of a company's property, he does not enjoy and cannot exercise the powers provided in Schedule 11 to the CAMA without the leaveor direction of the Court.⁸²² Where such leave or direction is not obtained, the receiver or manager enjoys and exercises the rights and powers as contained in the debenture instrument appointing him.

⁸²² CAMA, op cit, ss. 393 (3), 391.

3.9 Remuneration of Receiver and/or Manager

The receiver and/or manager's remuneration is usually payable by the company where there is a rebuttal of the statutory provision that the receiver and/or manager appointed out of court is a deemed agent of the person or persons on whose behalf he is appointed. On the contrary, where the receiver and/or manager remains a deemed agent of the person or persons on whose behalf he is appointed it is the later that shall be responsible for the remuneration.⁸²³

Instructively, the court is vested with the power on the application of the company or the liquidator of a company, by order, fix the amount to be paid by way of remuneration to any person who, under the powers contained in any instrument, has been appointed as receiver or manager of the property of the company.⁸²⁴ Furthermore, it should be noted that the court may on an application made either by the company or the liquidator or by the receiver or manager, vary or amend an earlier order fixing the remuneration of the receiver or manager.⁸²⁵

3.10 Liabilities and Indemnities of Receivers and/or Managers

The trite position of the law is that a receiver and/or manager of any property or undertaking of a company shall be personally liable on any contract entered into by him.⁸²⁶ However, the personal liability of a receiver and/or manager in a contract can be avoided where there is an express provision in the contract⁸²⁷to the contrary. However, it is also established that a receiver and/or manager may be liable in tort, negligent acts⁸²⁸ and trespass⁸²⁹ where the receiver or manager exercises his duty to a property belonging *bona fide* to a third party.

⁸²³ Picarda, *op cit*, p. 177.

⁸²⁴ CAMA, *op cit*, s. 395 (1).

⁸²⁵ *Ibid*, s. 395 (3).

⁸²⁶*Ibid*, s. 394 (1); Abugu, *op cit*, p. 240.

⁸²⁷ CAMA, *ibid*.

⁸²⁸ Fadeyibi v I. H. (Beverages) Ltd, supra.

Notwithstanding, a receiver and/or manager is entitled to indemnity in the following instances:

- i. As regards contracts entered into by him in the proper performance of his functions.⁸³⁰ It does not matter if he is appointed by court or out of court.⁸³¹
- ii. Where a receiver and/or manager appointed out of court, he shall be entitled to indemnity as regards contracts entered into by him with the express or implied authority of the person that appointed him under the debenture instrument.⁸³²

The persons who appointed the receiver and/or manager shall be responsible for the indemnity, though only to the extent in which the receiver and/or manager is unable to recover from the assets of the company.⁸³³

3.11 Termination of Receivership

A receivership may be determined:

1. By the Parties

The most obvious and frequent mode for determination of receivership is where the receiver and/or manager's duties are fully performed and he is discharged. Such determination is usually consensual and occurs where the receivership has run its course. However, it is pertinent to state that termination of receivership may take place prematurely in the event of the death of the receiver and/or manager⁸³⁴ or where the receiver and/or manager ceases to act.⁸³⁵ A receiver and/or manager may cease to act by reason of resignation or removal where such is provided in the debenture instrument. But where there is no provision for resignation or removal of a receiver

⁸²⁹ Adetona v I.G. Enterprise Ltd [2011] 7 N.W.L.R (pt 1247) 1535.

⁸³⁰ CAMA, *op cit*, s. 394 (2).

⁸³¹ Abugu, *op cit*.

⁸³² CAMA, *op cit*, s. 394 (3).

⁸³³ Ibid.

 $^{^{834}}Ibid$, s. 396 (4).

⁸³⁵ *Ibid*, s. 206 (2).

and/or manager in the debenture instrument, in practice, a receiver and/or manager may lawfully resign his appointment by notice in writing to the person or persons on whose behalf he is appointed and who must consent or accept the resignation.⁸³⁶ But it should be noted that where a receiver and/or manager ceases to act, he shall give notice to the CAC to that effect.⁸³⁷

2. By the Court

The court may terminate a receivership for so many reasons. It can be for breach of any of the receiver and/or manager duties; invalidity of appointment and abuse of powers and misconduct. Also, the court may in the exercise of its inherent jurisdiction remove a receiver and/or manager appointed by a debenture holder.

3. By the Operation of Law

It is trite that the title of a receiver and/or manager is no better than the title of the debenture holder who appointed him. Thus, where a prior incumbrancer⁸³⁸ subsequently appoints his own receiver and/or manager, the receiver and/or manager appointed by the later incumbrancer will perforce be replaced. This replacement is automatic and requires no order of the court except there is a dispute as to priorities.⁸³⁹

3.12 Administrative Receiver in the United Kingdom

The main innovation made by the Insolvency legislation was the introduction of a special category of receiver and/or manager known as the "Administrative Receiver". An Administrative Receiver is defined as a "receiver or manager of the whole or substantially the whole of the property of a company appointed by or on behalf of the

⁸³⁶ Picarda, *op cit*, p.188.

⁸³⁷ CAMA, *op cit*, s. 206 (2).

⁸³⁸ *Ibid*, s. 393 (1).

⁸³⁹ Picarda, *op cit*, p. 189.

holders of a debenture of the company secured by a floating charge"⁸⁴⁰ or a composite fixed and floating charge.⁸⁴¹Also, the floating charge must be enforceable and the primary condition for enforcing a charge is that the debt secured by the charge is due and unpaid.⁸⁴² Unlike common law receivership, an administrative receiver must report to the unsecured creditors on the progress of the receivership.⁸⁴³ Furthermore, an administrative receiver of a company is deemed to be an agent of the company unless and until the company goes into liquidation.⁸⁴⁴

Effects of the Appointment of Administrative Receiver 1.

i. **On Directors**

From the time of appointment, an administrative receiver of a company has sole authority to deal with the charged property and the directors of the company no longer have any authority to deal with that property. However, they continue in office and are still liable, for instance, to submit returns and documents to Companies House.⁸⁴⁵

ii. Contracts of Employment

It is the position of the law that when the court appoints a receiver of the company's property, the company's contracts of employment are automatically terminated.⁸⁴⁶ However, the appointment of an administrative receiver of a company, as agent for the company, does not alter the personal relationship and so does not terminate the company's contract of employment.⁸⁴⁷ Where an administrative receiver of a company can see no hope of selling its business as a going concern, then he/she

⁸⁴⁰ Insolvency Act, *op cit*, s. 29 (2) (a).

⁸⁴¹ S W Mayson, et al, Mayson, French and Ryan on Company Law (25th edn, Oxford: OUP, 2008 -2009) p. 311. ⁸⁴² *Ibid*, p. 305.

⁸⁴³ Insolvency Act, op cit, s. 48.

⁸⁴⁴ *Ibid*, s, 44 (1) (a); Mayson, *et al*, *op cit*, p. 643.

⁸⁴⁵ Mayson, et al, ibid, pp. 643 – 644.

⁸⁴⁶ Reid v Explosives Co. Ltd [1887] 19 QBD 264.

⁸⁴⁷ Griffiths v Secretary of State for Social Services, supra; Nicolle v Cutts [1985] BCLC 322.

may dismiss the employees forthwith.⁸⁴⁸ If an administrative receiver wants any of the company's employee to continue working for it during the receivership, may either adopt their existing contracts of employment, or acting as agent of the company, negotiate new contract between them and the company.

2. **Liability and Indemnity**

An administrative receiver is personally liable on any contract entered into in the performance of his or her functions, except where the contract otherwise provides.⁸⁴⁹ Exclusion of liability may be express or implied.⁸⁵⁰ For many years, the position regarding an administrative receiver appointed as agent of the company was that in general, he was not liable for contracts entered into by him, for and on behalf of the company. He was no more personally liable than was a director who entered into a contract, for and behalf of his company.⁸⁵¹ The position now is that an administrative receiver is not personally liable for contracts entered before his appointment. But an administrative receiver is personally liable on any contract entered into by him in the carrying out of his functions, except so far as the contract otherwise provides, but he is entitled in respect of that liability to an indemnity out of the assets of the company.⁸⁵²

Duties and Powers of an Administrative Receiver 3.

An administrative receiver carries out duties similar to that of common law receiver and/or manager and exercises a plethora of powers.⁸⁵³ Instructively, the statutory powers of an administrative receiver are *in tandem* with that of a receiver

⁸⁴⁸ Mayson, et al, op cit, p. 645.

⁸⁴⁹ Insolvency Act, op cit, s. 44 (1).

⁸⁵⁰ Mayson, et al, op cit, p. 646.

⁸⁵¹ *Re: Atlantic Computer Systems Plc.* [1992] Ch 505.

⁸⁵² Insolvency Act, op cit, s. 44 (1); CAMA, op cit, s. 394 (1) (2); Atlantic's case, ibid; Mayson, et al, loc *cit.*⁸⁵³ Insolvency Act, *ibid*, ss. 42, 43 & Schedule 1.

and/or manager in the CAMA.⁸⁵⁴ An administrative receiver owes the duty to demand from the directors of the company a statement in a prescribed form of the affairs of the company and in turn, make a report and send a copy to the registrar of companies, to any trustees for secured creditors of the company (in so far as he is aware of their addresses) and to all unsecured creditors of the company (in so far as he is aware of their addresses).⁸⁵⁵

An administrative receiver of a company does not owe any duty of care to the company or to persons who have charges over the company's property ranking under one over which the administrative receiver was appointed. Thus, an administrative receiver appointed under a charge given by a company also owes no duty of care to guarantors of the obligation secured by the charge.⁸⁵⁶

4. Changes in the Law on Administrative Receivership

The Enterprise Act⁸⁵⁷ made amendments to the Insolvency Act, which introduced substantial reforms. The amendments introduced by the Enterprise Act were to the effect that, (i) when the assets of a company subject to a floating charge are realised, a certain proportion must be set aside for the unsecured creditors,⁸⁵⁸ (ii) an administrative receiver cannot be appointed under a floating charge created on or after 15 September 2003,⁸⁵⁹ except as in the cases specified thereunder.⁸⁶⁰ The cases in which an administrative receiver can be appointed to enforce a floating charge created on or after 15 September 2003 are:

- i. Where a debt of £50 million or more is financed by a marketable loan;
- ii. Where a person has step-in rights in relation to a project company (that is,

⁸⁵⁴CAMA, *op cit*, s. 393 & Schedule 11.

⁸⁵⁵Insolvency Act, op cit, ss. 47 – 48; CAMA, ibid, ss. 396 – 398.

⁸⁵⁶ Downsview Nominees Ltd v First City Corporation Ltd [1993] AC 295; Mayson, et al, op cit, p. 648. ⁸⁵⁷ 2002 (UK).

⁸⁵⁸ Insolvency Act, op cit, s. 176 A.

⁸⁵⁹Insolvency Act, *ibid*, s, 72A; Enterprise Act, *op cit*, s. 250.

⁸⁶⁰ Insolvency Act, *ibid*, s. 72B – H.

rights to take over responsibility for carrying out the project or arranging for it to be carried out)⁸⁶¹ and the project is a public-private-partnership project,⁸⁶² a utility project,⁸⁶³ an urban regeneration project⁸⁶⁴, or will involve £50 million or more of financing.⁸⁶⁵ However, a right to appoint an administrative receiver does not amount to step-in rights;⁸⁶⁶

- iii. To enforce charges which secure payment for purchases on recognised investment exchanges;⁸⁶⁷
- iv. Where the company is a registered social landlord;⁸⁶⁸ and,
- v. Where the company is a railway, water or air traffic control company.⁸⁶⁹

Suffice to state that while a security holder of assets secured by a floating charge created before 15 September 2003, retained the rights to appoint an administrative receiver and also the option to appoint an administrator, a chargee whose floating charge does not cover the whole or substantially the whole of the company's assets, can still appoint a receiver, who will, by definition, not be an administrative receiver and cannot also appoint an administrator. Thus, chargees of floating charges created on or after 15 September 2003, has the right to appoint an administrator.⁸⁷⁰

In a nutshell, the present nature and features of administrative receivership were described thus:

⁸⁶¹Insolvency Act, *op cit*, schedule 2 A, para 6.

⁸⁶²*Ibid*, s. 72C.

⁸⁶³ *Ibid*, s.72D.

⁸⁶⁴ *Ibid*, s. 72DA.

⁸⁶⁵ *Ibid*, s. 72E.

⁸⁶⁶ Feetum v Levy [2005] EWCA Civ 1601; [2006] 2 BCLC 102.

⁸⁶⁷ Insolvency Act, *op cit*, s. 72F.

⁸⁶⁸*Ibid*, s. 72G.

⁸⁶⁹ *Ibid*, s. 72 GA; L Sealy and S Worthington, *Cases and Materials in Company Law* (10th edn, Oxford: Oxford University Press, 2013) p. 780; P L Davies and S Worthington, *Gower and Davies' Principles of Modern Company Law* (9th edn, London: Sweet and Maxwell, 2012) p. 1252.

⁸⁷⁰ Insolvency Act, *ibid*, schedule B1, para 14 (1).

The aim stated in the White Paper preceding the Enterprise Act, 2002, was that "administrative receivership should cease to be a major insolvency procedure". The administrative receiver would instead be replaced by an administrator, who is an officer of the court, and must "perform his functions in the interests of the company's creditors as a whole.⁸⁷¹

It is the contention that the Enterprise Act reforms, largely doing away with administrative receivership, were intended to favour the rescue culture. An administrative receiver is entitled to put the secured creditor's interest first, and a suspicion prevailed that the interest of unsecured creditors and the company itself, were not always best served. By contrast, an administrator must manage the company's affairs for the benefit of everyone concerned.⁸⁷²

Notwithstanding the advantages of receiverships, it has some shortcomings. Significantly, receiverships as a product essentially of private law, is not clothed with moratorium during the period of receivership. Conversely, in administration, there is a moratorium on the enforcement of creditors' rights against the company. This means that so long as the administration is in force, it is not possible to commence winding up proceedings or any other process against the company or enforce any charge except with the consent of the administrator or leave of court. Hence, the concept of receivership seems to thrive better in developed capital market economies where there is market awareness and the judicial machinery is efficient and effective.⁸⁷³ In this respect, the provisions of sections 387 to 400 of the CAMA are

⁸⁷¹ Davies and Worthington, *op cit*, p 1251 – 1252.
⁸⁷² Sealy and Worthington, *op cit*, p. 780.

⁸⁷³ Abugu, *op cit*, p. 246.

adjudged to offer only a legal remedy, the availability of which is subject to thorny and frustrating legal arguments, in motions and counter motions and invariably resulting in delay of justice.⁸⁷⁴

⁸⁷⁴Abugu, *op cit*, p. 246.

CHAPTER FOUR

COMPARATIVE ANALYSIS OF CORPORATE RESCUE MECHANISMS AND PROCEDURES IN NIGERIA AND OTHER JURISDICTIONS

The words of the Supreme Court of Nigeria in *Air Via Ltd v Oriental Airlines Ltd*⁸⁷⁵ clearly capture the essence or rationale for corporate rescue mechanisms. In his usual lucid manner, Onu JSC stated thus:

Furthermore, care and utmost caution must be exercised by the institution of justice in proceedings involving the termination of the life of a company with responsibility not only to the society but also to the section of the public namely, its employees who may be thrown into economic hardship of unemployment. Hence the need for the court to halt a petition of the type brought by the appellant likely to cause irreversible and incalculable damage especially when the circumstances of the case show that the respondent has disputed the debt alleged to be owed. For the proposition that a court must be slow in terminating the life of a company by way of winding up, this is given credence and support by the Court of Appeal case of *Union Bank of Nigeria Ltd v Tropic Foods Ltd.*⁸⁷⁶

In this chapter, the discussion will focus on the available corporate rescue mechanisms and procedures in Nigeria and other jurisdictions such as the United

⁸⁷⁵ (2004) 4 SC (pt II) 37; P L Davies& S Worthington, *Gower and Davies Principles of Modern Company Law* (9thedn, London: Sweet and Maxwell, 2012) p. 1272.
⁸⁷⁶ Air Via Ltd v Oriental Airlines Ltd, [2004] 4 SC (pt II) 37.

Kingdom (UK), the Republic of South Africa (South Africa), the Republic of India (India) and the United States of America (US).

4.1 Nigeria

The available corporate rescue mechanism under the Companies and Allied Matters Act (CAMA) is arrangements and compromise. However, some have posited that merger or acquisition is also a common rescue option available to ailing companies and their shareholders.⁸⁷⁷ In this chapter, other mechanisms adopted in the rescue of failing companies, particularly in the banking sector, suffering from illiquidity, would be examined.

1. Arrangements and Compromise

Arrangement means "any change in the rights or liabilities of members, debenture holders or creditors of a company or any class of them or in the regulation of a company other than a change effected under any other provisions of this Act or by the unanimous agreement of all parties affected thereby".⁸⁷⁸Arrangements and Compromise are provided for under Part XVI of the CAMA. It has been stated that the Arrangements and Compromise mechanism in Nigeria stems from what is known in England and Wales as "Schemes of Arrangement".⁸⁷⁹ It is a corporate rescue mechanism by which a company may propose any scheme to its members, creditors or both, to restructure shareholding or arrive at a compromise on debts.⁸⁸⁰ Although the words, "arrangements" and "compromise" are often used interchangeably, they do

⁸⁷⁷ J E O Abugu, *Company Securities: Law and Practice* (2ndedn, Lagos: MIJ Professional Publishers Limited, 2014) p. 303; O O Oladele, 'The Legal Intricacies of Corporate Restructuring and Rescue in Nigeria', (2007) the Calabar Law Journal, XI, 136.

⁸⁷⁸ Companies and Allied Matters Act Cap C20 Laws of the Federation of Nigeria, 2000 (CAMA), ss. 537, 567 (1); Abugu, *ibid*, p. 304.

⁸⁷⁹ B Adebola, 'The Nigerian Business Rescue Model: An Introduction' (2014) NIALS Journal of Legal Studies, 9; also available at <<u>http://papers.ssrn.com/sol3/papers.cfm?abstract_id=2297824</u>.> accessed on 10 May 2015.

⁸⁸⁰ CAMA, *op cit*, s. 539 (1); Abugu, *op cit*, p. 305; Candide-Johnson & A Alex-Adedipe, 'Debt Recovery: Corporate Insolvency – Receivership, Winding up and other Arrangements' in O Olanipekun (ed), *Banking: Theory, Regulation, Law and Practice* (Lagos: Au Courant, 2016) p. 453.

not necessarily mean the same thing. While 'compromise' implies some element of accommodation on each side, it is not just a total surrender of rights,⁸⁸¹ 'arrangements' on the other hand, implies some elements of give and take. Hence, confiscation is not an arrangement.⁸⁸²

The procedural steps for arrangements and compromise are provided as follows:

- A compromise or arrangement is proposed between a company and its members or creditors or any class of them;⁸⁸³
- An application is made to the court in summary by the company or any of its creditors or members.⁸⁸⁴ The application to the court in summary

is by originating summons⁸⁸⁵ supported by an affidavit, praying for an order to summon a meeting of the creditors or class of creditors or of the members or class of members, as the case may be;⁸⁸⁶

- If the application is granted, the meeting is summoned pursuant to the order of the court;
- 4. The notice summoning the meeting which is sent to a member or creditor, must be accompanied by a statement explaining the effect of the compromise or arrangement⁸⁸⁷ and in particular stating any material interests of the directors of the company, whether as directors or as members or as creditors of the company or otherwise and the effect thereon of the compromise or arrangement in so far as it is different from

⁸⁸¹ H Y Bhadmus, *Corporate Insolvency Law and Practice in Nigeria* (Enugu: Chenglo Limited, 2009) p. 25.

^F⁸⁸² *Ibid.*

⁸⁸³ CAMA, *op cit*, s. 539 (1).

⁸⁸⁴ *Ibid*, s. 539 (1).

⁸⁸⁵ Companies Proceedings Rules, (CPR) 2004, r. 2.

⁸⁸⁶ CAMA, *op cit*, s. 539 (1).

⁸⁸⁷ *Ibid*, s. 540 (1) (a).

the effect on the like interest of other persons;⁸⁸⁸

- 5. At the meeting, the scheme can only be ratified by holders of at least three-quarters in value of the shares of members or class of members or of the interest of creditors or class of creditors, as the case may be, present and voting either in person or by proxy.⁸⁸⁹Consequently, the scheme of compromise or arrangement (the 'Scheme') will be referred to the court for sanction.
- 6. Prior to the sanction of the scheme of arrangement by the court, it must pass the fairness test.⁸⁹⁰ In order to satisfy the fairness test, the court may refer the Scheme to the Securities and Exchange Commission (SEC), which shall appoint one or more inspectors to investigate the fairness of the said compromise or arrangement and to make a written report thereon to the court within a time specified by the court.⁸⁹¹ It is pertinent to state that the Scheme in the case of a publicly quoted company is usually referred to the SEC.⁸⁹² In the case of companies other than a publicly quoted

company, the court examines the fairness of the scheme.⁸⁹³

7. Where the court is satisfied as to the fairness of the Scheme, it shall sanction same.⁸⁹⁴ Hence, making it binding,⁸⁹⁵ but subject to a certified true copy of the order of the court sanctioning the Scheme being delivered by

⁸⁸⁸CAMA, op cit, s. 539 (2) (3); Abugu, op cit, p. 307.

⁸⁸⁹*Ibid*, (2); *Oceanic International Bank Ltd v Victor Odili & Ors* (unreported) Suit No. FHC/L/CS/1361/2005; CA/L/171M/2008; Abugu, *op cit*, p. 305.

⁸⁹⁰CAMA, *ibid*, s. 539 (2) (3); Candide-Johnson & Alex-Adedipe, *op cit*, p. 453.

⁸⁹¹CAMA, *ibid*, s. 539 (2).

⁸⁹²Candide-Johnson & Alex-Adedipe, *loc cit*.

⁸⁹³CAMA, op cit, s. 539 (2) (3); Candide-Johnson & Alex-Adedipe, ibid

⁸⁹⁴CAMA, *ibid*, s. 539 (3); Candide-Johnson & Alex-Adedipe, *ibid*.

⁸⁹⁵CAMA, *ibid*.

the company to the Corporate Affairs Commission for registration.⁸⁹⁶

Schemes of arrangement and compromise under the CAMA are subject to the exercise of discretion of the court. Unfortunately, the CAMA did not provide for the factors to be considered by the court when looking at schemes of arrangement. This may account for the misconceived position of the Federal High Court in *Re: Interfirst Finance and Securities Limited*⁸⁹⁷ and *Andruche Investment Plc .v. Financial Mediators.*⁸⁹⁸ In the two matters decided by the same judge, the Federal High Court appeared to have taken a restrictive view of the notion of arrangements when it held that the proposals by which the companies in the matter set out the manner by which their respective debt would be repaid were not arrangements or compromise because there was no evidence of mutuality between the respective company and its creditors. It is pertinent to note that in the matters under reference, the application that was before the court was for an order to convene a meeting and not a consideration of the proposal or sanction of the scheme of arrangement.

In light of the foregoing, the arrangements and compromise mechanism under the CAMA had not passed unchallenged. Firstly, the procedure for enforcing arrangements and compromise in the CAMA has been held to be difficult and complex, hence the suggestion for a simpler procedure known as the Company Voluntary Arrangement (CVA) by which distressed companies could negotiate simple compromises or schemes of arrangements with their creditors as contained in the extant Insolvency legislation of the United Kingdom.⁸⁹⁹Secondly, arrangements

⁸⁹⁶*Ibid*, s. 539(4).

⁸⁹⁷(1993) FHCLR 421.

⁸⁹⁸(1994) FHCLR 51.

⁸⁹⁹ B Adebola, 'A Few Shades of Rescue: Towards an Understanding of the Corporate Rescue Concept in England and Wales' (2014) <u>http://papers.ssrn.com/sol3/papers.cfm?abstract_id=2524488</u> accessed on 10 May, 2015.

and compromise under the CAMA have been ineffectual because of the absence of moratorium.⁹⁰⁰ In the words of Adebola:-

In Nigeria, the absence of a stay on creditor's right exacerbates the impact of the complex judicial system on the already complex arrangement and compromise procedure.... Giving the structure of the court system, practitioners find that an arrangement and compromise can be derailed in various ways. Unsecured creditors may initiate parallel debt collection proceedings at a SHC after the scheme is proposed. Similarly, secured lenders may enforce their security in the absence of a moratorium. Furthermore, those with quasi security would also be outside of the control of the directors. As a result, the company's vital assets may be dismantled by recalcitrant creditors, intent on enforcing their individual rights at the detriment of other interest including those of the company.⁹⁰¹

Furthermore, it has been propounded that arrangements and compromise mechanism are not only cumbersome, costly, court intensive, bureaucratic but suffers

⁹⁰⁰ M B Halita, 'A Comparative Study of some Issues Relating to Corporate Insolvency Law in Nigeria and South Africa', unpublished LL.M Dissertation, University of Pretoria, May 2013, p. 62.
 ⁹⁰¹ B Adebola, 'Conflated Arrangements: A comment on the Company Voluntary Arrangements in the

Proposed Nigerian Insolvency Act', (2015) 3 NIBLeJ 2 also available at http://www.4,ntu.ac.uk/n/s/document_uploads/174811-pdf accessed on 20 November 2016.

from protracted delays.⁹⁰² Thus, there is a consensus of opinion that arrangement and compromise have not been a veritable corporate rescue option in Nigeria.⁹⁰³

2. Mergers or Acquisition

Merger or Acquisition is principally governed by the provisions of the Investment and Securities Act⁹⁰⁴(ISA) and regulated by the Securities and Exchange Commission Rules⁹⁰⁵ (SEC Rules). In relation to companies in the banking sector, the prior approval of the Governor of the Central Bank of Nigeria is mandatory before any merger or acquisition is consummated or announced⁹⁰⁶. Merger or Acquisition has been posited as a common rescue option available to failing or ailing companies and their shareholders⁹⁰⁷ and may help a company to remain an ongoing concern⁹⁰⁸. Although the terms 'Merger' and 'Acquisition' are often used interchangeably to mean the same thing and in a more common sense used in the twin form of 'Mergers and Acquisitions', they do not necessarily have the same meaning.⁹⁰⁹ While a 'merger' means any amalgamation of the undertakings or part of the undertakings of one or more companies and one or more bodies corporate,⁹¹⁰ 'acquisition' means where a person or group of persons buys most (if not all) of a company's ownership

⁹⁰²Halita, *op cit*; A Abdulai, 'Mergers, Acquisitions and Take-Overs in Nigeria: The Guiding Philosophy, Laws, Regulations and Practice' [1997] 2 MILBQ, 21-34; S Akinwunmi, 'Corporate Insolvency Law in Nigeria – A need for Reform', <u>http://www.akinwunmibusari.com</u> accessed on 17 July, 2015; This is because, the role of the courts in the process where a company opts to utilize the arrangement and compromise mechanism would necessarily have to consider the exorbitant professional fees, lack of expertise on the bench and delayed court hearing as a result of congested cause list.

⁹⁰³S U Mba, 'Recession Preventive Rehabilitation and the Nigerian Draft Insolvency Bill: Lessons for the EU?' LAWYER, a This Day Weekly Pull-out 06-09-2016, p. 7.

⁹⁰⁴2007.

⁹⁰⁵2013 as amended 2015.

⁹⁰⁶Banks and Other Financial Institutions Act (BOFIA) Cap B3, Laws of the Federation of Nigeria 2004, s. 7; F Ajogwu, 'Mergers, Takeovers and Reorganisation of Banks in Nigeria' in Olanipekun (ed), *op cit*, pp. 574 – 580.

⁹⁰⁷SEC Rules, *op cit*, r. 426 (1) (f); Abugu, *op cit*, p. 303.

⁹⁰⁸Oladele, *art cit*, 136.

 ⁹⁰⁹F Ajogwu, Mergers & Acquisitions in Nigeria: Law & Practice (2ndedn, Lagos: Centre for Commercial Law Development (CCLD), 2014) p. 5.
 ⁹¹⁰ISA, op cit, s. 119 (1).

stake in order to assume control of a or the target company.⁹¹¹ The SEC Rules have gone a long way in stating the distinguishing features of a merger and an acquisition, hence, clearing the nebulous interpretations of the terms.⁹¹² Furthermore, there appears to be a misunderstanding between 'acquisition' and 'take over'. Thus, 'takeovers' and 'acquisitions' are in some instances assumed to be same and treated as same restructuring scheme. This is probably because acquisition was not treated as a distinct restructuring scheme in the ISA but as a variant of merger⁹¹³ or take-over.⁹¹⁴ This is no more the case as acquisition is now regulated⁹¹⁵ differently and take over is also distinctly regulated.⁹¹⁶

Although, mergers or acquisition are rarely employed in the rescue of ailing companies, it had played a prominent role in the restructuring of companies,⁹¹⁷ particularly in the banking sector to meet up with the recapitalisation and consolidation policy announced by the Governor of the Central Bank of Nigeria in 2004 and 2005⁹¹⁸ to address the then state of illiquidity and mitigate the impact of the global financial crisis. However, the benefits of merger or acquisition transactions are not in doubt. Some of the benefits include the pursuit of growth strategy, risk diversification, economies of scale, efficiency in management, maximising shareholders income, reduction in competition, corporate leverage and corporate

⁹¹¹SEC Rules, *op cit*, r 433.

⁹¹²F Ajogwu, 'Mergers & Acquisitions: Opportunities and Pitfalls', Unib Law Journal, VOL 1, NO.1, October 2011, 5 – 10; Abugu, *op cit*, pp. 301 – 303.

⁹¹³ISA, op cit, s. 119 (2); Ajogwu, ibid, 10.

⁹¹⁴ISA, *ibid*, ss. 131 – 151; Ajogwu, *ibid*, 9.

⁹¹⁵SEC Rules, *op cit*, rr. 433 - 444.

⁹¹⁶*Ibid*, rr. 445 – 448 as amended 2015.

⁹¹⁷N C S Ogbuanya, *Essentials of Corporate Law Practice in Nigeria* (2ndedn, Lagos: Novena Publishers Ltd, 2014) pp. 622 – 624.

⁹¹⁸F Ajogwu, 'Mergers, Takeovers and Reorganisation of Banks in Nigeria' in Olanipekun (ed), *op cit*, pp. 574 – 580; Ajogwu, *art cit*,7 – 8.

survival.⁹¹⁹ It is important to state that the benefits of mergers or acquisition may not be circumscribed only to the foregoing benefits.

The merger combinations can be commonly classified as horizontal,⁹²⁰ vertical⁹²¹ and conglomerate.⁹²² In this study, we would examine the classifications of mergers as provided in the investment and Securities Act. The classifications of mergers in the ISA are by reference to thresholds of combined annual turnover and of assets: "a small merger",⁹²³ "an intermediate merger",⁹²⁴ and "a large merger",⁹²⁵ The thresholds provided for in the ISA⁹²⁶ stipulated lower threshold of N500, 000, 000. 000. and upper threshold of N5, 000, 000, 000. 00. Pursuant to the powers conferred on it in the ISA, the SEC has increased the lower thresholds to N1, 000, 000, 000. 00. and the upper threshold to above N5, 000, 000, 000. 00.⁹²⁷ Simply put, a small merger means a merger or proposed merger with a value below N1, 000, 000, 000. 00. of either combined assets or turnover of the merging companies,⁹²⁸ an intermediate merger means a merger or proposed merger with a value between N1, 000, 000, 000. 000, 000. 000, 000. 000, 000. 000, 000. 000, 000. 000, 000. 000, 000. 000, 000. 000, 000. 000, 000. 000, 000. 000, 000. 000, 000. 000, 000. 000, 000. 000, 000. 000, 000, 000. 000, 000. 000, 000. 000, 000. 000, 000. 000, 000. 000, 000. 000, 000. 000, 000. 000, 000. 000, 000. 000, 000. 000, 000. 000, 000. 000, 000. 000, 000. 000, 000. 000, 000, 000. 000, 000, 000. 000, 000, 000. 000, 000, 000. 000, 000, 000. 000, 000, 000. 000, 000, 000. 000, 000, 000, 000. 000, 000, 000. 000, 000, 000. 000, 000, 000. 000, 000, 000, 000. 000, 000. 000, 000, 000, 000. 000, 000, 000, 000. 000, 000, 000, 000, 000. 000, 000, 000, 000, 000, 000. 000, 000, 000, 000. 000, 00

- ⁹²²SEC Rules, *ibid* means other types of mergers; Abugu, *ibid*; Ogbuanya, *ibid*; Bhadmus, *ibid*.
- ⁹²³ISA, op cit, s. 120 (2) (a); Bhadmus, op cit, p. 561.

⁹¹⁹Abugu, *op cit*, pp. 302 – 303; Ogbuanya, *op cit*, pp. 616 – 617; Ajogwu, *op cit*, 10 – 15.

⁹²⁰SEC Rules, *op cit*, r 421 (1) - mergers involving direct competitors; Abugu, *ibid*, p. 302; Ogbuanya, *ibid*, p. 617 – 618; H Y Bhadmus, *BHADMUS on Corporate Law Practice* (2ndedn, Enugu: Chenglo Limited, 2013) p. 561.

⁹²¹SEC Rules, *ibid*- mergers involving firms in non-competitive relationship; Abugu, *ibid*; Ogbuanya, *ibid*; Bhadmus, *ibid*.

⁹²⁴ISA, *ibid*, s. 120 (2) (b).

⁹²⁵*Ibid*, s. 120 (2) (c).

⁹²⁶*Ibid*, s. 120 (4).

⁹²⁷SEC Rules, *op cit*, r. 427 (1).

⁹²⁸*Ibid*.

companies.⁹³⁰ It is important to state that the SEC is conferred with the powers to prescribe or vary the thresholds from time to time.⁹³¹

It is instructive to note that while a small merger may not require prior review, notification and formal approval of the SEC,⁹³² it shall inform the SEC at the conclusion of the merger,⁹³³ Furthermore, the parties to a small merger may voluntarily⁹³⁴ inform the SEC of the merger or where the SEC having regard to the provisions of section 121 of the ISA, is of the view that the merger may substantially prevent or lessen competition or cannot be justified on public interest grounds.⁹³⁵ These provisions appear difficult to enforce as it is unclear how the SEC would become aware of the merger before its conclusion if it was not notified.⁹³⁶ In the case of an intermediate merger and a large merger, they are subject to prior review notification and approval of the SEC and the sanction of the merger by the court.⁹³⁷

It is pertinent to state that the SEC's approval of a merger is not granted as a matter of course. In consideration of a merger, the SEC determines whether or not the merger is likely to substantially prevent or lessen competition by assessing the strength of competition in the relevant market and the probability that the company, after the merger, will behave competitively or cooperatively, taking into account any factor that is relevant to competition in that market.⁹³⁸ Furthermore, if the merger is likely to substantially prevent or lessen competition, the SEC must determine

⁹³⁸ISA, *ibid*, s. 121 (1) (a), (2); Abugu, *op cit*, p. 309.

⁹³⁰SEC Rules, op cit, r. 427 (1).

⁹³¹*Ibid*, s. 120 (1).

⁹³²*Ibid*, s. 122 (1); SEC Rules, *op cit*, rr. 422 (4), 423 (1), 424 (1) b.

⁹³³SEC Rules, *ibid*.

⁹³⁴ISA, op cit, s. 122 (2); Abugu, op cit, p. 308.

⁹³⁵ ISA, *ibid*, s. 122 (3). This requirement can only be invoked by the SEC, within 6 months after a small merger has commenced implementation. In other words, it is not a pre-implementation step.
⁹³⁶Ajogwu, *op cit*, p. 23.

⁹³⁷ISA, *op cit*, ss. 123 (1) (3), 124, 125 (1), 126; SEC Rules, *op cit*, r. 423 (1); SEC Rules (as amended 2015), r. 425 (1); N Dimgba, 'The Regulation of Competition through Merger Control: The Case under the Investment and Securities Act, 2007' A paper delivered at the Nigerian Bar Association Section on Business Law Conference on 16th April, 2009, p. 6.

whether, notwithstanding, the merger is likely to result in any technological efficiency or pro-competition gain which will be greater than or offset the effects of any prevention or lessening of competition that may result from the merger.⁹³⁹ In determining whether the merger can or cannot be justified on substantial public interest grounds⁹⁴⁰ the SEC will have to consider the effect of the merger on a particular industrial section or region; employment; the ability of small businesses to become competitive and the ability of National industries to compete in international markets.⁹⁴¹

The SEC has 20 working days after all parties to a merger have fulfilled all their notification requirements to consider all factors and make a determination approving the merger or approving the merger subject to any conditions or prohibiting the implementation of the merger.⁹⁴² Though, the SEC may extend consideration for a single period not exceeding 40 working days.⁹⁴³ If the SEC is unable to communicate its approval or otherwise within the prescribed time limit, the merger shall be deemed to have been approved.⁹⁴⁴ It should be noted that the SEC may revoke its own decision to approve a merger.⁹⁴⁵ In the case of a large merger, the SEC shall upon receipt of notification of same, refer the notice to the court and within 40 working days after all parties to the large merger have fulfilled all the notification requirements, forward to the court, a statement as to whether or not implementation of the merger is approved subject to any conditions or prohibited.⁹⁴⁶ This provision

⁹³⁹ ISA, op cit, s. 121 (1) (b); Abugu, op cit.

⁹⁴⁰ISA, *ibid*, s. 121 (1) (b) (ii), (c).

⁹⁴¹*Ibid*, s. 121 (3).

⁹⁴²Ibid, ss. 122 (5) (a small merger), 125 (1) (an intermediate merger).

⁹⁴³*Ibid*, ss. 122 (5) (a) (a small merger), 125 (2) (an intermediate merger).

⁹⁴⁴*Ibid*, s. 125 (3).

⁹⁴⁵*Ibid*, s. 127.

⁹⁴⁶*Ibid*, s. 126.

has been adjudged to be impracticable until the parties themselves apply to court for court ordered meetings and the sanction of the court.⁹⁴⁷

It is instructive to note that the procedure for approval of a merger involves a lot of documentation,⁹⁴⁸ it is cost intensive⁹⁴⁹ and entails much of bureaucratic process. In view of the essence of merger as an available option of corporate restructuring or rescue option in Nigeria, in this study, it is deemed appropriate to give a surmise of the procedural steps for approval of a merger, to wit:

- 1. **Pre-Merger Procedure**
- i. Preparation of the draft scheme document between the merging companies;
- Consideration and approval in principle of the draft scheme document by ii. the boards of directors of the merging companies;⁹⁵⁰
- File with the SEC a merger notification and a draft scheme for iii. Evaluation.⁹⁵¹ The requirements for merger notification are provided in the SEC Rules;⁹⁵²
- Upon notification of approval in principle in respect of the merger iv. notification and a draft scheme for evaluation, the merging companies file an application in the Federal High Court for an order to convene a court ordered meeting of the members of the merging companies;⁹⁵³

⁹⁴⁷Abugu, *op cit*, p. 310.

⁹⁴⁸SEC Rules (as amended), *op cit*, rr. 425, 426 (2) (4), 428; SEC Rules, *op cit*, r. 426 (1).

⁹⁴⁹SEC Rules (as amended), *ibid*, r. 426 (2), Appendix VII (c) - prescribed fees for (i) Public Companies - value of shares issued by the resultant company, calculated thus: 1st N500 million -0.3%; next N500 million - 0.225%, any sum thereafter - 0.15%; (ii) Private Companies - total consideration of the transaction calculated in same way as Public Companies; Companies Regulations 2012, reg. 53 (1) (h); SEC Rules, *ibid*, r. 426 (1) (ix) (g) – merger notification fee of N50, 000. 00. per merging company for intermediate and larger mergers. Furthermore, it is usual in practice to pay solicitors' fees, capital market operators' fees, financial advisers' fees, etc. ⁹⁵⁰SEC Rules, *op cit*, r. 426 (1).

⁹⁵¹SEC Rules (as amended 2015), op cit, r. 425 (1) (a).

⁹⁵²SEC Rules, *op cit*, r. 426 (1); SEC Rules (as amended), *ibid*, r. 426 (2) (4).

⁹⁵³SEC Rules (as amended), *ibid*, r. 425 (1) (b).

- v. The merging companies issue notice of the court ordered meeting to members and publish same in two national dailies and a copy filed with the SEC.⁹⁵⁴
- vi. The merging companies hold separate meetings and pass special resolution approving the scheme.⁹⁵⁵ The special resolution shall be filled with the Corporate Affairs Commission within 15 days of their passing.⁹⁵⁶
- 2. Procedure for Formal Approval
- Upon receipt of favourable response to a merger notification from the SEC, a formal application for approval of the merger is filed accompanied with relevant documents and compliance with requirements for formal approval of the merger;⁹⁵⁷
- ii. Approval or otherwise of the scheme document by the SEC;
- Upon approval of the merger the SEC shall inform the court by a statement in writing if the merger is approved wholly or subject to conditions or is prohibited;⁹⁵⁸
- iv. The court sanctions the scheme.⁹⁵⁹
- 3. Post approval procedure

After the approval given by the SEC and the court order sanctioning the scheme, there are certain requirements to be complied with by the merging companies, namely, (i) the merging companies shall cause a copy of the order sanctioning the

⁹⁵⁴SEC Rules (as amended), *op cit*,r. 425 (1) (c).

⁹⁵⁵Companies Regulation, *op cit*, reg 53 (1) (a).

⁹⁵⁶*Ibid*, reg 53 (2).

⁹⁵⁷SEC Rules, op cit, r. 429; SEC Rules (as amended), op cit, r. 428.

⁹⁵⁸SEC Rules, *ibid*, r. 428 (5).

⁹⁵⁹*Ojora v Agip Nigeria Plc* [2014] 1 N.W.L.R (pt 1387) 151. In the instant case, the court interpreted the provisions of section 100 of the ISA 1999, which applies *mutatis mutandis* to sections 121 and 122 of the ISA 2007.

scheme to be delivered to the SEC for registration within 7 days⁹⁶⁰ and filed with the Corporate Affairs Commission (CAC) within 15 days⁹⁶¹ of the making of the Order; (ii) a Notice of the Order shall be published in the official gazette of the Federation and at least one newspaper.⁹⁶²

4. Post-merger inspection

It is a statutory requirement that 3months after approval by the SEC, a post-merger inspection would be carried out by the SEC to ascertain the level of compliance with the provisions of the scheme documents. Procedurally, the documents⁹⁶³ to be inspected include;

- i. the board minute book;
- ii. original certificate of incorporation of the resultant company (where applicable);
- iii. copy of the amended Memorandum and Articles of Association (where applicable);
- iv. severance benefits of employees of the dissolved companies;
- v. final settlement of shareholders;
- vi. dispatch of share certificates;
- vii. settlement of debt;
- viii. report of shareholders' representatives on the merger;
- ix. any other document that may be required by the SEC from time to time,

It is important to state that upon a merger, the emerging company takes over the assets, rights, duties and liabilities of the scheme parties. However, it should be noted that the merger companies are not technically dead as to lose their legal personalities

⁹⁶⁰SEC Rules, *op cit*, r. 430 (1).

⁹⁶¹Companies Regulations, *op cit*, reg 53 (3).

⁹⁶²*Ibid*, reg 53 (1) (d); SEC Rules, *op cit*, r. 430 (2).

⁹⁶³SEC Rules, *ibid*, r. 431.

but become transformed into the new or emerging company.⁹⁶⁴ Although, merger have been proposed as an available corporate rescue option in Nigeria, it is seldom used in the rescue of failing or ailing companies because of the complexity and high transaction cost.

However, the SEC Rules⁹⁶⁵ provided for a separate procedure for acquisition of companies. Thus, acquisition is now treated as a distinct transaction. The requirements for acquisition is that an acquirer shall file a letter of intent⁹⁶⁶ accompanied with relevant documents, mainly, two draft copies of Information Memorandum⁹⁶⁷ and extracts of board resolutions of the acquirer and the acquiree agreeing to the acquisition. The filing of the letter of intent is to be filed by a registered capital market operator registered to function as an issuing house.⁹⁶⁸ Furthermore, the letter of intent has to be accompanied by evidence of payment of application fee of N50, 000. 00⁹⁶⁹ and processing fee based on the value to be acquisition in at least 2 national dailies after consummation is mandatory.⁹⁷¹ Further, the treatment of dissenting shareholders in acquisition is fully taken care of.⁹⁷²

Although, the express powers of the SEC to approve or reject an acquisition was inadvertently omitted in the SEC Rules, the unequivocal implications of the power conferred on the SEC to regulate acquisitions⁹⁷³ in both private and public quoted companies and to conduct a post-acquisition inspection 3 months after the approval of

⁹⁶⁸ *Ibid*, r. 434 (b).

⁹⁶⁴Commercial Bank (Credit Lyonnais) Nig. Ltd v Okoli [2009] 5 N.W.L.R (pt 1135) 446.

⁹⁶⁵SEC Rules, *op cit*, rr. 433 – 439.

⁹⁶⁶*Ibid*, r. 434 (b).

⁹⁶⁷*Ibid*, rr. 434 (b) (i); 436 (Contents of the Information Memorandum).

⁹⁶⁹*Ibid*, r. 434 (b) (xiii).

⁹⁷⁰*Ibid*, r. 434 (b) (xiv).

⁹⁷¹*Ibid*, rr. 434 (b) (xviii), 435 (Contents of Publication).

⁹⁷²ISA, *op cit*, ss. 146, 147; SEC Rules, *ibid*, r. 438; Abugu, *op cit*, pp. 320 – 326.

⁹⁷³SEC Rules, *ibid*, r. 434.

the application⁹⁷⁴ can only suggest that acquisition transactions are subject to approval or otherwise of SEC. Hence, one of the contents of the Information Memorandum includes the effects of the acquisition on the relevant industry.⁹⁷⁵

In light of the foregoing, acquisition procedure appears not to be a rescue option readily adopted to rescue failing or ailing companies in Nigeria. This may not be unconnected to the documentations and cost implication involved in acquisition transactions. It is not far-fetched that the expected outcome of an acquisition is the purchase of the target company and where this is the case, there is nothing left for the target company. In this vein, acquisition is not illustrative of quintessential corporate rescue option.

3. Bailouts

This was a functional mechanism that was used to stem the effects of the global financial crisis in 2007 - 2009. At first, the general impression was that Nigerian financial institutions or banks were sound and insulated from the 2007 – 2009 global financial crisis. This general impression was found to be wrong and misplaced as Nigerian financial institutions or banks were enmeshed in crisis. While some pointed to the global financial crisis as triggering the Nigerian banking crisis, a good number of analysts pointed to domestic events such as lax regulations, bank fraud, excessive margin lending and non-performing unhedged loans to oil importers and other credit malpractices.⁹⁷⁶

In order to forestall a deepening of the crisis and to mitigate and prevent the near collapse of banks, the Central Bank of Nigeria (CBN) introduced a variety of measures aimed at improving confidence in the banking sector, enforcing discipline

⁹⁷⁴SEC Rules, op cit, r. 439.

⁹⁷⁵*Ibid*, r. 436 (5).

⁹⁷⁶I Agabi & A Onayemi, 'Troubled Assets Resolution' in Olanipekun (ed), *op cit*, p. 466; K Ekwueme, 'Failure Resolution' in Olanipekun (ed), *ibid*, pp. 494 – 496.

in banking management and most of all, improving liquidity through the establishment of a special purpose vehicle, the Asset Management Corporation of Nigeria (AMCON).⁹⁷⁷ The basis for the establishment of AMCON was to help resolve the Non-Performing Loan (NPL)⁹⁷⁸ assets of banks and to recapitalise banks. In addition, to improve bank liquidity, the CBN injected N620 Billion of liquidity into the banking sector in the form of unsecured subordinated debt and guaranteed inter-bank deposits for over a two year period.⁹⁷⁹ The CBN's power to inject liquidity into a bank having liquidity issues was judicially approved by the Federal High Court in Izedonmwen v Union Bank of Nigeria Plc.⁹⁸⁰This is what is commonly known as 'bailouts' in corporate rescue lexicon in Nigeria. This has informed the postulation that there is an insolvency regime for banks different from general corporate insolvency.981

Notwithstanding, the opposition to the injection of the bailout funds and the establishment of the AMCON salvaging the financial system, particularly failing banks, have been acknowledged as a cheaper and more veritable means to rescue failing banks.⁹⁸²However, it has been posited that AMCON is not a special purpose vehicle strictu sensu as it does not have an identity totally distinct from that of its

⁹⁷⁷Asset Management Corporation of Nigeria Act, 2010, s. 1 (1). It should be noted that as a means of addressing the Asian financial crisis, the countries most affected by the crisis, Indonesia, South Korea, Malaysia and Thailand, respectively set up the Indonesia Bank Restructuring Agency (IBRA), Korea Asset Management Corporation (KAMCO), Danaharta Malaysia, and the Thai Asset Management Corporation (TAMC). Further, Asset Management Company has been used in a variety of jurisdictions such as France, Sweden, Mexico and the United States of America.

⁹⁷⁸Agabi & Onayemi, *op cit*, p. 470. NPL is defined in line with CBN Prudential Guidelines as loans for which (i) interest or principal is due and unpaid for 90 days or more; (ii) interest payments equal to 90 days' interest or more, have been capitalised, rescheduled, or rolled over into a new loan (except in cases where they have been reclassified as specified by the CBN). ⁹⁷⁹Agabi & Onayemi, *op cit*, p. 469; Ekwueme, *op cit*, p. 502; K Aina, 'Asset Management

Corporation of Nigeria - A Pretentious Special Purpose Vehicle'

<u>http://papers.ssrn.com/sol3/papers.cfm?abstract_id=2617500</u> accessed on 20 June 2015. ⁹⁸⁰Unreported Suit No. FHC/B/CS/209/2009, per Idris J (as he then was); *Gbajabiamila v CBN* (Unreported Suit No. FHC/L/CS/1125/2009), per Buba J in Ekwueme, *op cit*, pp. 503, 505. ⁹⁸¹Ekwueme, *op cit*, p. 491.

⁹⁸² Agabi & Onayemi, op cit, p. 472.

promoters and sponsors, lacks independence and without a definite life span unlike other asset management companies.⁹⁸³

The resort to bailout appears to be the trend in addressing insolvent situations in Nigeria. Hence, the allure of bailouts has found reception not only in the rescue of failing companies but corporate entities such as State Governments⁹⁸⁴ and have also necessitated the clamour of companies in the aviation industry for bailouts from the Federal Government of Nigeria to save them from collapse and distress.⁹⁸⁵

4. Regulatory Rescue

There are rescue procedures in various Statutes in Nigeria empowering regulatory bodies to address insolvency of some corporate entities towards rescuing such failing corporate entities.⁹⁸⁶ Such rescue procedures are provided for failing banks,⁹⁸⁷ for failing insured institutions,⁹⁸⁸ for failing insurance companies⁹⁸⁹ and for a failing capital market operator.⁹⁹⁰ The circumstances upon which the regulatory institutions can come to the rescue of the failing corporate entity when such a corporate entity:

- 1. is likely to become unable to meet its obligations under the enabling
 - Act,⁹⁹¹or
- is about to suspend payment to any extent,⁹⁹² or 2.
- is insolvent,⁹⁹³and/or 3.

⁹⁸³Aina, art cit; Agabi & Onayemi, op cit, p. 473.

⁹⁸⁴It is of public knowledge that about 28 States in Nigeria were able to pay backlog of salaries owed its workers from bailouts given to them by the Federal Government.

⁹⁸⁵ Messy Bailout Proposal for Aviation Industry' <<u>http://www.punchng.com/editorials/messy-bailout-</u> <u>proposal-for-aviation-industry</u>> accessed on 17 July 2015. ⁹⁸⁶Bhadmus, *op cit*, pp. 4 – 19.

⁹⁸⁷BOFIA, *op cit*, s. 35.

⁹⁸⁸Nigeria Deposit Insurance Corporation (NDIC) Act, 2006, ss. 36 – 39.

⁹⁸⁹H Y Bhadmus, Corporate Insolvency Law and Practice in Nigeria (Enugu: Chenglo Limited, 2009) pp. 11 – 17. ⁹⁹⁰ISA, *op cit*, ss. 48 – 52.

⁹⁹¹BOFIA, *op cit*, s. 35 (1) (a).

⁹⁹²*Ibid*, s. 35 (1) (b).

⁹⁹³*Ibid*, s. 35 (1) (c).

4. is in grave situation.⁹⁹⁴

However, it is worthwhile to point out that despite the statutory provisions for the regulatory rescue of failing corporate entities, such measures appear not to be effective. In the case of failing banks, it is not undeniable that none of the failed banks in Nigeria had survived without being liquidated.

5. Management Buy-Out

It refers to the acquisition by a management team of a company of controlling shares of that company or its subsidiaries, with or without third party financing.⁹⁹⁵ Management team usually refers to the Directors and other officers of the Company.

The Procedural Steps for Management buy-out includes an Application for the approval for management buy-out, filed by the management team, accompanied by the following:⁹⁹⁶

- 1. Resolution of the shareholders approving the Management Buy-out
- 2. Resolution of the management team to under-take the management buyout
- 3. A copy of Certificate of incorporation of the Company
- 4. A copy of Memorandum and article of association of the Company
- 5. Two copies of Prospectus, containing
- a. A profile of the Company
- b. Profile of management team buying over the Company
- c. Objectives of the management buy-out
- d. 5 years audited financial Statement of the Company or if less than 5 years,

the statement of affairs for the number of years in existence and

⁹⁹⁴BOFIA, op cit, s. 35 (1) (d).

⁹⁹⁵ Securities and Exchange Commission Rules 2013 ("SEC Rules") as amended, r 449(a)
⁹⁹⁶SEC Rules, *op cit*, r.449(a); NCS Ogbuanya, *Essentials of Corporate Law Practice in Nigeria* (2nd edn, Lagos: Novena Publishers Ltd (Best Practice Books Series), 2014) pp.613-614; H Y Bhadmus, *Bhadmus on Corporate Law Practice* (3rd edn, Enugu: Chenglo Limited, 2015) p.535

e. Claims and litigation

6 Sale Agreement between the Company and the management team containing (a) terms and conditions of sale and (b) Indemnity against contingent liabilities by the seller to 3^{rd} parties.

4.2. The United Kingdom

There is a plethora of corporate rescue mechanisms in the United Kingdom, formal and informal corporate rescue procedures. The available formal corporate rescue procedures are discussed below.

1. Schemes of Arrangement

A scheme of arrangement is a flexible and versatile mechanism of corporate rescue. A scheme of arrangement can also come in different forms and shades. Hence, it is impossible to identify a typical scheme of arrangement. In this part of our study, Schemes of Arrangement as contained in Part 26 of the extant Companies Act⁹⁹⁷ in the United Kingdom would be examined.

2. Arrangement and Reconstruction

The advantage of a scheme of arrangement in the form of arrangements and reconstructions is "a complicated alteration of rights or of capital structure to be implemented by a simple court order approving the scheme. There is no elaborate follow up action to be taken and in some cases, stamp duty is avoided since there are no instruments of transfer requiring to be stamped.⁹⁹⁸ However, there is a disadvantage of using a scheme of arrangement in the sense that it requires the preparation of a document setting out the scheme and of an explanatory circular, with two applications to the court at the start and at the end of the procedural sequence.

⁹⁹⁷2006 (United Kingdom)

⁹⁹⁸S Barber, Company Law (4thedn, London: Old Bailey Press, 2004) p. 303.

For that reason, a scheme of arrangement is likely to be uneconomic unless large companies, with millions of Pounds in assets, are involved.⁹⁹⁹

It is noteworthy that the mechanism of schemes of arrangements in the United Kingdom is almost on all fours with arrangements and compromise as contained in the CAMA. Section 895 (1) of the Companies Act, provides for compromise or arrangement between the company and its members or creditors. Section 896 of the Companies Act, provides for the convening and holding of meeting pursuant to the application to the court. Section 897 of the Companies Act, provides for every notice summoning the meeting that is sent to a creditor or member must be accompanied by a statement explaining the effect of the court may sanction the compromise or arrangement if a majority in number, representing 75% (that is three-quarters) in value of the creditors or class of creditors or members or class of members (as the case may be) present and voting either in person or by proxy, agree to a compromise or arrangement, which sanction by the court is binding on all creditors or the class of creditors or members or class of members (as the case may be) and the company, subject to the court's order having been delivered to the registrar of companies.

However, in *Re: Alabama, New Orleans, Texas v Pacific Function Railway* $Co.,^{1000}$ the court held that it should exercise its discretion in deciding whether to approve a scheme of arrangement as follows:

⁹⁹⁹S Barber, *op cit*, p. 303.

¹⁰⁰⁰[1891] 1 Ch 213.

The court must look at the scheme, and see whether the Act has been complied with, whether the majority are acting bona fide, and whether they are coercing the minority in order to promote interests adverse to the class whom they purport to represent; and then see whether the scheme is a reasonable one or whether there is any reasonable objection to it, or such objection that a reasonable man might say that he could not approve it.

Furthermore, the court in exercise of its discretion in deciding whether to approve a scheme of arrangement considers whether it is within the ambit of statutory provisions and whether the initial statutory requirements, in respect of documents, have been complied with.¹⁰⁰¹Arrangement and Reconstruction as a form of schemes of arrangement in the United Kingdom has also faced the same challenges as arrangements and compromise under the CAMA, such as being costly, complex, and time consuming.¹⁰⁰²

But the significant limitation in the use of schemes of arrangement mechanism in corporate rescue is that, just like receivership, there is no moratorium whatsoever.

3. Company Voluntary Arrangements (CVAs)

This procedure was added to the English Company Law at the behest of the Cork Committee and presently is contained in Part 1 of the Insolvency Act¹⁰⁰³ as a simple procedure whereby a company which is in financial difficulties may enter into a

¹⁰⁰¹*Re: Hellenic and General Trust Ltd* [1975] 3 All ER 382; *Re: Savoy Hotel Ltd* [1981] Ch 351; *Re: Telewest Communications Plc.* (*No.1*) [2005] 1 B.C.L.C 752.

¹⁰⁰² Bourne, *op cit*, p. 269; B Adebola, 'Conflated Arrangements: A comment on the Company Voluntary Arrangements in the Proposed Nigerian Insolvency Act', *op cit*; L Sealy & S Worthington, *Cases and Materials in Company Law* (10thedn, Oxford: Oxford University Press, 2013) p. 708. ¹⁰⁰³1986 as amended,ss.1-7

voluntary arrangement with its creditors. This arrangement may involve either: a composition in satisfaction of its debts, that is, provision for creditors to receive a percentage of what is due to them or a scheme of arrangement of its affairs.¹⁰⁰⁴

The procedure for CVAs is segmented into chapters:

- Chapter 2 of the Insolvency Rules applies when the proposal for the voluntary arrangement is made by the directors of the company and the company is either in liquidation or in administration, and no steps have been taken to obtain a moratorium under Schedule A1 to the Act in connection with the proposal.¹⁰⁰⁵
- 2. Chapter 3 of the Insolvency Rules applies when the proposal for the voluntary arrangement is made by either the liquidator (if the company is in liquidation) or the administrator (if the company is in administration) and in either case, the liquidator or the administrator is the nominee.
- Chapter 4 of the Insolvency Rules applies in the same case as Chapter 3 but where the nominee is not the liquidator or administrator.
- Chapter 5 of the Insolvency Rules applies in all the cases mentioned in Chapters 2, 3 and 4 above.
- 5. Chapter 6 was revoked by Insolvency (Amendment) (No.2) Rules, 2002.
- 6. Chapters 7 and 8 of the Insolvency Rules apply to all voluntary arrangements with or without a moratorium.
- 7. Chapter 9 of the Insolvency Rules applies where the proposal is made by the directors of an eligible company with a view to obtaining a moratorium.
 The glaring feature of CVAs procedure is that one should be informed of the chapter to apply for a proposal. In other words, a CVA may be proposed through the

¹⁰⁰⁴Bourne, *op cit*, p. 251.

¹⁰⁰⁵Insolvency Rules, op cit, r. 1.1 (2).

stand-alone procedure (proposal is made by directors of the company) or as an element of a main or parent procedure (proposal is made by liquidator or administrator in the case of liquidation or administration respectively).¹⁰⁰⁶ Where it is a main or parent procedure that is utilized, the CVAs benefits from the moratorium that accompanies the main or parent procedure.¹⁰⁰⁷ Unlike schemes of arrangement, a stand-alone CVA can obtain a moratorium,¹⁰⁰⁸ but only to companies that qualify as small by law.¹⁰⁰⁹ The procedure for obtaining a moratorium in stand-alone CVAs is stipulated in Chapter 9 of the Insolvency Rule.

The mechanisms of a CVA, whether by stand-alone procedure or as an element of a main or parent procedure is that:

- 1. A nominee is appointed for the purpose of supervising its implementation and must be a qualified insolvency practitioner.¹⁰¹⁰
- A proposal is prepared by the directors¹⁰¹¹ or liquidator¹⁰¹² or 2.

administrator¹⁰¹³or the nominee.¹⁰¹⁴ The proposal should contain the matters

as specified in the Insolvency Rules.¹⁰¹⁵

3. Written notice of the proposal is delivered to the nominee (in the case where the liquidator or administrator is not the nominee),¹⁰¹⁶ accompanied with a statement of the company's affairs (in the case where the liquidator or administrator is not the nominee),¹⁰¹⁷ which must be verified by a

¹⁰⁰⁶Adebola, 'Conflated Arrangements...', op cit.

¹⁰⁰⁷Insolvency Act, *op cit*, ss. 112, 130 (2), schedule B1, paras 42 - 43.

¹⁰⁰⁸*Ibid*, s. 1Å, schedule A1; the Insolvency Act, 2000.

¹⁰⁰⁹*Ibid*, s. 1A, schedule A1, para 3 (2); Companies Act, *op cit*, s. 382 – companies qualifying as small. ¹⁰¹⁰Insolvency Act, *ibid*, s. 1(2).

¹⁰¹¹*Ibid*, s. (1); Insolvency Rule, *op cit*, r. 1.1 (2) (a).

¹⁰¹²Insolvency Act, *ibid*, s. 1 (3) (b); Insolvency Rule, *ibid*, r. 1.1 (2) (b).

¹⁰¹³Insolvency Act, *ibid* s. 1 (3) (a); Insolvency Rule, *ibid*.

¹⁰¹⁴Insolvency Rule, *ibid*, r. 1.1 (2) (c).

¹⁰¹⁵*Ibid*, r. 1.3 – proposal by directors; r. 1.10 – proposal by liquidator or administrator and r. 1.12 – proposals by the nominee. ¹⁰¹⁶*Ibid*, r. 1.4.

statement of truth made by at least one director.¹⁰¹⁸ The statement of affairs shall be made up to a date not earlier than 2 weeks before the date of the notice to the nominee under Rule 1.4.¹⁰¹⁹

- 4. The nominee makes a report from the proposal and statement of the company's affairs submitted to him (in the case where the liquidator or administrator is not the nominee), which report he submits to the court stating whether, in his opinion, meetings of the company and of its creditors should be summoned to consider the proposal, and if in his opinion such meeting should be summoned, the date on which and place at which they shall be held.¹⁰²⁰
- 5. A meeting is summoned.¹⁰²¹ Unlike with schemes of arrangement, the court's role is only administrative.¹⁰²² It is not required to grant an order to convene the meeting, but where the nominee is the liquidator or administrator, he shall summon meetings of the company and of its creditors to consider the proposal without submitting his report to the court.¹⁰²³
- 6. The meeting summoned shall decide whether to approve the proposed CVAs (with or without modifications).¹⁰²⁴ Section 4 (3) and (4) of the Insolvency Act provides for instances when a meeting so summoned shall not approve a proposal except with the concurrence of the creditor or the preferential creditor concerned.

¹⁰¹⁷*Ibid*, r. 1.5.

¹⁰¹⁸Insolvency Rule, op cit, r. 1.5 (4).

¹⁰¹⁹*Ibid*, r. 1.5 (3).

¹⁰²⁰Insolvency Act, op cit, s. 2; Insolvency Rule, op cit, r. 1.7.

¹⁰²¹Insolvency Act, *ibid*, s. 3 (1); Insolvency Rule, *ibid*, chapter 5 – proceedings at meetings.

¹⁰²²Sealy & Worthington, *op cit*, p. 769.

¹⁰²³Insolvency Act, *op cit*, s. 3 (2); Insolvency Rule, *op cit*, r. 1.11.

¹⁰²⁴Insolvency Act, *ibid*, s. 4 (1).

- 7. A proposal or modification thereto is approved when a majority of threequarters or more "in value" of those present and voting in person or by proxy has voted in favour of it.¹⁰²⁵
- 8. If the CVAs are approved, the nominee becomes the supervisor of the CVAs.¹⁰²⁶ It is important to know that any person that is dissatisfied by any act, omission or decision of the supervisor may apply to the court for necessary directions.¹⁰²⁷

The CVAs has been applauded for certain innovations contained therein. It has been argued that since an insolvency practitioner is required to bring the proposals before the meetings for approval and to supervise their implementation if approved, an insolvency practitioner must be found to accept responsibility for the scheme. Hence, if the proposals were deceptive or unlikely to succeed, an insolvency practitioner would unlikely act in connection with the CVAs.¹⁰²⁸ Also, the CVAs can be turned into an administration for the purposes of obtaining a moratorium.¹⁰²⁹

The CVAs procedure has been argued to be defective not only because of the non-inclusion of the power to cram-down the rights of a dissenting minority 1030 but the lack of a moratorium for the stand-alone procedure, which means that despite the quite lengthy formalities, one impatient creditor could thwart the whole process. Although a way round the issue of moratorium was found by the introduction of CVAs with a moratorium,¹⁰³¹ this was achieved only at the cost of making the whole procedure more elaborate, public and expensive. In addition, the option is only

¹⁰²⁵Insolvency Rule, *op cit*, r. 1.19 (2). ¹⁰²⁶Insolvency Act, *op cit*, s. 7(1)(2).

¹⁰²⁷*Ibid*, s. 7(3).

¹⁰²⁸Bourne, *op cit*, p. 315.

¹⁰²⁹Insolvency Act, op cit, s. 8(3)(b).

¹⁰³⁰A CVA cannot be made binding on secured or preferential creditors without their consent by virtue of section 175 and schedule 6 to the Insolvency Act. 1031 Insolvency Act, *op cit*, schedule A1.

available to 'eligible companies' (primarily small private companies). The procedure allows for an automatic 28day moratorium to come into force as soon as the documents containing a proposal for a CVA are filed in court. During this period, which may be extended for a further 2 months, the company may not be wound up and no steps may be taken, at least without the leave of court, to enforce security over the company's property or to take proceedings against it. The success of the much sought after provision remains unclear.

However, the advantages of the moratorium may be outweighed by the relative complexity and cost in terms of the companies which are eligible, the role of the supervisor, and the restrictions on directors during the moratorium and the possible liabilities of the directors and the supervisor for actions taken during the moratorium. Furthermore, even the advantages of the moratorium provision, pale in significance now that the Enterprise Act permits the appointment of administrators out of court coupled with their own associated moratorium.¹⁰³²

The newly introduced moratorium provisions in CVAs are unlike administration, which comes with ubiquitous moratorium in the sense that the administrator also has the power to avoid the meeting of creditors.¹⁰³³

4. Administration

Following the report of the Review Committee on Insolvency Law and Practice, ¹⁰³⁴a recommendation for the establishment of a procedure to facilitate the rehabilitation or reorganisation of a company led to the incorporation of administration orders in the Insolvency Act of 1985, which in turn consolidated in the Insolvency Act 1986. In essence, an administration order is to make possible the rescue of a company by

¹⁰³²Sealy & Worthington, *op cit*, pp. 769 – 770.

 ¹⁰³³Adebola, 'Conflated Arrangements...', op cit.
 ¹⁰³⁴The CORK Report (CMMD 8558, 1982).

placing its management in the hands of an administrator. For as long as the administration is in force, it is not possible to commence winding up proceedings or any other process against the company or to enforce any charge, hire purchase or retention of title provision against the company without the leave of court.¹⁰³⁵

However, the Enterprise Act made substantial amendments¹⁰³⁶ which introduced fundamental changes in the purpose, appointment and powers of administrators, although the procedure remains open only to companies that are or are likely to become insolvent.¹⁰³⁷ Administration has now largely replaced receiverships, at least for floating charge holders, and has the merit of being conducted in the interest of all concerned rather than only the secured creditor. The process is associated with a moratorium,¹⁰³⁸ so it provides the company with breathing space, free from creditors' claims, where the administrator can determine the future of the company even where liquidation is inevitable, so that its assets can be realised to better advantage by selling the business as a going concern.¹⁰³⁹

1. Purpose of Administration

The purposes¹⁰⁴⁰ for which an administration order may be made are:

- 1. Rescuing the company as a going concern;
- 2. Achieving a better result for the company's creditors as a whole than would be likely if the company were wound up;
- Realising property in order to make a distribution to one or more secured or preferential creditors;

¹⁰³⁵Insolvency Act, *op cit*, s. 10; Bourne, *op cit*, p. 253.

¹⁰³⁶Insolvency Act, *op cit*, s. 8, schedule B1. The 'old' provisions on administration contained in Part 2 of the Insolvency Act, are retained and applied to special categories of companies (water companies, railway companies, air traffic services companies and public-private partnership companies). ¹⁰³⁷V. Final C. Companies and P. Statishing and P. Statishing and S. Statishing an

¹⁰³⁷V Finch, *Corporate Insolvency Law: Perspective and Principles* (Cambridge: Cambridge University Press, 2002) p. 392.

¹⁰³⁸Insolvency Act, *op cit*, schedule B1, para 43.

¹⁰³⁹Sealy& Worthington, *op cit*, p. 771.

¹⁰⁴⁰Insolvency Act, *op cit*, schedule B1, para 3 (1).

It should be noted that the administrator must always act in the interest of the creditors as a whole¹⁰⁴¹ and as quickly and efficiently as reasonably practicable.¹⁰⁴² Hence, administrators must pursue rescuing the company as a going concern rather than pursuing the objective of achieving a better result for the company's creditors as a whole than would be likely if the company was wound up unless it is not reasonably practicable to pursue rescuing the company as a going concern.

2. **Procedure for Appointment of Administrator**

An administrator may be appointed by the court on the application of the company itself or its directors or one or more creditors¹⁰⁴³ of the company or the designated officer for a Magistrate's Court in the exercise of the power conferred by Section 87A of the Magistrate's Courts' Act, 1980 (fine imposed on company) or a combination of the persons mentioned herein before.¹⁰⁴⁴ Where there is a difference in view as to the choice of administrator, all things being equal, the choice of the largest creditor (in terms of the value of the debt owed to it by the company) is regarded as being the tie breaker.¹⁰⁴⁵ However, there are certain general restrictions as to persons that may be appointed as administrator of a company.¹⁰⁴⁶

The appointment of an administrator out of court can be made by the holder of a floating charge relating to the whole or substantially the whole of the company's property.¹⁰⁴⁷ For appointment by a floating charge holder, the floating charge must simply be enforceable. However, an out of court appointment must be reported to the court. The preconditions for appointment of an administrator whether by the court or

¹⁰⁴¹Sealy & Worthington, op cit, pp. 770 – 771; Insolvency Act, *ibid*, schedule B1, para 3 (2).

¹⁰⁴²Sealy & Worthington, *ibid*; Insolvency Act, *ibid*, schedule B1, para 4.

¹⁰⁴³Insolvency Act, op cit, schedule B1, para 12 (4) - "Creditor" includes a contingent creditor and a prospective creditor. ¹⁰⁴⁴*Ibid*, s. 9, schedule B1, paras 2, 12, 14 and 22.

¹⁰⁴⁵Healthcare Management Services Ltd v Caremark Properties Ltd[2012] EWHC 1693.

¹⁰⁴⁶Insolvency Act, *op cit*, schedule B1, paras 6, 7,8, 9.

¹⁰⁴⁷*Ibid*, schedule B1, para 14.

out of court, is that the company must be unable to pay its debts or is likely to become unable to do so.¹⁰⁴⁸

The process of appointment of an administrator by a holder of a floating charge is contained in Schedule B1 to the Insolvency Act as it relates to power to appoint,¹⁰⁴⁹ restrictions on power to appoint,¹⁰⁵⁰ filing with the court a notice of appointment¹⁰⁵¹ (which must include a statutory declaration by and on behalf of the person who makes the appointment, the identification of the administrator and a statement of consent by the administrator), commencement of appointment¹⁰⁵² and entering of indemnity for invalid appointment.¹⁰⁵³

In the case of an appointment of an administrator by a company or its directors, the process is contained in Schedule B1 as it relates to power to appoint,¹⁰⁵⁴ restrictions on power to appoint,¹⁰⁵⁵ notice of intention to appoint,¹⁰⁵⁶ notice of appointment,¹⁰⁵⁷ commencement of appointment¹⁰⁵⁸ and indemnity for invalid appointment.¹⁰⁵⁹

For an administration order to be made by the court, the court must be satisfied that the company 'is or is likely to become' unable to pay its debts and the administration order is reasonably likely to achieve the purpose of administration.¹⁰⁶⁰

¹⁰⁴⁸Insolvency Act, *op cit*, schedule B1, para 11 (a).

¹⁰⁴⁹*Ibid*, schedule B1, para 14.

¹⁰⁵⁰*Ibid*, schedule B1, paras 15, 16, 17.

¹⁰⁵¹*Ibid*, schedule B1, para 18.

¹⁰⁵²*Ibid*, schedule B1, paras19, 20.

¹⁰⁵³*Ibid*, schedule B1, para 21.

¹⁰⁵⁴*Ibid*, schedule B1, para 22.

¹⁰⁵⁵*Ibid*, schedule B1, paras 23, 24, 25.

¹⁰⁵⁶*Ibid*, schedule B1, paras 26, 27 – Five business days' written notice is given to any person who is or may be entitled to appoint an administrative receiver or an administrator of the company or such other persons as may be prescribed. A copy of the notice of intention to appoint and any document accompanying it shall be filed with the court as soon as is reasonably practicable. Accompanying the notice of intention is a statutory declaration in accordance with Para. 27 (2) (3) and the identification of the proposed administrator.

¹⁰⁵⁷*Îbid*, schedule B1, para 29.

¹⁰⁵⁸*Ibid*, schedule B1, paras 31, 32, 33.

¹⁰⁵⁹*Ibid*, schedule B1, para 34.

¹⁰⁶⁰*Ibid*, schedule B1, para 11 (b).

In *Re: Harris Simons Construction Ltd*,¹⁰⁶¹ the court interpreted the provision of Section 8 (1) (b) of the Insolvency Act, 'would be likely to achieve' to mean a 'real prospect'. Vinelott, J., in considering whether to grant an order in respect of a company trading in Nigeria, said: "the question must always be, if there is a real prospect that one or more of the stated purposes would be achieved, is that prospect sufficiently likely in the light of all the other circumstances of the case to justify the making of the order.¹⁰⁶² In *Re: AA Mutual International Insurance Co. Ltd*,¹⁰⁶³ the court interpreted 'is or is likely to become' under Schedule B1, Para 11 (a) to the Insolvency Act to mean 'more probably than not' while 'is reasonably likely' under Schedule B1, Para 11 (b) to means 'real prospect'.

It is instructive to note that the precondition for an administration order, which is predicated on the company's inability to pay its debt, is determined by the definition of inability to pay debt as contained in Section 123 of the Insolvency Act.

3. Powers and Duties of Administrator

An administrator is an officer of the court (whether or not he is appointed by the court),¹⁰⁶⁴ and must be a qualified insolvency practitioner.¹⁰⁶⁵

An administrator is given exceptionally wide powers of management to do what is necessary for the smooth management of the affairs, business and property¹⁰⁶⁶ of the company. These powers include:

 Taking possession of the property, selling and otherwise disposing of it, raising or borrowing money, appointing a solicitor or accountant, bringing or defending legal proceedings, effecting and maintaining insurances,

¹⁰⁶¹[1989] 1WLR 368.

¹⁰⁶²*Re: Primlaks UK Ltd*[1989] BCLC 734.

¹⁰⁶³[2004] EWHC 2430.

¹⁰⁶⁴Insolvency Act, *op cit*, schedule B1, para 5.

¹⁰⁶⁵*Ibid*, schedule B1, para 6.

¹⁰⁶⁶*Ibid*, schedule B1, para 1 (1).

appointing agents, carrying on the business of the company, establishing subsidiaries, granting or accepting surrender of a lease or tenancy and the power to do all such things that are incidental to the powers set out in Schedule 1 of the Insolvency Act.¹⁰⁶⁷

 To dispose of property subject to a floating charge without the leave of court and for property subject to any other charge with the leave of court.¹⁰⁶⁸

However, the proceeds from such disposition of the property must be deployed according to the statutory priorities.¹⁰⁶⁹

3. To call any meeting of members or creditors of the company.¹⁰⁷⁰

Flowing from the wide powers given to the administrator of a company, he owes certain duties such as:

- To make a statement setting out the proposals for achieving the purpose of administration;¹⁰⁷¹
- 2. To request by notice to the relevant persons in the prescribed form, a statement of the affairs of the company, which must be verified by a statement of truth in accordance with Civil Procedure Rules;¹⁰⁷²
- 3. To summon creditors meetings;¹⁰⁷³
- 4. To take custody or control of all the property to which he thinks the company is entitled;¹⁰⁷⁴
- 5. To take reasonable steps to obtain a proper price for its assets if in the

¹⁰⁶⁷Insolvency Act, *op cit*, schedule 1; B1, paras 59, 60.

¹⁰⁶⁸*Ibid*, schedule B1, paras 70, 71.

¹⁰⁶⁹*Ibid*, s. 15 (4).

¹⁰⁷⁰*Ibid*, s. 14 (2) (b).

¹⁰⁷¹*Ibid*, schedule B1, para 49.

¹⁰⁷²*Ibid*, schedule B1, para 47.

¹⁰⁷³*Ibid*, schedule B1, paras 50, 51, 56.

¹⁰⁷⁴*Ibid*, schedule B1, para 67.

course of a sale;¹⁰⁷⁵

 To make distribution of the assets of a company as the circumstances determine.¹⁰⁷⁶

4. Effect of Administration Order

The appointment of an Administrator engenders certain consequences, to wit:

- The company in administration or an officer of the company in administration ceases to exercise a management power except with the consent of the administrator;¹⁰⁷⁷
- 2. Dismissal of pending winding up petition.¹⁰⁷⁸

The dismissal shall be upon the making of an administration order but a petition for the winding up shall be suspended while the company is in administration. However, such suspension does not apply to a petition presented under Section 124A (public interest) or Section 124B (SEs) of the Insolvency Act and Section 367 of the Financial Services and Market Act, 2000 (petition by Financial Services Authority);

- 3. Dismissal of administrative or other receiverships;¹⁰⁷⁹
- 4. Moratorium on insolvency proceedings.¹⁰⁸⁰

However, there are exceptions as provided in Paragraph 42 (4) of Schedule B1 to the Insolvency Act. By virtue of the provisions of paragraph 44 of Schedule B1 to the Insolvency Act, an interim moratorium will avail in the circumstances as provided in the said paragraph.

5. Moratorium on other legal process.¹⁰⁸¹

¹⁰⁷⁵Re: Charnley Davies Ltd (No. 2)[1990] BCLC 760.

¹⁰⁷⁶Insolvency Act, *op cit*, schedule B1, paras 65, 66.

¹⁰⁷⁷*Ibid*, schedule B1, para 64.

¹⁰⁷⁸*Ibid*, schedule B1, para 40.

¹⁰⁷⁹*Ibid*, schedule B1, para 41.

¹⁰⁸⁰*Ibid*, schedule B1, para 42.

This is to the effect that no step including the enforcement of security over the company's property;¹⁰⁸² or to repossess goods in the company's possession under a hire purchase agreement;¹⁰⁸³ or to exercise a landlord's right of forfeiture by peaceable re-entry in relation to premises let to the company¹⁰⁸⁴ and institution or continuation of legal proceedings, execution, distress and diligence against the company or property of the company, except with the consent of the administrator or with the permission of the court.¹⁰⁸⁵

6. Every business document, whether an invoice, order for goods or services, a business letter and an order form, issued by or on behalf of the company or the administrator and all the company's websites must have the name of the administrator and statement that the affairs, business and property of the company are being managed by the administrator.¹⁰⁸⁶

5. Termination of Administration

Administration ends at the happening of any of the following events:

- The appointment of an administrator ends automatically after one year beginning with the date on which it takes effect¹⁰⁸⁷ unless the term is extended¹⁰⁸⁸ by the court or by consent;¹⁰⁸⁹
- 2. Pursuant to an application by the administrator to the court.¹⁰⁹⁰ The grounds upon which an administrator can make an application to the court

¹⁰⁸¹Insolvency Act, *op cit*, schedule B1, para 43.

¹⁰⁸²*Ibid*, schedule B1, para 43 (2).

¹⁰⁸³*Ibid*, schedule B1, para 43 (3).

¹⁰⁸⁴*Ibid*, schedule B1, para 43 (4).

¹⁰⁸⁵*Ibid*, schedule B1, para 43 (6).

¹⁰⁸⁶ *Ibid*, schedule B1, para 45 (1) (3).

¹⁰⁸⁷*Ibid*, schedule B1, para 76 (1).

¹⁰⁸⁸*Ibid*, schedule B1, para 76 (2).

¹⁰⁸⁹*Ibid*, schedule B1, paras 76 (2), 78.

¹⁰⁹⁰*Ibid*, schedule B1, para 79.

to end the administration of a company is provided in Paragraph 79 (2) and (3) of Schedule B1 to the Insolvency Act;

- 3. Where the purpose of administration has been sufficiently achieved in relation to the company;¹⁰⁹¹
- 4. Pursuant to an application to the court by a creditor¹⁰⁹² on the grounds as provided in Paragraph 81 (2) of Schedule B1 to the Insolvency Act, and,
- 5. Public interest winding up.¹⁰⁹³

4.3 The Republic of South Africa

The available corporate rescue mechanisms in the Republic of South Africa are the business rescue proceedings and an arrangement or a compromise.

1. Business Rescue Proceedings

Business rescue proceedings is a home grown or an indigenous procedure adopted in the Republic of South Africa to facilitate the rehabilitation of a company that is financially distressed.¹⁰⁹⁴ The procedure governing business rescue proceedings is provided in the extant law.¹⁰⁹⁵

2. Purpose of Business Rescue Proceedings

The primary purpose for business rescue proceedings is to facilitate the rehabilitation of a company that is financially distressed.¹⁰⁹⁶ A company is financially distressed¹⁰⁹⁷when:-

1. It appears to be reasonably unlikely to pay all of its debt as they become

due and payable within the immediately ensuing six months;¹⁰⁹⁸ or

¹⁰⁹¹Insolvency Act, *op cit*, schedule B1, para 80.

¹⁰⁹²*Ibid*, schedule B1, para 81 (1).

¹⁰⁹³*Ibid*, schedule B1, para 82.

¹⁰⁹⁴Companies Act 2008, chapter 6 (South Africa) (CA).

¹⁰⁹⁵*Ibid*; Companies Amendment Act 2011, Part 6 (South Africa) (CAA).; E Levenstein, 'An Appraisal of the New South African Business Rescue Procedure', unpublished LLD Thesis, University of Pretoria, Pretoria, November 2015, pp 1-20

¹⁰⁹⁶CA, *op cit*, s 128 (1)(b).

¹⁰⁹⁷*Ibid*, s. 128 (1) (f).

 It appears to be reasonably likely to become insolvent within the Immediately ensuing six months¹⁰⁹⁹

Essentially, the purpose of business rescue proceedings is to put the financially distressed company into a temporary supervision.¹¹⁰⁰ This is achieved by imposing temporary moratorium on claims against the company¹¹⁰¹ and the development and implementation of a business rescue plan.¹¹⁰²

3. Modes of Commencement of Business Rescue Proceedings

The modes of commencement of business rescue proceedings are by company resolution and court order pursuant to the application of an affected person.¹¹⁰³

4. Business Rescue Proceedings: Company Resolution

The board of a company may resolve that the company voluntarily begin business rescue proceedings and place the company under supervision if the board has reasonable grounds to believe that the company is financially distressed and there appears a reasonable prospect of rescuing the company.¹¹⁰⁴It should be noted that a resolution to begin business rescue proceedings will not be adopted if liquidation proceedings has been initiated by or against the company and a resolution will have no effect until it has been filed.¹¹⁰⁵ Where a resolution has been adopted, the company shall within five business days after the adoption or such longer time as the Companies and Intellectual Property Commission¹¹⁰⁶ may permit, publish a notice of the resolution and its effective date in the prescribed manner to every affected person

¹⁰⁹⁸CA, *op cit*, s 128 (1)(f)(i); CAA, s 81 (a).

¹⁰⁹⁹CA, *ibid*, s. 128 (1) (f) (ii).

¹¹⁰⁰CA, *ibid*, s. 128 (1) (b)(i).

¹¹⁰¹*Ibid*, s. 128 (1) (b) (ii).

¹¹⁰²*Ibid*, s. 128 (1) (b) (iii)(c).

¹¹⁰³*Ibid*, s. 128 (1) (a)Affected person mean a shareholder or creditor of the company; any registered trade union representing employees of the company and if any employee of the company are not represented by a registered trade union, each of those employees or their respective representatives. ¹¹⁰⁴*Ibid*, s. 129 (1).

¹¹⁰⁵*Ibid*, s. 129 (2).

¹¹⁰⁶*Ibid*, ss. 1 & 185 (1).

accompanied with a sworn statement of the facts stating the grounds on which the board resolution was founded¹¹⁰⁷ and appoint a qualified business rescue practitioner, who has consented in writing to accept the appointment.¹¹⁰⁸

Upon the appointment of a business rescue practitioner, the company must file a notice of the appointment of the business rescue practitioner within two days after making the appointment¹¹⁰⁹ and publish a copy of the notice of appointment to each affected person within five business days after the notice was filed.¹¹¹⁰However, where the company fails to file the resolution upon adoption or publish a notice of the resolution to every affected person or appoint a business rescue practitioner or publish a copy of the notice of appointment of the business rescue practitioner or publish a copy of the notice of appointment of the business rescue practitioner to every affected person, the resolution to begin business rescue proceedings and place the company under supervision, lapses and it becomes a nullity.¹¹¹¹ Furthermore, if by reason of the foregoing, the resolution lapses or is a nullity, the company will not file a further resolution for a period of three months after the date on which the lapsed resolution was adopted, unless a court on good cause shown on an *ex parte* application, approves the company filing a further resolution.¹¹¹²

Where the board of a company has reasonable grounds to believe that the company is financially distressed but failed to adopt a resolution to begin business rescue proceedings, the board must deliver a written notice to each affected person, setting out the reasons for not adopting the resolution.¹¹¹³Notwithstanding, an affected person can object to company's resolution to begin business rescue proceedings by

¹¹⁰⁹*Ibid*, s. 129 (4) (a).

¹¹¹²*Ibid*, s. 129 (5) (b).

¹¹⁰⁷CA, *op cit*, s. 129 (3) (a).

¹¹⁰⁸*Ibid*, s. 129 (3) (b).

¹¹¹⁰ *Ibid*, s. 129 (4) (b).

¹¹¹¹*Ibid*, s. 129 (5) (a).

¹¹¹³*Ibid*, s. 129 (7) as amended by CAA, *op cit*, s. 82.

applying to the court¹¹¹⁴ for an order¹¹¹⁵ setting aside the resolution on the grounds that there is no reasonable ground for believing that the company is financially distressed¹¹¹⁶ or there is no reasonable prospect for rescuing the company¹¹¹⁷ or the company failed to satisfy procedural requirements for making a resolution.¹¹¹⁸ The order of court may be for setting aside of the appointment of business rescue practitioner on the ground that he is not qualified to act as a business rescue practitioner¹¹¹⁹ or is not independent of the company or its management¹¹²⁰ or lack the necessary skills required, having regards to the circumstances of the company.¹¹²¹

It is pertinent to state that the grant of an order setting aside the resolution for business rescue proceedings or the appointment of a business rescue proceedings is not granted as a matter of course. There are grounds to be established before the court can grant the order setting aside the resolution or the appointment of a business rescue practitioner.¹¹²² Where the court grants an order setting aside the resolution or the appointment of a practitioner, it may make any further necessary and appropriate order, including placing the company under liquidation¹¹²³ or an order of costs against any director who voted in favour of the resolution *mala fide*¹¹²⁴ or appointment of an alternate practitioner.¹¹²⁵

5. Business Rescue Proceedings: Court Order

Unless a company has adopted a resolution to begin business rescue proceedings, an affected person may apply to a court at any time for an order placing the company

¹¹¹⁴CA, *op cit*, s. 128 (1) (e). ¹¹¹⁵*Ibid*, s. 130 (1). ¹¹¹⁶*Ibid*, s. 130(1) (a) (i). ¹¹¹⁷*Ibid*, s. 130 (1) (a) (ii). ¹¹¹⁸*Ibid*, s. 130 (1) (a) (iii). ¹¹¹⁹*Ibid*, s. 130 (1) (b) (i). ¹¹²⁰*ibid*, s. 130 (1) (b) (ii). ¹¹²¹*Ibid*, s. 130 (1) (b) (iii). ¹¹²²*Ibid*, s. 130 (5) (a) (b). ¹¹²³*Ibid*, s. 130 (5) (c) (i). ¹¹²⁴*Ibid*, s. 130 (5) (c) (ii). ¹¹²⁴*Ibid*, s. 130 (6) (a).

under supervision and commencing business rescue proceedings.¹¹²⁶ In this respect, an applicant to the court must serve a copy of the application on the company, the Companies and Intellectual Property Commission and notify each affected person of the application in the prescribed manner.¹¹²⁷

The application by an affected person for an order of court placing the company under supervision and commencing business rescue proceedings is not granted as a matter of course. The application is granted where the court is satisfied that:¹¹²⁸

- i. The company is financially distressed;
- The company has failed to pay over any amount in terms of an obligation under or in terms of a public regulation or contract with respect to employment related matters; or
- It is just and equitable to do so for financial reasons and there is a reasonable prospect for rescuing the company.

Where the court pursuant to the application makes an order placing the company under supervision and commencing business rescue proceedings, it will make a further order appointing an interim practitioner nominated by the affected person, subject to ratification of the majority of the first meeting of creditors.¹¹²⁹ Furthermore, it is instructive to state that an application suspends any liquidation proceedings until the court has adjudicated upon the application.¹¹³⁰

6. Duration of Business Rescue Proceedings

Business rescue proceedings by company resolution begin when the company files a resolution to place itself under supervision or applies to the court for consent to file a

¹¹²⁶CA, *op cit*, s. 131 (1).

¹¹²⁷*Ibid*, s. 131 (2) (a) (b).

 $^{^{1128}}$ *Ibid*, s. 131 (4) (a).

 $^{^{1129}}Ibid$, s. 131 (5).

¹¹³⁰*Ibid*, s. 131 (6) (a).

further resolution.¹¹³¹ In the case of business rescue proceedings by court order, it commences when an affected person applies to the court for an order placing the company under supervision¹¹³² or a court makes an order placing a company under supervision during the course of liquidation proceedings or proceedings to enforce a security interest.¹¹³³ Thus, business rescue proceedings end when (i) the court sets aside the resolution or order that begun the proceedings¹¹³⁴ or has converted the proceedings to liquidation proceedings (ii) the practitioner files with the Companies and Intellectual Property Commission a notice of the termination of the business rescue proceedings¹¹³⁵ (iii) a business rescue plan is proposed and rejected without any affected person applying to extend the proceedings¹¹³⁶ or the business rescue plan is adopted but the practitioner has subsequently filed a notice of substantial implementation of that plan.¹¹³⁷

It should be noted that where a company's business rescue proceedings did not end within three months after the commencement of the proceedings or such longer time as the court on application by the practitioner may allow, the practitioner must prepare a report on the progress on the business rescue proceedings and update same at the end of each subsequent month until the end of those proceedings¹¹³⁸ deliver the report and each update in the prescribed manner to each affected person¹¹³⁹ and to the court if the proceedings have been the subject of a court order¹¹⁴⁰ or the Companies and Intellectual Property Commission , in any other case.¹¹⁴¹

¹¹³⁸*Ibid*, s.132(3)(a).

¹¹³¹CA, *op cit*, s. 132 (1) (a) (i) (ii).

¹¹³²*Ibid*, s. 132 (1) (b) as amended by CAA, *op cit*, s. 83 (a).

¹¹³³CA, *ibid*, s. 132 (1) (c) as amended by CAA, *op cit*, s. 83 (b).

¹¹³⁴CA, *ibid*, s. 132 (2) (a) (i).

¹¹³⁵*Ibid*, ss. 132 (2) (b), 153 (5).

¹¹³⁶*Ibid*, ss. 132 (2) (c) (i), 153 (5).

¹¹³⁷*Ibid*, s. 132 (2) (c) (ii).

¹¹³⁹*Ibid*, s. 132(3)(b).

¹¹⁴⁰*Ibid*, s 132(3)(b)(1).

¹¹⁴¹*Ibid*, s. 132(3)(b)(ii).

7. Effects of Business Rescue Proceedings

1. Legal proceedings

No legal proceeding, including enforcement action, against the company or in relation to any property belonging to the company or lawfully in its possession may be commenced or preceded, except¹¹⁴²:-

- a. With the written consent of the practitioner;¹¹⁴³
- b. With the leave of court;¹¹⁴⁴
- c. As a set off against any claim made by the company in any legal proceedings;¹¹⁴⁵
- Criminal proceedings against the company or any of the directors or Officers;¹¹⁴⁶
- e. Proceedings concerning any property or right over which the company exercises the powers of a trustee;¹¹⁴⁷
- f. Proceedings by a regulatory authority in the execution of its duties after
 Written notification to the business rescue practitioner;¹¹⁴⁸

Furthermore, during business rescue proceedings, any right to commence proceedings or otherwise, assert a claim against a company that is subject to time limit is suspended.¹¹⁴⁹

2. Shareholders and directors

There shall not be an alteration to the classification or status of any issued securities of a company other than by way of a transfer of securities in the ordinary

¹¹⁴²CA, *op cit*, s. 133(1).

¹¹⁴³*Ibid*,s.133(1)(a).

¹¹⁴⁴*Ibid*, s.133(1)(b)

¹¹⁴⁵*Ibid*, s. 133 (1) (c) as amended by CAA, *op cit*, s. 84(a).

¹¹⁴⁶CA, *ibid*, s. 133(1) (d) as amended by CAA, *ibid*, s. 84 (b).

¹¹⁴⁷CA, *ibid*, s. 133(1) (e) as amended by CAA, *ibid*, s. 84 (b).

¹¹⁴⁸CAA, *ibid*, s. 84(c).

¹¹⁴⁹CA, *op cit*, s. 133(3).

course of business except as the court otherwise directs or as contemplated in the approved business rescue plan.¹¹⁵⁰

A director of the company continues to exercise the functions of a director, subject to the authority of the business rescue practitioner.¹¹⁵¹ Furthermore, each director of the company has a duty to the company to exercise any management function within the company in accordance with the express instructions or direction of the practitioner.¹¹⁵² It is pertinent to state that the director of company continues to function as director but subject to instructions, direction and authority of the business rescue practitioner, who is empowered to apply to a court for an order removing a director from office on the ground of incompetence or obstruction of the business rescue proceedings and the development or implementation of a business rescue plan.¹¹⁵³

3. Company's Property

The company retains title to its property during a business rescue proceedings.¹¹⁵⁴ However, it may dispose or agree to dispose of its property only, in the ordinary course of its business¹¹⁵⁵ or in a *bona fide* transactions at arm's length for fair value approved in advance and in writing by the business rescue practitioner ¹¹⁵⁶ or in a transaction contemplated within and undertaken as part of the implementation of an approved business rescue plan.¹¹⁵⁷

¹¹⁵⁰CA, *op cit*, s. 137(1) (a) (b).

¹¹⁵¹*Ibid*, s. 137(2)(a).

¹¹⁵²*Ibid*, ss. 137 (2) (b), 140 (1) (b).

¹¹⁵³*Ibid*, s. 137(5).

¹¹⁵⁴*Ibid*, s. 134(1); CAA, *op cit*, s. 85(a).

¹¹⁵⁵CA, *ibid*, s. 134(1)(a)(i).

 $^{^{1156}}Ibid$, s. 134(1)(a)(ii).

¹¹⁵⁷*Ibid*, s.134(1)(a)(iii).

4. Employees and Contracts

The employees of the company immediately before the beginning of business rescue proceedings continue to be employed on the same terms and conditions, except to the extent that changes occur in the ordinary course of attrition or the employees and the company in accordance with applicable labour laws agree to different terms and conditions.¹¹⁵⁸

It should be noted that where the business rescue plan provides for the retrenchment of employees, it is subject to the relevant provision of Labour Relations Act¹¹⁵⁹ and other applicable employment related legislation.¹¹⁶⁰ However, subject to sections 35A and 35B of the Insolvency Act,¹¹⁶¹ the business rescue practitioner may entirely, partially or conditionally suspend for the duration of the business rescue proceedings, any obligation of the company that arises under an agreement to which the company was a party at the commencement of the business rescue proceedings.¹¹⁶² Nevertheless, a business rescue practitioner cannot suspend any provision of an employment contract ¹¹⁶³ or an agreement to which sections 35A or 35B of the Insolvency Act would have applied had the company been liquidated.¹¹⁶⁴

6. Appointment of Business Rescue Practitioner

The appointment of a business rescue practitioner is provided for in business rescue proceedings, whether by company resolution¹¹⁶⁵ or by court order.¹¹⁶⁶ The purpose of

¹¹⁵⁸CA, *op cit*, s. 134(1)(a)(i)(ii).

¹¹⁵⁹Act No. 66 of 1995.

¹¹⁶⁰CA, *op cit*, s. 136 (1)

¹¹⁶¹Act No. 24 of 1936.

¹¹⁶²CA, *op cit*, s. 136(2) as amended by CAA, *op cit*, s. 87(b).

¹¹⁶³CA, *ibid*, s. 136(2A) (a)(i) as amended by CAA, *ibid*, s. 87(c).

¹¹⁶⁴CA, *ibid*, s. 136(2A) (a)(ii) as amended by CAA, *ibid*.

¹¹⁶⁵CA, *ibid*, s. 129(3)(b).

¹¹⁶⁶*Ibid*, s. 131(5).

the appointment of a business rescue practitioner is to oversee a company during business rescue proceedings¹¹⁶⁷ and it is usually an officer of the court.¹¹⁶⁸

7. Qualification of Business Rescue Practitioners

A person may be appointed as a business rescue practitioner of a company only if the person:¹¹⁶⁹

- is a member in good standing of a legal, accounting or business management profession accredited by the Companies and the Intellectual Property Commission;
- has been licensed as such by the Companies and Intellectual Property Commission;¹¹⁷⁰
- 3. is not subject to an order of probation in terms of section 162(7);
- would not be disqualified from acting as a director of the company in terms of section 69(8);
- 5. does not have any other relationship with the company such as would lead a reasonable and informed third party to conclude that the integrity, impartiality or objectivity of that person is compromised by that relationship;
- is not related to a person who has a relationship as contemplated in 5 above.

8. Effects of the Appointment of Business Rescue Practitioner

The primary effect of the appointment of business rescue practitioner is that the management powers of the company is divested from the board and invested in the

¹¹⁶⁷CA, *op cit*, s. 128(1)(d).

¹¹⁶⁸*Ibid*, s. 140(3)(a).

¹¹⁶⁹*Ibid*, s. 138(1) as amended by CAA, *op cit*, s. 88.

¹¹⁷⁰CA, *ibid*, s. 138(2) as amended by CAA, *ibid*.

business rescue practitioner.¹¹⁷¹ Furthermore, the board ceases to act except as the business rescue practitioner may approve. Hence, upon the commencement of business rescue proceedings, the directors of a company must provide the business rescue practitioner with a statement of affairs.¹¹⁷²

9. Cessation of Appointment of Business Rescue Practitioner

The appointment of business rescue practitioner may come to an end if he or she dies,¹¹⁷³ resigns or is removed from office by the court on the following grounds:

- i. The practitioner is not qualified and independent of the company or its management or lacks the necessary skills;¹¹⁷⁴
- ii. Incompetence or failure to perform his or her duties;¹¹⁷⁵
- Failure to exercise the proper degree¹¹⁷⁶ of care in the performance of his or her duties;
- iv. Engaging in illegal acts or conduct;¹¹⁷⁷
- v. Conflict of interest or lack of independence;¹¹⁷⁸
- vi. Incapacitation.¹¹⁷⁹

10. Powers and Duties of Business Rescue Practitioners

In order to achieve the purpose of his appointment, business rescue practitioners are conferred with far-reaching powers and duties. In addition to any other powers and duties set out in the applicable provisions of the extant legislation, the business rescue practitioner has such power and duties to exercise, either with or without the approval of the court as follows:

¹¹⁷¹CA, *op cit*, s. 140(1).

¹¹⁷²*Ibid*, s. 142(3) as amended by CAA, *op cit*, s. 92.

¹¹⁷³CA, *ibid*, s. 139(3).

¹¹⁷⁴*Ibid*, s. 130(1)(b).

¹¹⁷⁵*Ibid*, s. 130(2) (a) as amended by CAA, *op cit*, s. 89.

¹¹⁷⁶CA, *ibid*, s. 130(2)(b).

¹¹⁷⁷*Ibid*, s. 130(2)(c).

¹¹⁷⁸*Ibid*, s. 130(2)(e).

¹¹⁷⁹*Ibid*, s. 130(2)(f).

- Power to exercise full management powers or control of the company in substitution for its board and pre-existing management.¹¹⁸⁰ This management power can be delegated by the practitioner.¹¹⁸¹
- 2. Power to appoint a person as part of the management of the company or an adviser to the company or the practitioner where such person has a relationship with the company that impugns his integrity, impartiality or objectivity.¹¹⁸² This power can only be exercised with the approval of the court.¹¹⁸³
- Power to investigate the company's affairs, business, property and financial situation so as to ascertain whether there is any reasonable prospect of the company being rescued.¹¹⁸⁴
- 4. Duty to inform all relevant regulatory authorities having authority in respect of the activities of the company to the fact that the company has been placed under business rescue proceedings and of his or her appointment.¹¹⁸⁵
- 5. Duty to develop a business rescue plan to be considered by affected persons and to implement such plan.¹¹⁸⁶
- 6. Duty to report to the court as an officer of the court¹¹⁸⁷ and to inform the court, the company and all affected persons, to wit, (i) where there is no reasonable prospect for the company to be rescued and hence, apply to the court for an order discontinuing the business rescue proceedings and

¹¹⁸⁰CA, *op cit*, s. 140(1)(a).

¹¹⁸¹*Ibid*, s. 140(1)(b)(c).

 $^{^{1182}}_{1182}$ *Ibid*, s. 140(2)(a).

¹¹⁸³*Ibid*, s. 140(2).

¹¹⁸⁴*Ibid*, s. 141(1).

¹¹⁸⁵*Ibid*, s. 140(1A) as amended by CAA, *op cit*, s. 90.

¹¹⁸⁶CA, *ibid*, ss. 140(1)(a); 150, 151 & 152.

¹¹⁸⁷ *Ibid*, s. 140(3)(a).

placing the company into liquidation¹¹⁸⁸ (ii) where there are no longer reasonable grounds to believe that the company is financially distressed and hence, apply to the court for an order terminating the business rescue proceedings or otherwise to file a notice of termination of the business rescue proceedings.¹¹⁸⁹

- Duty to take necessary steps to rectify any void transactions or the failure by the company or any director to perform any material obligation relating to the company.¹¹⁹⁰
- 8. Duty to forward the evidence of any reckless trading, fraud or other contravention of any law relating to the company to the appropriate authority for further investigation and possible prosecution.¹¹⁹¹
- 9. Duty to convene and preside over the first meeting of creditors¹¹⁹² and the first meeting of employees' representatives.¹¹⁹³
- Power to exercise all the duties of directors as set out in sections 75 to 77 of the CA.¹¹⁹⁴

11. Committee of Affected Persons

The Committee of Affected Persons (Committee) consisting of the committee of creditors ¹¹⁹⁵or of employees¹¹⁹⁶ is constituted during the business rescue proceedings to carry out the following functions or duties, that is:

i. Consult with the practitioner about any matter relating to the business

 $^{^{1188}}$ CA, *op cit*, s.141(2)(a).

¹¹⁸⁹*Ibid*, s. 141(2)(b).

¹¹⁹⁰*Ibid*, s. 141(c)(i) as amended by CAA, *op cit*, s. 91.

¹¹⁹¹CA, *ibid*, s. 141(2)(c) (ii).

¹¹⁹²*Ibid*, s.147(1).

¹¹⁹³*Ibid*, s. 148(1).

¹¹⁹⁴*Ibid*, s. 140(3)(b).

¹¹⁹⁵*Ibid*, s. 147(1).

¹¹⁹⁶*Ibid*, s. 148(1).

rescue proceedings.¹¹⁹⁷ However, the Committee cannot direct or instruct the practitioner in relation to the exercise of his or her powers and duties;

- ii. Consider reports relating to the business rescue proceedings;¹¹⁹⁸
- iii. Act independently of the practitioner to ensure fair and unbiased representation of creditors' or employees' interests;¹¹⁹⁹
- iv. Approve or reject the business rescue plan developed by the practitioner.¹²⁰⁰

12. Business Rescue Plan

The whole essence of business rescue proceedings is the entrenchment or development of a business rescue plan for a financially distressed company as contemplated in section 150 of the CA. It is the duty of the business rescue practitioner to prepare a business rescue plan for consideration and possible adoption at a meeting held in terms of section 151 of the CA.¹²⁰¹ The business rescue plan is to be published by the company within 25 business days after the date on which the practitioner was appointed or longer time as may be allowed by the court on the application by the company, or the holder of the majority of the creditors voting interests.

1. Contents of Business Rescue Plan

The business rescue plan (Plan) contains all the information reasonably required to facilitate affected persons in deciding whether or not to accept the plan. Thus, the Plan is divided into three parts,¹²⁰² as follows:

¹¹⁹⁷CA, *op cit*, s. 149(1)(a).

¹¹⁹⁸*Ibid*, s. 149(1)(b).

¹¹⁹⁹*Ibid*, s. 149(1)(c).

¹²⁰⁰*Ibid*, s. 150(2).

¹²⁰¹*Ibid*, s. 150(1).

¹²⁰²*Ibid*, s. 150(2).

Part A- Background

This part¹²⁰³ must include the following:

- a. A complete list of all the material assets of the company;
- b. A complete list of the creditors of the company;
- c. The probable dividend that will be received by creditors, in their specific classes, if the company were to be placed in liquidation;
- d. A complete list of the holders of the company's issued securities;
- e. A copy of the written agreement concerning the practitioner' remuneration;
- f. A statement whether the Plan includes a proposal made informally by a creditor of the company.
- Part B- Proposals

This part must include the following:

- a. The nature and duration of the moratorium attached to the Plan;
- b. The extent to which the company is to be released from the payment of its debts, and the extent to which any debts is to be converted to equity in the company or another company;
- c. The on-going role of the company, and the treatment of any existing agreements;
- The property of the company that is to be available to pay creditors' claims;
- e. The order of preference in which the proceeds of property will be applied to pay creditors if the Plan is adopted
- f. The benefits of adopting the Plan as opposed to benefits that be received

¹²⁰³CA, *op cit*, s. 150(2)(a).

by creditors if the company were to be placed in liquidation;

g. The effect of the Plan on the holders of each class of the company's issued securities.

Part C- Assumptions and Conditions

This must include the following:

- a. A statement of the conditions that must be satisfied, if any, for the Plan to come into operation and be fully implemented;
- b. The effect, if any, that the Plan contemplates in the number of employees, and their terms and conditions of employment;
- c. The circumstances in which the Plan will end;
- d. A projected balance sheet for the company; and statement of income and expenses for the ensuing three years, prepared on the assumption that the proposed Plan is adopted.

The projected balance sheet and statement must include a notice of any material assumptions on which the projections are based; and may include alternative projections based on varying assumptions and contingencies.¹²⁰⁴

Furthermore, a proposed Plan must conclude with a certificate by the practitioner stating that any:¹²⁰⁵

- Actual information provided appears to be accurate, complete and up to date; and
- Projections provided are estimates made in good faith on the basis of factual information and assumptions as set out in the statement.

¹²⁰⁴CA, *op cit*, s. 150(3). ¹²⁰⁵*Ibid*, s. 150(4).

2. Consideration of Proposed Business Rescue Plan

The practitioner, must within 10 business days after publishing a Plan, commence and preside a meeting of creditors and any other holders of a voting interest for the purpose of considering the Plan.¹²⁰⁶ At the meeting convened for consideration of the Plan, the practitioner must:

- i. Introduce the proposed Plan for consideration by the creditors and if applicable, by the shareholders;¹²⁰⁷
- ii. Inform the meeting whether the practitioner continues to believe that there is reasonable prospect of the company being rescued;¹²⁰⁸
- iii. Provide an opportunity for the employees' representatives to address the meeting¹²⁰⁹
- iv. invite discussion, entertain and conduct a vote on any motion to amend the proposed Plan or direct the practitioner to adjourn the meeting in order to revise the Plan for further consideration;¹²¹⁰
- v. Call for a vote for preliminary approval of the proposed Plan, as amended, if applicable.¹²¹¹

The proposed Plan will be approved on a preliminary basis if it was supposed by the holders of more than 75% of the creditors' voting interests that were voted and the votes in support of the proposed Plan included at least 50% of the independent creditors' voting interest, if any, that were voted.¹²¹²An independent creditor is a person who is a creditor of the company, including an employee of the company who

¹²⁰⁸CA,*ibid*,s.152(1)(b).

¹²⁰⁶CA, *op cit*, s. 151(1) as amended by CAA, *op cit*, s. 95.

¹²⁰⁷CA, *ibid*, s. 152(1)(a) as amended by CAA, *ibid*, s. 96(a).

 $^{^{1209}}Ibid$, s. 152(1)(c).

¹²¹⁰*Ibid*, s. 152(1)(d).

¹²¹¹*Ibid*, s. 152(1)(e).

¹²¹²*Ibid*, s. 152(2)(a)(b).

is a creditor in terms of section 144(2) of the Companies Act; and is not related to the company, a director or the practitioner.¹²¹³

It should be stated that where a proposed Plan is not approved on a preliminary basis, the Plan is deemed rejected, and may be reconsidered subject to section 153 of the CA.¹²¹⁴ On the other hand, where the meeting approves a proposed Plan on a preliminary basis and does not alter the rights of the holders of any class of the company's security, such approval constitutes the final adoption of the plan, subject to the satisfaction of any contingent conditions.¹²¹⁵ But if the meeting approves a proposed Plan on a preliminary basis and does alter the rights of the holders of any class of any class of the company's securities, the practitioner must commence and hold a meeting of holders of the class or classes of securities, whose rights would be altered by the plan, and call for a vote by them to approve the adoption of the Plan.¹²¹⁶ If a majority of the voting rights that were exercised, either support adoption of any contingent conditions¹²¹⁷ or oppose the adoption of the Plan, the Plan is deemed to have been rejected, and may be considered further only in terms of section 153 of the CA.¹²¹⁸

In a situation where a proposed Plan is rejected, the practitioner may seek a vote of approval from the holders of voting interests to prepare and publish a revised Plan or advise the meeting that the company will apply to a court to set aside the result of the vote by the holders of voting interests or shareholders on the grounds that it was inappropriate.¹²¹⁹ It should be noted that where the practitioner fails or neglects to take any action as contemplated herein before, any affected person present at the

¹²¹³CA, *op cit*, s. 128 (1) (g) (2) as amended by CAA, *op cit*, s. 81(b).

¹²¹⁴CA, *ibid*, s. 152 (3) (a).

¹²¹⁵*Ibid*, s. 152 (3) (b).

¹²¹⁶*Ibid*, s. 152 (3) (c) (i).

¹²¹⁷*Ibid*, s. 152 (3) (c) (ii) (aa).

¹²¹⁸*Ibid*, s. 152 (3) (c) (ii) (bb).

¹²¹⁹*Ibid*, s. 153 (1) (a) (i) (ii).

meeting may stand in place of the practitioner to so do.¹²²⁰ However, any affected person, or combination of affected persons may make a binding offer to purchase for valuable consideration the voting interests of one or more persons who opposed the adoption of the Plan.¹²²¹

3. Effect of Adoption of Proposed Business Rescue Plan

The adoption of a Plan has the following effects:

- 1. It is binding on the company, and on each of the creditors of the company and every holder of the company' securities, whether or not such a person was present at the meeting where the plan was adopted; voted in favour of adoption of the Plan; or in the case of creditors, had proven their claims against the company.¹²²²
- Any prior rights to enforce debts and claims by a creditor is abated and is only exercised in accordance with the terms and conditions in the plan¹²²³. Suffice to state that if a Plan has been approved and adopted, a creditor is not entitled to enforce any debt owed by the company, except to the extent provided for in the Plan.¹²²⁴
- 3. The company under the direction of the practitioner must take all necessary steps to attempt to satisfy any conditions on which the Plan is contingent; and implement the Plan as adopted.¹²²⁵ In furtherance to this, when the Plan has been substantially implemented, the practitioner must file a notice of the substantial implementation of the Plan.¹²²⁶

¹²²⁰CA, *op cit*, s. 153(1)(b) (i) (aa) (bb).

¹²²¹*Ibid*, s. 153(1)(b)(ii).

¹²²²*Ibid*, s. 152(4).

¹²²³*Ibid*, s. 154.

¹²²⁴*Ibid*, s. 154(2).

¹²²⁵*Ibid*, s. 152(5) (6).

¹²²⁶*Ibid*, s. 152(8).

A moratorium on legal proceedings against the company is provided.¹²²⁷ 4.

13. Arrangement or Compromise with Creditors

Another corporate rescue mechanism available under the extant law in the Republic of South Africa is scheme of arrangement or compromise. The unique feature of this mechanism is that it can be invoked by a company, irrespective of whether or not it is financially distressed, unless the company is engaged in business rescue proceedings.¹²²⁸

1. Procedure

The board of a company or the liquidator of such a company if it is being wound up may propose an arrangement or a compromise of its financial obligations to all of its creditors or to all of the members of any class of its creditors.¹²²⁹ An arrangement or a compromise is commenced by delivery of a copy of the proposal and notice of meeting to consider the proposal to every director of the company or every member of the relevant class of creditors whose name or address is known to, or can reasonably be obtained by the company and the Companies and Intellectual Property Commission.¹²³⁰

It is the procedure that a proposal contemplated in section 155(2) of the CA must contain all information reasonably required to facilitate creditors in deciding whether or not to accept or reject the proposal and is divided into three parts, namely (i) Part A- Background; (ii) Part B- Proposal; (iii) Part C -Assumptions and Conditions.¹²³¹ A salient feature of the proposal in an arrangement or a compromise is that it makes provision for the nature and duration of any proposed debt

¹²²⁷CA, *op cit*, s. 150 (2) (b) (i). ¹²²⁸*Ibid*, s. 155(1).

¹²²⁹*Ibid*, s. 155(2).

 $^{^{1230}}$ *Ibid*, s. 155(2)(a)(b). 1231 *Ibid*, s. 155(3)(a)(b)(c).

moratorium.¹²³² This feature is not applicable in arrangement and compromise under the CAMA.

The proposal by the board of a company or the liquidator of such a company if under liquidation proceedings can be either approved or rejected at the meeting convened or called to consider the proposal. Where the proposal is supplied by a majority representing at least 75% in value of the creditors or class, as the case may be, present and voting in person or by proxy at the meeting called to consider the proposal, such a proposal would be deemed to have been adopted by the creditors.¹²³³

Where a proposal is adopted at the meeting called to consider the proposal, the company may apply to the court for an order approving the proposal¹²³⁴ and the court may sanction the compromise as set out in the adopted proposal if it considers it just and equitable to do so.¹²³⁵ In this respect, the court applies the just and equitable principle in the case of proposal by the board of the company, having regard to the number of creditors or any affected class of creditors, who were present or represented at the meeting, and who voted in favour of the proposal.¹²³⁶ In the case of a proposal by a liquidator of a company, the court applies the just and equitable principle, having regard to the report of the master.¹²³⁷ The master¹²³⁸ means the person holding the office of that name in terms of the Supreme Court Act.¹²³⁹

Where the court sanctions a compromise, a copy of the order of the court has to be filed within five business days and attached to each copy of one company's'

¹²³² CA, *op cit*, s. 155(3)(b)(i).

¹²³³*Ibid*, s. 155 (6).

 $^{^{1234}}Ibid$, s. 155 (7) (a).

¹²³⁵*Ibid*, s. 155 (7) (b). ¹²³⁶*Ibid*, s. 155 (7) (b) (i).

 $^{^{1237}}$ *Ibid*, s. 155 (7) (b) (ii).

¹²³⁸*Ibid.* s.1.

¹²³⁹No. 59 of 1959.

Memorandum of Incorporation.¹²⁴⁰ Hence, the order of one court sanctioning a compromise is final and binding on all of the company's creditors or all of members of the relevant class of creditors, as the case may be, as of the date in which it is filed.¹²⁴¹

The corporate rescue mechanisms available in the Republic of South Africa are not only home-grown but dynamic, flexible and simple to facilitate the rehabilitation of a company that is financially distressed. Furthermore, the procedure for business rescue proceedings is reasonably holistic to the extent that the interests of all the stakeholders in a company were incorporated. However, while the procedure for business rescue proceedings made obvious references to the rejection of a proposed Plan and thereby dismissal of the business rescue proceedings and placing the company under liquidation proceedings,¹²⁴² the law and procedure relating to the liquidation proceedings after the conversion is still under the provisions of the Companies Act¹²⁴³ by virtue of item 9, schedule 5 of the CA seems a descent into retrogression. Also the role of court in the respective corporate rescue mechanism in the Republic of South Africa is minimal and thus saves time and cost.

4.4 The Republic of India

There are two main modes of corporate rescue mechanisms in the Republic of India. These are the Corporate Insolvency Resolution Process¹²⁴⁴ and Fast Track Corporate Insolvency Resolution Process.¹²⁴⁵

¹²⁴⁰CA, *op cit*, s. 155(8)(a)(b).

¹²⁴¹*Ibid*, s. 155(8)(c).

¹²⁴²*Ibid*, ss. 131(4) (b), 140 (4), 130(5)(c)(1).

¹²⁴³No 61of 1973, chapter xiv.

¹²⁴⁴Insolvency and Bankruptcy Code 2016 (IBC), chapter 11, ss. 6–32 (India).

¹²⁴⁵*Ibid*, chapter iv, ss. 55–58.

1. Corporate Insolvency Resolution Process (CIRP)

These are rescue mechanisms provided in the Insolvency and Bankruptcy Code to resolve corporate insolvency matters.

2. Modes of Initiation of CIRP

CIRP may be initiated by a financial creditor,¹²⁴⁶ an operational creditor¹²⁴⁷ or the corporate applicant ¹²⁴⁸ itself where a corporate debtor commits a default.¹²⁴⁹

A financial creditor¹²⁵⁰ either by itself or jointly with other financial creditors may file an application to initiate corporate resolution process against a corporate debtor before the Adjudicating Authority.¹²⁵¹ Adjudicating Authority means the National Company Law Tribunal constituted under section 408 of the Companies Act 2013 when a default has occurred.¹²⁵² The application filed by a financial creditor shall be accompanied with a record of the default recorded with the information utility,¹²⁵³ or such other record or evidence of default as may be specified; the name of the resolution professional¹²⁵⁴ proposed to act as an intern resolution professional; and any other information as may be specified by the Board.¹²⁵⁵

The Adjudicating Authority shall within fourteen days of the receipt of the application filed by the financial creditor, ascertain the existence of a default from the records of an information utility or on the basis of other evidence furnished by the financial creditor.¹²⁵⁶ Where the Adjudicating Authority is satisfied that a default has

¹²⁴⁶IBC, *op cit*, s. 5(7).

¹²⁴⁷*Ibid*, s. 5(20).

¹²⁴⁸*Ibid*, s. 3(8).

¹²⁴⁹Ibid, s. 3 (12). "Default" means non-payment of debt when while or any part or instalment of the amount have become due and payable and is not repaid by the debtor or corporate debtor. ¹²⁵⁰*Ibid*, s. 5(7).

¹²⁵¹*Ibid*, s. 5(1).

¹²⁵²*Ibid*, s. 7(i).

¹²⁵³Ibid, s. 3(21) -means a person who is registered with the Board as an information utility under section 210

¹²⁵⁴*Ibid*, s. 3(19).

¹²⁵⁵*Ibid*, s. 7(3).

¹²⁵⁶*Ibid*, s. 7(4).

occurred and the application filed by the financial creditor is complete, and there is no disciplinary proceedings pending against the proposed resolution professional, it may, by order, admit the application.¹²⁵⁷ Where the Adjudicating Authority is satisfied that the default has not occurred or the application filed by the financial creditor is incomplete or any disciplinary proceedings is pending against the proposed resolution professional, it may, by order, reject such application.¹²⁵⁸ It should be noted that where the Adjudicating Authority rejects an application on the ground of incomplete application, it shall give notice to the applicant to rectify the defect in the application within three days of receipt of such notice from the Adjudicating Authority.¹²⁵⁹ In all cases, the Adjudicating Authority within two days of an order of admission or rejection of an application, communicate same to the financial creditor and the corporate debtor.¹²⁶⁰

An "Operational Creditor"¹²⁶¹may, on the occurrence of a default, deliver a demand notice of unpaid operational debt or copy of an invoice demanding payment of the amount involved in the default to the corporate debtor in such form as may be prescribed, through an information utility, wherever applicable, or by registered post or courier or by such electronic mode of communication, as may be specified. After the expiry period of ten days from the date of delivery of the notice or invoice demanding payment, if the operational creditor does not receive payment from the corporate debtor or notice of the dispute under sub-section (2) of section 8 of the IBC, such operational debtor may file an application before the Adjudicating Authority for

¹²⁵⁷IBC, *op cit*, s. 7(5)(a).

¹²⁵⁸*Ibid*, s. 7(5)(b).

¹²⁵⁹*Ibid*, s. 7(5) proviso.

¹²⁶⁰*Ibid*, s. 7(7).

¹²⁶¹*Ibid*, s. 5(20)- means a person to whom an operational debt is owed and includes any person to whom such debt has been legally assigned or transferred (including a person resident outside India), ¹²⁶¹ "Operational debt" means a claim in respect of the provision of goods or services including employment or a debt in respect of these payment of dues arising under any law for the time being in force and payable to the central government, any state government or any local authority..

initiating corporate insolvency resolution process.¹²⁶² The application shall be accompanied with a copy of the invoice demanding payment or demand notice delivered by the operational creditor to the corporate debtor; an affidavit to the effect that the corporate debtor has not given notice of dispute of the operational debt; a copy of the certificate from the operational creditor's financial institution confirming that there is no payment of an unpaid operational debt by the corporate debtor and such other information as may be specified.¹²⁶³ The operational creditor may propose a resolution professional to act as an interim resolution professional.¹²⁶⁴

The Adjudicating Authority shall, within fourteen days of the receipt of the application filed by an operational creditor, by an order, admit the application and communicate such decision to the operational creditor and the corporate debtor if¹²⁶⁵:-

- (i) The application filed by the operational creditor is complete;
- There is no repayment of the unpaid operational debt; (ii)
- (iii) The invoice or notice for payment to the corporate debtor has been delivered by the operational creditor;
- (iv) No notice of dispute has been received by the operational creditor or there is no record of dispute in the information utility;
- There are no disciplinary proceedings pending against any proposed (v) resolution professional to act as interim resolution professional.

However, where the converse to the above is the case, the Adjudicating Authority shall by order, reject the application and communicate such decision to the operational creditor and the corporate debtor.¹²⁶⁶ It should be noted that before the Adjudicating Authority rejects an application, the operational creditor, shall be given

¹²⁶² IBC, *op cit*, s. 9(1). ¹²⁶³*Ibid*, s. 9(3).

¹²⁶⁴*Ibid*, s. 9(4).

¹²⁶⁵*Ibid*, s. 9(5)(I).

¹²⁶⁶*Ibid*, s. 9(5)(ii).

notice to rectify the defect in the application within three days of the date of receipt of such notice from the Adjudicating Authority.¹²⁶⁷

A corporate Applicant¹²⁶⁸ may where a corporate debtor¹²⁶⁹ has committed a default, file an application for initiating corporate insolvency resolution process with the Adjudicating Authority.¹²⁷⁰ The corporate applicant shall along with the application furnish the information relating to its books of account and such other documents relating to such period as may be specified and the resolution professional proposed to be appointed as an interim resolution professional.¹²⁷¹

The Adjudicating Authority shall within a period of fourteen days of the receipt of the application, by an order, admit the application, if it is complete or reject the application, if it is incomplete.¹²⁷²

3. Disqualification from Initiating CIRP

Although a financial creditor or an operational creditor or a corporate debtor are generally qualified to initiate corporate insolvency resolution process, the following persons are not entitled to make an application to initiate CIRP, namely:

- (i) A corporate debtor undergoing a CIRP; 1273
- (ii) A corporate debtor having completed CIRP in last twelve months preceding the date of making the application;¹²⁷⁴
- (iii) A corporate debtor or a financial creditor who has violated any of the terms of resolution plan which was approved twelve months before the

¹²⁶⁷IBC, *op cit*, s. 9(5)(ii) proviso.

¹²⁶⁸*Ibid*, s. 5(5)- means (a) corporate debtor; or (b) a member or partner of the corporate debtor who is authorised to make an application for the corporate insolvency resolution process under the constitutional document of the corporate debtor; or (c) an individual who is in charge of managing the overall operations and resources of the corporate debtor; or (d) a person who has the control and supervision over the financial affairs of the corporate debtor.

¹²⁶⁹*Ibid*, s.3(8)- means a corporate person who owes a debt to any person.

¹²⁷⁰*Ibid*, s. 10(1).

¹²⁷¹*Ibid*, s. 10(3).

¹²⁷²*Ibid*, s. 10(4).

¹²⁷³*Ibid*, s. 11(a).

¹²⁷⁴*Ibid*, s. 11(b).

date of making of an application;¹²⁷⁵

 (iv) A corporate debtor in respect of whom a liquidation order has been made.¹²⁷⁶

4. Duration of CIRP

CIRP commences from the date of admission of the application.¹²⁷⁷ CIRP shall be completed within a period of one hundred and eighty days from the date of the application to initiate such process.¹²⁷⁸ However, the resolution professional shall file an application to the Adjudicating Authority to extend the period of the CIRP if instructed to do so by a resolution passed at a meeting of the committee of creditors by a vote of 75% of the voting shares.¹²⁷⁹ It should be noted that while the Adjudicating Authority may by order extend the duration of such process by further period as it thinks fit, it cannot exceed ninety days¹²⁸⁰ and not for more than once.¹²⁸¹

5. Effects of Admission of Application for Initiation of CIRP

The admission of an application for initiation of CIRP has the following effects:

- (1) The Adjudicating Authority by an order, shall declare a moratorium for the purposes set out in section 14 of the IBC¹²⁸². A moratorium is declared to prohibit all of the following:¹²⁸³
- (i) the institution of suits or continuation of pending suits or proceedings against the corporate debtor including execution of any judgment, decree or order in any court of law, tribunal, arbitration panel or other

¹²⁷⁵IBC, *op cit*, s. 11(c).

¹²⁷⁶*Ibid*, s. 11(d).

¹²⁷⁷*Ibid*, ss. 7(6) (Application filed by financial creditor), 9 (6) (Application filed by operational creditor), 10 (5) (Application filed by corporate applicant).

¹²⁷⁸*Ibid*, s. 12(1).

¹²⁷⁹*Ibid*, s. 12(2).

¹²⁸⁰*Ibid*, s. 12(3).

¹²⁸¹*Ibid*, s. 12(3) proviso.

¹²⁸²*Ibid*, s. 13(1)(a).

¹²⁸³*Ibid*, s. 14(1).

authority;1284

- (ii) the corporate debtor from transferring, encumbering, alienating or disposing of any of its assets or any legal right or beneficial interests;¹²⁸⁵
- (iii) any action to foreclose, recover or enforce any security interest created by the corporate debtor in respect of its property including any action under the Securitisation and Reconstruction of Financial Assets and Enforcement of Security Interest Act;¹²⁸⁶
- (iv) the recovery of any property by an owner or lesser where such property is occupied by or in the possession of the corporate debtor;
- (v) termination or suspension of the supply of essential goods or services to the corporate debtors as may be specified.¹²⁸⁷

It is worthwhile to state that the order of moratorium declared by the Adjudicating Authority shall have effect from the date of order until the completion of the CIRP.¹²⁸⁸ However, such moratorium will not apply to such transactions as may be notified by the Central Government of India in consultation with any financial sector regulator¹²⁸⁹ and will also cease, if the Adjudicating Authority approves the resolution plan or passes an order for liquidation of the corporate debtor.¹²⁹⁰

2. The Adjudicating Authority by an order shall cause a public announcement of the initiation of The CIRP and call for the submission of claims.¹²⁹¹ Such public announcement of the CIRP shall contain the information set out in section (15)(1) of the IBC. It should be mentioned that the public announcement of the corporate

¹²⁸⁴ IBC, op cit, s. 14(1)(a).
¹²⁸⁵ Ibid, s. 14(1)(b).
¹²⁸⁶2002.
¹²⁸⁷IBC, op cit, s. 14(2).
¹²⁸⁸ Ibid, s. 14(4).
¹²⁸⁹ Ibid, s. 14(4).
¹²⁹⁰ Ibid, s. 14 proviso.
¹²⁹¹ Ibid, ss. 13(1) (b), 15(1)(c)(e).

insolvency resolution process is made immediately after the appointment of the interim resolution professional.¹²⁹²

3. The Adjudicating Authority, by an order, shall appoint an interim resolution professional in the manner as laid down in section 16 of the IBC.¹²⁹³

6. Appointment of Interim Resolution Professional

The Adjudicating Authority shall within fourteen days from the date of commencement of the insolvency resolution process, appoint an interim resolution professional.¹²⁹⁴ Where the application for CIRP was made by a financial creditor or the corporate debtor, the resolution professional, as proposed respectively in the application, shall be appointed as the interim resolution professional.¹²⁹⁵ Where the application was filed by an operational creditor, if there was no proposal for an interim resolution professional, the Adjudicating Authority shall make a reference to the Board for the recommendation of an insolvency professional, who may act as an interim resolution professional, but if there was a proposal for an interim resolution professional.¹²⁹⁶ It is important to note that an interim resolution professional acts in that capacity for a maximum period of thirty days from the date of his appointment.¹²⁹⁷

An interim resolution professional is insolvency professional¹²⁹⁸ enrolled with an insolvency professional agency as its member¹²⁹⁹ and also registered with the

¹²⁹² IBC, *op cit*, s. 13 (2).

¹²⁹³ *Ibid*, s. 13 (1) (c).

¹²⁹⁴*Ibid*, ss. 16 (1), 5 (27) interim resolution professional has same meaning as resolution professional.

¹²⁹⁵*Ibid*, s. 16 (2).

¹²⁹⁶*Ibid*, s. 16 (3).

¹²⁹⁷*Ibid*, s. 16 (5).

¹²⁹⁸*Ibid*, s. 5 (27).

¹²⁹⁹*Ibid*, s. 207 (1).

Board as insolvency professional.¹³⁰⁰ Thus, a person can only be qualified to be appointed an interim resolution professional if he is insolvency professional.

7. Effects of Appointment of Interim Resolution Professional.

The appointment of interim resolution professional in corporate insolvency resolution process from the date of appointment begets the following effect:-

- 1. The management of the affairs of the corporate debtor is invested in the interim resolution professional.¹³⁰¹
- The powers of the board of directors of the corporate debtor, stands suspended, and is to be exercised by the interim resolution professional.¹³⁰²
- 3. The officers and managers of the corporate debtor are to report to the interim resolution professional and provide access to such documents and records of the corporate debtor as may be demanded by the interim resolution professional.¹³⁰³
- 4 The financial institutions maintaining accounts of the corporate debtor shall act on the instruction of the interim resolution professional in relation to such accounts and furnish all information relating to the corporate debtor available to them to the interim resolution professional.¹³⁰⁴

8. Duties and Authority of Interim Resolution Professional

The interim resolution professional is conferred with some far-reaching duties and authority in order to achieve the purpose of the CIRP. Some of the duties of the interim resolution professional include:

 collect all information relating to the assets, finances and operations of the corporate debtor for determining the financial position of the corporate

¹³⁰⁰IBC, *op cit*, s. 3 (19).

¹³⁰¹ *Ibid*, s. 17 (1) (a).

¹³⁰²*Ibid*, s. 17 (1) (b).

¹³⁰³*Ibid*, s. 17 (1) (c).

¹³⁰⁴*Ibid*, s. 17 (1) (d).

debtor;¹³⁰⁵

- receive and collect all the claims submitted by creditors to him, pursuant to the public announcement made under sections 13 and 15 of the IBC;¹³⁰⁶
- 3. constitute the committee of creditors;¹³⁰⁷
- 4. monitor the assets of the corporate debtor and manage its operation until a resolution professional is appointed by the committee of creditors;¹³⁰⁸
- 5. file information collected with the information utility, if necessary;¹³⁰⁹

take

control and custody of any asset over which the corporate debtor has ownership rights as recorded in the balance sheet of the corporate debtor, or with the information utility or the depository of securities or any other registry that records the ownership of assets;¹³¹⁰

- provide all the information, documents and records pertaining to the corporate debtor in his possession and knowledge to the resolution professional;¹³¹¹
- 7. to perform such other duties as may be specified by the Board; 1312
- 8. to take reasonable care and diligence while performing his duties.¹³¹³

In order to perform his duties, an interim resolution professional is vested with the authority:

 to appoint accountants, legal counsel or such other professionals as may be necessary,¹³¹⁴

- ¹³⁰⁶*Ibid*, s. 18 (b).
- ¹³⁰⁷*Ibid*, s. 18 (c).
- ¹³⁰⁸*Ibid*, s. 18 (d).
- ¹³⁰⁹*Ibid*, s. 18 (e). ¹³¹⁰*Ibid*, s. 18 (f).

 101a , s. 25 (3). 1312 *Ibid*, s. 18 (g).

¹³⁰⁵ IBC, *op cit*, s. 18 (a).

 $^{^{1311}}$ *Ibid*, s. 23 (3).

 $^{^{1313}}$ *Ibid*, s. 208 (2) (a).

- to enter into contracts on behalf of the corporate debtor or to amend or modify the contracts or transactions which were entered into before the commencement of CIRP,¹³¹⁵
- 3. to raise interim finance provided that no security interest shall be created over any encumbered property of the corporate debtor without the prior consent of the creditors whose debt is secured over such encumbered property,¹³¹⁶
- 4. to issue instructions to its personnel as may be necessary for keeping the corporate debtor as a giving concern,¹³¹⁷
- 5. to take all such actions as are necessary to keep the corporate debtor as a going concern.¹³¹⁸

9. Committee of Creditors

After the collation of all claims received against the corporate debtor and determination of the financial position of the corporate debtor, the interim resolution professional, shall constitute a committee of creditors.¹³¹⁹ The committee of creditors shall comprise all financial creditors of the corporate debtor.¹³²⁰ All decisions of the committee of creditors shall be taken by a vote of not less than seventy-five per cent of voting shares of the financial creditors.¹³²¹ However, where a corporate debtor do not have financial creditors, the committee of creditors shall be constituted and

 $^{1317}_{1218}$ *Ibid*, s. 20 (2) (d).

 $^{1319}Ibid$, s. 21 (1).

¹³¹⁴*Ibid*, s. 20 (2) (a).

¹³¹⁵IBC, *op cit*, s. 20 (2) (b).

 $^{^{1316}}Ibid$, s. 20 (2) (c).

¹³¹⁸*Ibid*, s. 20 (2) (e).

¹³²⁰*Ibid*, s. 21 (2).

¹³²¹*Ibid*, s. 21 (8).

comprise of such persons to exercise such functions in such a manner as may be specified by the Board.¹³²²

The first meeting of the committee of creditors shall be held within three days of being constituted.¹³²³ It should be noted that subsequent meetings of the committee may be by such electronic means as may be specified.¹³²⁴ All meetings of the committee shall be conducted by the resolution professional,¹³²⁵ who shall give notice of each meeting of the committee to the members of the suspended board of directors of the corporate debtor.¹³²⁶ Although, the directors may attend the meetings of the committee, they do not have any right to vote in such meetings.¹³²⁷ Furthermore, the absence of any such director at the meeting of the committee shall not invalidate proceedings of such meeting.¹³²⁸ Instructively, any creditor who is a member of the committee may appoint an insolvency professional other than the resolution professional to represent such creditor in a meeting of the committee.¹³²⁹

The Committee of Creditors exercise the following functions:

- 1. Appoint the resolution professional;¹³³⁰
- Make request for financial information in relation to the corporate debtor during the CIRP from the resolution professional;¹³³¹
- 3. Give prior approval of certain actions to be taken by the resolution professional as laid down in section 28 of the IBC;
- 4. Replace a resolution professional found wanting in the performance of

¹³²²*Ibid*, s. 21(8) proviso.

¹³²³IBC, *op cit*, s. 22 (1).

 $^{^{1324}}$ *Ibid*, s. 24 (1).

¹³²⁵ *Ibid*, s. 24 (2).

¹³²⁶*Ibid*, s. 24 (3). ¹³²⁷*Ibid*, s. 24 (4).

¹³²⁸*Ihid*.

¹³²⁹*Ibid*, s. 24 (5).

¹³³⁰*Ibid*, s. 22 (2).

 $^{^{1331}}$ *Ibid*, s. 21 (9).

his duties;¹³³²

5. Approve the resolution plan presented to it by the resolution professional.¹³³³

10. Appointment of Resolution Professional

It is the duty of the committee of creditors in the first meeting to appoint the resolution professional.¹³³⁴ The appointment of the resolution professional¹³³⁵ by the committee of creditors may either be by a resolution to appoint the interim resolution professional as the resolution professional or to replace the interim professional by another resolution professional.¹³³⁶ Where the committee of creditors resolve to continue with the interim resolution professional as resolution professional, it shall communicate the decision to the interim resolution professional, corporate debtor and Adjudicating Authority.¹³³⁷ But where the committee of creditors resolve to replace the interim resolution professional, it shall file an application before the Adjudicating Authority for the appointment of the proposed resolution professional.¹³³⁸

In a situation where the committee filed an application, the Adjudicating Authority shall forward the name of the proposed resolution professional to the Board for its confirmation. Where the Board confirms the proposed resolution professional, the Adjudicating Authority shall then make the appointment.¹³³⁹ If the Board refuses to confirm the name of the proposed resolution professional within ten days of the receipt of the name, the Adjudicating Authority shall, by order, direct the interim resolution professional to continue to function as the resolution professional until

¹³³²*Ibid*, s. 26.

¹³³³IBC, *op cit*, s. 30 (4).

¹³³⁴ *Ibid*, s. 22 (2).

¹³³⁵*Ibid*, s. 5 (27).

¹³³⁶*Ibid.*

¹³³⁷*Ibid*, s. 22 (3) (a).

¹³³⁸*Ibid*, s. 22 (3) (b).

such time as the Board confirms the appointment of the proposed resolution professional.¹³⁴⁰

The primary function of the resolution professional upon appointment is to conduct the entire CIRP and manage the operations of the corporate debtor during the period of the CIRP.¹³⁴¹

A person is qualified to be a resolution professional if he or she is enrolled with an insolvency professional agency as its member and also registered with the Board as insolvency professional.

11. Powers and Duties of the Resolution Professional

The resolution professional shall exercise powers and perform duties as are provided in the extant law. These powers and duties are exercised by the resolution professional, either on his own volition or with the prior approval of the committee of creditors,¹³⁴² which duty, in the main, is to preserve and protect the assets of the corporate debtor, including the continued business operations of the corporate debtor.¹³⁴³ In order to achieve this duty, the resolution professional shall undertake the following actions, namely:

- Take immediate custody and control of all assets of the corporate debtor, including the business records of the corporate debtor;¹³⁴⁴
- Represent and act on behalf of the corporate debtor with third parties, exercise rights for the benefit of the corporate debtor in judicial, quasijudicial or arbitration proceedings;¹³⁴⁵
- 3. Raise interim finances, subject to the approval of the committee of

¹³⁴⁰*Ibid*, s. 22 (5).

¹³⁴¹IBC, *op cit*, s. 23 (1).

¹³⁴²*Ibid*, s. 28 (1).

¹³⁴³*Ibid*, s. 25 (1).

 $^{^{1344}}_{1245}$ *Ibid*, s. 25 (2) (a).

¹³⁴⁵*Ibid*, s. 25 (2) (b).

Creditors;¹³⁴⁶

- Appoint accountants, lawyers and other advisors in the manner as specified by the Board;¹³⁴⁷
- 5. Maintain an updated list of claims;¹³⁴⁸
- 6. Convene and attend all meetings of the committee of creditors;¹³⁴⁹
- Prepare the information memorandum in accordance with section 29 of the IBC;¹³⁵⁰
- 8. Invite prospective lenders, investors and any other persons to put forward resolution plans;¹³⁵¹
- Present all resolution plans at the meetings of the committee of Creditors;¹³⁵²
- 10. File application for avoidance of transactions;¹³⁵³
- 11. Any further actions as may be specified by the Board.¹³⁵⁴

12. Resolution Plan

Resolution plan¹³⁵⁵ is a plan proposed by any person for insolvency resolution of the corporate debtor as a going concern. A resolution plan is normally prepared on the basis of the information memorandum made by the resolution professional, and submitted by a resolution applicant¹³⁵⁶ to the resolution professional.¹³⁵⁷

Although, a resolution plan is not mandated to be in any prescribed form under the extant legislation, every resolution plan must:

¹³⁴⁶*Ibid*, s. 25 (2) (c).

¹³⁴⁷IBC, *op cit*, s. 25 (2) (d).

¹³⁴⁸*Ibid*, s. 25 (2) (e). ¹³⁴⁹*Ibid*, s. 25 (2) (f).

 $^{^{1350}}$ *Ibid*, s. 25 (2) (1).

 $^{^{1351}}Ibid$, s. 25 (2) (g).

 $^{^{1352}}Ibid$, s. 25 (2) (i).

 $^{^{1353}}$ *Ibid*, s. 25 (2) (j).

 $^{^{1354}}$ *Ibid*, s. 25 (2) (J).

 $^{^{1355}}Ibid$, s. 5 (26).

¹³⁵⁶*Ibid*, s. 5 (25) - means any person who submits a resolution plan to the resolution professional.

¹³⁵⁷*Ibid*, s. 30 (1).

- provide for the payment of insolvency resolution process costs¹³⁵⁸ in a manner specified by the Board in priority to the repayment of other debts of the corporate debtor;¹³⁵⁹
- 2. provide for the repayment of the debt of operational creditors in such manner specified by the Board.¹³⁶⁰ However, the provision for repayment of the debts in the resolution plan shall not be less than the amount to be paid to the operational creditors in the event of a liquidation of the corporate debtor under section 53 of the IBC;¹³⁶¹
- 3. not contravene any of the provisions of the law for the time being in force;¹³⁶²
- 4. confirm to such other requirements as may be specified by the Board.¹³⁶³

Upon receipt of any resolution plan, the resolution professional shall examine such resolution plan and confirm that it meets with the requirements or conditions of a resolution plan.¹³⁶⁴ Where the resolution professional is satisfied that a resolution plan meets the requirements of a resolution plan, he or she shall present same to the committee of creditors for its approval.¹³⁶⁵ It should be noted that the resolution applicant may attend the meeting of the committee of creditors in which the resolution plan is considered¹³⁶⁶ but shall not have a right to vote at the meeting unless such resolution applicant is a financial creditor.¹³⁶⁷

- ¹³⁶⁰*Ibid*, s. 30 (2) (b).
- 1361 *Ibid*, s. 30 (2) (b).
- ¹³⁶²*Ibid*, s. 30 (2) (c). ¹³⁶³*Ibid*, s. 30 (2) (d).
- 1364 *Ibid*, s. 30 (2) (C
- 1365 *Ibid*, s. 30 (3).
- ¹³⁶⁶*Ibid*, s. 30 (5).

¹³⁵⁸*Ibid*, s. 5 (13).

¹³⁵⁹IBC, *op cit*, s. 30 (2) (a).

¹³⁶⁷*Ibid*, s. 30 (5) proviso.

Where the committee of creditors approve a resolution plan, the resolution professional shall submit the resolution plan as approved to the Adjudicating Authority.¹³⁶⁸ If the Adjudicating Authority is satisfied that the resolution plan as approved by the committee of creditors met the requirements of a resolution plan and there is no material irregularity in exercise of powers by the resolution professional during the corporate insolvency resolution period, shall by an order, approve the resolution plan.¹³⁶⁹

Consequently, from the date of order approving the resolution plan, it shall be binding on the corporate debtor and its employees, members, creditors, guarantors and other stake holders involved in the resolution plan. Furthermore, the moratorium order passed by the Adjudicating Authority under section 14 of the IBC shall cease to have effect¹³⁷⁰ and the resolution professional shall forward all records relating to the conduct of the CIRP and the resolution plan to the Board for record in its database.¹³⁷¹ It is important to state that an appeal against can be made to the National Company Law Appellate Tribunal against an order approving the resolution plan.¹³⁷² Any appeal from an order approving the resolution plan may be filed on the following grounds¹³⁷³:-

- The approved resolution plan is in contravention of the provisions of any law for the time being in force;
- 2. There has been material irregularity in exercise of the powers by the resolution professional during the corporate insolvency resolution period;
- 3. The debts owed to operational creditors of the corporate debtor have not

¹³⁶⁸*Ibid*, s. 30 (6).
¹³⁶⁹IBC, *op cit*, s. 31 (1) (a) (b).
¹³⁷⁰*Ibid*, s. 31 (3) (a).
¹³⁷¹*Ibid*, s. 31 (3) (b).
¹³⁷²*Ibid*, s. 32.
¹³⁷³*Ibid*, s. 61 (3).

been provided for in the resolution plan in the manner specified by the Board;

- The insolvency resolution process costs have not been provided for repayment in priority of all other debts;
- 5. The resolution plan does not comply with any other criteria specified by the Board.

A corporate insolvency resolution process shall come to an end in the happening of the following events:

- Approval or rejection by the Adjudicating Authority of a resolution plan as approved¹³⁷⁴ by the committee of creditors;
- Expiration of the corporate insolvency resolution process period as provided for in section 12 of the IBC;
- 3. The Adjudicating Authority did not receive a resolution plan from the resolution professional before the expiry period of the CIRP;¹³⁷⁵
- 4. The committee of creditors passed a resolution that the corporate debtor be placed in liquidation.

13. Fast Track CIRP

A fast track CIRP is a corporate rescue mechanism conducted in accordance with chapter IV of the IBC¹³⁷⁶ and is to be completed within a period of ninety-nine days from the insolvency commencement date.¹³⁷⁷ However, the resolution professional may file an application to the Adjudicating Authority for extension of the period of the fast track CIRP if instructed to do so by the committee of creditors.¹³⁷⁸ In this case, if the Adjudicating Authority determines that the case is of such complexity that

¹³⁷⁴IBC, *op cit*, s. 31 (1) (2).

¹³⁷⁵*Ibid*, s. 31 (1).

¹³⁷⁶*Ibid*, s. 55 (1).

¹³⁷⁷*Ibid*, s. 56 (1).

¹³⁷⁸*Ibid*, s. 56 (2).

an orderly fast track CIRP cannot be completed within a period of ninety-nine days, by an order, extend the duration of the process for a period not exceeding forty-five days¹³⁷⁹ and not more than once.¹³⁸⁰

An application for fast track CIRP may be initiated by a creditor or corporate debtor to the Adjudicating Authority by furnishing the proof of the existence of default as evidenced by records available with information utility or such other means as may be specified by the Board.¹³⁸¹ However, an application for fast track CIRP may be made in respect of the following corporate debtors, namely:

- 1. A corporate debtor with assets and income below a level as may be notified by the Central Government;¹³⁸² or
- 2. A corporate debtor with such class of creditors or such amount of debt as may be notified by the Central Government;¹³⁸³ or
- Such other category of corporate persons as may be notified by the Central 3. Government.¹³⁸⁴

It is pertinent to state that the procedure for conducting a corporate insolvency resolution process as earlier on discussed in this study, applies to fast track CIRP as the context may require.¹³⁸⁵ A fast track CIRP is distinct from a CIRP under chapter 11 of the IBC by the shorter duration of the process.

The available corporate rescue mechanisms in the Republic of India, that is, the Corporate Insolvency Resolution Process (CIRP) and the Fast Track CIRP, stands out

¹³⁷⁹ *Ibid*, s. 56 (3).

¹³⁸⁰IBC, op cit, s. 56 (3) proviso.

¹³⁸¹*Ibid*, s. 57 (a).

¹³⁸²*Ibid*, s. 55 (2) (a).

¹³⁸³*Ibid*, s. 55 (2) (b). ¹³⁸⁴*Ibid*, s. 55 (2) (c).

¹³⁸⁵*Ibid*, s. 58.

with the remarkable feature that provided for the exhaustion of any of the CIRP as condition precedent for initiation of liquidation proceedings.¹³⁸⁶

4.5 The United States of America

In addition to the various State Laws, the Bankruptcy Code¹³⁸⁷ (herein after referred to as the 'USBC') governs the cause of bankruptcy in the United States of America. It is important to note that the word "bankruptcy" is used in the USBC to define when an individual or corporate entity is unable to pay debts or discharge obligations, as and when due. Accordingly, the USBC is divided into 8 chapters. While chapters 1, 3 and 5 are administrative rules that generally apply to all types of bankruptcy proceedings, chapters 7, 9, 11, 12 and 13 are substantive rules for different types of bankruptcies, with one or two objectives, rehabilitation or liquidation. In this study, it is chapter 11 of the USBC which deals with rehabilitation of corporate entities that will be examined.

1. Chapter 11 Reorganization or Proceeding

The general purpose of Chapter 11 proceeding is to provide a meaningful opportunity to preserve the debtor's business by restructuring its debts and equity interest through a plan of reorganisation. The procedure for Chapter 11 proceeding follows almost the same steps as Chapter 7 proceedings that deals with liquidation of companies under the USBC.

Chapter 11 proceeding is initiated or commenced by a petition, either voluntary or involuntary. Voluntary petition is mostly filed by a debtor¹³⁸⁸ and an involuntary

¹³⁸⁶*Ibid*, s. 33.

¹³⁸⁷1978 (as amended).

¹³⁸⁸11 USC, *op cit*, § 301; J F Beatty & S S Samuelson, *Business Law for A New Century* (New York: Little, Brown and Company Ltd, 1996) p. 626.

petition usually filed by creditors.¹³⁸⁹ However, three creditors with "bona fide unsecured claims" of at least \$14, 425 in the aggregate are needed to file an involuntary petition against a debtor.¹³⁹⁰

The commencement of Chapter 11 proceeding by way of a petition triggers an 'automatic stay'.¹³⁹¹ An automatic stay operates as an injunction or prohibition against all actions affecting the debtor or its property. It also provides a respite and breathing spell for the debtor and for the creditors, it protects them by bringing to a halt all actions by individual creditors to obtain satisfaction of their claims using the remedies available under State law. However, there are some limited exceptions to the automatic stay that is; (i) it does not protect parties who are not in bankruptcy such as guarantors; and (ii) it does not stay the continuation or commencement of various types of regulatory actions.¹³⁹²

Upon filing a petition, the Bankruptcy Court (Court) issues an order for relief. This order is an official acknowledgment that the debtor is under the jurisdiction of the court and it is, in a sense, the commencement of the whole bankruptcy process, that is, Chapter 11 proceeding.

2. Primary Parties in Chapter 11 Reorganization Proceeding

i. Debtor in Possession or Trustee

This is one of the striking features of the Chapter 11 proceeding. In a Chapter 11 proceeding, the debtor entity and its existing management continue to operate the business as a "debtor in possession".¹³⁹³ In other words, the debtor entity and its existing management assume the role of a liquidator under insolvency proceedings in

¹³⁸⁹11 USC, *ibid*, § 303 (a); Beatty & Samuelson, *ibid*, p. 627.

¹³⁹⁰11 USC, *ibid*, § 303 (b) (1).

 ¹³⁹¹11 USC, *op cit*, § 362 (a). Automatic stay is akin to moratorium in the corporate rescue mechanisms of the United Kingdom, the Republic of India and the Republic of South Africa.
 ¹³⁹² *Ibid*, § 362 (b).

 $^{^{1393}}$ *Ibid*, § 1107 – 8.

the CAMA. In addition, the debtor in possession has the first prerogative to develop a plan of reorganisation. Where a debtor in possession is incompetent or uncooperative, the Court, the creditors or the United States Trustee can appoint a trustee to take over management of the debtor's affairs.¹³⁹⁴

ii. Creditors' Committee and other Official Committees.

Creditors are not permitted a direct role in operating the on-going business of the debtor in Chapter 11 proceeding. However, since in most cases, there is no neutral trustee, the creditors committee monitors the debtors' on-going operations, consults with the debtor on major business decisions and has a right to develop a plan of reorganisation if the debtor in possession fails to do so. It is pertinent to state that the creditors' committee is constituted of all the secured creditors and seven of the largest unsecured creditors willing to serve and appointed by the United States Trustee.¹³⁹⁵ Where the debtor is a corporation, the United States Trustee may also appoint a committee of equity shareholders to represent shareholders' interest.¹³⁹⁶

3. Plan of Reorganisation ("the Plan")

The debtor has the exclusive right to propose and file a plan during the first 120 days after the order for relief.¹³⁹⁷ The 120 days period is called the debtor's "exclusivity period" and it is often extended by the Court¹³⁹⁸ for a cause. However, the exclusivity period cannot be extended beyond 18months following the order for relief.¹³⁹⁹ A creditor may file a Plan only if the debtor's exclusivity period for filing a Plan has

¹³⁹⁴11 USC, op cit, § 1104.
¹³⁹⁵ Ibid, § 1102.
¹³⁹⁶Ibid, § 1102 (a) (2).
¹³⁹⁷Ibid, § 1121 (b).
¹³⁹⁸Ibid, § 1121 (d).
¹³⁹⁹Ibid, § 1121 (d) (2) (A).

expired or been terminated by the Court or if a Trustee has been appointed in the proceeding.¹⁴⁰⁰

The debtor is required to file schedules listing all of its creditors and shareholders. If a particular claim is described as "disputed", "contingent" or "unliquidated", then proof of claim must be filed by the creditor in order to preserve its rights. Otherwise, the creditor need not file a separate proof of claim unless it disagrees with the amount of the claim as listed or if the Chapter 11 proceeding is converted to a Chapter 7 proceeding.¹⁴⁰¹ It is important to note that where a proof of claim needs to be filed, it must be filed prior to the "bar date"¹⁴⁰² established by the Court. Furthermore, all creditors and shareholders are entitled to receive advance notice of the bar date.

Creditors assert their claims by filing a "proof of claim"¹⁴⁰³ in accordance with the form prescribed in the Bankruptcy Rules and attach the relevant supporting documentation such as the written contract or Note, or evidence of perfected security interest. In respect of a holder of an equity interest in the debtor, it asserts its claim by filing a "proof of interest". It is often the case that because the schedule of equity security holders filed by the debtor is usually sufficient evidence of the validity and amount of shareholder interest in the debtor, it is often unnecessary for holders of equity securities to file proofs of interest.

Under Chapter 11 proceeding, all claims of creditors are placed into one or three classes, namely (i) secured claim, (ii) priority claims, (iii) unsecured claims. A

¹⁴⁰⁰*Ibid*, § 1121.

¹⁴⁰¹11 USC, *op cit*, § 1111 (a).

¹⁴⁰²It means the last date stipulated by the Court as the last date by which proofs of claim will normally be accepted. In Chapter 11 proceedings that involves a large amount, the bar date is commonly set by the Court for six months or more.

¹⁴⁰³A proof of claim is a written statement setting forth the nature and amount of the creditor's claim against the debtor.

secured claim is a claim that is secured or backed by collateral; priority claims¹⁴⁰⁴ are those claims after the payment of secured claims that are entitled to payment priority over unsecured claims. In the general group of priority claims, it is worthy to note that certain priorities are higher than others. In other words, there are "super-super priority claims", "super-priority claims" and claims for "administrative expenses".¹⁴⁰⁵ Shareholders are also divided into classes depending on their interests, that is, preferred or common shareholders.

Instructively, it is required that before voting on a Plan, holders of claims and equity interests must receive a Court approved disclosure statement containing adequate information concerning the debtor and the Plan.¹⁴⁰⁶ Subsequently, the creditors and shareholders vote for or against the Plan. Acceptance by a class of claims require consent of at least 2/3 majority of the class of claims.¹⁴⁰⁷ Where there is an acceptance of the Plan, it is taken to the Court for confirmation and to make it binding on all parties.¹⁴⁰⁸

The confirmation of a plan by the Bankruptcy Court is not granted as a matter of course. The Court makes a number of determinations, to wit:

- i. That the Plan complies with all applicable law and has been proposed in good faith;
- That the Plan is feasible, that is, that the debtor has a credible business plan and can reasonably be expected to perform its obligations and accomplish the objectives set forth in the Plan;
- iii. If any individual creditor voted against the Plan and objects to its

¹⁴⁰⁴This is akin to preferential debt under s. 494 of the CAMA.

¹⁴⁰⁵11 USC, op cit, § 364, 503, 507.

¹⁴⁰⁶*Ibid*, §1125; Beatty and Samuelson, *op cit*, pp. 638 – 639.

¹⁴⁰⁷11 USC, *ibid*, § 1126 (c).

¹⁴⁰⁸ *Ibid*, § 1129 (a).

confirmation, that the Plan passes the "Best Interests of Creditors Test".¹⁴⁰⁹

Upon consideration of the acceptance of the Plan by the creditors and shareholders, if the Court makes the above determinations in the affirmative, it proceeds to confirm the Plan. Nevertheless, it is instructive to note that the Court can still confirm the Plan even where some classes voted against it under the principle of "crammed down".

This principle applies if at least one impaired class has voted to accept the Plan and the court finds that the treatment provided for the objecting class under the Plan does not discriminate unfairly¹⁴¹⁰ and is fair and equitable¹⁴¹¹. However, where the court rejects the Plan made by the debtor in possession, the creditors would be entitled to develop a new one.

4. Consequences of Confirmation of Plan under Chapter 11 Reorganization Proceeding

¹⁴⁰⁹*Ibid*, § 1129 (a) (7)-This test requires the Court to determine that the dissenting creditors or shareholders are receiving under the Plan at least as much as in present value terms as they would receive if the debtor were instead liquidated under Chapter 7 proceeding. Simply put, the test requires the court to compare the probable distribution to the dissenting creditors or equity holders if the debtor were liquidated with the present value of the payment or property to be received or retained by the same creditors or equity holders under the Plan. If the Court determines that the distribution under the Plan is equal to or more than the holders would have received in Chapter 7 proceeding, then the Plan would be held to be in the best interest of creditors as they would receive at least as much as they would have received had the business been liquidated under Chapter 7 proceeding.

¹⁴¹⁰The prohibition against "unfair discrimination" means that, ordinarily, similar claims or equity interests must be treated similarly.

¹⁴¹¹11 USC, *op cit*, § 1129 (b) (1), (2) (A) (B) (C); The precise determinations required for meeting the fair and equitable test turn on whether the class is secured or unsecured. Cram down of a secured class will be permitted if the Plan provides: (1) the secured creditors in the class will retain a lien to the extent of their secured claims and will receive deferred cash payments which have a present value equal to at least the value of their interest in the collateral, (2) for the sale of the secured creditor's collateral with the creditor's security interest attaching to the proceeds, or (3) for the realisation by the secured creditors of the "indubitable" equivalent of their secured claims. For the unsecured creditors and shareholders, cram down can be permitted if the Plan provides either that the creditors in the class receive (over time) cash payments equal to the present value of their full unsecured claims (i.e payment in full), or if creditors in the Plan are not recovering (over time) payment in full that at least classes junior to the class in question such as subordinated creditors or stockholders.

A confirmed Plan is binding on the debtors, creditors and shareholders. The debtor now owns the assets in the bankrupt estate, free of all obligations, except those listed in the Plan.¹⁴¹²

While the Chapter 11 reorganization or proceeding have been criticised for being too friendly to debtors and unfair to creditors because of the veto power conferred on shareholders over any rescue plan,¹⁴¹³ the ambit and purpose of the Chapter 11 proceeding which is not limited to just rescue of ailing companies but to entrench reorganisation or restructuring of a company even where solvent, is a very laudable step in the pursuit of corporate rescue. In this study, however, it is submitted that Chapter 11 proceedings may not be readily applicable as a corporate rescue option in Nigeria without modification.¹⁴¹⁴

4.6 Informal Corporate Rescue Procedures

An informal corporate rescue procedure focuses on the engagements of corporate restructuring outside formal statutory insolvency procedure. Such procedures give debtor companies a flexible opportunity to attempt to resolve their troubles outside the rigid, lengthy and often time consuming formal insolvency procedures. Informal corporate rescue procedures are by informal 'workouts', 'pre-packs' and 'debt for equity swaps', cherry picking and purchase and assumption.

Workouts is a 'loose' term used to include a variety of approaches implemented differently in respective jurisdictions and equally are different in each restructuring circumstance. Informal workouts depict consensual out of court restructuring through which a financially distressed company and its creditors reach an agreement adjusting

¹⁴¹²Beatty & Samuelson, *op cit*, p. 640.

¹⁴¹³*Ibid*.

¹⁴¹⁴The principal feature of Chapter 11 proceeding is the debtor being the person in possession, develops the plan of reorganisation. It is a fact that in Nigeria, most of the corporate failures have been linked to corporate mismanagement and misfeasance. In other words, it would be difficult for such directors to continue to steer the affairs of a company which they failed to keep afloat.

their respective obligations. Fundamentally, the creditors agree on a 'co-ordinated approach' by seeking to work with each other to devise a collective response which is in their mutual self-interest. Notable divergences exist in so far as the implementation of workouts in different jurisdictions is concerned but not without similar main characteristics.¹⁴¹⁵ An in-depth analysis of how a workout works and its principles are elaborated and expressed under the 'London Approach', 'pre-packs' and 'debts for equity swaps', purchase and assumption and cherry picking.

1. The 'London Approach'

The origin of the 'London Approach' can be traced back to the early 1970's, when a large number of companies, especially multi-bank based companies, encountered financial difficulties during the economic recession and concomitantly the crisis had a bad impact on the secondary banking sector of the United Kingdom.¹⁴¹⁶ The 'London Approach' is a non-statutory and informal framework introduced with the support of the Bank of England for dealing with temporary support operations mounted by banks and other lenders to a company or group in financial difficulties, pending a possible restructuring.¹⁴¹⁷

Essentially, the 'London Approach' established a flexible and co-operative informal rescue arrangement for corporate rescue which relies on negotiation with banks and their collective financial support.¹⁴¹⁸ Furthermore, the approach does not require the sanction of the court for its efficacy and implementation. Although, there is no such approach in Nigeria, the procedure that is somehow analogous, is the

¹⁴¹⁵UNCITRAL, 'Legislative Guide on Insolvency Law', 2005, p.22; The World Bank, 'Principles for Effective Insolvency and Creditor Rights System' (Revised Draft, 2005); INSOL, 'A Statement of Principles for a Global Approach to Multi-Creditor Workouts' (London: INSOL International, 2000). ¹⁴¹⁶J Armour & S Deakin, 'Norms in Private Insolvency: the 'London Approach' to the Resolution of Financial Distress' (2001) 1JCLS 21.

¹⁴¹⁷ C Baird, 'The London Approach' (1996) 12 Insolvency Law &Practice, 87; British Bankers Association, Description of the 'London Approach', Unpublished Memo, 1996, p.1

¹⁴¹⁸A Belcher, Corporate Rescue: A Conceptual Approach to Insolvency Law (London: Sweet & Maxwell, 1997) p.118; Armour & Deakin, art cit, 35.

approach adopted by the AMCON in the restructuring of its purchased NPL's and toxic loans.¹⁴¹⁹

2. Pre-packs

This basically involves a period of pre-insolvency negotiation with a prospective purchaser of the business of an insolvent company. The assets required by the purchaser will be agreed and a price for the sale of the business settled, invariably by reference to an independent valuation. Where an administration is then entered into, the business comprising the agreed assets and goodwill, contracts, employees and the like, are transferred to the purchaser. Pre-packs may be a business sale to a third party or a 'phoenix' sale where the previous directors take over the new business. While in the US, Pre-pack is a preliminary drafted agreement, in the UK, Pre-pack is a prepared business sale.¹⁴²⁰ In a nutshell, Pre-pack is a method of selling the business of an insolvent company as a going concern.¹⁴²¹

The essential features of a Pre-pack include the following:

- 1. Engagement of an insolvency practitioner to provide pre-appointment advice and assistance with the planning and negotiation of the business;
- Limited or "clandestine" marketing of the business in pre-appointment period on the basis that the core value of the business (goodwill, employees, customers, suppliers, etc) could quickly dissipate if there was public knowledge of the company's situation (or if an external administration itself were to come to pass before a sale was concluded);
- 3. Appointment of the same practitioner as voluntary administrator of the

¹⁴¹⁹Agabi & Onayemi, op cit, pp.472 – 473.

¹⁴²⁰K Crawford, 'The Law and Economics of Orderly and Effective Insolvency', unpublished PhD Thesis University of Nottingham, December 2012, pp.170-193

¹⁴²¹S Frisby, 'The Second Chance Culture and Beyond: Some Observations on the Pre-pack Contribution' (2009) Law & Financial Markets Review, 242.

company whereupon the administrator executes the contract on the first day of his appointment or immediately thereafter.

3. Debt for Equity Swap

A debt for equity swap can be employed as an informal corporate rescue procedure. This refers to where a creditor agrees to exchange a debt for an equity share in the company with the hope that it will produce a greater return in future. It is an approach that can enable a suitable compromise to be reached with creditors. However, this approach can be constrained as it may be time consuming and expensive to negotiate because the consent of the company shareholders, as well as majority of the creditors are usually required.¹⁴²² Worse still, in Nigeria, debt for equity swap as a form of compromise with creditors require the intervention of the Federal High Court and in deserving circumstance, reference to the SEC.¹⁴²³

4. Purchase and Assumption

This is where another Company or investors purchase the liabilities of a failing company and assume ownership of its assets, usually at auction price. The assumed Company does not go through a formal winding up proceedings but is dissolved through a judicial sale of its assets and liabilities to the purchasing Company or investor through a judicial accent.¹⁴²⁴

The transaction is perfected by way of duly executed Deed of Purchase and Assumption by the parties and the resolutions approving the transaction or Government white paper in case of Government Company or Company acquired or

¹⁴²²V Finch, *Corporate Insolvency Law: Perspective and Principles* (Cambridge: CUP, 2002) p.230. ¹⁴²³CAMA, *op cit*, s 539 (2) (3).

¹⁴²⁴Ogbuanya, *op cit*, p.670; A N Ibrahim, 'Resolution of Problem Banks: Purchase and Assumption Option', A Paper delivered at the International Association of Deposit Insurers (IADI) African Regional Committee Workshop at Abuja, Nigeria from May 9-13, 2011, p.3

taken over by a regulatory agency such as the AMCON, Central Bank of Nigeria (CBN) and NDIC.¹⁴²⁵

In the case of purchase and assumption in relation to reorganisation of banks, there are variants, that is, whole bank purchase and assumption, partial purchase and assumption, loss sharing purchase and assumption and bridge bank.¹⁴²⁶

5. Cherry Picking

This is akin to Purchase and Assumption but is differentiated by the fact that in this case, the purchasing company or investor is not taking up or purchasing all liabilities of the failing company but is allowed to inspect the books and assets or operations or business activities of the failing company with a view to picking out those aspects it could save by integrating into its own operations.¹⁴²⁷

¹⁴²⁵ Ogbuanya, *ibid*.

¹⁴²⁶ Ibrahim, *op cit*, p.7.

¹⁴²⁷ Ogbuanya, *op cit*, p.671.

CHAPTER FIVE

ISSUES AND CHALLENGES IN CORPORATE RESCUE AND INSOLVENCY PRACTICE IN NIGERIA

It is quite indubitable that the gloomy economic climate in Nigeria is taking its toll on business. The protracted fall in oil prices, skyrocketing foreign exchange rate and unfavourable government policies have driven the economy into its worst recession in about three decades. Consequently, corporate insolvencies are becoming rampant. In order to stem the effects of the on –going recession, many companies have adopted diverse restructuring, cost- cutting and rescue measures to stay afloat.

Unfortunately, there are certain issues and challenges that debilitate or enervate corporate rescue and insolvency practice in Nigeria. Thus, the essence of this chapter is to examine the issues and challenges in corporate rescue and insolvency practice in Nigeria.

5.1 Multifarious Statutory Framework

The primary Statute governing corporate rescue and insolvency law and practice in Nigeria is the Companies and Allied Matters Act.¹⁴²⁸ In the CAMA, copious provisions were made on corporate insolvency.¹⁴²⁹ However, the procedure on corporate insolvency as provided in the CAMA seems out-dated and not in conformity with global trends. Hence, there have been urgent suggestions for the review of the CAMA so as to bring it in conformity with global trends.¹⁴³⁰

Furthermore, the inadequacy of statutory framework as relating to corporate insolvency in Nigeria had been attributed to the fact that there are no law specifically

¹⁴²⁸ Cap C 20 Laws of the Federation of Nigeria 2004 (CAMA).

¹⁴²⁹*Ibid*, part XV, chapters 1, 2 and 5; Companies Winding up Rules 2001(CWR).

¹⁴³⁰ O Abayomi-Olukunle 'Reforms, Revolutions and the Ministry of Trade & Investment: Amendment of the Companies and Allied Matters Act', The Lawyer, a ThisDay Weekly Pullout13.09. 2016,p. 12.

dedicated to the cause of corporate insolvency.¹⁴³¹ Rather, there are disparate laws on corporate insolvency as contained in other statutes.¹⁴³² This explains the seeming excitement that stakeholders felt when the 8th Senate of the Federal Republic of Nigeria passed into law the Bill,¹⁴³³ which Bill purports amongst other things to be for the rehabilitation of businesses.¹⁴³⁴ However, in this study, it is respectfully submitted that the Bill appears not helpful to the entrenchment of orderly and effective corporate rescue and insolvency law and practice in Nigeria. It has been rightly observed that the reference to the Dollar instead of the Naira as the currency of choice in the Bill¹⁴³⁵ smacks of bad piece of Legislation.¹⁴³⁶It is difficult to decipher the rationale for the adoption of a foreign currency in the Bill.

In this study, it is respectfully submitted that there is no specific Law on corporate rescue and insolvency in Nigeria and the provisions of the CAMA as it relates to corporate rescue and insolvency are out-dated and at best, inadequate. The available formal corporate rescue option is Arrangement and Compromise¹⁴³⁷ under Part XVI of the CAMA, which option seems to be hardly used.¹⁴³⁸ It is pertinent to state that since there cannot be a single corporate rescue framework or option that holds all the answers to the rehabilitation of distressed companies, the available corporate rescue option of arrangement and compromise only in the CAMA is quite

¹⁴³⁵The Bill, *op cit*, ss. 30(1)(c), 106.

¹⁴³¹ CA Candide-Johnson and A Alex- Adedipe, 'Debt Recovery: Winding up and other Arrangement' in O Olanipekun(ed), *Banking: Theory, Regulation, Law and Practice* (Lagos; Au Curant, 2016) p.433 ¹⁴³²Asset Management Corporation of Nigeria Act 2010 (as amended), ss.6(1)(e) (l), 48 & 52(1); Investment and Securities Act 2007 (ISA), ss.197(1), 198(a), 204(a) & 212(b) & Banks and other Financial Institutions Act, Revised Edition, Laws of the Federation of Nigeria (as amended), ss.35(1)(c), 40 & 55.

¹⁴³³ Bankruptcy and Insolvency (Repeal and Re-Enactment) Bill 2015("the Bill").

¹⁴³⁴ S U Mba, 'Recession Preventive Rehabilitation and the Nigerian Draft Insolvency Bill: Lessons from the EU? LAWYER, a ThisDay Weekly Pull-out 06.09.2016,p. 7.

¹⁴³⁶ Mba, *loc cit*.

¹⁴³⁷ Candide-Johnson &Alex-Adedipe, *op cit*, p.453; Mba, *op cit*; B Adebola, 'The Nigerian Business Rescue Model: An Introduction'(2014) NIALS Journal of Legal Studies, 9; also available at <u>http://papers.ssrn.com/sol3/papers.cfm?abstract_id=2297824</u>; B Adebola, 'A Law of Insolvency in Nigeria: A Critical Analysis of the Present Legal Framework', Bripan-NIALS Round-Table, May 2013, p. 8.

¹⁴³⁸ Mba, *ibid*.

inadequate. Worse still, the procedure for arrangement and compromise as provided in the CAMA had been lampooned as absurd, slow and dysfunctional.¹⁴³⁹

Unfortunately, the Bill did not address or deal with the cause of corporate rescue. Suffice to state that beyond the title and preamble of the Bill, the cause of corporate insolvency and rescue were not within the purview of the 269 sections of the Bill. It is instructive to note that only the Bankruptcy Act was repealed.¹⁴⁴⁰Hence, it is difficult to understand the basis for introduction of insolvency matters in the Bill when there have never been any prior insolvency legislation, except as provided in the CAMA.¹⁴⁴¹

Unlike the position in Nigeria, in other common law countries such as the United Kingdom, the Republic of India and the Republic of South Africa, there are statutes specifically enacted and dedicated to the cause of corporate rescue and insolvency. A glaring and peculiar feature of the corporate rescue mechanisms as provided in the legislation under review is the spirited effort to enact home grown and indigenous corporate rescue mechanisms that are genuinely focused on the rehabilitation of distressed companies. Thus, the inadequate and disparate statutory and legal frameworks on corporate insolvency and rescue do not only obfuscate the ease of doing business in Nigeria but constitutes a challenge to the rescue of insolvent companies.

5.2 Paucity of Corporate Rescue Mechanisms

The global trend is to develop and allow for the development and survival of companies in distress, to seek to "rescue" a defaulting company rather than winding

 ¹⁴³⁹ Adebola, The Nigerian Business Rescue Model, *op cit*, p. 18.
 ¹⁴⁴⁰The Bill, *op cit*, s. 269.

¹⁴⁴¹ CAMA, *op cit*, ss. 408(d), 409, 567(1); Candide-Johnson & Alex-Adedipe, *op cit*, p. 433.

it up.¹⁴⁴² It has been correctly stated that the focus of corporate insolvency in modern day law, deviates from the previous trend of dissolution through winding up of insolvent companies and currently focuses on restructuring of insolvent companies to allow for recovery and continuity in place of dissolutions.¹⁴⁴³Notwithstanding the imperative for viable corporate rescue mechanisms, the only available formal remedy to encourage business recovery under the extant law in Nigeria is the procedure for arrangement and compromise.¹⁴⁴⁴

It is important to state that corporate rescue mechanisms that provide the means by which a company can restructure itself or its business to avoid ultimate failure is not same as corporate restructuring through mergers, amalgamation, take – overs and acquisitions, purchase and assumption and cherry picking.¹⁴⁴⁵ While corporate restructuring mechanisms through mergers, take - over and acquisitions may incidentally rescue distressed companies, the procedural steps and costs for such corporate restructuring mechanisms as provided in the extant legislations¹⁴⁴⁶ do not make such restructuring mechanisms to be readily conducive to rescue distressed companies.¹⁴⁴⁷ However, in spite of the shortcomings, merger or acquisition and

¹⁴⁴² O Ajayi, Legal Aspects of Finance in Emerging Markets (London: LexisNexis Butterworth Tolley, 2005) p.603.

¹⁴⁴³ Candide-Johnson & Alex-Adedipe, op cit, p.436; PL Davies & S Washington, Gower and Davies Principles of Modern Company Law (9th edn, London: Sweet and Maxwell, 2012) p.1272.

¹⁴⁴⁴ Candide-Johnson & Alex- Adedipe, *ibid*; Adebola, The Nigerian Business Rescue Model, op cit, p.6. ¹⁴⁴⁵Adebola, The Nigerian Business Rescue Model, *op cit*, p.6.

¹⁴⁴⁶Investment and Securities Act 2007, part XII; Securities and Exchange Commission (SEC) Rules and Regulation 2013, part 1, regulation 421 - 432 (Mergers), 433 - 439 (Acquisition), 445 - 448 (Take – over);SEC Amendment Rules 2015 ¹⁴⁴⁷ ISA, *op cit*, schedule1, part C(8). The processing fee for Schemes of Merger, Acquisition and

Take – Over, which fee is different from the professional fee to be paid to the solicitor for handling the brief from the application stage to the sanction of the court cannot be corporate rescue "friendly".

other informal rescue mechanisms have been common options available to rescue ailing companies.¹⁴⁴⁸

When juxtaposed with the position in some other jurisdictions such as the United Kingdom, the Republic of India and the Republic of South Africa, the available corporate rescue mechanisms in Nigeria, whether arrangement and compromise or receivership or merger or acquisition, pale into insignificance. In the United Kingdom, in addition to the corporate rescue mechanisms of scheme of arrangements,¹⁴⁴⁹ receivership¹⁴⁵⁰ administrative receiver,¹⁴⁵¹ are company voluntary arrangements (CVA's)¹⁴⁵² and administration¹⁴⁵³ with moratorium.¹⁴⁵⁴ In the Republic of India, corporate rescue mechanism of corporate insolvency resolution process¹⁴⁵⁵ and fast track corporate insolvency resolution process are not only coupled with moratorium¹⁴⁵⁶ but a mandatory process to be exhausted before the initiation of liquidation,¹⁴⁵⁷ if at all, against insolvent companies. In the Republic of South Africa, there is a dedicated corporate rescue mechanism in the form of business rescue proceedings¹⁴⁵⁸ that is coupled with a moratorium.¹⁴⁵⁹ In the United States of America, the Chapter 11 proceedings are a veritable and available corporate rescue mechanism.

¹⁴⁴⁸ JEO Abugu, *Company Securities: Law and Practice* (2nd edn, Lagos: MIJ Professional Publishers limited, 2014) p.303;HY Bhadmus, *Corporate Insolvency Law and Practice in Nigeria* (Enugu: Chenglo Limited, 2009) p.48.

¹⁴⁴⁹ Companies Act 2006, part 26(UK); Davies & Worthington, op cit, pp. 1105-1113.

¹⁴⁵⁰ Insolvency Act 1986, as amended, part III (UK).

¹⁴⁵¹*Ibid.* ss. 72A to 72G, schedule 2A.

¹⁴⁵²*Ibid*, part 1; Enterprise Act 2002, schedule 17, paras 10,11,12 (UK)

¹⁴⁵³Insolvency Act, *ibid*, schedule B1; Enterprise Act, *ibid*, schedule 17.

¹⁴⁵⁴ Insolvency Act, *ibid*, schedule B1, paras 42, 43.

¹⁴⁵⁵ Insolvency and Bankruptcy Code 2016, chapter II (India)

¹⁴⁵⁶Insolvency Act, op cit, chapter IV, ss. 13(i)(a), 14.

¹⁴⁵⁷ Insolvency and Bankruptcy Code, *op cit*, chapter III,s. 33.

¹⁴⁵⁸ Companies Act 2008, chapter 6 (South Africa)

¹⁴⁵⁹*Ibid*, s.133; Companies Amendment Act 2011(CAA) (South Africa), s.84 (a) (b) (c).

Flowing from the above, it appears incontrovertible that the paucity of corporate rescue mechanisms in Nigeria poses a big challenge in the quest for rescue of insolvent companies.

5.3 Absence of Efficient and Pragmatic Institutions

While the enactment of good and functional statutory framework appears to be desiderata for corporate rescue and insolvency practice, an indispensable prerequisite is also the existence of pragmatic, functional, dynamic and capable regulatory institutions. In this vein, the role of the court, particularly the Federal High Court in corporate rescue and insolvency proceedings cannot be over-looked.

The prominent institutions that regulate the cause of corporate rescue and insolvency in Nigeria are the Corporate Affairs Commission (CAC) and the Securities and Exchange Commission. By virtue of the provisions of the CAMA, the CAC is vested with far-reaching functions and powers to administer the provisions of the CAMA, including corporate insolvency.¹⁴⁶⁰ In this respect and in this study, it is submitted that the CAC seems to lack the capacity¹⁴⁶¹ to combine its functions of regulation and supervision of the incorporation, registration and management of companies with that of regulating the cause of corporate rescue and insolvency.

Unlike the position in Nigeria, in other countries such as the United Kingdom and the Republic of India, there is a distinct body that regulates the cause of corporate rescue and insolvency. In the United Kingdom, the Insolvency Service, an Executive Agency of the Department for Business Innovation & Skills (BIS),¹⁴⁶² regulates insolvency matters. In the Republic of India, the Insolvency and

¹⁴⁶⁰CAMA, *op cit*, s. 7 (1) (a).

¹⁴⁶¹Lack of funds not unconnected to the economy now in recession; reduction of budgetary allocation to statutory bodies including the CAC; insufficient personnel ¹⁴⁶² Davis & Worthington, *op cit*, p.1276.

Bankruptcy Board of India was established¹⁴⁶³ to exercise powers and functions, including the regulation of insolvency and bankruptcy matters.¹⁴⁶⁴

Without gainsaying, it is obvious that the CAC and the SEC do not possess the requisite capacity and expertise to carry out their routine statutory functions and also regulate the cause of corporate rescue and insolvency.

The court is another institution at the hub of regulation, even peripherally, of corporate rescue and insolvency proceedings in Nigeria. Instructively, the Federal High Court is vested with jurisdiction to the exclusion of any other court in respect of civil causes and matters arising from the operation of the CAMA¹⁴⁶⁵ and bankruptcy and insolvency.¹⁴⁶⁶ Regrettably, the role of the court in corporate rescue and insolvency matters has not been laudable. A situation where the determination of a winding up petition lasts up to eight to eleven years¹⁴⁶⁷ is unsatisfactory, regrettable and constitutes an albatross to the effectual dispensation or determination of corporate insolvency matters. Worst still, a situation where a motion for leave to advertise Petition lasts up to four years¹⁴⁶⁸ and suffers unnecessary adjournments is condemnable.

Furthermore, the role of the court in matters of corporate rescue and insolvency is not immune from the malaise associated with the Nigerian court system¹⁴⁶⁹ such as

¹⁴⁶³ Insolvency and Bankruptcy Code, *op cit*, s. 188(1).

¹⁴⁶⁴*Ibid*, s. 196.

¹⁴⁶⁵ Constitution of the Federal Republic of Nigeria, s. 251 (1)(e).

¹⁴⁶⁶*Ibid*, s. 251(1)(j).

¹⁴⁶⁷Air Via Limited v Oriental Airlines Limited(2004) 4 SC (pt 11) 37.

¹⁴⁶⁸*Pharma – Deko Plc v F.D.C Ltd* [2015] 10 N.W.L.R. (pt 1467)225 - the winding up petition was filed on 4th June 1993 at the Federal High Court and finally determined by the Supreme Court of Nigeria on the 23rd of April 2004, a period of about 10 years.

¹⁴⁶⁹G C Nwakoby, *The Law and Practice of Commercial Arbitration in Nigeria*(2nd edn, Enugu: Snapp Press Ltd, 2014) pp. 467–468; Nigeria: 'Publication of Financial Sector Assessment Program Documentation-Technical Note on Crisis Management and Crisis Preparedness' (May 2013 IMF Country Report No.13/143) pp.33 & 38.

slow administration of justice, incessant adjournments, ignorance, incapacity, slow proceedings and rigidity.¹⁴⁷⁰

5.4 Absence of Moratorium Provisions

One of the foremost challenges that beset the cause of corporate rescue mechanisms in Nigeria is the absence of moratorium¹⁴⁷¹ provisions in the extant legislation providing for rescue of insolvent companies.

As already examined in the preceding aspects of this study, the seeming available corporate rescue options under extant statute in Nigeria are arrangement and compromise,¹⁴⁷² receivership¹⁴⁷³ and merger or acquisition.¹⁴⁷⁴

The declaration of a moratorium is one of the most important requisites for a viable corporate rescue. Hence, most of the legislation providing for corporate rescue in different variants have entrenched a moratorium clause.¹⁴⁷⁵ The salient effect of a moratorium is that it amongst others and from the initiation of a corporate rescue mechanism until the sooner determination of same, set to prohibit the institution of suits or continuation of pending suits or proceedings against the debtor company including execution of any judgment, decree or order or prohibits the taking of any action or step to foreclose, recover or enforce any security interest created by the

¹⁴⁷⁰Abugu, op cit, p. 307.

¹⁴⁷¹SU Mba, 'Rethinking Business Rescue in Nigeria: Borrowing Virtues from Chapter 11 of the US Bankruptcy Code' Unpublished LLM Thesis, Central European University, Budapest, Hungary, March 2015, p.43. Moratorium is the temporary suspension of legal proceedings, actions or claims against an insolvent company during rescue proceedings.

¹⁴⁷²CAMA *op cit*, part XVI, ss. 537,539, 540; Candide-Johnson & Alex –Adedipe, *op cit*, p.453; Abugu, *op cit*, pp. 304–308; A Abdulai, 'Mergers, Acquisitions and Take-overs in Nigeria: The Guiding Philosophy, Laws, Regulations and Practice', June 1997 Vol 2 No 2 MILBQ, 33 .The challenge posed by the lack of capacity of regulatory institutions, particularly the Federal High Court can be surmised thus: Of what value is the Federal High Court's involvement in take-over bids, mergers, etc, in view of the fact that this court does not have sufficient expertise with regard to these kinds of truncations? Is it true that reference to Federal High Court has in the past made merger exercises more expensive and frustrating to all concerned?

¹⁴⁷³ CAMA, *ibid*, part XIV; O O Oladele 'The Legal Intricacies of Corporate Restructuring and Rescue in Nigeria', The Calabar Law Journal Volume XI ISSN 0189.0020, 2007, 153.

¹⁴⁷⁴ ISA, op cit, part XII; SEC Rules, op cit, part 1; Abugu, op cit, p.303.

¹⁴⁷⁵ Companies Act, *op cit*, s. 133 (South Africa); CAA, *op cit*, s. 84 (South Africa); The Insolvency and Bankruptcy Code, *op cit*, ss.13(1) (a), 14; Insolvency Act, *op cit*, schedule B1, paras 42 & 43(UK).

debtor company.¹⁴⁷⁶ It is important to state that the declaration of a moratorium under a corporate rescue mechanism can be absolute or subject to exceptions as may be provided in the enabling statute. However, the usual practice is for moratorium provisions to be subject to exceptions. In other words, some of the matters that are under moratorium can be invoked with the permission of the court,¹⁴⁷⁷ or with the consent of the insolvency practitioner.¹⁴⁷⁸

Notwithstanding the obvious desideratum for a moratorium provisions so as to ensure a viable corporate rescue procedure, it is unfortunate that moratorium declarations are manifestly absent in the corporate rescue options in Nigeria such as arrangements and compromise, receivership and merger or acquisition. Hence, against the challenge posed by the absence of moratorium provisions in the extant legislation on corporate rescue, the CAC is in the fore front for a review of the CAMA to provide for Business Rescue and Recovery Proceedings that will give temporary moratorium to financially distressed companies.¹⁴⁷⁹

5.5 Absence of Minimum Qualification for and/or Regulation of Corporate Rescue and Insolvency Practitioners

The drivers of any corporate rescue and insolvency system are the practitioners. Corporate rescue or insolvency practitioner is a person appointed to conduct the corporate rescue or insolvency process. In the United Kingdom, a person who is appointed to or acts as a liquidator, provisional liquidator, administrator or administrative receiver or as a supervisor of a company voluntary Arrangement is

¹⁴⁷⁶ Ian F Fletcher, *The Law ofInsolvency* (2nd edn, London: Sweet & Maxwell, 1996) pp. 435–437; JH Farrar & B Hannigan, *Farrah's Company Law* (4th edn, London: LexisNexis Butterworths Tolley, 1998) p.687; SW Mayson, *et al, Mayson, French & Ryan on Company Law* (25th edn, Oxford: Oxford University Press, 2008-2009) p. 654.

¹⁴⁷⁷ Companies Act, *op cit*, s. 133 (South Africa); Insolvency Act, *op cit*, schedule B1, para43(2)(b). ¹⁴⁷⁸ Companies Act, *ibid*; Insolvency Act, *ibid*.

¹⁴⁷⁹Mahmud, B, Report of the Corporate Affairs Commission to the 2016 Annual General Conference of the Nigeria Bar Association at Port Harcourt, Nigeria from 19th to the 2th day of August 2016, p. 6.

known as an "insolvency practitioner".¹⁴⁸⁰ In the Republic of India, a person that conducts corporate insolvency matters is generally known as an "insolvency professional".¹⁴⁸¹ Where the insolvency professional is acting to rescue a financially distressed company, such a person is known as a "resolution professional",¹⁴⁸² or an interim-resolution professional.¹⁴⁸³ In the Republic of South Africa, a person appointed to oversee a company during business rescue proceedings is known as a "business rescue practitioner".¹⁴⁸⁴ In Nigeria, the drivers of the corporate rescue or insolvency process are the receivers and/or managers for receivership¹⁴⁸⁵ and a liquidator, whether provisional or otherwise,¹⁴⁸⁶ in the case of winding up.

Because of the prominent role played by persons appointed or acting as receivers and/or managers or provisional liquidator or liquidator in corporate rescue or insolvency matters, the global trend is the stipulation of certain requisite minimum qualification for persons to be appointed or to act as receivers and/or managers or provisional liquidator or liquidator. Although, the usual practice is the prescription in the various legislation of persons who are disqualified from being appointed or prevented from acting as a receiver and/or manager,¹⁴⁸⁷ or liquidator whether as provisional or otherwise.¹⁴⁸⁸ In the Republic of South Africa, it should be noted that in spite of the repeal of the Companies Act,¹⁴⁸⁹ by reason of section 224(1), schedule

¹⁴⁸⁰ Insolvency Act, *op cit*, s. 388(1).

¹⁴⁸¹ Insolvency and Bankruptcy Code, *op cit*, s. 3(19).

¹⁴⁸²*Ibid*, s. 5(27).

¹⁴⁸³Insolvency and Bankruptcy Code, op cit.

¹⁴⁸⁴ Companies Act, *op cit*, s. 128(1)(d)(South Africa).

¹⁴⁸⁵ CAMA, op cit, part XIV, ss. 389, 390.

¹⁴⁸⁶ Ibid, ss. 422-432, 453-454; CWR, op cit, rr. 21,41; A Aliyu and MA Saulawa, 'The Role of a Liquidator in Winding up Proceedings in Nigeria: Examining The Inadequacies under Companies and Allied Matters Act (CAMA)', PHJBL 2.1 (2016), 225-236.

¹⁴⁸⁷ CAMA, *ibid*, s. 387(1); Insolvency Act, *op cit*, ss. 30, 31.

¹⁴⁸⁸ CAMA, *ibid*, s. 509(1); Insolvency Act, *ibid*, ss. 388(1), 390 (UK) - for liquidator or administrator or administrative receiver; Companies Act 1973, ss. 372, 373 (South Africa). ¹⁴⁸⁹No 61 of 1973.

5, item 9 of the Companies Act,¹⁴⁹⁰ it did not apply to proceedings in respect of winding up or liquidation of companies or the appointment of business rescue practitioner.¹⁴⁹¹

The prevailing practice is for the extant legislation to provide for the regulation of corporate rescue and insolvency practitioners. In other words, a person is not appointed or acts as a practitioner in corporate rescue and insolvency matters merely because he or she is not disqualified under the extant statute. In the United Kingdom, a person can only be appointed or act as insolvency practitioner if he or she is qualified or meets with further requisites.¹⁴⁹² Such further requisite is that for a person to be an insolvency professional or carry out any insolvency related work in the UK, he must pass the practical examination of the Joint Examination Insolvency Board (JIEB).¹⁴⁹³ In the Republic of India, a person can only render services as insolvency professional if he or she has enrolled as a member of an insolvency professional agency and thereafter register with the insolvency and Bankruptcy Board of India.¹⁴⁹⁴In the Republic of South Africa, a person can only be appointed as a business management profession accredited by the Companies and Intellectual Property Commission of the Republic of South Africa.¹⁴⁹⁵

In a nutshell, the global trend and requisites for a person to be appointed or act as a corporate rescue and insolvency practitioner is that he or she must be a member

¹⁴⁹⁰No 71 of 2008.

¹⁴⁹¹ Companies Act, *op cit*, s. 138(1); CAA, *op cit*, s. 88 (South Africa).

¹⁴⁹² Insolvency Act, *op cit*, ss. 390(2), 391; Small Business, Enterprise and Employment Act 2015, ss. 137–146(UK); Mayson, et al, *op cit*, pp. 640–641. A person can only act as an insolvency practitioner if he is authorised to so act by virtue of membership of a recognised professional body or by the authorisation of the Secretary of State; having being satisfied that he is a fit and proper person so to act and meets acceptable requirements as to education and practical training and experience.

¹⁴⁹³<u>http://www.icaew.com/en/technical/insovency/qualifying-as-an-insolvency-practitioner</u> accessed on 19 November 2016; Insolvency Practitioners Association, Insolvency Code of Ethics; Background and Overview (November 2008).

¹⁴⁹⁴Insolvency and Bankruptcy Code, *op cit*, s. 207.

¹⁴⁹⁵Companies Act, op cit, s. 138 (1) (a); CAA, op cit, s. 88.

of recognised or licensed professional body or an insolvency professional agency. Furthermore, persons who are appointed or act as corporate rescue and insolvency practitioners are subject to regulation.¹⁴⁹⁶

The rationale for further requisites and the regulation of persons appointed to or act as corporate rescue and insolvency practitioners cannot be over-emphasized.¹⁴⁹⁷ The practice is not only intended to restrain malpractice, abuse but to promote a stringent review by competent and qualified corporate rescue and insolvency practitioners of the distressed or ailing companies.¹⁴⁹⁸

Unfortunately, in Nigeria a person can be appointed or act as a receiver and/or manager, liquidator or provisional liquidator in corporate rescue and insolvency matters if he is not disqualified under the extant provisions in the CAMA. In other words, for a person to practise as a corporate rescue and insolvency practitioner in Nigeria, he or she does not require any minimum educational qualification or requisite knowledge and experience to discharge the functions of a corporate rescue and insolvency practitioner. The situation where to practise as a corporate rescue and insolvency practitioner is unregulated is not only regrettable but absolutely unsatisfactory.¹⁴⁹⁹

5.6 Corruption and other Sharp Practices

¹⁴⁹⁶Insolvency and Bankruptcy Code, *op cit*, s. 204; Insolvency Act, *op cit*, s. 391; Small Business, Enterprise and Employment Act, *op cit*, s. 144. The recognised professional body regulates the members acting as insolvency practitioners, pending when the Secretary of State establishes a single regulator of insolvency practitioners pursuant to the Small Business, Enterprise and Employment Act. An insolvency professional agency regulates the members enrolled with it as insolvency professionals. ¹⁴⁹⁷Bhadmus, *op cit*, p.304

¹⁴⁹⁸ S Barber, *Company Law* (4th edn, London: Old Bailey Press, 2003) p. 324.

¹⁴⁹⁹ U Melah, 'AMCON, DELTA STEEL AND ARIBISALA's BASIC INSTINCT', This Day, The Saturday Newspaper, July 4.2015. p.48. This article exemplified the abuse and malpractice that can be occasioned by having a corporate rescue and insolvency system where the practitioners are not subject to regulation

The concept of corruption is not subject to comprehensive, precise or concise definition.¹⁵⁰⁰ However, the word "corruption" has been used to describe conduct that reflects abuse of public office for private gain.¹⁵⁰¹ It should be noted that public office is abused for private gain when an official accepts, solicits or extorts a bribe. Private agents actively offer bribes to circumvent public policies and processes for competitive advantage and profit.¹⁵⁰² It is therefore pertinent to state that public office can be abused for personal benefit without bribery, through patronage and nepotism, the theft of state assets or the diversion of state resources.¹⁵⁰³

It may appear from the above descriptions of corruption that corrupt practices are prevalent only in the public sector, but this is not correct. Corrupt practices can occur in the private sector.¹⁵⁰⁴ Hence, corruption has been described as the 'abuse of entrusted power for private gain.¹⁵⁰⁵ In the same vein, corruption has been described as including 'bribery, fraud and other related offences.¹⁵⁰⁶ Furthermore, the definition of official corruption was not limited to only the public sector, but mentions official duties or in relation to any matter connected with the functions, affairs or business of a government department or corporate body or other organisation or institution in which the person is an official.¹⁵⁰⁷

¹⁵⁰⁰ CC Ani, 'Corruption in Criminal Justice Administration in Nigeria: The Role of the Legal Profession', Nigeria Bar Journal, Vol 7, 2011, 110.

¹⁵⁰¹G Etomi, An Introduction to Nigerian Commercial Law: Text, Cases and Materials (Lagos: MIJ Professional Publishers Limited, 2014) p. 450; OECD Convention on Combating Bribery of Foreign Public Officials in International Business Transactions; United Nations Convention against Corruption.

¹⁵⁰² K Yusuf 'Financial Crime' in O Olanipekun, Banking: Theory, Regulation, Law & Practice (Lagos: AU Courant, 2016) p.707. ¹⁵⁰³*Ibid*.

¹⁵⁰⁴ NS, Okogbule, 'An Appraisal of the Legal and Institutional Framework for Combating Corruption in Nigeria', Journal of Finance Crime Volume 13 Number 1 2006, 96; Etomi, op cit, p.457. ¹⁵⁰⁵NS Okogbule, art cit.

¹⁵⁰⁶Corrupt Practices and other Related Offences Act, 2000, Cap C31Laws of the Federation of Nigeria (LFN) 2004 (as amended), s. 2.

Corrupt Practices and other Related Offences Act, op cit, s. 8; C Wigwe, 'Corruption as an Economic Barrier to Development in Third World Countries' PHJBL 2.1(2016),328.

Notwithstanding the difficulty in defining corruption, there is consensus on the effects or impact of corruption on economic and socio-political development of any nation.¹⁵⁰⁸ It is said that corruption makes business expensive, uncompetitive and results in negative return on investments,¹⁵⁰⁹ subversion of fundamental rules of economic relations, institution and structure, good governance, an equitable and just legal order and other institutions of state, particularly the Judiciary.¹⁵¹⁰

In consideration of the prominent role of the court in causes of corporate rescue and insolvency in Nigeria, namely, summoning meeting of the creditors, or class of creditors, or of the members of the company or class of creditors, or of the members of the company or class of members in respect of compromise or arrangement;¹⁵¹¹ sanctions compromise or arrangement;¹⁵¹² makes a winding up order in respect of compulsory winding up,¹⁵¹³ an issue of corruption as it affects the court must be of interest. Instructively, it has been opined that:-

> where a state is characterised by corruption endemic, it is most manifest in the judicial sector as legal disputes emanating in the course of economic relationships are increasingly resolved in favour of the party who is able bid the highest for the conscience of the judges rather than by established laws and judicial proceedings.¹⁵¹⁴

It is a fact, albeit in the realm of conjecture, that the judiciary in Nigeria is riddled with corruption or corrupt acts. There are incidences of case files been

¹⁵⁰⁸ Etomi, op cit, p.453; Ekwenugo v Federal Republic of Nigeria [2001] 6N.W.L.R (pt 703) 187.

¹⁵⁰⁹ O, Ajayi, *Legal Aspects of Finance in Emerging Markets* (London: LexisNexis Butterworths Tolley, 2005) p.120.

¹⁵¹⁰ Wigwe, art cit, 329.

¹⁵¹¹ CAMA, *op cit*, s. 539(1); Abugu, *op cit*, p.307.

¹⁵¹²CAMA, *ibid*, s. 539 (3).

¹⁵¹³*Ibid, Part xv, chapter 2, s. 407 (1).*

¹⁵¹⁴ Wigwe, art cit, 329

hidden, officials of the court, whether clerk or bailiffs or registrars, demanding and receiving bribes or other forms of gratification so as to manipulate the process to the advantage of the giver, perversion of justice by judicial officers as a result of monetary inducements, nepotism or external influence.

In order to combat corruption that has been referred to as a 'virus',¹⁵¹⁵ a good number of domestic and international initiatives and legislations¹⁵¹⁶ have been developed and enacted to tackle corruption.¹⁵¹⁷ Notwithstanding the efforts being made to combat corruption, it still remains a cankerworm and a challenge to sustainable economic and socio-political development in Nigeria.

Another challenge to corporate rescue and insolvency practice in Nigeria is the issue of corporate fraud.¹⁵¹⁸ It should be noted that there is no general offence of fraud under Nigerian criminal law.¹⁵¹⁹ However, there are a number of offences linked to the concept of fraud, including stealing,¹⁵²⁰ obtaining by false pretences,¹⁵²¹ false accounting, and false statement by directors with intent to deceive members or

¹⁵¹⁵ E Chianu, *Company Law* (Abuja: Law Lords Publications, 2012) p. 317

¹⁵¹⁶Corrupt Practices and Related Offences Act, *op cit*; Economic and Financial Crimes Commission (EFCC) Act 2004 (as amended) ; Advance Fee Fraud and Other Related Offences Act 2006; Money Laundering (Prohibition) Act 2011; Foreign Corrupt Practices Act 1977(as amended) (US); Bribery Act 2010 (UK); Fraud Act 2006 (UK); Criminal Code Act, Cap C 38, Laws of the Federation of Nigeria 2004; Criminal Code Laws of different States ; Penal Code; United Nations Convention Against Corruption (General Assembly Resolution 58/4 of 31 October 2003): African Union Convention on the Preventing and Combating of Corruption (adopted in 2003); ECOWAS Protocol on the Fight against Corruption(not yet in force)

¹⁵¹⁷ Ajayi, *op cit*, p.120.

¹⁵¹⁸*Ibid*, pp. 211, 212.

¹⁵¹⁹ Yusuf, *op cit*, p.175.

¹⁵²⁰ Criminal Code Act, op cit, s. 390

¹⁵²¹*Ibid*,s.419.

creditors,¹⁵²² fraudulent trading.¹⁵²³ It is instructive to note that the issue of corporate fraud has been linked informally with the cause of corporate failure.¹⁵²⁴

The inability to define the word "fraud" as it is a term of so wide an import remains a challenge in addressing the cause of corporate rescue and insolvency law and practice. However, it has been held that any act which may amount to an infraction of fair dealing, or abuse of confidence, or unconscionable conduct, or abuse of power as between a trustee and his shareholders in the management of a company, constitutes fraud.¹⁵²⁵ On the other hand, it should be noted that "blameworthy irresponsibility" is not sufficient to establish fraud. A deliberate intent to defraud must be established¹⁵²⁶ in all cases of fraud. Thus, the concept of fraud involves actual dishonesty.¹⁵²⁷ Nevertheless, the standard of proof required to establish fraud still remains ambiguous, ambivalent and unsettled. This is not unconnected to the challenge arising from not having a general offence of fraud under extant Nigerian Law. In most of the statutory offences on corporate fraud, the elements are not specifically and clearly spelt out.

5.7 Continuing Financing during Corporate Insolvency and Rescue Proceedings.

Continuing financing¹⁵²⁸ is fundamental to the success of any corporate rescue plan as well as a hindrance when it is in short supply. Without an established provision which clearly outlines prospective post-petition financing, that is, funds needed by

¹⁵²²J Ulph, *Commercial Fraud: Civil Liability, Human Rights and Money Laundering* (Oxford: Oxford University Press, 2006) p.26.

¹⁵²³CAMA, *op cit*, s. 506.

¹⁵²⁴ JEO Abugu, *Principles of Corporate Law in Nigeria*(2nd edn, Lagos: MU Professional Publishers Limited, 2014) p. 677; Farrar' *op cit*, p. 622.

 ¹⁵²⁵Yalaju- Amaye v Associated Registered Engineering Contractors Ltd[1990] 4 N.W.L.R (pt 145)
 422 at 466; Virgin Technologies Ltd v Mohammed[2009] 11 N.W.L.R (pt 1151)136 at 150, 153.
 ¹⁵²⁶ Chianu, op cit, p.313.

¹⁵²⁷*Ibid*.

¹⁵²⁸It is the continuing financing mechanisms, schemes and options for insolvent companies during rescue proceedings.

the debtor company to enable it to continue business during the rescue period and how these new creditors who often demand priority payments over pre-existing creditors will fit into the debtor's repayment plans, efforts to effectively rescue an insolvent company may amount to an exercise in futility.¹⁵²⁹

Continuing financing in corporate rescue proceedings has remained a seminal issue in corporate rescue and insolvency law and practice.¹⁵³⁰ While in other jurisdictions such as the UK, the US and the Republic of India, there seems to be commendable efforts towards providing for continuing financing during corporate rescues, in Nigeria, there are no formal continuing financing schemes for corporate entities at times of corporate insolvency or rescue. In the UK, the issue of continuing financing is addressed by the powers conferred on an administrator to borrow funds, grant security,¹⁵³¹ and prioritise repayment of debts owing under contracts entered during administration.¹⁵³² Although the powers conferred on an administrator to borrow are helpful as regards obtaining funds and continuing the business, it nevertheless lacked the incentives required to encourage fresh lending.

In the US, continuing financing during Chapter 11 corporate reorganization is achieved by the new priority financing, debtor-in-possession funding.¹⁵³³ In this vein, and in order to incentivise creditors, there are a hierarchy of claims in the Bankruptcy Code so as to ensure creditors are more amenable to the idea of advancing credits to

¹⁵²⁹ A Aruoriwo, 'Financing Corporate Rescues, Where Does the UK Stand?' (2014) 1(2) IALS Student Law Review, 10.

¹⁵³⁰J Armour & A Walters, 'Funding Liquidation: A Functional View' (2006) LQR 295; G McCormack, 'Super-Priority Financing and Corporate Rescue' (2007) J.B.L, 701-732. ¹⁵³¹Insolvency Act 1986 (as amended), sch 1, para 14, 15 (UK).

¹⁵³²Ibid, sch B1, para 99; Bibby Trade Finance Ltd v Mckay [2006] All ER 266; Frenkley v *Reinsurance International Co* [2006] BCC, 971. ¹⁵³³Bankruptcy Code 1978 (as amended), s.364 (US).

the corporate entity under reorganization. In the case of India, there is a Fund established to finance corporate insolvency resolution process.¹⁵³⁴

It is pertinent to state that the viability of any business is often dependent on the availability of credit and more so when the company is within the vicinity of insolvency. Hence, continuing financing is fundamental in corporate insolvency and rescue proceedings. In Nigeria, however, there is abysmal absence of this fundamental prerequisite of continuing finance in corporate insolvency and rescue proceedings under extant statute.

Meanwhile, this Study acknowledges the essence of the establishment of the Investor Protection Fund¹⁵³⁵ with the objective to compensate investors who suffer pecuniary loss arising from the insolvency, bankruptcy and negligence of a dealing member firm of a securities exchange or capital trade point.¹⁵³⁶

5.8 Absence of Uniform Guideline for Informal Corporate Rescue Procedures

It is common for informal corporate rescue options such as bailout, debt for equity swap, purchase and assumption and cherry picking to be invoked for resolution of insolvency matters in Nigeria. Despite the advantages¹⁵³⁷ of such informal corporate rescue options, it has not been impactful and enduring. This is not unconnected to the absence of clear, certain and transparent guidelines and procedures for such informal corporate rescue options. At present, the guidelines and procedures, if any, for informal corporate rescue options are disparate and sector-based. The end result of having disparate uniform guideline for informal corporate rescue proceedings in Nigeria is incongruity

¹⁵³⁴Insolvency and Bankruptcy Code 2016, s.224 (India).

¹⁵³⁵ISA, op cit, s. 197(1).

¹⁵³⁶ISA, *op cit*, ss. 198(a), 204(a) & 212(b); K Udofia, 'Contribution to Investors' Protection Fund and the Fraudulent Preference Risk', LAWYER, a This Day Weekly Pull-out, 18.04.2017, p.11.

¹⁵³⁷Voluntariness, Dynamism, Speedy Transaction, Cheap and Inclusiveness of parties.

and complexity with the potential for added cost and complication in the resolution of insolvency.

CHAPTER SIX

CONCLUSION AND RECOMMENDATIONS

6.1 Conclusion

One of the incidents of incorporation is that a company in furtherance of its business may borrow money by mortgage or charge its undertaking, property and uncalled capital, or any part thereof and issue debentures, debenture stock and other securities whether outright or as security for any debt, liability or obligation for any debt, liability or obligation of the company.

However, the situation may arise where a company may not be able to pay back borrowed capital and in such a circumstance, a creditor would be interested in the mechanism for enforcing or realising the security or borrowed capital. Where a company is unable to pay its debt, the company is deemed insolvent. Under the extant laws in Nigeria, the available procedures to address the cause of corporate insolvency are through the appointment of receiver and/or manager or the commencement of winding up proceedings.

Corporate insolvency proceedings by way of winding up is inexorably terminal and with irreversible consequences. The winding up of insolvent companies is not just a process leading to ultimate dissolution of a company but one that irreparably terminates the life of the insolvent company, throws the employees of the insolvent company into unemployment, causes societal and family breakdown and a remote trigger for the incidences of violence and disorientation in values.

It is against the background of the unviability of winding up petition in addressing the corporate insolvency, the resort to corporate rescue options becomes exigent. However, the formal corporate rescue options that are seemingly available in Nigeria are receivership, arrangements and compromise and mergers or acquisitions. Such corporate rescue options appear not to have impacted positively in the cause of rescuing insolvent companies. This is not unconnected to the fact that the corporate rescue options are hampered by complex procedure, high processing cost, the absence of moratorium provisions and the non-regulation of corporate rescue and insolvency practitioners have been propounded as the reason for the abuses, sharp practices, malpractices and unethical conducts and behaviour in corporate rescue and insolvency matters in Nigeria.

In view of the essence of a well-defined and viable corporate rescue and insolvency system for development of any economy and the constraints highlighted in this study, the need for reform of corporate rescue and insolvency law and practice in Nigeria is not only imperative but urgent.

The summary of findings of this Study are:

1. The principal statute governing corporate rescue and insolvency in Nigeria is the Companies and Allied Matters Act.¹⁵³⁸ Furthermore, it was found that the provisions of the CAMA are not robust and comprehensive to deal with the cause of corporate rescue and insolvency because CAMA enacted in 1990 is old and the provisions as relating to corporate rescue and insolvency are out-dated, obsolete and neither indigenous nor in line with international best practices. Also, the legislative¹⁵³⁹ efforts of the National Assembly to address corporate rescue and insolvency were found to be a sham as it did not provide for corporate rescue and insolvency at all.

2. Corporate rescue and insolvency provisions were also contained in some other specific statutes such as the Asset Management Corporation of Nigeria Act.¹⁵⁴⁰

¹⁵³⁸ Cap C20 Laws of the Federation of Nigeria 2004 (CAMA).

¹⁵³⁹ Bankruptcy and Insolvency (Repeal and Re- Enactment) Bill 2015.

¹⁵⁴⁰ 2010 as amended (AMCON Act).

Investment and Securities Act,¹⁵⁴¹ Bank and Other Financial Institutions Act,¹⁵⁴² Insurance Act,¹⁵⁴³ and Nigeria Deposit Insurance Corporation Act.¹⁵⁴⁴

3. The discernible procedure in Nigeria for winding up proceedings is mainly provided in the CAMA¹⁵⁴⁵ and the Companies Winding up Rules.¹⁵⁴⁶ Furthermore, it was found that winding up proceedings as a means to address the cause of corporate insolvency is not a viable option after all.¹⁵⁴⁷ This is not unconnected amongst others to the obfuscating court system.¹⁵⁴⁸

4. The available formal corporate rescue options that have been adopted or available to address issues of corporate insolvency in Nigeria are arrangements and compromise,¹⁵⁴⁹ receivership,¹⁵⁵⁰ mergers or acquisition,¹⁵⁵¹ and the amorphous bail outs through the vehicle of Assets Management Corporation of Nigeria. It was found that arrangements and compromise has not been an attractive rescue tool for insolvent companies.¹⁵⁵² The unviability of arrangements and compromise as a rescue option to insolvency have been linked amongst others to the absence of moratorium which

¹⁵⁴¹ 2007 (ISA).

¹⁵⁴² Revised Edition, Laws of the Federation of Nigeria 2007 as amended (BOFIA).

¹⁵⁴³ Cap 118 Laws of the Federation of Nigeria 2004 as amended.

¹⁵⁴⁴ 2006 as amended.

¹⁵⁴⁵ CAMA, *op cit*, part XV, chapters 2,5; Companies Regulation 2012, reg 43. ¹⁵⁴⁶ 2001.

¹⁵⁴⁷O Ajayi, Legal Aspects of Finance in Emerging Markets (London: LexisNexis Butterworth Tolley, 2005) p.603.

¹⁵⁴⁸ Congested cause lists; adjournments, delays and longer period of determination of a winding up from the Federal High Court to the Supreme Court of Nigeria.

¹⁵⁴⁹ CAMA, *op cit*, part XVI, ss. 537, 539, 540.

¹⁵⁵⁰Ibid, s. 209, part XIV; Companies Regulation, op cit, regs 36, 37, 41, 42.

¹⁵⁵¹ ISA, *op cit*, part XII, ss. 119–130; Securities and Exchange Commission Rules and Regulation 2013, (as amended as) part 1, rr. 421–432(mergers); 433–439 (Acquisition).

¹⁵⁵² Nigerian Law Reform Commission Report in the Reform of Nigerian Company Law and Related Matters Volume 1, Renew and Recommendation.1988, pp.321-322; B Adebola Conflated Arrangements; A comment on the company Voluntary Arrangements in the proposed Nigerian Insolvency Act 2014'(2015) 3 NIBLeJ 2 also available at http://www.4,ntu.ac.uk/n/s/document_uploads/174811-pdf accessed on 20 November 2016.B Adebola, 'The Nigerian Business Rescue Model: An Introduction'(2014) NIALS Journal of Legal Studies, 18; also available at http://papers.ssrn.com/sol3/papers.cfm?abstract id=2297824. accessed on 10 May 2015.

creates an opening for unsecured creditors to initiate parallel debt collection proceedings at a State High Court after the scheme is proposed.¹⁵⁵³

Furthermore, it was found that the impact of receivership as a corporate rescue option is limited. While receivership is mainly invoked to preserve, receive and protect the interest of a security holder in a company or manage a going concern for the purpose of the receivership, the appointment of a receiver and/or manager is usually restricted to secured charges.¹⁵⁵⁴ Although the court may invoke its power to grant an injunction or appoint a receiver by interlocutory order in all cases in which it appears just and convenient to do so,¹⁵⁵⁵ receivership even where successful, resolves the interests of the security holder only, ailing company remained comatose or in an ailing state, other security holders' interest still unresolved and the employees of the ailing company on the verge of losing their jobs.

It was further found that the rescue of failing banks as a result of the 2009 Nigerian Banking crisis was through some form of mergers and acquisition.¹⁵⁵⁶ However, mergers or acquisition appear not a feasible rescue option for insolvent companies that are not in strictly regulated sectors of the economy. This is not unconnected to the complex procedure for approval of mergers or acquisitions and the prescribed fees that is based on the value of the merging companies.¹⁵⁵⁷ Suffice to state that insolvent companies cannot afford to utilise corporate rescue option in form of merger or acquisition. It was also found that the legal framework for corporate rescue in Nigeria is grossly inadequate to address the cause of corporate insolvency.

 ¹⁵⁵³Adebola, The Nigerian Rescue Model, *op cit*, 9.
 ¹⁵⁵⁴ CAMA, *op cit*, s. 180(3).

¹⁵⁵⁵ Federal High Court Act, cap 134 Laws of the Federation of Nigeria 2004, s. 13(1); Abugu, Company Securities, op cit, p.227. ¹⁵⁵⁶ I Agabi & A Onayemi; 'Troubled Assets Resolution' in O Olanipekun (ed), Banking; Theory,

Regulation, Law and Practice (Lagos: AU Courant, 2016) p.469.

¹⁵⁵⁷SEC Rules, op cit, part 1, appendix VII(c); r. 434 (xiv) – in the case of an acquisition, the value to be acquired on the graduation fee.

Hence, the corporate rescue mechanisms available in Nigeria are more or less puerile as such rescue mechanisms do not debar the rights of an affected creditor to seek for the termination of the life of the insolvent company by winding up proceedings.

5. Corporate rescue and insolvency practitioners in Nigeria are not strictly subject to regulation of any regulatory institution in matters of insolvency. It was found that a person can be appointed a receiver and/or manager, provisional liquidator and liquidator in the cause of corporate rescue and insolvency in Nigeria insofar as he or she is not disqualified under relevant provisions in the extant laws governing such appointment.¹⁵⁵⁸ He or she is not expected to meet any minimum educational requirements or experience. This is against the findings in other jurisdictions where there is not only a requirement for minimum qualifications for corporate rescue and insolvency practitioners.¹⁵⁵⁹

It was found that because of the absence of minimum qualification and the nonregulation of corporate rescue and insolvency practitioners in Nigeria, there have been incidences of abuse, malpractices and exhibition of ineptitude and sheer incompetence in corporate rescue and insolvency proceedings.

6. There is clear absence of moratorium in the corporate rescue options available in Nigeria. In effect, where any of the corporate rescue option of receivership or arrangements and compromise or merger or acquisitions or even the amorphous construct of bailout applies to an insolvent company, it does not extinguish the right that any third party creditor may have against such insolvent company. Such creditor's right includes that of commencement of action to enforce

¹⁵⁵⁸ CAMA, op cit, ss. 387(1)- (Receiver and Manager); 509(1) (Liquidator).

¹⁵⁵⁹ Insolvency Act 1986 (as amended), ss. 389, 390 (United Kingdom); Insolvency and Bankruptcy Code 2016, ss. 200,204, 207(India); Companies Act 2008, s 138(1) (South Africa); Companies Amendment Act 2011, s 88 (South Africa).

security or recover the debt owed it, continuation of action, institution of winding up petition and the execution of judgment debt and attachment. This is against the finding in other jurisdictions where moratorium provision is intrinsic to the specific corporate rescue option.¹⁵⁶⁰

7. The implications of a winding up order under the extant law in Nigeria are rigid, inflexible and cannot be a means to rescue or restructure insolvent companies. Simply, it was found that the powers of the court and the liquidator in winding up of insolvent companies are circumscribed and do not extend to the initiation of restructuring or rescue plans of such insolvent companies.¹⁵⁶¹ This is against the finding in the United Kingdom where the liquidator is vested with the power to initiate corporate rescue proceedings¹⁵⁶² and the Republic of South Africa, the court, that is the "Companies Tribunal", is vested with the power to commence business rescue proceedings during the course of any liquidation proceedings.¹⁵⁶³

8. It was found that the initiation of winding up petition does not require prior exhaustion of any of the corporate rescue options. Contrarily, it was found that in the Republic of India, rescue option in the form of "corporate insolvency resolution process" or "fast track corporate insolvency resolution process" must be exhausted before the initiation of any liquidation proceeding.¹⁵⁶⁴

9. It was found that there are no prescribed time limits for carrying out rescue and insolvency proceedings in Nigeria. Contrarily, it was found that in the Republic of India, there are prescribed time limits for corporate rescue and insolvency

¹⁵⁶⁰Insolvency Act, op cit, (United Kingdom); Insolvency and Bankruptcy Code, op cit, ss. 13, 14; Companies Act, op cit, s. 133; Companies Amendment Act, op cit, s. 84.

¹⁵⁶¹ CAMA, op cit, s. 425; A Aliyu and M A Saulawa, 'The Role of a Liquidator in Winding up Proceedings in Nigeria: Examining the Inadequacies Under Companies and Allied Matters Act (CAMA)', PHJBL 2.1 (2016)235. ¹⁵⁶² The Insolvency Rules 1986 (as amended), part 1, chapter 3 (Proposal for Company Voluntary

Arrangements by Liquidator). ¹⁵⁶³ Companies Act, *op cit*, s. 131(7).

¹⁵⁶⁴ Insolvency and Bankruptcy Code, *op cit*, s. 33.

proceedings.1565

10. It was found that the regulatory institutions lack the requisite capacity to enthrone a viable corporate rescue and insolvency practice in Nigeria. The institutions, that is, the Courts, the Corporate Affairs Commission (CAC) and even the Securities and Exchange Commission (SEC), were found to be over burdened with functions under their respective enabling statute that they do not possess the requisite for regulating corporate rescue and insolvency matters. Contrarily, it was found that in the Republic of India, there is a body corporate established amongst others to make regulations and guidelines on matters relating to corporate rescue and insolvency.¹⁵⁶⁶

11. It was found that there are no standing financial mechanisms under CAMA or any other statute to address corporate insolvency matters in Nigeria. Rather, it was found that where the spate of corporate insolvency affects or is likely to affect companies in some perceived critical sectors of the economy such as the banking and aviation sectors, the Federal Government of Nigeria utilised informal corporate rescue options such as bailout and purchase and assumption.

Unlike the case in Nigeria where there is an absence of a statutory standing financial mechanism or fund to address corporate Insolvency matters, in the Republic of India, it was found that a Fund¹⁵⁶⁷ was created to address the cause of corporate insolvency. Furthermore, it was found that sources of the Insolvency and Bankruptcy Fund were clearly spelt out, namely, the grants made by the Central Government of

¹⁵⁶⁵Insolvency and Bankruptcy Code, *op cit*, s. 5 (14), 12 (1) (3) - Corporate insolvency resolution process shall be for a period of one hundred and eighty days but subject to extension by the Adjudicating Authority, that is National Company Tribunal, for a period not exceeding ninety days and such extension can be granted once only.

¹⁵⁶⁶Insolvency and Bankruptcy Code, *op cit*, ss. 188(1)(2). The establishment of 'the Insolvency and Bankruptcy Board of India" (the Board) as a body corporate; s. 196 - the powers and functions of the Board.

¹⁵⁶⁷*Ibid*, s. 224(1)- creation of the "Insolvency and Bankruptcy Fund". For the purposes of insolvency resolution, liquidation and bankruptcy

India, the amount deposited by persons as contribution, the amount received in the Fund from any other source, and the interest or other incomes received out of the investment made from the Fund.¹⁵⁶⁸

6.2 **Recommendations**

In light of the statement of problem and the summary of findings in this study, it is pertinent to make the following recommendations:

1. The National Assembly as a matter of urgency should enact a single holistic and comprehensive statute to consolidate the extant multifarious laws on corporate rescue and insolvency to be known as the "Corporate, Insolvency and Rescue Act".

2. The National Assembly should abort the passing into Law of the Insolvency and Bankruptcy (Repeal and Re-enactment) Bill 2015, which Bill is at the conference stage.

In the unlikely event that the National Assembly passes the Bill into Law, the President of the Federal Republic of Nigeria should withhold assent.

3. The National Assembly in the recommended Corporate Insolvency and Rescue Act should create a home grown and holistic corporate rescue option to be known as "Corporate Insolvency Rescue Proceedings".

The proposed Corporate Insolvency Rescue Proceedings should be a mix of the best provisions in the business rescue proceedings under Chapter 6 of the Companies Act of the Republic of South Africa and Part II, chapter II of the Insolvency and Bankruptcy Code of the Republic of India and Chapter 11 reorganisation proceedings of the Bankruptcy Code of the United States of America.

¹⁵⁶⁸Insolvency and Bankruptcy Code, op cit, s. 224(2).

4. In the recommended Corporate Insolvency and Rescue Act, the National Assembly should establish a special insolvency regime for corporate entities in the banking, aviation, telecommunication, insurance and other sectors of the economy as the National Assembly may from time to time determine.

5. In the recommended Corporate Insolvency and Rescue Act, the National Assembly should establish a body corporate to be known as the "Insolvency Commission of Nigeria", which body corporate shall have the power to make regulations and guidelines on matters relating to corporate rescue and insolvency.

6. Corporate rescue and insolvency practitioners should only be subject to regulatory supervision and authority of the Insolvency Commission of Nigeria.

7. The unbundling of the Corporate Affairs Commission (CAC), Asset Management Corporation of Nigeria (AMCON), Nigeria Deposit Insurance Corporation (NDIC), Securities and Exchange Commission (SEC), Central Bank of Nigeria (CBN), National Insurance Commission of Nigeria (NAICOM) and any other regulatory institutions other than the Insolvency Commission of Nigeria of all powers and functions relating to regulation and supervision of corporate insolvency matters.

8. A first degree in Law with at least seven (7) years post-call experience as the minimum qualification for a person to be appointed or to act as corporate rescue and insolvency practitioner in Nigeria, whether as receiver and/or manager or provisional liquidator or liquidator.

9. The creation of a specialised Court to be known as the 'Companies Court of Nigeria' to handle in exclusion to any other court or tribunal, all matters relating to or connected with companies, corporate rescue and insolvency matters.

10. A person shall not be eligible to be appointed a judge of the Companies Court unless the person is qualified to practice as a legal practitioner in Nigeria and

303

has been so qualified with not less than ten (10) years post-call experience in corporate law and practice or administration for not less than eight (8) years.

11. The National Assembly should amend section 251(1)(j) of the Constitution of the Federal Republic of Nigeria 1999 by removing the exclusive jurisdiction vested on the Federal High Court in respect of Bankruptcy and Insolvency matters.

12. In the interim, the Chief Judge of the Federal High Court should by directive, create divisions of the Federal High Court, including the Companies and insolvency Division.

13. Alternative dispute resolution mechanisms such as Mediation, Negotiation, Conciliation and Arbitration should be statutorily entrenched as means of resolving insolvency matters.

14. Moratorium provisions should be embedded in the recommended Corporate Insolvency and Rescue Proceedings. The moratorium period should be for an initial period of 12 months, subject to extension for a further period of 12 months.

15. Corporate rescue option as provided in the recommended Corporate Insolvency and Rescue Act must be exhausted before the commencement or initiation of winding up proceedings.

16. Informal corporate rescue options such as debt for equity swap, purchase and assumption and cherry picking should be strengthened and encouraged by the National Assembly making uniform and harmonised guidelines for the invocation of such informal corporate rescue options

17. The Corporate Affairs Commission (CAC) should ensure and enforce total compliance with the extant laws and regulations on financial statements and audits of a company, filing of annual returns and maintenance of reserve funds.

304

18. The statutory offences connected in respect of corporate insolvency as stipulated in the extant laws such as the CAMA should be enforced by the prosecution of offenders, whether directors or other officers of a company.

19. There should be prescribed time limits within which to exploit any of the available corporate rescue and insolvency procedure. In the interim, the Companies Regulation 2012 should be reviewed so as to prescribe time limits within which to exploit any of the available corporate rescue and insolvency procedures. Winding up proceedings should not exceed 180 days from the time of commencement at the court of first instance.

20. There should be regular public lectures, seminars, sensitisation and training programmes for judges, government officials, legislators, regulatory officers, legal practitioners, company executives and students on the essence, dynamics and rudiments of corporate rescue and insolvency.

21. Corporate rescue and insolvency law and practice should be accredited and taught as a compulsory course at the undergraduate level in the respective Faculty of Law of the various Universities in Nigeria.

22. The words, phrases and terms to be used in the Corporate Insolvency and Rescue Act should be simple, clear, unambiguous, modern and devoid of any form of anachronistic language.

23. Application, Processing or Filing fees in corporate rescue transactions and proceeding should be waived.

24. The National Assembly should re-order the list of preferential payments as provided in section 494 of the CAMA by amending same to confer priority to creditors in respect of claims during insolvency proceedings.

305

25. The creation of a special fund to be known as the "Corporate Insolvency Fund of Nigeria" (CIFN) in the recommended Corporate Insolvency and Rescue Act. The sources of income for the Corporate Insolvency Fund of Nigeria shall be from:

i. grants by the Federal Government of Nigeria;

ii. 3 percent of the profits declared by every company in a relevant financial year;

iii. Such monies as may from time to time be granted or lent to or deposited with the recommended Insolvency Services Commission of Nigeria by the Federal or State Government, any other body or institution, whether local or foreign;

iv. All monies raised for the purposes of the recommended Insolvency

Services Commission by way of gifts, loan, grants-in-aid, testamentary deposition or otherwise;

v. The amount received in the CIFN from other sources; and

vi. The interest or other income received out of the investment made from the CIFN.

BIBLIOGRAPHY

Books

Abugu, JEO, *Company Securities: Law and Practice* (2ndedn, Lagos: MIJ Professional Publishers Limited, 2014).

----- *Principles of Corporate Law in Nigeria* (2ndedn, Lagos: MIJ Professional Publishers Limited, 2014).

- Adeleye, B (ed), *Nigerian Company Law Cases* (Lagos: Alexander Payne Company Limited, 2016).
- Agabi, I & Onayemi, A, 'Troubled Assets Resolution' in Olanipekun, O (ed), Banking: Theory, Regulation, Law and Practice (Lagos: Au Courant, 2016) pp.457-488.
- Ajayi, O, *Legal Aspects of Finance in Emerging Markets* (London: LexisNexis Butterworths Tolley, 2005).
- Ajogwu, F, Mergers & Acquisitions in Nigeria: Law & Practice (2ndedn, Lagos: Centre for Commercial Law Development (CCLD), 2014).
- ------ 'Mergers, Takeovers and Reorganisation of Banks in Nigeria' in Olanipekun, O(ed), *Banking: Theory, Regulation, Law and Practice* (Lagos: Au Courant, 2016) pp.565-586.
- Akinwunmi, S 'Chapter 21-Nigeria' in Mallon, C (ed), *The Restructuring Review* (6th edn, London: Law Business Research Limited, 2013) pp.266-276.

------ 'Chapter 20-Nigeria' in Mallon, C (ed), *The Restructuring Review* (7th edn, London: Law Business Research Limited, 2014) pp.255-265.

------ 'Chapter 19-Nigeria' in Mallon, C (ed), *The Restructuring Review* (8th edn, London: Law Business Research Limited, 2015) pp.251-262.

Altman, EI, Corporate Financial Distress (New York: Wiley, 1983).

Amadi, J, *Modern Civil Procedure Law and Practice in Nigeria* Vol I (Port Harcourt: Pearl Publishers, 2014).

-----*Modern Civil Procedure Law and Practice in Nigeria* Vol II (Port Harcourt: Pearl Publishers, 2014).

- Ama Oji, EE & Amucheazi, OD, *Employment & Labour Law in Nigeria* (Lagos: Mbeyi & Associates (Nig) Ltd, 2015).
- Amupitan, JO, *Principles of Company Law in Nigeria* (Lagos: Innovative Communications, 2013).

- Argenti, J, Corporate Collapse: The Causes and Symptoms (London: Mcgraw-Hill, 1976).
- Bachner, T, Creditors Protection in Private Companies: The Anglo-German Perspective (Cambridge: Cambridge University Press, 2009).
- Barber, S, Company Law (4thedn, London: Old Bailey Press, 2003).
- Baldwin, R & Cave, M, *Understanding Regulation* (Buckingham: Open University Press, 1999).
- Baxt, Rand Fletcher, K, *Afterman and Baxt's Cases and Materials on Corporations and Associations* (6thedn, London: Butterworth & Co Publisher Ltd, 1992).
- Beatty, JF & Samuelson, SS, *Business Law for New Century* (New York: Little, Brown and Company, 1996).
- Bhadmus, HY, *Corporate Insolvency Law and Practice in Nigeria* (Enugu: Chenglo Limited, 2009).

-----Bhadmus on Corporate Law Practice (2ndedn, Enugu: Chenglo Limited, 2013).

----- Bhadmus on Corporate Law Practice (3ndedn, Enugu: Chenglo Limited, 2015).

- Bhandari, JS & Weiss, LA (eds), *Corporate Bankruptcy* (Cambridge: Cambridge University Press, 1996).
- Blair, MM, Ownership and Control: Rethinking Corporate Governance for the Twenty First Century (Washington: Brooking Institute, 1995).
- Bailey, E & Groves, H, *Corporate Insolvency-Law and Practice* (4thedn, London: LexisNexis Butterworths, 2014).
- Birds, J, Boyle, AJ, Clark, B, MacNeil, I, McCormack, G, Twigg-Flesner, C & Villiers, C (eds), *Boyle & Birds' Company Law* (9thedn, Bristol: Jordan Publishing Ltd, 2014).
- Birds, J, Boyle, AJ, Ferran, E & Villiers, C, *Boyle & Birds' Company Law* (4thedn, Bristol: Jordan Publishing Ltd, 2000).
- Black, J, Rules and Regulators (Oxford: Clarendon Press, 1997).
- Broc, KG & Parry, R (eds), Corporate Rescue: An Overview of Recent Developments (2ndedn, Hague Netherlands: Kluwer Law International, 2006).
- Brown, D, *Corporate Rescue: Insolvency Law in Practice* (Chichester: John Wiley & Sons, 1996).

- Bourne, N, *Principles of Company Law* (3rdedn, London: Cavendish Publishing Limited, 1998).
- Burton, WC, Burton's Legal Thesaurus (3rdedn, New York: Mcgraw-Hill, 1980).
- Butterworth's Company Precedent Vol 3.
- Candide-Johnson, CA & Alex-Adedipe, A, 'Debt Recovery: Corporate Insolvency Receivership, Winding up and other Arrangements' in Olanipekun, O (ed), *Banking: Theory, Regulation, Law and Practice* (Lagos: Au Courant, 2016) pp.429-456.
- Carruthers, BG & Halliday, TC, Rescuing Business: The Making of Corporate Bankruptcy in England and the United States (Oxford: Clarendon Press, 1998).
- Campbell, C & Underdown, B, Corporate Insolvency in Practice: An Analytical Approach (London: Chapman, 1991).
- Clark, LS, Aalberts, R & Kinder, P, *Law and Business: The Regulatory Environment* (4thedn, New York: McGraw- Hill, Inc, 1994).
- Clarke, F, Dean, G and Oliver, K, *Corporate Collapse: Regulatory, Accounting and Ethical Failure* (Cambridge: CUP,1997).
- Cheffins, C, *Company Law: Theory, Structure and Operation* (Oxford: Clarendon Press, 1997).
- Chinwuba, NN, 'Re-Examining the Use of Credit Balances as Security' in Smith, IO (ed), Secured Credit in a Global Economy-Challenges and Prospects (Lagos: The Department of Private and Property Law, Faculty of Law, University of Lagos' (2003) pp. 267-282.
- Chukwu, LOC, 'The Role Courts in the Enforcement of Securities' in Smith, IO (ed), Secured Credit in a Global Economy-Challenges and Prospects (Lagos: The Department of Private and Property Law, Faculty of Law, University of Lagos,2003) pp.229-247.
- Cork, K, Cork on Cork (London: Macmillan, 1988).
- Curzon, LB, *Dictionary of Law* (5thedn, London: Financial Times Pitman Publishing, 1998).
- Dada, AJ, *Principles of Nigerian Company Law* (Rev edn, Calabar: Wusen Publishers, 2001).
- Davidson, DV, Knowles, BE, Forsythe, LM & Jespersen, RR, *Business Law: Principles and Cases* (4thedn, California: Wadsworth Publishing Company, 1993).

- Davies, PL & Worthington, S, *Gower & Davies' Principles of Modern Company Law* (9thedn -Asian edn, London: Sweet & Maxwell, 2012).
- Davies, S, Insolvency and Enterprise Acts (Bristol: Jordan Publishing Ltd, 2003).
- Davis G and Haywood M, *Butterworths Insolvency Law Handbook* (18thedn, London: LexisNexis Butterworths, 2016).

Donnell, JD, Barnes, AJ & Metzger, MB, *Law for Business* (Rev edn, Illinois: Irwin Inc, 1983).

- Doyle, L & Keay, A, *Insolvency Legislation: Annotations and Commentary* (Bristol: Jordans, 2007).
- Ekwueme, KUK, 'Failure Resolution' in Olanipekun, O (ed), *Banking: Theory, Regulation, Law and Practice* (Lagos: Au Courant, 2016) pp. 489-528.
- Enterprise Chambers, Annotated Guide to Insolvency Legislation and Practice (London: LexisNexis Butterworths, 2006).
- Essien, E, *Law of Credit and Security in Nigeria* (2ndedn, Uyo: Toplaw Publishments Ltd, 2012).
- Fletcher, IF, The Law of Insolvency (4thedn, London: Sweet & Maxwell, 2009).
- Farrar, JH & Hannigan, B, *Farrar's Company Law* (4thedn, London: LexisNexis Butterworths Tolley, 1998).
- Finch, V, Corporate Insolvency Law: Perspective and Principles (Cambridge: CUP, 2002).
- -----*Corporate Insolvency Law: Perspective and Principles* (2ndedn, Cambridge: CUP, 2009).
- Finch, V & Worthington, S, 'The Pari Passu Principle and Ranking Restitutionary Rights' in Rose, F (ed), *Restitution and Insolvency* (London: Mansfield Press, 2000) pp.1-20.
- Flood, J, 'The Vultures Fly East; the Creation and Globalization of the Distressed Debts Market' in Nelken, D (ed), Adapting Legal Cultures (Oxford: Hart Publishing, 2001) pp. 257-278
- Ford, CHJ &Austin, RP, '*Ford's Principles of Corporations Law* (6thedn, London: Butterworth & Co (Publishers) Ltd, 1992).
- French, D, Mayson, DS & Ryan, CL, *Mayson, French and Ryan on Company Law* (25thedn, Oxford: OUP, 2008 -2009).
- Frisby, S, Corporate Rescue: Law and Practice: Contemporary Studies in Corporate

Law (Oxford: Hart Publishing, 2014).

----- *Kerr and Hunter on Receivers and Administrators* (20th edn, rev, London: Sweet & Maxwell, 2012).

- Frisby, S, Davis-White, M, Jones, M & Bennett, D, *Kerr and Hunter on Receivers and Administrators* (19th edn, London: Sweet & Maxwell, 2009).
- Fuller, G, Corporate Borrowing: Law and Practice (3rdedn, Bristol: Jordan, 2006).
- Garner, BA(ed), Black's Law Dictionary (9th edn, USA: Thomas Reuters, 2009).
- Gasiokwu, MOU, *Legal Research and Methodology* (Enugu: Chenglo Limited, 2006).
- Gauthier, D, Morals by Agreement (Oxford: Clarendon Press, 1986).
- Goode, R, *Principles of Corporate Insolvency Law* (4thedn, London: Sweet & Maxwell, 2011).
- Gilson, S, Creating Value Through Corporate Restructuring: Case Studies in Bankruptcy, (Buy outs and Breakups (New York: John Wiley and Sons, 2001).
- Gordan, J & Roe, M (eds), *Convergence and Persistence in Corporate Governance* (Cambridge: CUP, 2004).
- Girvin, S, Frisby, S & Hudson, A, *Charlesworth's Company Law* (18thedn, London: Sweet & Maxwell, 2010).
- Gullick, JM, *Ranking and Spicer's Company Law* (13thedn, London: Butterworths, 1989).
- Halliday, TC & Carruthers, BG, *Bankrupt, Global Law Making and Systemic Financial Crisis* (Standford University Press, 2009).
- Halsbury's Laws of England, volume 15 (5thedn, United Kingdom: LexisNexis, 2009).
- Hicks, A & Goo, SH, *Cases & Materials on Company Law* (2nd edn, London: Blackstone Press Limited, 1997).
- Insolvency Practitioners' Association, *Insolvency Practitioners Handbook* (4thedn, London: LexisNexis Butterworths, 2015).
- Jackson, TH, *The Logic and Limits of Bankruptcy Law* (Cambridge: Mass: Harvard University Press, 1996).
- James, JS, *Stroud's Judicial Dictionary of Words and Phrases*(4thedn, London: Sweet & Maxwell Limited, 1972).

- Jennings, M, *Business, Its Legal, Ethical and Global Environment* (4thedn, Ohio: International Thomson Publishing, 1997).
- Keay, A, *Company Directors' Responsibilities to Creditors* (Abingdon: Routledge-Cavendish, 2007).
- Keay, A, and Walton, P, *Insolvency Law: Corporate and Personal* (Harlow: Pearson Education Limited, 2003).
- Keenan, D, *Smith & Keenan's Company Law* (7thedn, London: Pitman Publishing, 1987).
- Lightman, D & Mokal, RJ, 'Duties and Liabilities of Receivers and their Appointors' In Lightman, G & Moss, G (eds), *The Law of Administrators and Receivers of Companies* (4thedn, London: Thomsom/Sweet & Maxwell, 2007) pp.234-272.
- Lightman, G & Moss, G (eds), *The Law of Administrators and Receivers of Companies* (5thedn, London: Thomsom/Sweet & Maxwell, 2011).
- Lingard, J, Corporate Rescue and Insolvency (2ndedn, London: Butterworths, 1989).
- Lynch-Fannon, I and Murphy, G,*Corporate Insolvency and Rescue* (2ndedn, London: Bloomsbury Professional, 2012).
- Magnus, SW and Estrin M, *Companies Law and Practice*(5thedn, London: Butterworths, 1978).
- Mallon, C (ed), *The Restructuring Review* (7thedn, United Kingdom: Law Business Research Limited, 2014).
- Martin, EA & Law, J (eds), A Dictionary of Law (6thedn, OUP, 2006).
- McCormack, G, Secured Credit under English and American Law (Cambridge: CUP, 2004).
- -----*Corporate Rescue Law An Anglo-American Perspective* (London: Edward Elgar Publishing, 2008).

McLaughlin, S, Unlocking Company Law (London: Hodder Education, 2009).

-----*Unlocking Company Law* (2ndedn, London and New York: Routledge, 2003)

- Meinhardt, P, *Company Law in Europe* (3rdedn, England: Gower Publishing Company Limited, 1981).
- Miller, S & Bailey, E, *Personal Insolvency: Law and Practice* (4thedn, London: LexisNexis Butterworths, 2008).

- Mokal, RJ, *Corporate Insolvency Law: Theory and Application* (Oxford: OUP, 2005).
- Morse, G, *Charlesworth's Company Law* (13thedn, London: Steven& Sons Ltd, 1987).
- Morgan, B & Yeung, K, *An Introduction to Law and Regulation* (Cambridge: CUP, 2007).
- Morris, C & Murphy, C, *Getting a PhD in Law* (Oxford: Hart Publishing Limited, 2011).
- Mustoe, NE, *Bankruptcy Liquidation and Receivership* (London: Butterworth & Co (Publishers) Ltd, 1939).
- Narang, U & Salwan SL, *Legal Dictionary* (New Delhi: Academic (India) Publishers/EPP Books Services Nigeria Ltd, 2007).
- Nnabude, PC, Nkamnebe, AD & Ezenwa, MO (eds), *Readings in Research Methodology & Grant Writing* (Nimo: Rex Charles & Patrick Ltd, 2009).
- North, DC, Institutions, Institutional Change and Economic Performance (Cambridge: CUP, 1990).
- Nwakoby, GC, *The Law and Practice of Commercial Arbitration in Nigeria* (2ndedn, Enugu: Snaap Press Limited, 2014).
- Offe, C, Disorganised Capitalism (Cambridge: Polity Press, 1985).
- Ogbuanya, NCS, *Essentials of Corporate Law Practice in Nigeria* (2ndedn, Lagos: Novena Publishers Ltd, 2014).
- Ola, CS, *Company Law in Nigeria* (Ibadan: Heinemann Educational Books (Nigeria) Plc,2002).
- Onuoha, RA, 'The Role of a Receiver in the Preservation, Realisation and Enforcement of Security' in Smith, IO (ed), *Secured Credit in a Global Economy-Challenges and Prospects* (Lagos: The Department of Private and Property Law, Faculty of Law, University of Lagos, 2003) pp.118-144.
- Orojo, JO, *Company Law and Practice in Nigeria* (5thedn, London: LexisNexis ButterworthsTolley, 2008).
- Palmer, FB, *Palmer's Company Law Volume 3* (London/Edinburgh: Sweet & Maxwell/W Green).
- Parry, R, Corporate Rescue (London: Sweet & Maxwell,2008).
- Parry, R, Ayliffe, J & Shivji, S, Transactions Avoidance in Insolvencies (2ndedn,

Oxford: OUP, 2011).

- Payne, J Schemes of Arrangement, Theory, Structure and Operation (Cambridge: CUP, 2014).
- Pennington, RR, Pennington's Company Law (7thedn, London: Butterworths, 1995).
- Picarda, H, The *Law Relating to Receivers and Managers* (London: Butterworths & Co (Publishers) Ltd, 1984).
- ----- The Law Relating to Receivers, Managers and Administrators (4thedn, London: Tottel Publishing, 2006).
- Platt, HD, Why Companies Fail: Strategies for Detecting, Avoiding and Profiting from Bankruptcy (Lexington: Lexington Books, Mass, 1985).
- Power, M, The Risk Management of Everything (London: Demos, 2004).
- Rajak, H(ed), *Insolvency Law: Theory and Practice* (London: Sweet & Maxwell, 1993).
- Reimann, M & Zimmerman, R, *Oxford Handbook on Comparative Law* (Oxford: OUP, 2008).
- Sanni,O, 'Securities in a Global Economy: New Trends' in Smith, IO (ed), Secured Credit in a Global Economy-Challenges and Prospects (Lagos: The Department of Private and Property Law, Faculty of Law, University of Lagos,2003) pp.24-41.
- Sasegbon, D, Sasegbon's Laws of Nigerian Law and Practice, Vol 3 (Lagos: DSC Publications 2005).
- Sealy, L & Worthington, S, *Cases and Materials in Company Law* (10thedn, Oxford: OUP, 2013).
- Singh, A, *Company Law* (16thedn, Lucknow/New Delhi: Eastern Book Company, 2015).
- Smith, IO, *Nigerian Law of Secured Credits* (Lagos: Ecowatch Publications (Nigeria) Limited, 2001).
- -----(ed), Secured Credit in a Global Economy-Challenges and Prospects (Lagos: The Department of Private and Property Law, Faculty of Law, University of Lagos,2003).
- Sofowora, MO, Modern Nigerian Company Law (2ndedn, Lagos: Soft Associates, 2002)
- Solomon, LD & Palmiter, AR, *Corporations Examples and Explanations* (2ndedn, Toronto: Little, Brown & Company, 1994).

- Stahl, Gk & Mendenhall, ME (eds), *Mergers and Acquisitions* (USA: Standford University Press, 2005).
- Taiwo, A, *Basic Concepts in Legal Research Methodology* (Ibadan: St. Paul's Publishing House, 2011).
- Tomasic, R (ed), *Insolvency Law in East Asia* (London: Ashgate Publishing Ltd, 2006).
- Ulph, J, Commercial Fraud: Civil Liability, Human Rights and Money Laundering (Oxford: Oxford University Press, 2006).
- Uzodike, ENU, 'Liability of Receiver/Manager in Nigeria' in Smith, IO (ed), Secured Credit in a Global Economy-Challenges and Prospects (Lagos: The Department of Private and Property Law, Faculty of Law, University of Lagos,2003) pp.145-156.
- Vagts, DF, *Basic Corporation law: Materials-Cases-Text* (10thedn, New York: The Foundation Press, Inc, 1989).
- Warren, E &Westbrook, J, *The Law of Debtors and Creditors: Texts, Cases and Problems* (6thedn,Hague, Netherlands, Wolters Kluwer Law & Business, 2008).
- Webster's Legal Speller(Springfield, Massachusetts: Merriam-Webster Inc, 1978).
- Wigwe, CC, *Introduction to Company Law and Practice* (Osu-Accra: Mountcrest University Press, 2016).
- Wood, P, Principles of International Insolvency (London: Sweet & Maxwell, 2007)
- Woodley, M(ed), Osborn's Concise Law Dictionary (12thedn-South Asianedn, London: Sweet & Maxwell /Thomson Reuters, 2013).

Journal Articles

- Abdulai, A, 'Mergers, Acquisitions and Take-overs in Nigeria: The Guiding Philosophy, Laws, Regulations and Practice' (1997) 4(2)MILBQ,21-34.
- Adegbite, LO & Carew, KO, 'Practice and Procedure of Mergers' (1989) The Gravitas Review of Business & Property Law,47-55.
- Adesanya, O, 'Negotiating Shareholder Disputes: Principled Negotiation in Cross-Cultural Shareholder Disputes' JCL, Vol 1, No 1, March 2015, 42-58

Adler, BE, 'A World Without Debt' (1994) 72(3) Wash.U.L.Q, 811-828.

- Aina, K, 'The Machinery for Raising Capital by Companies through Debt Finances in Nigeria' (2011) 2(1) International Journal of Advanced Legal Studies and Governance, 85-104.
- ------ 'Rethinking the Duties of a Receiver and Powers of Directors of Companies in Receivership under the Nigerian Law' (2015) The Gravitas Review of Business and Property Law XI. II, 60-73.
- Ajayi, K, 'Legal Aspects of Securitisation'(1998) 3(1)MILBQ, 30-43
- Ajayi, K & Opasanya, S, 'New Developments in Law and Practice' (1989) The Gravitas Review of Business Property Law, 28-32
- Ajogwu, F 'Mergers & Acquisitions: Opportunities and Pitfalls' Unib Law Journal, VOL 1, NO.1, October 2011, 5 10.
- Akume, AA, 'Predicting Corporate Collapse in Nigeria' (1999) MILBQ 4(2), 43-49
- ----- 'The Phenomenon of Corporate Collapse and its Impact on Developing Economies' ABUJCL, vol. 1 No. 1 of 2002, 39-51.
- Alfter, M, Bishop, S and Mahr, E, 'New Horizontal Merger Guidelines: Plugging the Gap or Opening the Floodgates' (2004) Business Law International Vol 5 No 3, 385-399.
- Aliyu, A & Saulawa, MA, 'The Role of a Liquidator in Winding Up Proceedings in Nigeria: Examining the Inadequacies under Companies and Allied Matters Act (CAMA) PHJBL 2.1 (2016), 225-236.
- Anderson, C, 'The Australian Corporate Rescue Regime: Bold Experiment or Sensible Policy' (2001) 10(2) Int.Insolv.Rev, 81-113.
- Anderson, H, 'Insolvent Insolvencies' (2001) 17(3) Insolvency Law and Practice, 87.
- ------ 'Directors' Liability for fraudulent Phoenix Acting A Comparison of the Australia and UK Approaches' (2014) 14(1) Journal of Corporate Law Studies, 139-173.
- Ani, CC, 'Corruption in Criminal Justice Administration in Nigeria: The Role of the Legal Profession' (2011)Nigeria Bar Journal, Vol 7, 110.
- Apinega, SA, 'Accessing the Capital Market by States, Local Government and Their Agencies under the Investments and Securities Act, 1999' J.P.C.L VOL.1.NO.22006, 179-199.
- Armour, J, 'Personal Insolvency Law and the Demand for Venture Capital' (2004/1) 5 E.B.O.R, 87-118.

Armour, J & Frisby, S, 'Rethinking Receivership' (2001) OJLS, 20-26.

- Armour, J & Deakin, S, 'Norms in Private Insolvency Procedure: The "London Approach" To the Resolution of Financial Distress' (2001) 1 JCLS, 21.
- Armour, J & Walters, A, 'Funding Liquidation: A Functional View' (2006) LQR, 295.
- Arora, A, 'The Global Financial Crisis; A new Global Regulatory Order' (2010) 8 JBL 670-699.
- Aruoriwo, A, 'Financing Corporate Rescues, Where Does the UK Stand?' (2014) 1(2) IALS Student Law Review, 10-19.
- Asomugha, EM 'A Review of the Borrowing Powers of the Company under the 1990 Companies and Allied Matters Decree'(1990/91) 13, 14 & 15 J.P.P.L, 41-52
- ------'A Re-Appraisal of the Functions of the Court in Scheme of Arrangements, Mergers and Take-Overs' (1991) A Journal of Contemporary Legal Problems Volume 2 Number 1 ISSN 1115-1277, 41-53.
- Baird, DG, 'The Uneasy Case for Corporate Reorganizations' (1986) 15 Journal of legal Studies, 127.

----- 'Loss Distribution, Forum Shopping and Bankruptcy; A Reply to Warren' (1987) 54(3) University of Chicago Law Review, 815-834.

Baird, DG & Jackson, TH, 'Corporate Reorganizations and the Treatment of Diverse Ownership Interests: A Comment on Adequate Protection of Secured Creditors' (1984) 51(1) University of Chicago Law Review, 97-130.

----- 'Bargaining After the Fall and the Contours of the Absolute Priority Rule' (1985) 55(3) University of Chicago Law Review, 738-789.

- Baird, DG & Rasmussen, RK, 'Anti Bankruptcy' (2000) 119(4) Yale LJ, 648-699.
- Block-Lieb, S & Halliday, TC, 'Harmonization and Modernization in UNCITRAL's Legislative Guide on Insolvency Law' (2007) 42(3) Texas International Law Journal, 1-37.
- Brouwer, M, 'Reorganization in US and European Bankruptcy Law' (2006) European Journal of Law & Economics, 1-14.
- Bolodeoku, IO, 'Receivership and Enforcement of Tax Liability' (2004) Vol. 24 The Journal of Private and Property Law, University of Lagos, 29-43.
- Boshkoff, DG, 'The Attack on Chapter II' (1993) 2(2)Int.Insolv.Rev,173-186.

- Bufford, SL, 'What is Right about Bankruptcy Law and Wrong About its Critics' (1994) 72(3) Wash.U.L,Q, 829-848.
- Busari, T, 'A Case for Insolvency Law & Practice' (October2000) MPJFIL Vol 4, No.4, 288-296.
- Chan, TE, 'Scheme of Arrangement as a Corporate Rescue Mechanism; The Singapore Experience' (2009) 18(1) Int. Insolv.Rev, 37-58.
- Chianu, E, 'Preferential Payments after Bank Failures: The Fate of Depositors, Shareholders and Landlords' (1998) *MILBQ* vol.3, No 1, 49-59.
- Cooke, C, 'Balance Sheet Insolvency: A Commercial Approach' (2011) 8(3) International Corporate Rescue, 228- 230.
- Dal Pont, G and Griggs L, 'A Principled Justification for Business Rescue Laws: A Comparative Perspective (part 1)' (1995) 4(2) Int.Insolv.Rev, 189-198.

------'A Principled Justification for Business Rescue Laws: A Comparative Perspective (part II)' (1996) 5(1) Int.Insolv.Rev, 47-79.

- Davies, P, 'Employee Claims in Insolvency: Corporate Rescues and Preferential Claims' (1994) 23(2) Ind Law J, 141 151.
- Dickfos, J, 'The Costs and Benefits of Regulating the Market for Corporate Insolvency Practitioner Remuneration' (2016) 25(1) Int.Insolv.Rev, 56-71.
- Elayan, A & Meyer, T, 'The Impact of Receiving Debtor-in-Possession Financing on the Probability of Successful Emergence and the Time Spent Under (CHAP II)' (2001) 28(7) BJBFA, 905-942.
- Etukakpan, S, 'Business Rescue and Continuity of Employment: Analysing Policy Through the Lens of Theory' (2011) Company Lawyer, 99 – 113.

----- 'The Lost Voice in Insolvency: Theories of Insolvency Law and their Implications for the Employees (2014) 23 Nott L.J, 34-65.

- Farrar, JH, 'The Bankruptcy of the Law of Fraudulent Preference' (1983) J.B.L, 390-408.
- Finch, V, 'Company Directors: Who Cares about Skill and Care' (1992) 55 M.L.R, 179 210.
- ----- 'Controlling the Insolvency Professionals' (1999) Insolvency Lawyer, 228 239.
- ----- 'Security, Insolvency and Risk: Who Pays the Price?' (1999) 62(5) M.L.R, 633 670.

----- 'Is Pari Passu Passe?' (2000) Insolvency Lawyer. 194 - 210.

'Re-Invigorating Corporate Rescue'(2003) J.B.L, 527 – 557.
'Control and Co-ordination in Corporate Rescue' (2005) 25(3) Int. Insolv. Rev, 374–403.
'Doctoring in the Shadows of Insolvency' (2005) J.B.L, 690 – 696.
'The Recasting of Insolvency Law' (2005) 68(5) M.L.R, 713 – 732.
'Regulating Insolvency Practitioners: Rationalisation on the Agenda' (2005) Insolvency Intelligence, 17 – 20.
'Pre- Packaged Administrations: Bargains in the Shadow of Insolvency or Shadowy Bargains?' (2006) J.B.L, 568 – 588.
'Corporate Rescue in a World of Debt' (2008) J.B.L, 756 – 777.
'Corporate Rescue Processes: The Search for Quality and the Capacity to Resolve'(2010) 6 J.B.L, 502 – 521.
'Corporate Rescue: A Game of Three Halves' (2012) 32(2) Legal Studies, 302 – 324.
'Corporate Rescue: Who is Interested?' (2012) 3 J.B.L, 190 – 212.
'Insolvency Practitioners: The Avenues of Accountability' (2012) 8 J.B.L, 645 – 667.
Fletcher, IF 'The Reform of Insolvency Law -I' (1983) 3 J.B.L, 94 – 103.
'The Reform of Insolvency Law- II' (1983) 5 J.B.L, 200 – 217.
'International Insolvency: The Way Ahead' (1993) 2(1)Int.Insolv.Rev, 7 – 28.
'UK Corporate Rescue: Recent Developments- Changes to Administrative Receivership, Administrative Company Voluntary Arrangement – The Insolvency Act 2000, The White Paper 2001 and the Enterprise Act 2002' (2004/1) 5 <i>E.B.O.R</i> , 119-151.
'Spreading the Gospel: The Mission of Insolvency Law and Insolvency Practitioners in the Early 21 st Century' (2014) 7 J.B.L, 523 – 540.
Frisby, S, 'In Search of a Rescue Regime: The Enterprise Act 2002' (2004) 67(2) M LR, 247 – 272.
'The Second-Chance Culture and Beyond: Some Observations on the Pre- pack Contribution' (2009) Law and Financial Markets Review, 242 – 247.

- Garrido, JM, 'The Distributional Question in Insolvency: Comparative Aspects' (1995)4(1)Int.Insolv.Rev, 25 53.
- Gelpern, A, 'Bankruptcy, Backwards: The Problem of Quasi-Sovereign Debt (2012) 121(4) Yale L.J, 888 942.
- Girgis, J, 'Corporate Restructuring, the Evolution of Corporate Assets and the Public Interest' (2013) 22 (1) Int.Insol.Rev, 29 54.
- Godwin, A 'Corporate Rescue in Asia-Trends and Challenges' (2012) 34 Sydney Law Review, 163 187.
- Golding, J & Phillips, M, 'Rescue and Reconstruction' (2002) Insolvency Intelligence, 75 – 76.
- Gross, K, 'Taking Community Interest into Account in Bankruptcy: An Essay' (1994) 72 Wash.U.L.Q, 1031 1048.
- Halliday, C & Babalola, M, 'Quest for Reform of Corporate Rescue and Insolvency Procedure in Nigeria' P.H.J.B.L 2.1 (2016), 352 373.
- Halliday, TC & Carruthers, BG, 'The Moral Regulation of Markets: Professions, Privatization and the English Insolvency Act 1986' (1996) 21(4) Accounting, Organisations and Society, 371 – 413.
- Hamilton, *R*, *et al*, 'Back from the Dead-Survival Potential in Administrative Receiverships' (1997) 13 Insolvency Legal & Practice, 78 – 81.
- Harmer, RW, 'Insolvency Law and Reform in the People's Republic of China' (1996) 64 Fordham L Rev, 2563 2583.
- Haywood, M, 'Pre-pack Administration' (2010) 23(2) Int.Insolv.Law, 17 22.
- Henderson, D, 'Inability to Pay Debts; Where are We Now? (2011) 24(4) Insolvency Intelligence, 54 59.
- Hunter, M, 'The Nature and Functions of a Rescue Culture' (1999) J.B.L, 491 498.
- Igwenyi, B, 'Appraisal of Anti-Corruption'(1999) 5 PCLJ,172 180.
- Jackson, TH & Scott, RE, 'On the Nature of Bankruptcy: An Essay on Bankruptcy Sharing and the Creditors' Bargain (1989) 75(2) Virginia Law Review, 155 – 204.
- Jackson, TH, 'Bankruptcy, Non-Bankruptcy and the Creditors' Bargain' (1982) 91(5) Yale Law Journal, 857 – 907.
- Jones, B & Wilhelm, E, 'Past the Point of no Return: (2011) 4(2) Corporate Rescue & Insolvency, 43 44.

- Kastrinou, A 'Comparative Analysis of the Informal Pre-Insolvency Procedures in the UK and France (2016) 25(2), Int.Isolv.Rev, 99 118.
- Keay, A, 'Directors' Duties to Creditors: Contractarian Concerns Relating to Efficiency and Over-Protection of Creditors' (2003) 66 M.L.R, 665 – 680.
- Keay, A, 'The Shifting of Directors' Duties in the Vicinity of Insolvency' (2015) 24(2)Int.Insolv.Rev, 140 164.
- Keay, A, Boraine, A and Burdette, D, 'Preferential Debts in Corporate Insolvency: a Comparative Study' (2001) 10(3) Int.Insolv.Rev, 167 194.
- Keay, A, and Murray, M, 'Making Company Directors Liable: A Comparative Analysis of Wrongful Trading in the United Kingdom and Insolvent Trading in Australia' (2005) 14 Int. Insolv. Rev, 27 – 45.
- Korobkin, DR 'Contractarianism and the Normative Foundations of Bankruptcy Law' (1993) 71 Tex.L.Rev, 541-554.
- Leno, ND, 'Rwanda Legal Framework on Insolvency: Problem and Proposals for Reform' (2015) 24(2), Int.Insolv.Rev, 122 139.
- LoPucki, LM, & Whitford, WC, 'Bargaining over Equity's Share in the Bankruptcy Reorganization of Large, Publicly Held Companies' (1991) 139 University of Pennsylvania Law Review, 125 – 196.
- LoPucki, LM, 'Reorganization Realities, Methodological Realities and the Paradigm Dominance Game' (1994) 72(3) Wash.U. L.Q, 1307 1318.
- Loubser, A, 'The Interaction between Corporate Rescue and Labour Legislation: Lessons to be drawn from the South Africa Experience' (2005) 14(1), Int.Insolv.Rev 57 – 69.
- Martin, N, 'Common-Law Bankruptcy Systems: Similarities and Differences' (2003) American Bankruptcy Institute Law Review, 367 – 380.
- McCormack, G, 'Control and Corporate Rescue- An Anglo- American Evaluation' (2007)56(3) I.C.L.Q, 515 551.
- ----- 'Super-Priority Financing and Corporate Rescue' (2007) J.B.L, 701 732.
- ------ 'Apples and Orange? Corporate Rescue and Functional Convergence in the US and UK' (2009) 18(2) Int.Insolv.Rev, 109 134.
- ----- 'Rescuing Small Business: Designing an 'Efficient' Legal Regime' (2009) J.B.L, 299 330.
- ------ 'Universalism in Insolvency Proceedings and the Common Law' (2012)

32(2) O.J.L.S, 325 – 347.

-----'Schemes of Arrangement: Theory, Structure and Operation' (2015) 16(1) E.B.O.R, 167 – 171.

------ 'Something Old, Something New: Recasting the European Insolvency Regulation' (2016) 79(1) M.L.R, 121 – 146.

McCormack, G & Hargovan, A, 'Australia and the International Insolvency Paradigm' (2015) 37 Sydney Law Review, 389 – 416.

Meadows, F, 'OECD Bribery Convention Five Years On: How is it working and how is it Monitored?' (2004) 5(3) Business Law International,371 – 384.

Milman, D, 'Test of Commercial Insolvency Rejected' (1983) 4 Law, 231.

------ 'Moratoria on Enforcement Rights: Revisiting Corporate Rescue' (2004) Conveyancer and Property Law, 89 – 91.

Mokal, RJ, 'The Authentic Consent Model: Contractarianism, Creditors' Bargain and Corporate Liquidation' (2001) 21(3) Legal Studies, 400 – 443.

----- 'Priority as Pathology: The *PariPassu* Myth' (2001) 60(3) Cambridge Law Journal, 581 – 621.

----- 'The Search for Someone to Save: A Defensive Case for the Priority of Secured Credit' (2002) 22(4) OJLS, 687 – 728.

-----, 'Review Article on Fairness and Efficiency' (2003) 66(3) M.L.R, 452 – 467.

------'Contractarianism, Contractualism and the Law of Corporate Insolvency' (2007) Singapore Journal of Legal Studies, 51 – 95.

------'What Liquidation Does for Secured Creditors and What It Does for You' (2008) 71 (5) M.L.R, 699 – 733.

Mokal, RJ, & Armour J, 'The New UK Rescue Procedure-The Administrator's Duty to Act Rationally' (2004) 1(3) International Corporate Rescue, 136 – 142.

Mooney, C W, 'A Normative Theory of Bankruptcy Law: Bankruptcy Law AS (IS) Civil Procedure' (2004) 61 Washington and Lee Law Review, 931 – 1060.

Moss, G, 'Comparative Bankruptcy Cultures: Rescue or Liquidation? Comparison of Trends in National Law-England' (1997-1998) 23 Brooklyn or Brook. J. Int'l L, 115

-----' 'Chapter II: An English Lawyer's Critique' (1998) 11 Insolvency Intelligence, 17 – 21.

- Nwauche, ES, 'The Duties of a Receiver/Manager in Nigeria and Ghana (2005) 14(1) Int. Insolv. Rev, 71 91.
- Nyombi, C, 'Employees' Rights during Insolvency' (2013) 55(6) International Journal of Law and Management,417 428.
- Nyombi, C, Kibandama, A & Bakibinga, DJ, 'The Motivations behind the Uganda Insolvency Act 2011' (2014) 8 J.B.L, 651 – 666.
- Oditah, F, 'Assets and the Treatment of Claims in Winding up' (1992) 108 LQR, 459 500.
- Oladele, OO, 'The Legal Intricacies of Corporate Restructuring and Rescue in Nigeria' (2007) the Calabar Law Journal, Xi, 136 153.
- Ofili, PN, 'Mergers of Companies under the Investment and Securities Act, 2007' (2012) IJCL, 177 193.
- Oghoghomeh, T & Odita, A, 'Entrepreneurs' Nightmare-Corporate Failure; Consequences and Probable Solutions' (2013) Research Journal of Finance and Accounting Vol. 4, No.19, 45 – 52.
- Okoli, HSN, 'Corporate Insolvency-Options and Strategies for Lending Institutions' (1998) M.P.J.F.I.L Vol.2 No. 1, 55 64.
- Okoli, P, 'Rescue Culture in the United Kingdom: Realities and the Need for a Delicate Balancing Act' (2012) 23(2) I.C.C.L.R, 61.
- Okogbule, NS, 'An Appraisal of the Legal and Institutional Framework for Combating Corruption in Nigeria' (2006) 13(1) Journal of Financial Crime, 92 – 106.
- ----- 'Official Corruption and the Dynamics of Money Laundering in Nigeria' (2007) 14(1) Journal of Financial Crime, 49 63.
- Omar, P, 'The Landscape of International Insolvency Law' (2002) 10(2) Int.Insolv.Rev, 173 – 200.
- ----- 'The Mutual Influence of French and English Commercial Laws in Insolvency' (2008) 19(4) I.C.C.L.R, 136 146.
- ----- 'Insolvency Law and Human Rights: An Update' (2009) 6(6) International Corporate Rescue, 369 373.
- Osunbor, OA, 'The Nigeria Law of Take-Overs and Mergers' (1989) The Gravitas Review of Business & Property Law, 40 46.

Peniel, AO and Babatope, OA, 'Assessment of Bribery and Corruption in Nigeria

Socio-Political Milieu (2006) International Journal of Law and Contemporary Studies Vol.I No 1 & 2, 140 – 147.

- Pennington, RR, 'The Genesis of the Floating Charge' (1960) 23(6) M.L.R, 630 646.
- Prest, M, 'The Credit Crunch and Restructuring Techniques: A Basic Guide' (2008) Corporate Rescue and Insolvency, 8.
- Prentice, D, 'Bargaining in the Shadow of the Enterprise Act 2002'(2004/1) 5 E.B.O.R, 153 158.
- Rahmani, A, 'Shareholder Control and its Nemesis', (2012) 23(1) ICCLR, 12 23.
- Rajak, H, 'Determining the Insolvent Estate- A Comparative Analysis' (2011) 20(1)Int.Insolv.Rev, 1 28.
- Riley, CA, 'Understanding and Regulating the Corporation', (1995) 58(4) M.L.R, 595 612.
- Rose-Ackerman, S, 'Risk Taking and Ruin: Bankruptcy and Investment Choice' (1991) 20 Journal of Legal Studies, 277 310.
- Sancho, DC, 'Reform of Spanish Bankruptcy Law', (2004) 32(2) International Business Lawyer,53 59.
- Schermer, BS, 'Response to Professor Gross: Taking the Interests of the Community into Account in Bankruptcy: A Modern Day Tale of Belling the Cat' (1994) 72(3) Wash.U.L.Q, 1049 1054.
- Schwehr, B, 'Corporate Rehabilitation Proceedings in the United States and Germany', (2003) 12(1) Int.Insolv.Rev, 11 35.
- Steyn, L, 'Human Rights Issues in South African Insolvency Law' (2004) 13(1) Int.Insolv.Rev, 1 – 25.
- Stout, LA, 'Bad and Not-So-Bad Arguments for Shareholder Primacy' (2002) 75 Southern California Law Review, 1189 – 1209.
- Suchak, R, 'Corporate Rescue Proceedings and the Enforcement of *Ipso Facto* Termination Clauses: A Comparison of the English and US Approaches' (2011) International Corporate Rescue, 131 – 136.
- Tettenborn, AM, 'Reservation of Title: Insolvency and Priority Problems' (1981) J.B.L, 173 – 177.
- Tribe, JP, 'The Morality of Romalpa Clauses in Corporate Insolvency: A Case for Reform?' (2001) 15(5) Tolley's Insolvency Law and Practice, 166 175.

-----'Companies Act Schemes of Arrangement and Rescue: The Lost Cousin or Restructuring Practice?' (2009)24(7) Butterworths Journal of International Banking and Financial Law, 386 – 392.

-----'Company Voluntary Arrangements and Rescue: a new Hope and a Tudor Orthodoxy' (2009) 5 J.B.L, 454 – 487.

- ----- 'Crystal Ball and Insolvency: What Does the Future Hold' (2012) 23(12) ICCLR 405 411.
- Tomasic, RA, 'The Rescue of Northern Rock: Nationalisation in the Shadow of Insolvency' (2008) 1(4) Corporate Rescue and Insolvency, 109 111.
- ------ 'Between a Rock and a Hard Place: A Post Northern Rock Regime for the Rescue of Failing Banks' (2008) 5(1) Corporate Rescue and Insolvency, 141 – 143.
- ------ 'Creating a Template for Banking Insolvency Law Reform after the Collapse of Northern Rock: Part I (2009) 22(5) Insolvency Intelligence, 65 – 70.
- ------ 'Creating a Template for Banking Insolvency Law Reform after the Collapse of Northern Rock: Part 2' (2009) 22(6) Insolvency Intelligence, 81 – 88.
- ----- 'Establishing a UK Rescue Regime for Failed Investment Banks' (2010) 3(2) Corporate Rescue and Insolvency, 60 – 64.
- ----- 'The Financial Crisis and the Haphazard Pursuit of Financial Crime' (2011) 18(1) Journal of Financial Crime, 7 31.
- Tschentscher, F, 'The Modern German Insolvency Regime: Process, Reform, Financing and Creditor Ranking' (2013) 10(3) International Corporate Rescue. 167 – 171.
- Qi, L, 'Availability of Continuing Financing in Corporate Reorganisation: the UK and US Perspective' (2008) 29 Comp.Law, 162 167.

Walton, P, 'Pre-packing' in the UK' (2009)18(2)Int.Insolv.Rev, 85 – 108.

Warren, E, 'Bankruptcy Policy' (1987) 54(3)U.Chil.Rev,775 – 814.

-----'The Untenable Case for Repeal of Chapter 11' (1992) 102(2) Yale LJ, 437

-479.

----- 'In Bankruptcy Policy Making in an Imperfect World' (1993) 92(2) Michigan Law Review, 336 – 387.

Warren, E & Westbrook, JL, 'Searching for Reorganization Realities' (1994) 72

Wash.U.L.Q, 1257 – 1289.

----- 'The Success of Chapter 11: A Challenge to the Critics' (1994) 107 Michigan Law Review, 603 – 641.

- Wellard, M & Walton, P, 'A Comparative Analysis of Anglo-Australian Pre-Packs: Can the Means Be Made to Justify the End?' (2012) 21(3) Int.Insolv.Rev, 243 – 281.
- Westbrook, JL, 'The Globalisation of Insolvency Reform' (1999) N.Z.L.R, 401-413.
- Whincop MJ, 'Painting the Corporate Cathedral: The Protection of Entitlement in Corporate Law' (1999) OJLS, 19 33.
- White, MJ, 'Does Chapter 11 Save Economically Inefficient Firms?' (1994) 72(3) Wash.U.L.Q, 1319 – 1340.
- -----, 'Corporate Bankruptcy as a Filtering Device: Chapter II Reorganizations and Out-of-Court Debt Restructurings' (1994) 10 (2) The Journal of law, Economics &Organization,268.
- Whitford, WC, 'What's Right about Chapter 11'(1994) 72(3) Wash.U.L.Q, 1379-1406.
- Wigwe, C, 'Corruption as an Economic Barrier to Development in Third World Countries' P.H.J.B.L 2.1 (2016), 324 – 336.
- Wijaya, A, 'Pre-Pack Administration Sale: A Case of Sub Rosa Debt Restructuring' (2016) 25(2) Int.Insolv.Rev, 119-137.
- Wosu, EM, 'The Legal Effect of Revocation of Banking Licence and Winding up Order on the Legal Personality of a Bank' (2016) P.H.J.B.L 2.2, 577 – 588.
- Xie,B, 'Role of Insolvency Practitioners in the UK Pre-Pack Administrations: Challenges and Control' (2012) 21(2) Int.Insolv.Rev, 85 – 103.
- Zacaroli,A,' The Powers of Administration Under Schedule BI Prior to the Creditors' meeting Transbus International Limited' (2004) 1 (4) International Corporate Rescue, 208.
- Zhao, J and Tribe, J, 'Corporate Social Responsibility in an Insolvent Environment: Directors Continuing Obligations in English Law' (2010) ICCLR, 305, 645-658.

Online Materials

Adebola, B 'A Few Shades of Rescue: Towards an Understanding of the Corporate
Rescue Concept in England and Wales' (2014).
< <u>http://papers.ssrn.com/sol3/papers.cfm?abstract_id=2524488>accessed</u>
on 10 May 2015.
'The Duty of the Nigerian Receiver to "Manage" the Company',
<a>http://papers.ssrn.com/sol3/papers.cfm?abstract_id=2297805> accessed
on 10 May 2015.
'The Nigerian Business Rescue Model: An Introduction' (2014) NIALS
Journal of Legal Studies. Also available at
">http://papers.ssrn.com/sol3/papers.cfm?abstract_id=2297824> accessed
on 10 May 2015.
'Conflated Arrangements: Comments on the Company Voluntary
Arrangements in the Proposed Nigerian Insolvency Act'(2015) 3 NIBLeJ
2 also available at< <u>http://www.4,ntu.ac.uk/n/s/documentuploads/174811</u> >
accessed on 20 November 2016.
'Corporate Rescue and the Nigerian Insolvency System'
on 19 November 2016.
oli 19 November 2010.
Aina, K, 'Asset Management Corporation of Nigeria- A Pretentious Special Purpose
Vehicle'< <u>http://papers.ssrn.com/sol3/papers.cfm?abstract_id=2617500</u> >ac
cessed on 20 June 2015.
'The Role and Duties of a Receiver in Nigeria Law' (2015)
http://www.ssrn.com/abstract=2605189 accessed on 7 December 2016.
Altimeter and C. (Compared Incolor of Long in Nicolais A. Nood Compared Comp
Akinwunmi, S 'Corporate Insolvency Law in Nigeria – A Need for Reform'
http://www.akinwunmibusari.comaccessed on 17 July 2015.
Alam, SM, 'The Enterprise Act 2002: Past, Present and Future'
http://www.shossainandassociates.com accessed on 5 December 2016.
<u>http://www.shossanandassociates.com</u> accessed on 5 December 2010.
Arnold, P, Cooper, C, and Sikka, P, 'Insolvency, Market Professionalism and the
Commodification of Professional Expertise (2005)
http://www.ibranan.net/naron/paper/insolvency-
commodification.pdf?paperid=360736 accessed on 18 November 2016.
BusinessDay Newspaper, Tuesday, 15 th March, 2016. <u>www.businessdayonline.com</u>
accessedon 15 March 2016.
'Code of Ethics for Insolvency Practitioners' <u>http://www.icaew.com/en</u> accessed on 18
November 2016.

- Enriques, L & Gatti, M, 'Creeping Acquisitions in Europe: Enabling Companies to be Better Safe than Sorry' (2015) 15:1 Journal of Corporate Law Studies, 55 – 101, also available at <u>http://www.tandfonline.com/doi/abs/10.52351...5970.15.1.127</u>accessed on 19 November 2016.
- Fennessy, M(ed), International Corporate Rescue <u>http://www.chasecambria.com</u> accessed on 17 November 2016.
- Habi, HM, 'A Comparative Study of some Issues Relating to Corporate Insolvency Laws in Nigeria and South Africa' (2013) <u>http://www.repository.up.ac.za/bitsbeam/handle/22</u> accessed on 22 November 2016.
- Kate-Iouzou, D & Siems, M, 'Disappearing Paradigms in Shareholder Protection: Leximetric Evidence for 30 Countries, 1990-2013' (2005) Journal of Corporate Law Studies. 15:1,127–160. Also available at <u>http://www.tandfonline.com/dol/abs/10.5235/14735970.15.1.127</u> accessed on 19 November 2016.
- Idigbe, A, 'The Nigerian Insolvency Law and the Rights of Creditors and Account Holders of Intermediated Securities vis-a-visThe Insolvent Intermediary' <u>http://www.punuka.com/uploads/THE-NIGERIA</u> accessed on 20 November 2016.
- Kirkpatrick, G, 'The Corporate Governance lessons from the Financial Crisis' Financial Market Trends (2009) <u>http://www.oecd.org/dataoecd/32/1/42229620.pdf</u> accessed on 18 November 2016.
- Levenstein, E, 'An Appraisal of the New South African Business Rescue Procedure' (2016) LLD Thesis, University of Pretoria, Pretoria <u>http://respository.up.ac.za</u> accessed on 21 November 2016.
- Loubser, A, 'The Interaction between Corporate Rescue and Labour Legislation: Lessons to be drawn from the South African Experience' (2005) 14 (1) International Insolvency Review, 57 – 69 http://www.onlinelibrary.wiley.com
- Messy Bailout Proposal for Aviation Industry' <u>http://www.punchng.com/editorials/messy-bailout-proposal-for-aviation-industry/</u> accessed on 17 July 2015.
- Nyombi, C, 'The Development of Corporate Rescue Laws in Uganda and in UK (2015) 57:2 International Journal of Law and Management, also available at <u>http://www.emeraldinsight.com</u> accessed on 17 November 2016.
- Okolo, EO, 'Insolvency Law in Nigeria'

http://www.investadvocate.com.ng/2016/10/11/insolvency-law-nigeria accessed on 20 November 2016.

- Oluwa, M, 'The Need for Corporate Rescue in Nigeria' <u>http://www.modupeolualegalconsultants</u> accessed on 19 November 2016.
- Opara, LC, Okere, LI, & Opara, CC, 'The Legal Regime of Bankruptcy and Winding up Proceedings as a Tool for Debt Recovery in Nigeria. An Appraisal' (2014) <u>http://www.cscanada.net/index.php/ccs/article</u> /view/4151/pdf-106accessed on 20 November 2016.
- Real Business Rescue Licensed Insolvency Practitioners, 'Corporate Administration Rules And The Pre-Pack Pool'<u>http://www.realbusinessrescue.co.uk/pre-pack-administration/administration-rules-and-the-pre-pack-pool</u>accessed on 19 November 2016.
- Sengupta, R & Sharma, A, 'Corporate Insolvency Resolution in India' (2016) 51(15) Economic & Political Weekly. Also available at <u>http://www.epwin</u> accessed 5 December 2016.
- ------ 'Corporate Insolvency Resolution in India: Lessons from a Cross-Country Comparison' (2015) <u>http://www.igidr.ac.n/pdf/publications/wp.2015-</u> 029.pdf accessed on 6 December 2016.
- Svoronou, E & Downey, M, 'Rescue Procedures- A Comparative Outlook' (2014) Digesta Online Law Review,348 – 371. Also available at <u>http://www.digestaonline.gr.pdfs</u> accessed on 10 December 2016.
- Taddia, M, 'Pre-Pack' Administrations Rule Changes Face Trouble' Law Gazette, 1 November 2011 <u>http://www.lawgazette.co.uk/anlysis/pre-packadministrationsrule-change-face-trouble/6295.full</u> accessed on 19 November 2016.
- Uchechukwu, MS, 'Rethinking Business Rescue in Nigeria: Borrowing Virtues from Chapter II of the US Bankruptcy Code' <u>http://www.etd.ceu.hu/2015/mba_sanford.pdf</u>accessed on 19 November 2016.
- Wood, JM, 'Corporate Rescue: A Critical Analysis of its Fundamentals and Existence', unpublished PhD Thesis, University of Leeds, May 2013 <u>http://www.ethesis.whiterose.ac.uk/4963/1...ood_JM-Law-PhD-2013.PDF.pdf</u> accessed on 17 November 2016.
- Zwieten, KV, 'Corporate Rescue in India: The Influence of the Courts' (2015) 15:1 Journal of Corporate Law Studies, 1-31, also available at <u>http://www.tandfonline.com/doi/abs/10.52351....</u> accessed on 19 November 2016.

Unpublished Works

- Aina, AA, 'Legal and Institutional Framework for Debt Finance by Companies in Nigeria', A PhD Pre-Field Seminar presented at the Faculty of Law, University of Ibadan, Ibadan, Nigeria on October, 2014.
- Alemika, EI, *et al*, 'NALT'S Uniform Format & Citation Guides for Scholarly Writings and Researches in Law Faculties and Law Schools in Nigeria' in Omaka, CA (ed), Law in Nigeria @ 100: Reviewing the Past, Projecting the Future(Proceedings of the 47th Annual Conference of the Nigerian Association of Law Teachers (NALT) held 2nd to 6th June, 2014 at Ebonyi State University, Abakaliki-Nigeria) pp.42 – 71.
- Armour, J, 'The Law and Economics of Corporate Insolvency' (2001) ESRC Centre for Business Research, University of Cambridge Working Paper No 197.
- British Bank Association, 'Banks and Business Working Together (London: 1997).
- Cork, K, 'Report of the Review Committee on Insolvency Law and Practice' (cmnd.8558, 1982).
- Crawford, K, 'The Law and Economics of Orderly and Effective Insolvency', unpublished PhD Thesis, University of Nottingham, United Kingdom, December 2012.
- Dimgba, N 'The Regulation of Competition Through Merger Control: The Case Under the Investment and Securities Act, 2007', A paper delivered at the Nigerian Bar Association Section on Business Law Conference at Abuja on 16th April, 2009.
- Etukakpan, SE, 'Transfer of Undertakings: The Tension Between Business Rescue and Employment Protection in Corporate Insolvency', unpublished PhD Thesis, Nottingham Trent University, United Kingdom, July 2012.
- Halita, MB, 'A Comparative Study of Some Issues Relating to Corporate Insolvency Law in Nigeria and South Africa', unpublished LL.M Dissertation, University of Pretoria, South Africa, May 2013.
- Ibrahim, A N, 'Resolution of Problem Banks: Purchase and Assumption Option', A Paper delivered at the International Association of Deposit Insurers (IADI) African Regional Committee Workshop at Abuja, Nigeria from May 9-13, 2011, p.3.
- Idigbe, A, 'Using Existing Insolvency Framework to Drive Business Recovery in Nigeria: The Role of the Judges' A Paper delivered at the Federal High Court Judges Conference at Sankuru Hotel Sokoto, 2011.

Levenstein, E, 'An Appraisal of the New South African Business Rescue Procedure',

unpublished LLD Thesis, University of Pretoria, Pretoria, November 2015.

- LoPucki, LM, 'A Team Production Theory of Bankruptcy Reorganization' (2003) UCLAW Law School, Law and Econ Research Paper.
- Mahmud, B, Report of the Corporate Affairs Commission to the 2016 Annual General Conference of the Nigeria Bar Association at Port Harcourt, Nigeria from 19th to the 2th day of August 2016.
- Omaka, CA (ed), Law in Nigeria @ 100: Reviewing the Past, Projecting the Future (Proceedings of the 47th Annual Conference of the Nigerian Association of Law Teachers (NALT) held 2nd to 6th June, 2014 at Ebonyi State University, Abakaliki-Nigeria).
- Paterson, S, 'Rethinking the Role of the Law Corporate Distress in the Twenty-First Century' LSE Law, Society and Economy Working Papers 27/2014.
- Pomerleano, M & Shaw, W (eds), 'Corporate Restructuring: Lessons from Experience' (the World Bank, 2005).
- Too, FC, 'A Comparative Analysis of Corporate Insolvency Laws: Which is the Best Option for Kenya?', unpublished PhD Thesis, Nottingham Trent University, September 2015.
- The World Bank, 'Principles for Effective Insolvency and Creditor Rights System' (Revised Draft, 2005).

-----'Principles for Effective Insolvency and Creditor/Debtor Regime'.

UNCITRAL, Legislative Guide on Insolvency Law, 2005.

----- Report of Working Group V (Insolvency Law), 2001 – 2016.

Wood, JM, 'Corporate Rescue: A Critical Analysis of its Fundamentals and Existence', unpublished PhD Thesis, University of Leeds, May 2013.

World Bank, Doing Business Report 2015: 'Going Beyond Efficiency'.

----- Doing Business 2016 Report: 'Measuring Regulatory Quality and Efficiency'.

Zhang, H, 'Making an Efficient and Well-Functioning Corporate Rescue System in Chinese Bankruptcy Laws: From the Perspective of a Comparative Study between England and China', unpublished PhD Thesis, School of Law, University of Leicester, September 2008.

Magazines and Newspapers

- Abayomi-Olukunle, O, 'Reforms, Revolutions and the Ministry of Trade & Investment: Amendment of the Companies and Allied Matters Act', LAWYER, a Thisday Weekly Pull-out, 13-09-2016, p.12.
- Abdulrasaq, A, 'Create Judicial Ombudsman Office', LAWYER, a Thisday Weekly Pull-Out, p.7.
- Akinmurele, L, 'Increasing Importance of Remittances as Bailout for Nigerian Banks', BusinessDay Monday 14 March 2016, p.17.
- Akintunde, A, 'Honeywell v Ecobank: Judge Refuses to Recuse Himself from Case', LAWYER, a Thisday Weekly Pull-out, 22-11-2016, p.6.
- Akintola, K, 'CBN Confirms N4.5trn Exposure to AMCON'S N6.6trn Liabilities, Business Day Friday 18 March 2016, p.7.
- Amaechi, N, 'CAC, In Innovative Strategies to Keep Pace with Global Trends, Best Practices', Business Day Tuesday, 18 October 2016, p. 15.
- AMCON Publication of List of Delinquent Debtors, BusinessDay, February 15, 2016, p. 13.
- Atuanya, P, 'AMCON: Building Investor Confidence is a Marathon Race', Business Day Friday, 12 August 2016, p.28.
- Bassey, J, et al, 'Bailouts Reveal Precarious Finances of Nigerian States', Business Day Monday 25 April 2016, p.1, 4 & 51.
- Boyi, I, 'AMCON'S Intervention into Peugeot Automobile Nigeria was Timely', Business Day Wednesday, 22 June 2016, p.27.
- Brunsden, J, 'Eurozone Ministers Breakthrough', Business Day Financial Times, Monday 25 April 2016, p.A1.
- Candide-Johnson, Y, 'Nothing is more Dangerous than a Corrupt Judge', The Nation Law, Tuesday, November 22, 2016, p.21 22.
- Ease of Doing Business', BusinessDay Editorial Monday 18 April 2016, p.12.
- Egbe, R, 'Special Anti-Graft Courts' II be Financial Burden', The Nation Tuesday, April, 19, 2016, p.24.
- Eyo, B, 'E-Commerce Industry Needs Government Bailout', BusinessDay Wednesday 05 October 2016, p. 23.

- Fidelity Bank Plc Publication of Names of Delinquent Customers, BusinessDay, February 15, 2016, p.19.
- 'FG to give Fresh Bailout to States', The Nation News Sunday June 12, 2016, p.4.
- First Bank of Nig Plc, 'List of First Bank Delinquent Debtors' The Punch Monday, May 9 2016, pp. 38 – 44.
- Hassan, BI, 'Why NASS Must Support AMCON', The Nation Tuesday, November 15 2016, p.20.
- How Best to Fight Corruption in Nigeria', Business Day Editorial, Friday 28 October 2016, p.12.
- Ikhilaee, E, 'Waiting for Buhari's Judicial Reforms', The Nation Law Tuesday, June 7, 2016 pp.21, 22, 29&30.
- Ikhilae, E, 'Acting CJN to Judges: Back Fed. Govt's Anti-Corruption Effort', The Nation Tuesday, November 22, 2016, p.7.
- Kuru, AL, 'If I had the Formula to Wind down AMCON in 3 years, I will do so', Business-Day Monday 22 February 2016, p.32.
- Jibueze, J, 'Alleged N2.1b Debt: Sky Bank Takes over Obat Oil's Depot', The Nation Saturday, November 12, 2016, p.8.
- Mba, SU 'Recession Preventive Rehabilitation and the Nigerian Draft Insolvency Bill: Lessons for the EU? LAWYER,a Thisday Weekly Pull-out 06-09-2016, p. 7.
- Melah, U 'Amcon, Delta Steel and Aribisala's Basic Instinct', ThisDay Saturday Newspaper, July 4.2015. p.48.
- Mowoe, KM, 'The Fight against Corruption: Matters Arising', The Nation Tuesday, November 15, 2016, p.27.
- Nyong, C, 'Fraud Examiners Can Help Nigeria Curb Corruption, The Guardian Saturday, November, 12, 2016, p.59.
- Okojie, J, 'States' Insolvency Threatens GES Agric Support', BusinessDay Friday 29 April 2016, pp.1 & 4.
- Onanuga, A, 'Court Reverses Receiver/Manager's Appointment', The Nation Law Tuesday, June 7, 2016, p.28.
- Onyekwere, J, 'Bank Takes OverObat Oil's Depot Over Alleged Debt', The Guardian Saturday, November 12, 2016, p.56.

Punch Editorial, Nigeria's Tough Businesses Terrain', Monday, May 9, 2016, p. 24.

- Skye Bank Plc Publication of List of Loan Debtors, BusinessDay, February 11, 2016, pp. 33 35.
- Udofia, K, 'Quasi-Security Transactions: Sidestepping the Adverse Effects of Formal Insolvency Processes', LAWYER, a Thisday Weekly Pull-Out 19-07-2016, p.14.
- ----- 'Fraudulent Trading Rule: Asset Recovery Mechanism in Winding-up Proceedings', LAWYER, a Thisday weekly Pull-Out 06-12-2016, p. 10.
- ------ 'Contribution to Investors' Protection Fund and the Fraudulent Preference Risk', LAWYER, a This Day Weekly Pull-out, 18.04.2017, p.11
- Uroko, C, 'Embracing Mediation as Alternative Business Dispute Resolution Instrument, Business Day Thursday 21 April 2016, p.26.
- Uwaleke, J, 'Improving Nigeria's Ranking in Doing Business Index', The Punch Monday, May 9, 2016, back page.
- Uzuegbunan, F, 'Ecobank Loses in Court against Honeywell', BusinessDay 29 June 2016 p.6.